UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-K

[X] ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(D) OF THE SECURITIES EXCHANGE ACT OF 1934 For the fiscal year ended December 31, 2000

0R

[] TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(D) OF THE SECURITIES EXCHANGE ACT OF 1934

> For the transition period from to Commission file number 1-8993

WHITE MOUNTAINS INSURANCE GROUP, LTD. (Exact name of Registrant as specified in its charter)

BERMUDA (State or other jurisdiction of incorporation or organization) 94-2708455 (I.R.S. Employer Identification No.)

80 SOUTH MAIN STREET, HANOVER, NEW HAMPSHIRE (Address of principal executive offices)

03755-2053 (Zip Code)

Registrant's telephone number, including area code: (603) 643-1567

Securities registered pursuant to Section 12(b) of the Act:

Title of each class

Common Shares, par value \$1.00

per share

Name of each exchange on which registered New York Stock Exchange

Securities registered pursuant to section 12(g) of the Act: None

Indicate by check mark whether the Registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the Registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes X No

Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K is not contained herein, and will not be contained, to the best of the Registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K. [X]

The aggregate market value of voting shares (based on the closing price of those shares listed on the New York Stock Exchange and the consideration received for those shares not listed on a national or regional exchange) held by non-affiliates of the Registrant as of March 23, 2001, was \$1,825,187,696.

As of March 23, 2001, 5,880,115 Common Shares, par value of \$1.00 per share, were outstanding (excludes 21,000 restricted Common Shares which were not vested as of such date).

DOCUMENTS INCORPORATED BY REFERENCE None

TABLE OF CONTENTS

PART I

ITEM 1.	Busi	ness	1
	a.	General	1
	b.	Reinsurance and Insurance Operations	1
	c.	Discontinued Operations	8
	d.	Investing Operations	8
	e.	Regulation	8
	f.	Employees	9
	g.	Forward-Looking Statements	9
ITEM 2.	Prop	erties	9
ITEM 3.	Lega	l Proceedings	10
ITEM 4.	Subm	ission of Matters to a Vote of Security Holders	10

PART II

ITEM 5.	Market for the Company's Common Equity and Related Stockholder Matters	10
ITEM 6.	Selected Financial Data	11
ITEM 7.	Management's Discussion and Analysis of Financial Condition and Results of Operations	12
ITEM 7A.	Quantitative and Qualitative Disclosures About Market Risk	22
ITEM 8.	Financial Statements and Supplementary Data	22
ITEM 9.	Changes in and Disagreements with Accountants on Accounting and Financial Disclosure	22
	PART III	
ITEM 10.	Directors and Executive Officers	22
ITEM 11.	Executive Compensation	25
ITEM 12.	Security Ownership of Certain Beneficial Owners and Management	28
ITEM 13.	Certain Relationships and Related Transactions	29

PART IV

ITEM 14.	Exhibits,	Financial	Statement	Schedules,	and Reports on	Form 8-k	К	29
----------	-----------	-----------	-----------	------------	----------------	----------	---	----

PART I

Item 1. Business

GENERAL

White Mountains Insurance Group, Ltd. (the "Company") was originally formed as a Delaware corporation in 1980 and became a Bermuda corporation during 1999. Within this report, the consolidated organization is referred to as "White Mountains". White Mountains' principal operations are conducted through its subsidiaries and affiliates in the businesses of property and casualty insurance and reinsurance, each of which is described in detail herein. The Company's headquarters is located at Crawford House, 23 Church Street, Hamilton, Bermuda HM11, its principal executive office is located at 80 South Main Street, Hanover, New Hampshire, 03755-2053 and its registered office is located at Clarendon House, 2 Church Street, Hamilton, Bermuda HMDX.

In June 1999 the Company changed its name from "Fund American Enterprises Holdings, Inc." to "White Mountains Insurance Group, Inc."

In October 1999 the Company completed a corporate reorganization whereby it changed its domicile from Delaware to Bermuda (the "Redomestication"). The Redomestication was primarily undertaken to improve the Company's ability to compete in international markets by creating a corporate structure which is more favorable for the formation and growth of internationally-based insurance and reinsurance operations and which has an enhanced ability to pursue business combinations with non-United States entities. In connection with the Redomestication, the Company's name was further changed to "White Mountains Insurance Group, Ltd." to comply with Bermuda law.

In September 2000 White Mountains entered into a definitive agreement with London-based CGNU plc ("CGNU") to purchase its U.S. property and casualty insurance operations ("CGU"). CGU, headquartered in Boston, is a large U.S. property & casualty insurer which operates in all 50 states. Completion of the CGU acquisition is subject to, among other matters, the receipt of regulatory approvals, the completion of financing and the satisfaction of other customary conditions. Further details concerning this pending acquisition are contained herein.

REINSURANCE AND INSURANCE OPERATIONS

REINSURANCE OPERATIONS

FOLKSAMERICA HOLDING COMPANY, INC. ("FOLKSAMERICA")

Folksamerica, through its wholly-owned subsidiary, Folksamerica Reinsurance Company (a New York-domiciled reinsurance company), is a multi-line broker-market reinsurer which provides reinsurance to insurers of property and casualty and accident and health risks in the United States, Canada, Latin America and the Carribean. Folksamerica is rated "A" (Excellent) by A.M. Best Company. During 2000, 1999 and 1998, Folksamerica had net written premiums of \$332.6 million, \$201.7 million and \$212.6 million, respectively. At December 31, 2000 and 1999, Folksamerica had total assets of \$2.8 billion and \$1.3 billion, respectively, and shareholder's equity of \$339.7 million and \$249.4 million,

In June 1996 White Mountains purchased a 50.0% economic interest in Folksamerica for \$79.9 million from a group of European mutual insurance companies (the "European Mutuals") who continued to own the remaining 50.0% interest. In November 1997 White Mountains and the European Mutuals each purchased equal interests in Folksamerica for \$20.8 million which maintained White Mountains' 50.0% economic ownership position. In August 1998 White Mountains acquired all of the remaining capital stock of Folksamerica from the European Mutuals for \$169.1 million which resulted in Folksamerica becoming a wholly-owned consolidated subsidiary of White Mountains at that time.

In June 1999 Folksamerica acquired USF Re Insurance Co. ("USF Re") from the Centris Group. Significant assets and liabilities acquired through USF Re included \$204.1 million of cash and investments and \$106.5 million of net loss and loss adjustment expense reserves.

On March 31, 2000, Folksamerica acquired PCA Property & Casualty Insurance Company ("PCA") from Humana Inc. Significant assets and liabilities acquired through PCA included \$339.8 million of cash and investments and \$253.8 million of net loss and loss adjustment expense reserves.

On May 5, 2000, Folksamerica acquired substantially all the reinsurance operations of Risk Capital Reinsurance Company ("Risk Capital"), a wholly-owned subsidiary of Risk Capital Holdings, Inc. Significant assets and liabilities acquired through the Risk Capital transaction included \$249.9 million of cash and investments, \$108.6 million of premiums receivable, \$312.6 million of net loss and loss adjustment expense reserves and \$82.0 million of net unearned reinsurance premiums.

On October 10, 2000, Folksamerica acquired an 80% interest in Esurance Inc. ("Esurance") for \$9.0 million in cash. Esurance is an internet-based insurance provider which sells personal auto insurance in 26 states.

REINSURANCE OVERVIEW

Reinsurance is an arrangement in which a reinsurance company (the "reinsurer") agrees to indemnify an insurance company (the "ceding company") for all or a portion of the insurance risks underwritten by the ceding company under one or more insurance policies. Reinsurance can benefit a ceding company in a number of ways, including reducing net liability exposure on individual risks, providing catastrophe protections from large or multiple losses, stabilizing financial results and assisting in maintaining acceptable operating leverage ratios. Reinsurance also provides a ceding company with additional underwriting capacity by permitting it to accept larger risks and underwrite a greater number of risks without a corresponding increase in its capital or surplus. Reinsurers may also purchase reinsurance, known as retrocessional reinsurance, to cover their own risks assumed from primary ceding companies. Reinsurance companies enter into retrocessional agreements for many of the same reasons that ceding companies enter into reinsurance agreements.

Folksamerica writes both treaty and facultative reinsurance. Treaty reinsurance is an agreement whereby the ceding company is obligated to cede, and the reinsurer is obligated to assume, a specified portion or category of risk under all qualifying policies issued by the ceding company during the term of a treaty. In the underwriting of treaty reinsurance, the reinsurer does not evaluate each individual risk assumed and generally accepts the original underwriting decisions made by the ceding insurer. Facultative reinsurance is underwritten on a risk-by-risk basis whereby Folksamerica applies its own pricing to an individual exposure. Facultative reinsurance is normally purchased by insurance companies for individual risks not covered under reinsurance treaties or for amounts in excess of limits on risks covered under reinsurance treaties. The majority of Folksamerica's premiums are derived from treaty reinsurance contracts both on an excess of loss and quota share basis, which in 2000 amounted to 23.2% and 67.1% of its total earned premiums, respectively. Folksamerica derives its business from a spectrum of ceding insurers including national, regional, specialty and excess and surplus lines writers. Folksamerica selects transactions based solely on anticipated underwriting results of the transaction which are evaluated on a variety of factors including the quality of the reinsured, the attractiveness of the reinsured's insurance rates, policy conditions and the adequacy of the proposed reinsurance terms.

A significant period of time normally elapses between the receipt of reinsurance premiums and the disbursement of reinsurance claims ("float"). The claims process generally begins upon the occurrence of an event causing an insured loss followed by: (i) the reporting of the loss to the ceding company; (ii) the reporting of the loss by the ceding company to Folksamerica; (iii) the ceding company's adjustment and payment of the loss; and (iv) the payment to the ceding company by Folksamerica. During this time, Folksamerica earns investment income on the float. Therefore, Folksamerica's combined ratio can generally be higher than that of White Mountains' consolidated property and casualty insurance operations and yet may still earn an equivalent or superior return on equity.

Folksamerica's net written premiums by line of business follow:

	Year Ended December 31,							
Millions	2000	1999	1998	1997	1996			
Liability Property	\$208.4 91.6	\$122.6 68.9	\$122.0 87.2	\$123.6 104.9	\$ 79.6 85.9			
Accident and Health Other	26.4 6.2	10.2	- 3.4	- 3.9	- 6.4			
Total	\$332.6	\$201.7	\$212.6	\$232.4	\$171.9			

Folksamerica's net written premiums by geographic location follow:

	Year Ended December 31,								
Millions	2000	1999	1998	1997	1996				
United States	\$296.7	\$172.2	\$190.9	\$208.6	\$155.2				
Canada Latin America and the Carribean	21.6 14.3	21.0 8.5	13.9 7.8	10.1 13.7	4.2 12.5				
Total	\$332.6	\$201.7	\$212.6	\$232.4	\$171.9				

UNDERWRITING

Folksamerica's primary underwriting objective is to carefully assess reinsurance opportunities to determine the probability of a particular transaction providing an underwriting profit. Those risks that do not provide a reasonable likelihood of delivering an underwriting profit are rejected. Underwriting opportunities presented to Folksamerica are evaluated based on a number of factors including historical analysis of results, estimates of future loss costs, a review of other programs displaying similar exposure characteristics, the primary insurer's underwriting and claims experience and the primary insurer's financial condition. Folksamerica regularly conducts underwriting and claims audits of ceding companies to assist it in evaluating the information submitted by the ceding companies.

Folksamerica's most senior underwriters and executives are responsible for its underwriting policy and quality standards and informing Folksamerica's board of directors of current and anticipated market conditions and underwriting results.

MARKETING

Folksamerica generally obtains all its reinsurance business through brokers and reinsurance intermediaries which represent the ceding company in negotiations for the purchase of reinsurance. The process of effecting a brokered reinsurance placement typically begins when a ceding company enlists the aid of a reinsurance broker in structuring a reinsurance program. Often the ceding company and the broker will consult with one or more lead reinsurers as to the pricing and contract terms for the reinsurance protection being sought. Once the ceding company has approved the terms quoted by the lead reinsurer, the broker will offer participations to qualified reinsurers until the program is fully subscribed by reinsurers at terms agreed to by all parties.

Folksamerica pays its reinsurance brokers' commissions based on negotiated percentages of the premium it writes. These commissions, which generally average 5% of premium, constitute a significant portion of Folksamerica's total acquisition costs and are included in its underwriting expenses. During the year ended December 31, 2000, Folksamerica received approximately 56.4% of its gross reinsurance premiums written from three major reinsurance brokers as follows: (i) E.W. Blanch - 21.6%; (ii) Guy Carpenter and affiliates - 17.6%; and (iii) AON Re, Inc. - 17.2%. During the year ended December 31, 2000, Folksamerica received no more than 10% of its gross reinsurance premiums from any individual ceding company.

LOSS AND LOSS ADJUSTMENT EXPENSE RESERVES

Insurers and reinsurers establish loss and loss adjustment expense reserves representing estimates of future amounts needed to pay claims and related expenses with respect to insured events that have occurred. Loss and loss adjustment expense reserves have two components: case reserves, which are reserves for reported losses, and incurred but not reported ("IBNR") reserves, which are reserves for losses not yet reported. Reserve estimates reflect the judgement of both the ceding company and the reinsurer, based on the experience and knowledge of its claims personnel, regarding the nature and value of the claim. The ceding company may periodically adjust the amount of the case reserves as additional information becomes known or partial payments are made.

Upon notification of a loss from a ceding company, Folksamerica establishes case reserves, including loss adjustment expense reserves, based upon Folksamerica's share of the amount of reserves established by the ceding company and Folksamerica's independent evaluation of the loss. Where appropriate, Folksamerica establishes case reserves in excess of its share of the reserves established by the ceding company.

Folksamerica uses a combination of actuarial methods to determine its IBNR reserves. These methods fall into two general categories: (1) methods by which ultimate claims are estimated based upon historical patterns of paid and reported claim development experienced by Folksamerica, as supplemented by reported industry data, and (2) methods in which the level of Folksamerica's IBNR claim reserves are established based upon the IBNR claim reserves relative to earned premium applied by accident year, line of business and type of reinsurance written by Folksamerica. Due to the inherent uncertainties of estimating claim reserves, actual losses and loss adjustment expenses may deviate, perhaps substantially, from estimates of Folksamerica's reserves reflected in its consolidated financial statements. During the claims settlement period, which may extend over a long period of time, additional facts regarding claims and trends may become known which may cause Folksamerica to adjust its estimates of the ultimate liability. The revised estimates of ultimate liability may prove to be less than or greater than the actual settlement or award amount for which the claim is finally discharged.

REINSURANCE INDUSTRY AND COMPETITION

Folksamerica commenced writing business in 1980 as one of a host of newly formed, foreign-owned reinsurers capitalized with minimum surplus. In 1991, recognizing that surplus size would become an increasingly important business issue, Folksamerica launched an aggressive strategy to increase its resources and capacity through the acquisition of select broker-market reinsurance and property and casualty insurance companies. Since 1991, Folksamerica has acquired several other reinsurers which has served to raise Folksamerica's surplus and contributed a number of important business relationships.

In general, competition among primary companies has caused primary insurers to reduce their own premium writings or restructure their reinsurance programs, thereby reducing the amount of reinsurance they purchase. As a result of consolidation within the industry, many ceding companies are now larger and financially stronger, thereby enabling them to retain more risk. In addition, increasingly intense competition in the reinsurance markets has historically driven reinsurance prices on a number of accounts below pricing levels which Folksamerica will accept. Folksamerica's management believes that the reinsurance industry, including the intermediary market, will continue to undergo further consolidation. Management further believes that, although size and financial strength will continue to be factors in selecting reinsurance partners, product pricing has become the most telling competitive factor.

CONSOLIDATED INSURANCE OPERATIONS

Over the past several years White Mountains has been acquiring and developing various property and casualty insurance operating companies. These companies are described below:

INTERNATIONAL AMERICAN GROUP ("IAG")

In October 1999 White Mountains completed its acquisition of IAG, a collection of insurance companies for \$86.7 million in cash. White Mountains acquired the following subsidiaries through its acquisition of IAG:

PENINSULA INSURANCE COMPANY ("PIC") PIC, which was established in 1960, is a Maryland-domiciled property and casualty insurer which writes both personal and commercial lines, primarily private passenger auto, homeowners, commercial auto and commercial multiple peril. Most of PIC's insurance products are sold in Maryland, Delaware and Virginia. PIC is rated "A" (Excellent) by A.M. Best Company.

In the United States, property and casualty insurance can be obtained through national and regional companies that use an agency distribution system, direct writers, brokers or through self-insurance including the use by corporations of subsidiary captive insurers. PIC markets insurance products principally through independent agents. PIC's primary business focus is to establish strong long-term relationships with its agents and insured customers by focusing on providing quality insurance products to families and small private businesses. PIC pays their independent agents commissions representing negotiated percentages of the premium they write. These commissions, which currently range from 5.0% to 20.0% of premium, depending on the line of business, constitute a significant portion of total acquisition costs and are included in underwriting expenses.

The long-term relationships cultivated by PIC with its agents and insured customers have produced a relatively high level of renewal persistency, particularly in PIC's standard private passenger auto and commercial auto books of business. Renewal persistency can be a significant indicator of an insurance company's long-term prospects for successful underwriting. An insurance company typically incurs more marketing and underwriting costs to write new business (e.g., policies written for new customers) than it does to write "seasoned" business (e.g., policy renewals). Additionally, losses and loss adjustment expenses are typically higher and less predictable for new business than for seasoned business.

The principal competitive factors that affect PIC are: (i) pricing; (ii) underwriting; (iii) quality of claims and policyholder services; (iv) appointing and retaining high quality independent agents; (v) operating efficiencies; (vi) product differentiation and availability; and (vii) increased competition from national direct writers. No single company or group of affiliated companies dominates the insurance industry. The highly competitive environment in the property and casualty insurance market during the past several years has intensified due to increased capacity resulting from growing capital supporting the industry and robust investment returns achieved in recent years. PIC maintains a disciplined approach to pricing and underwriting of insurance risks. Application of this disciplined approach in a highly competitive environment results in a lower volume of insurance premiums than would result from a less disciplined approach, but should produce better overall financial returns from the business over long periods of time.

Selected financial information for PIC is as follows:

	Year Ended December 31,					
Statutory Basis (a) in Millions	2000	1999	1998			
Net written premiums, by line of business:	* 10.0	* * * *	• • • •			
Private passenger auto	\$16.2	\$16.9	\$ 20.2			
Commercial auto	3.5	2.7	2.2			
Commercial multiple peril	1.0	.9	1.0			
Homeowners	.9	1.0	1.0			
Other	1.1	1.0	1.3			
Total net written premiums	\$22.7	\$22.5	\$ 25.7			
Ending total admitted assets	======================================	======================================	\$ 57.3			
Ending policyholders' surplus	\$20.0	\$27.5	\$ 24.2			

(a) The term "statutory" as contained herein refers to a basis of accounting other than generally accepted accounting principles that is prescribed or permitted by the individual states that an insurance company is domiciled in.

AMERICAN CENTENNIAL INSURANCE COMPANY ("ACIC") ACIC is a Delaware-domiciled property and casualty insurance company in run-off. ACIC was incorporated in 1970 for the purpose of underwriting primary and excess liability insurance for many large national and international chemical, manufacturing and pharmaceutical companies, as well as for the purpose of underwriting facultative and treaty reinsurance for the same types of risks. In 1983, in response to the poor profitability of these books of business and substantial difficulties in the collection of its reinsurance recoverables due principally to financial problems of its reinsurers, ACIC stopped actively writing insurance and reinsurance and is currently in run-off. Since 1983, ACIC has concentrated its run-off efforts on commuting its loss exposures with its insureds and on settling the ultimate amount of its reinsurance recoverables with its reinsurers. In 1997, ACIC entered into a retrospective excess of loss reinsurance treaty with a highly rated reinsurer whereby substantially all of its remaining loss exposure has been reinsured. At December 31, 2000 and 1999, ACIC had \$47.1 million and \$51.4 million of total statutory admitted assets, respectively, and \$40.5 million and \$43.2 million of statutory policyholders' surplus, respectively.

BRITISH INSURANCE COMPANY OF CAYMAN ("BICC") BICC is a Cayman Island-domiciled property and casualty insurance company in run-off. BICC was established in 1997 as a means to improve IAG's ability to recover reinsurance recoverables from insolvent or near insolvent international reinsurers. BICC consists principally of certain reinsurance recoverables and loss reserves assumed from ACIC and invested assets. At December 31, 2000 and 1999, BICC had \$22.1 million and \$36.1 million of total assets and \$4.5 million and \$17.2 million of shareholder's equity, respectively.

WATERFORD INSURANCE COMPANY ("WATERFORD")

Waterford is a Kansas-domiciled property and casualty insurance company. Waterford was purchased in 1996 (at which time it was an inactive insurance company) and is licensed to write property and casualty insurance in 48 states. As of December 31, 1999, Waterford ceased writing new business. At December 31, 2000 and 1999, Waterford had \$11.6 million and \$13.3 million of total statutory admitted assets, respectively, and \$11.3 million and \$12.2 million of statutory policyholders' surplus, respectively. Waterford was sold to a third party on January 5, 2001 for cash proceeds of \$23.8 million.

VALLEY GROUP, INC. ("VGI")

In 1995 White Mountains acquired the Valley Insurance Companies ("Valley") of Albany, Oregon and Charter Group, Inc. ("CGI") of Richardson, Texas for \$41.7 million in cash less \$3.0 million of purchase price adjustments. In September 1995 White Mountains formed White Mountains Insurance Company ("WMIC"), a New Hampshire- domiciled mid-size commercial property and casualty company. Valley, CGI and WMIC are collectively referred to herein as "VGI".

In June 1999 White Mountains completed the sale of VGI to Unitrin (the "VGI Sale") and received net proceeds of \$139.0 million in cash after receiving a special dividend prior to the closing of \$76.6 million (net of related tax liabilities) consisting of cash, investment securities and the common stock of Waterford.

INVESTMENTS IN UNCONSOLIDATED INSURANCE AFFILIATES

White Mountains' investments in unconsolidated insurance affiliates represent operating investments in other insurers in which White Mountains has a significant voting and economic interest but does not own more than 50.0% of the entity.

MAIN STREET AMERICA HOLDINGS, INC. ("MSA")

MSA is a subsidiary of National Grange Mutual Insurance Company ("NGM"), a New Hampshire-domiciled property and casualty insurance company, which insures risks located primarily in New York, Massachusetts, Connecticut, Pennsylvania, New Hampshire, Virginia and Florida. NGM's principal lines of business and approximate percentage of total direct written premiums are personal automobile (41.5%), commercial multi-peril (18.0%), homeowners (14.4%) and commercial automobile (13.9%). MSA, through its subsidiaries, participates in NGM's property and casualty business through a reinsurance agreement. MSA's net written premiums totalled \$265.4

million, \$242.7 million and \$258.5 million in 2000, 1999 and 1998, respectively, and its net income was \$3.8 million, \$25.8 million and \$13.4 million, respectively. MSA's total assets as of December 31, 2000 and 1999 were \$608.7 million and \$582.3 million, respectively, and its shareholders' equity was \$253.8 million and \$233.4 million, respectively.

From 1994 to 1997 White Mountains owned 33.1% of the common stock of MSA ("MSA Common Stock"). During that period MSA participated in 40% NGM's property and casualty business through a reinsurance agreement.

In 1998 White Mountains acquired an additional 131,487 shares of MSA Common Stock for \$74.4 million (which includes a purchase price adjustment of \$4.1 million paid to MSA during 2000) which raised White Mountains ownership of MSA to 50.0%. As a result of White Mountains' additional investment in MSA, MSA's reinsurance pooling agreement was increased from 40.0% to 60.0% and NGM contributed certain of its insurance, reinsurance and financial services subsidiaries to MSA. White Mountains' investment in MSA Common Stock is accounted for using the equity method.

FINANCIAL SECURITY ASSURANCE HOLDINGS LTD. ("FSA")

FSA, through its wholly-owned subsidiary, Financial Security Assurance Inc., guarantees scheduled payments of principal and interest on municipal bonds and asset-backed securities, including residential mortgage-backed securities.

On July 5, 2000 White Mountains concluded the sale of its indirect, wholly-owned subsidiary, White Mountains Holdings, Inc. (which controlled a substantial amount of its holdings of FSA) and all its other holdings of the common stock of FSA ("FSA Common Stock") to Dexia S.A. ("Dexia") for proceeds of \$620.4 million (the "Dexia Sale") which resulted in a pretax gain of \$391.2 million.

In 1994 White Mountains purchased 2,000,000 shares of FSA Common Stock from MediaOne Capital Corp. ("MediaOne", formerly U S WEST Capital Corp.), a wholly-owned subsidiary of MediaOne Group, Inc. (formerly U S WEST, Inc.). The purchase was part of an initial public offering of FSA Common Stock at the offering price of \$20.00 per share.

White Mountains also acquired various fixed price options ("FSA Options") and shares of convertible preferred stock ("FSA Preferred Stock") during 1994 which, in total, gave White Mountains the right to acquire up to 4,560,607 additional shares of FSA Common Stock for aggregate consideration of \$125.7 million.

White Mountains purchased an additional 460,200 shares of FSA Common Stock on the open market for \$8.8 million during 1995 and an additional 1,000,000 shares of FSA Common Stock in a private transaction for \$26.5 million during 1996.

In May 1999, White Mountains exercised FSA Options to acquire 666,667 shares of FSA Common Stock for \$15.7 million in cash. In September 1999, White Mountains exercised FSA Options to acquire 1,893,940 shares of FSA Common Stock in exchange for White Mountains' \$50.0 million investment in MediaOne preferred stock.

In December 1999, White Mountains purchased an additional 922,509 shares of FSA Common Stock at a price of 54.20 per share in a private transaction with FSA.

Prior to the Dexia Sale, White Mountains' accounted for its investment in FSA Common Stock using the equity method. White Mountains' accounted for its investment in FSA Preferred Stock and FSA Options under the provisions of Statement of Financial Accounting Standards ("SFAS") No. 115 whereby the investment was reported at fair value as of the balance sheet date, with related unrealized investment gains and losses, after tax, reported as a net amount in a separate component of shareholders' equity and reported on the income statement as a component of other comprehensive net income.

DISCONTINUED OPERATIONS

In 1991 White Mountains sold Fireman's Fund Insurance Company ("Fireman's Fund"), a large property and casualty insurance company, to Allianz of America, Inc. Since 1991 the Company has carried a reserve related to various outstanding tax issues involving the sale. In September 2000, the Company was informed that the Internal Revenue Service agreed with the position taken by White Mountains in its 1991 tax return and, on October 19, 2000, the Company received the Technical Advice Memorandum from the Internal Revenue Service's National Office confirming this conclusion. As a result, the Company released a \$95.0 million reserve during 2000 to income which is presented as a gain from discontinued insurance operations.

In May 1999 White Mountains concluded the sale of substantially all the mortgage banking assets (the "Mortgage Banking Sale") of White Mountains Services Corporation ("WMSC" formerly Source One Mortgage Services Corporation) and received net proceeds totalling \$180.6 million. Mortgage banking assets and liabilities that were not part of the Citibank sale principally included WMSC's investments in derivative instruments (which were fully liquidated in 1999), various residual mortgage assets and preferred stock, each of which were substantially liquidated during 1999. White Mountains recorded a \$17.9 million pretax (\$11.6 million after tax) gain on the sale of its mortgage banking net assets (which is net of anticipated future liabilities) during 1999. During 2000 WMSC was liquidated and all remaining mortgage banking assets were transferred to White Mountains Services L.L.C., a wholly- owned subsidiary of White Mountains. As a result of White Mountains' decision to dispose of its net mortgage banking assets, these activities are shown as discontinued operations herein.

INVESTING OPERATIONS

White Mountains' philosophy is to invest all assets to maximize their after tax total return over extended periods of time. Under this approach, each dollar of after tax investment income, realized gains and unrealized gains is valued equally. Management further believes that the investment assets of its reinsurance and insurance operations should be invested in a "balanced portfolio" consisting of a mixture of fixed income investments, equity securities and occasionally other investments in order to maximize returns over extended periods of time. The Company's Investment Committee, headed by John D. Gillespie and comprised of certain other members of the Company's Board of Directors (the "Board"), key management and investment professionals, oversee the Company's investment activities which are more extensive than in the recent past. The Investment Committee regularly monitors the overall investment results of White Mountains, reviews the results of each of White Mountains' various investment managers, reviews compliance with established investment guidelines, approves all purchases and sales of investment securities and ultimately reports the overall investment results to the Board.

As previously stated, the investment portfolios of White Mountains' reinsurance and insurance operations consist, in part, of common equity securities and other investments. At December 31, 2000, the investment portfolios of White Mountains' reinsurance and insurance operations contained \$181.5 million of common equity securities and other investments which represented approximately 14% of their total portfolio, excluding short-term investments. Management believes that modest investments of common equity securities within its investment portfolio will enhance after tax returns without significantly increasing the risk profile of the portfolio when considered over long periods of time.

REGULATION

White Mountains' insurance and reinsurance operations are subject to regulation and supervision in each of the jurisdictions where they are domiciled and licensed to conduct business. Generally, regulatory authorities have broad supervisory and administrative powers over such matters as licenses, standards of solvency, premium rates, policy forms, investments, security deposits, methods of accounting, form and content of financial statements, reserves for unpaid losses and loss adjustment expenses, reinsurance, minimum capital and surplus requirements, dividends and other distributions to shareholders, periodic examinations and annual and other report filings. Over the last several years most states have, and continue to implement, laws which establish standards for current, as well as continued, state accreditation. In addition, the National Association of Insurance Commissioners ("NAIC") has adopted riskbased capital ("RBC") standards for property and casualty companies as a means of monitoring certain aspects affecting the overall financial condition of insurance companies. The RBC ratios for Folksamerica Reinsurance

Company, PIC, ACIC and Waterford at December 31, 2000 and 1999, were above the levels which would require regulatory action.

White Mountains is not aware of any current recommendations by regulatory authorities that would be expected to have a material effect on its results of operations or liquidity or any other matters that would require disclosure herein.

EMPLOYEES

As of December 31, 2000, White Mountains employed approximately 255 persons (consisting of 10 persons at the Company, 150 persons at Folksamerica, 70 persons at PIC, 15 persons at ACIC and BICC and 10 persons at subsidiary holding companies). Management believes that White Mountains' employee relations are good.

FORWARD-LOOKING STATEMENTS

White Mountains relies upon the safe harbor for forward looking statements provided by the Private Securities Litigation Reform Act of 1995. This safe harbor requires that White Mountains specify important factors that could cause actual results to differ materially from those contained in forward-looking statements made by or on behalf of White Mountains. Accordingly, forward-looking statements by the Company and its affiliates are qualified by reference to the following cautionary statements.

In its filings with the Securities and Exchange Commission (the "SEC"), reports to shareholders, press releases and other written and oral communications, White Mountains from time to time makes forward-looking statements. Such forward-looking statements include, but are not limited to, (i) projections of revenues, income (or loss), earnings (or loss) per share, dividends, market share or other financial forecasts, (ii) statements of plans, objectives or goals of White Mountains or its management, including those related to growth in book value and deferred credit per share or return on equity and (iii) expected losses on, and adequacy of loss reserves for, insurance in force. Words such as "believes", "anticipates", "expects", "intends" and "plans" and similar expressions are intended to identify forwardlooking statements but are not the exclusive means of identifying such statements.

White Mountains cautions that a number of important factors could cause actual results to differ materially from the plans, objectives, expectations, estimates and intentions expressed in forward-looking statements made by White Mountains. These factors include: (i) competitive forces, including the conduct of other property and casualty insurers and reinsurers, (ii) changes in domestic or foreign laws or regulations applicable to White Mountains, its competitors or its clients, (iii) an economic downturn or other economic conditions (such as a rising interest rate environment) adversely affecting White Mountains' financial position, and (iv) loss reserves established by White Mountains subsequently proving to have been inadequate and (v) the failure of pending transactions to be consummated. White Mountains cautions that the foregoing list of important factors is not exhaustive. In any event, such forward-looking statements made by White Mountains speak only as of the date on which they are made, and White Mountains does not undertake any obligation to update or revise such statements as a result of new information, future events or otherwise.

ITEM 2. PROPERTIES

The Company maintains two small professional offices in Hamilton, Bermuda which serve as its headquarters and registered office. In addition, the Company and certain of its subsidiaries lease 8,600 square feet of office space at 80 South Main Street, Hanover, New Hampshire, under a lease expiring in 2006, which serves as its principal executive office. Folksamerica leases 60,000 square feet of office space in New York, New York, under a lease expiring 2009, which serves as its principal office. Folksamerica also leases 12,000 square feet of office space in Greenwich, Connecticut, under a lease expiring in 2010, which serves as its satellite office. PIC owns a 20,000 square foot office building in Salisbury, Maryland which serves as its principal office. ACIC and BICC lease 8,000 square feet of office space in Wilmington, Delaware, under a lease which expires in 2004, which serves as their principal office. White Mountains leases several other office facilities and operating equipment under cancellable and noncancellable agreements.

ITEM 3. LEGAL PROCEEDINGS

White Mountains, in common with the insurance and reinsurance industry in general, is subject to litigation and arbitration in the normal course of its business. As of December 31, 2000, White Mountains was not a party to any material litigation or arbitration other than as routinely encountered in claims activity, none of which is expected by management to have a material adverse effect on the Company's financial condition.

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

There were no matters submitted to a vote of White Mountains' shareholders during the fourth quarter of 2000.

PART II

ITEM 5. MARKET FOR THE COMPANY'S COMMON EQUITY AND RELATED STOCKHOLDER MATTERS

As of March 26, 2001, there were 407 registered holders of common shares of the Company, par value \$1.00 per share ("Shares").

During 2000 and 1999 the Company declared and paid cash dividends totalling \$1.20 and \$1.60 per Share, respectively. Through the third quarter of 2000, dividends were declared and paid on a quarterly basis. In August 2000 the Board announced that, beginning in the 2000 fourth quarter, it would no longer pay quarterly dividends and expected to reduce future dividends to \$1.00 per Share, payable annually in the first quarter of each year, dependent on the Company's financial position and the regularity of its cash flows.

The Company's Shares (symbol WTM) are listed on the New York Stock Exchange (the "NYSE"). The quarterly range of the daily closing price for Shares during 2000 and 1999 is presented below:

HIGH					
		LOW	High	Low	
1040		\$ 222	¢104.1.(0	0110	
	7 /0				1/10
	1/8				
	10/10				3/8 1/2
	162	272 7/8	272 7/8 155 162 126	272 7/8 155 143 162 126 149	2727/8155143130162126149131

Selected consolidated income statement data and ending balance sheet data for each of the five years ended December 31, 2000, follows:

	Year Ended December 31,							
Millions, except per Share amounts	2000(a) 1999(b)	1998(c	:) 1997	1996			
INCOME STATEMENT DATA:	• • • • •	• • • •	• • • • • •	• • • • • •	• 100			
Revenues Expenses	\$ 848 493	\$ 579 418	\$ 390 311	\$293 192	\$ 186 164			
2,461363								
Pretax earnings	355	161	79	101	22			
Income tax provision	(42)	(53)	(28)	(36)	(9)			
Net income from continuing operations	\$ 313	\$ 108	\$ 51	\$ 65	\$ 13			
Not income from continuing operations per Charol	========							
Net income from continuing operations per Share: Basic	\$ 53.08	\$ 19.25	\$ 8.71	\$ 9.88	\$ 1.74			
Diluted	\$ 52.84				\$ 1.59			
ALANCE SHEET DATA:								
Total assets	\$ 3,545	\$2,049	\$2,164	\$ 1,156	\$ 1,120			
Short-term debt	-	4	52	2	2			
ong-term debt	96(d		186	132	133			
Deferred credits	92	101(e)	37(f		-			
Shareholders' equity (g)	1,046	614	703	659	687			
Book value per Share (h)	\$ 177.07	\$103.32	\$109.68	\$100.08	\$ 90.81			
Tangible Book value per Share (h) (i)	\$ 187.65	\$120.23	\$115.11	\$100.08	\$ 90.81			
SHARE DATA:								
ash dividends paid per Share	\$ 1.20	\$ 1.60	\$ 1.60	\$.80	\$.80			
Ending common and equivalent Shares (000's)	5,880	5,946	6,831	6,983	7,908			

(a) Includes the acquisitions of PCA and Risk Capital as well as the gain on the Dexia Sale.

(b) Includes gains resulting from the VGI Sale and the Mortgage Banking Sale.

(C) Includes the interim period income statement and ending balance sheet of

Folksamerica which was consolidated during 1998. Reflects a significant repayment of long-term debt by Folksamerica during (d) 2000. See Note 7.

(e) Deferred credits added during 1999 resulted from the purchase of IAG and exercises of FSA Options. See Note 1.

- Deferred credits added during 1998 resulted from the consolidation of (f) Folksamerica. See Note 1.
- (g) Reflects reductions in shareholders' equity resulting from significant repurchases of Shares from 1996 to 1999.
 (h) As adjusted for the dilutive effects of outstanding options and warrants to
- acquire Shares.
- Book value per Share plus unamortized deferred credits less goodwill per (i) Share.

CONSOLIDATED RESULTS

White Mountains reported net income from continuing operations of \$312.9 million for the year ended December 31, 2000, which compares to \$108.4 million for 1999 and \$51.1 million for 1998. Net income from continuing operations for 2000 and 1999 includes \$385.8 million and \$103.9 million, respectively, in pretax gains from sales of subsidiaries and insurance assets.

Net income was \$407.9 million for the year ended December 31, 2000, which compares to \$121.0 million for 1999 and \$78.5 million for 1998. Net income for 2000, 1999 and 1998 includes income from discontinued operations of \$95.0 million, \$12.6 million and \$27.4 million, respectively.

Comprehensive net income was \$447.6 million for 2000, which compares to \$3.0 million for 1999 and \$69.6 million for 1998. Comprehensive net income for 2000 benefitted by unrealized gains in the fixed maturity investments of the Company's insurance and reinsurance operations. Comprehensive net income for 1999 was adversely affected by unrealized fixed maturity investment losses and an accounting write-down resulting from exercises of FSA Options. Comprehensive net income for 1998 benefitted by modest unrealized gains from both fixed maturity investments and common stocks.

Book value per Share was \$177.07 at December 31, 2000, which compares to \$103.32 at December 31, 1999. Tangible book value per Share (which includes unamortized deferred credits less goodwill per Share) was \$187.65 at December 31, 2000, which compares to \$120.23 at December 31, 1999. White Mountains' unamortized deferred credits and goodwill represent the remaining difference between the cost of companies acquired by White Mountains during the periods presented and the fair value of the net identifiable assets acquired.

INSURANCE OPERATIONS

REINSURANCE OPERATIONS

Folksamerica provided a net loss of \$10.5 million during 2000 versus \$44.9 million of net income during 1999 and \$10.0 million of net income for 1998. Folksamerica's contribution for 1998 included \$4.5 million of net income as an unconsolidated insurance affiliate.

Folksamerica's results for the three years ended December 31, 2000, 1999 and 1998 included \$312.5 million, \$211.0 million and \$238.1 million of earned reinsurance premiums, respectively, and \$275.9 million, \$182.2 million and \$170.3 million of losses and loss adjustment expenses, respectively. For 2000 Folksamerica's statutory combined ratio was 126.3% versus a statutory combined ratio of 122.5% and 108.0% for the comparable 1999 and 1998 periods.

A summary of Folksamerica's 2000, 1999 and 1998 underwriting results follows. Folksamerica's underwriting results presented for 1998 reflects activity for the full calendar year which includes activity prior to Folksamerica becoming a consolidated subsidiary of White Mountains.

	Year	Year Ended December 31				
Dollars in millions	2000	1999	1998			
Net written premiums	\$ 332.6	\$ 201.7	\$ 212.6			
Earned premiums	312.5	211.0	238.1			
Losses and loss adjustment expenses	270.8	182.2	170.3			
Underwriting expenses	120.9	81.4	92.6			
Underwriting loss	\$(79.2)	\$(52.6)	\$(24.8)			
Statutory combined ratios:						
Loss and loss adjustment expense	88.3%	86.5%	71.5%			
Underwriting expense	38.0	36.0	36.5			
Combined	126.3%	122.5%	108.0%			

During 2000 Folksamerica acquired Risk Capital which served to significantly increase its written and earned premiums in that year. During 1999, Folksamerica's written premium volume decreased 5% versus 1998 despite its acquisition of USF Re during mid-1999. The premium volume decrease in 1999 excluding USF Re primarily reflects increased non-renewed business due to deteriorating terms and conditions.

Folksamerica's 2000 statutory combined ratio of 126.3% reflects the effects of \$111.1 million in premiums ceded in the 2000 fourth quarter to a third party insurer coupled with prior year reserve strengthening and increased loss estimates for the current underwriting year. Prior year reserve strengthening of \$22.9 million affected most of Folksamerica's segments including the portfolios acquired with USF Re and Risk Capital. In the fourth quarter of 2000 Folksamerica purchased reinsurance coverage from Imagine Re which was designed to reduce its statutory operating leverage and provide adverse development protection on the run-off of loss and unearned premium reserves acquired in the Risk Capital transaction and any additional premium reported for the 2000 underwriting year, as well as the reserves remaining from the USF Re acquisition and remaining environmental and asbestos exposures. Folksamerica holds a letter of credit and funds held as collateral for amounts due from Imagine Re.

Folksamerica's 1999 statutory combined ratio of 122.5% included approximately \$20.1 million in pretax losses associated with USF Re's loss reserves, \$4.0 million of pretax property catastrophe losses and higher than anticipated asbestos and environmental losses. These significant 1999 adverse loss developments resulted primarily from business acquired through Folksamerica's prior acquisitions (mainly USF Re).

The structure of Folksamerica's acquisitions often provides effective economic protections to offset potential post-acquisition loss development. Folksamerica's statutory combined ratios do not reflect such protections which are recorded at the holding company level. For 2000 and 1999, Folksamerica recorded \$27.8 million and \$20.3 million of net after tax benefits, respectively, resulting from favorable purchase structures which mitigate the impact of future adverse loss development from such acquisitions. These benefits for 2000 consisted of a \$6.8 million after tax reduction in the USF Re seller note, \$10.0 million of various purchase price adjustments and \$11.0 million of net after tax deferred credit amortization. These benefits for 1999 consisted of a \$14.0 million after tax reduction in the USF Re seller note and \$6.3 million of net after tax deferred credit amortization.

Folksamerica's 1998 statutory combined ratio of 108.0% included two property events experienced during the year (Canadian ice storms and Hurricane Georges) and higher than anticipated asbestos and environmental losses.

As previously mentioned, Folksamerica underwrites each reinsurance contract anticipating an element of underwriting profit. The anticipated degree of underwriting profit varies by contract and is based on a variety of factors which can include some degree of float. Despite this expectation on an individual contract basis, Folksamerica's reported results for the years ended December 31, 2000, 1999 and 1998 yielded underwriting losses due to the following: (i) actual results on some accounts or classes have produced higher than anticipated loss costs (considering the highly competitive market conditions, there has been insufficient margin in profitable accounts to absorb higher loss costs produced by other accounts); (ii) higher than anticipated property catastrophe losses and (iii) continued strengthening of reserve portfolios relating to acquired companies. However, as previously mentioned, Folksamerica has various protections into its prior acquisition structures at its holding company which are designed to mitigate such losses but are not recorded in a manner that offsets Folksamerica's underwriting results.

Since reinsurance claims settlement periods generally extend over long periods of time, Folksamerica earns significant amounts of investment income on the float generated by its reinsurance operations. When considering investment income and other comprehensive income items at the Folksamerica holding company level, the Company's reinsurance operations reported comprehensive net income of \$24.2 million, \$13.4 million and \$54.3 million during those periods, respectively. This resulted in an after tax return on Folksamerica's equity of 9.7%, 5.7% and 20.9% for 2000, 1999 and 1998, respectively.

The following table presents the subsequent development of the year-end reinsurance losses for the ten-year period from 1990 to 2000. Section I of the table shows the estimated liabilities that were recorded at the end of each of the indicated years for all current and prior year unpaid losses and loss adjustment expenses ("LAE"). Section II shows the re-estimate of the liabilities made in each succeeding year. Section III shows the cumulative liabilities paid of such previously recorded liabilities. Section IV shows the cumulative deficiency representing the aggregate change in the liability from the original balance sheet dates:

			Reins		ses and Lo ear Ended		ment Expen 31	ses (a)			
ollars in Millions	1990	1991	1992	1993	1994	1995	1996	1997	1998	1999	2000
. Liability for unpaid											
losses and LAE	\$ 431.8	\$ 506.3	\$ 758.3	\$ 798.4	\$ 856.2	\$ 981.5	\$1,578.7	\$1,461.3	\$1,437.6	\$1,210.7	\$1,500.
I. Liability re-estimat as of:	ed										
1 year later	466.3	543.4	804.6	850.4	928.4	1,044.3	1,599.7	1,560.9	1,435.8	1,289.8	
2 years later	494.3	542.9	843.0	910.3	979.3	1,074.4	1,675.5	1,532.3	1,451.0		
3 years later	489.2	567.6	882.9	951.5	1,002.4	1,090.7	1,652.6	1,526.1	,		
4 years later	508.0	589.2	915.6	975.5	1,015.3	1,104.3	1,646.3	,			
5 years later	522.9	610.2	943.0	989.7		1,091.1	,				
6 years later	539.1	623.0	956.3	1,001.2	1,020.9						
7 years later	551.3	630.6	964.6	995.9							
8 years later	559.3	637.1	962.1								
9 years later	567.1	635.7									
10 years later	566.7										
II. Cumulative amount o	 f										
liability paid thro	uah:										
1 year later	113.0	136.7	244.7	263.1	253.1	273.4	396.5	343.5	382.8	128.9	
2 years later	179.8	202.6	400.3	406.7	411.7	435.7	632.0	597.6	511.5		
3 years later	219.8	290.9	494.2	515.4	522.5	548.1	833.9	733.3			
4 years later	286.3	343.4	572.1	593.2	600.0	655.2	971.2				
5 years later	322.5	382.1	629.4	646.5	674.2	719.4					
6 years later	352.3	415.8	668.7	706.9	713.3						
7 years later	377.4	439.1	718.7	734.1							
8 years later	394.4	468.7	738.5								
9 years later	421.1	488.4									
10 years later	439.1										
V.Cumulative deficiency	\$134.9	\$ 129.4	\$ 203.8	\$ 197.5	\$ 164.7	\$ 109.6	\$ 67.6	\$ 64.8	\$ 13.4	\$ 79.1 \$	
Percent deficient	31%	26%	27%	25%		11%			1%		

(a) For the years 1990 and 1991 liabilities are shown net of reinsurance recoverable. For the years 1992 through 2000 liabilities are shown without regard to reinsurance recoverable in accordance with SFAS No. 113. The table excludes the insurance operations of VGI and IAG whose liability for unpaid losses and LAE totalled \$55.6 million, \$68.9 million, \$88.5 million, \$71.9 million and \$65.4 million as of December 31, 2000, 1999, 1998, 1997 and 1996, respectively.

The table above has been prepared in accordance with prescribed instructions, however, management believes that this information is not indicative of Folksamerica's actual loss development history for the following reasons: (i) with respect to 1992 through 2000, the information is presented prior to considering the benefit of significant amounts of ceded reinsurance recovered and recoverable from Folksamerica's reinsurers; (ii) the information includes the complete loss development history (whether favorable or unfavorable) for companies acquired by Folksamerica for all periods presented, including periods prior to Folksamerica's acquisition of such companies; and (iii) as previously described, the structure of Folksamerica's acquisitions often provides effective economic protections to offset potential post-acquisition loss development. In consideration of each of these factors, the table presented below is management's attempt to adjust the cumulative deficiencies presented above for the most recent five years:

	Year I	Ended December	31,		
Percent of deficit to carried reserves:	1996	1997	1998	1999	2000
Deficiency as reported Deficiency as adjusted for the effects described above	4% 2%	4% 1%	1% 2%	7% 2%	- % - %

INSURANCE OPERATIONS

On October 15, 1999, the Company concluded its acquisition of the IAG companies (PIC, ACIC and BICC) for \$86.7 million in cash. Because the cost of these companies was less than the fair value of their net identifiable assets at October 15, 1999, the Company recorded a \$62.0 million deferred credit (\$37.0 million and \$57.7 million as of December 31, 2000 and 1999, respectively) that will be amortized to income over the estimated period of benefit of three years.

For the year ended December 31, 2000, PIC, ACIC and BICC provided \$5.8 million of net income which resulted principally from favorable recoveries on prior years' reinsurance at BICC. For the period from October 15, 1999 to December 31, 1999, PIC, ACIC and BICC provided a \$3.6 million net loss which was due primarily to strengthening of prior year reserves at ACIC.

On June 17, 1999, the Company completed the VGI Sale and recorded a pretax gain of \$88.1 million (\$53.8 million after tax) on the transaction. As part of the VGI Sale, White Mountains has provided Unitrin, Inc. with certain adverse loss development protections that will be settled as of December 31, 2002. During 2000 White Mountains provided \$5.4 million in reserves for such adverse loss development protections.

During 2000, Waterford provided a \$1.2 million net loss which principally reflects goodwill amortization related to its purchase in 1996. For the 1999 period through the date of VGI Sale, the operations of VGI (which included Waterford) provided \$3.6 million of net income which primarily represented realized investment gains during the period. For the year ended December 31, 1998 the operations of VGI (which included Waterford) contributed \$5.0 million to net income.

INVESTMENT IN UNCONSOLIDATED INSURANCE AFFILIATE(S)

MSA contributed \$1.0 million to net income during the year ended December 31, 2000 versus \$9.8 million for 1999 and \$3.2 million for 1998. MSA's net income for 2000 was adversely impacted by significant realized losses on its investment portfolio whereas MSA's net income for 1999 benefitted by significant realized investment gains.

During 2000 FSA provided a \$3.6 million net loss for the interim period though July 5, 2000 which resulted from realized investment losses in FSA's investment portfolio and increased expenses for employee equity-based compensation programs, which rose significantly after FSA's announcement of its acquisition by Dexia. FSA contributed \$15.7 million to net income during the year ended December 31, 1999 versus \$9.3 million during 1998. The significant increase in FSA-related net income during 1999 resulted from increased equity in earnings resulting from the additional purchase of \$50.0 million of FSA Common Stock and the exercise of FSA Options, each occurring during 1999.

INVESTMENT OPERATIONS

Net realized gains on investments and the total net investment return from White Mountains' investment activities (excluding net unrealized investment holding gains from White Mountains' investments in unconsolidated insurance affiliates) are shown below:

Millions	Ye	ar Ended December	31,
	2000	1999	1998
Net realized investment gains (losses), before tax	\$(8.8)	\$ 69.6	\$ 71.0
Net investment income	\$ 85.9	\$ 61.9	\$ 36.8
Net change in unrealized investment gains (losses) for investments held	65.1	(11.7)	25.0
Total net investment return, before tax	\$151.0	\$ 50.2	\$ 61.8

Net realized investment losses for the year ended December 31, 2000 resulted principally from sales of fixed maturities in preparation for Folksamerica's acquisitions of PCA and Risk Capital during the first quarter of 2000. Net realized investment gains of \$69.6 million for 1999 included \$23.9 million of pretax gains from sales of San Juan Basin Royalty Trust units and \$28.0 million of pretax gains from sales of various other common stocks and other investments in Folksamerica's operating portfolio. In addition, \$9.4 million of pretax gains on sales of common stocks and fixed maturity investments were recorded in anticipation of or in connection with the VGI Sale. Net realized investment gains of \$71.0 million for 1998 resulted principally from the sale of White Mountains' investment in White River Corporation.

White Mountains' net investment income is comprised primarily of interest income associated with the fixed maturity investments of its consolidated insurance and reinsurance operations and dividend income from its equity investments. The significant increase in net investment income from 1998 to 2000 is mainly attributable to White Mountains' growing portfolio of fixed maturity investments resulting from the consolidation of Folksamerica in August 1998, the 1999 acquisition of USF Re and the 2000 acquisitions of PCA and Risk Capital.

Net unrealized gains for investments held during 2000 of \$65.1 million resulted principally from unrealized gains on fixed maturity investments whereby net unrealized investment losses for investments held during 1999 of \$11.7 million resulted principally from losses on fixed maturity investments. The fluctuations in net investment gains from fixed maturity investments from 1999 to 2000 principally resulted from the effects of changes in market interest rates on Folksamerica's sizable fixed income portfolio. Net unrealized gains on both fixed maturity investments and common stocks.

EXPENSES

Insurance losses and loss adjustment expenses totalled \$287.7 million for 2000 versus \$242.3 million for 1999 and \$174.8 million for 1998. During 2000, 1999 and 1998, losses and loss adjustment expenses relating to prior years developed unfavorably by \$23.6 million, \$31.9 million and \$7.8 million, respectively. Insurance and reinsurance acquisition expenses totalled \$101.1 million for 2000 versus \$73.4 million for 1999 and \$54.8 million for 1998. The increase in these insurance expenses from 1999 to 2000 is primarily attributable to the acquisitions of PCA and Risk Capital during 2000. The increase in these insurance argues to 1999 is primarily attributable to the acquisitions of USF Re and IAG during 1999.

Compensation and benefits totalled \$59.5 million for 2000 versus \$67.8 million for 1999 and \$51.5 million for 1998. Compensation and benefits expenses for 2000 include increased Share-based contingent compensation accruals resulting from the attainment of above-plan results as well as a significant increase in the market value of Shares during the year. The increase in compensation and benefits from 1998 to 1999 is due both to the inclusion of Folksamerica in the Company's consolidated results for the entire 1999 period and expenses incurred in connection with the Redomestication. See "Liquidity and Capital Resources".

General expenses totalled \$28.4 million for 2000 versus \$19.5 million for 1999 and \$15.9 million for 1998. The increase in general expenses during 2000 is primarily attributable to the acquisitions of PCA and Risk Capital as well as various contingencies and expenditures associated with certain of the Company's acquisition and disposition activities during the period. The increase in general expenses during the 1999 period are primarily attributable to both the inclusion of Folksamerica in the Company's consolidated results for the entire 1999 period and expenses incurred in connection with the Redomestication.

Interest expense totalled \$16.1 million for 2000 versus \$14.7 million for 1999 and \$13.7 million for 1998. The increase in interest expense from 1999 to 2000 reflects higher average levels of indebtedness at Folksamerica for the period. The increase in interest expense from 1998 to 1999 reflects higher average levels of indebtedness at Folksamerica for the period, partially offset by the repayment of \$15.0 million of indebtedness in May 1999 in connection with the VGI Sale.

INCOME TAXES

In connection with the Redomestication, the Company and certain of its subsidiaries changed their domicile to either Bermuda or Barbados (the "Offshore Companies") while certain other subsidiaries remained domiciled in the United States (the "Onshore Companies"). As a result, income earned by the Offshore Companies will generally be subject to an effective overall tax rate lower than that imposed by the United States, however, no tax benefits will be attained in the event of net losses incurred by such companies. Prior to the Redomestication, the Company filed a consolidated United States income tax return with its subsidiaries. The Onshore Companies continue to file United States tax returns but may no longer do so on a group-wide consolidated basis. As a result, the aggregate United States Federal income tax liability of the Onshore Companies may be higher than it otherwise would have been if part of a consolidated tax return. These factors may serve to increase or decrease White Mountains' effective tax rate for 1999 and beyond, depending on the events and circumstances occurring during such periods.

The income tax provision related to pretax earnings for 2000, 1999 and 1998 represents an effective tax rate of 12.0%, 32.9% and 35.8%, respectively. The reduction in the effective rate for 1999 and 2000 resulted from an increase in White Mountains' non-United States net earnings to \$395.9 million in 2000 versus \$9.0 million in 1999.

Deferred income taxes reflect the net tax effects of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts received for tax purposes. White Mountains recorded a deferred tax asset of \$105.1 million (relating primarily to discounting of insurance loss reserves and net operating loss carryforwards) and a deferred tax liability of \$2.8 million on its balance sheet as of December 31, 2000. White Mountains recorded a deferred tax asset of \$52.5 million (relating primarily to discounting of insurance loss reserves) and a deferred tax liability of \$37.5 million (relating primarily to unrealized investment gains) on its balance sheet as of December 31, 1999. Deferred income tax assets and liabilities are shown net in circumstances where a consolidated income tax return is filed.

In 1991, White Mountains sold Fireman's Fund to Allianz of America, Inc. Since 1991 the Company had carried a reserve related to various outstanding tax issues involving the sale. In September 2000, the Company was informed that the Internal Revenue Service agreed with the position taken by White Mountains in its 1991 tax return. On October 19, 2000, the Company received the Technical Advice Memorandum from the Internal Revenue Service's National Office confirming this conclusion. As a result, the Company released a \$95.0 million reserve during 2000 which represents a gain from discontinued insurance operations.

LIQUIDITY AND CAPITAL RESOURCES

THE COMPANY, INSURANCE OPERATIONS AND OTHER

The primary sources of cash inflows for the Company are investment income, sales of investment securities and dividends received from its operating subsidiaries. Under the insurance laws of the states and countries under which the Company's insurance subsidiaries are domiciled, an insurer is restricted with respect to the amount of dividends it may pay without prior approval by regulatory authorities. Accordingly, there is no assurance regarding the amount of such dividends that may be paid by such subsidiaries in the future.

During 1993 the Company issued \$150.0 million in principal amount of medium-term notes. The Company has repurchased certain of its medium-term notes from time to time and during 2000 and 1999 repurchased \$.1 million and \$15.9 million in principal amount of the notes due in February 2003, respectively. At December 31, 2000 the Company had \$96.3 million in principal amount of medium-term notes outstanding which had an average maturity of 2.6 years and an average yield to maturity of 7.87%.

The Company has historically secured revolving credit agreements whereby it could borrow funds at short-term market interest rates. As of December 31, 1999 the Company could borrow up to \$50.0 million under a short-term debt facility which the Company let expire in 2000. There were no borrowings outstanding under this facility at December 31, 1999.

During 2000, 1999 and 1998 the Company repurchased 65,838 Shares for \$8.3 million, 1,020,150 Shares for \$139.5 million and 151,916 Shares for \$19.8 million, respectively. In conformance with Bermuda law, the Company retires all Shares repurchased. During 2000 the Company declared and paid cash dividends totalling \$1.20 per Share. During 1999 and 1998 the Company declared and paid cash dividends totalling \$1.60 per Share. Shares repurchased and dividends paid from 1998 to 2000 represented returns of excess capital to shareholders. White Mountains had typically paid its dividends quarterly but in August 2000 announced that it would pay future dividends annually at an expected rate of \$1.00 per Share.

On July 5, 2000 White Mountains concluded the sale of its indirect, wholly-owned subsidiary, White Mountains Holdings, Inc. (which controlled a substantial amount of its holdings of FSA) and all its other holdings of FSA Common Stock to Dexia for proceeds of \$620.4 million.

In May 1999, White Mountains exercised FSA Options to acquire 666,667 shares of FSA Common Stock for \$15.7 million in cash. In September 1999, White Mountains exercised FSA Options to acquire 1,893,940 shares of FSA Common Stock in exchange for White Mountains' \$50.0 million investment in MediaOne preferred stock.

During 1999 White Mountains purchased an additional 922,509 shares of the common stock of FSA at a price of \$54.20 per share. The transaction was part of a private offering by FSA pursuant to which it sold a total of \$140.0 million of its common stock to White Mountains, XL Capital, Ltd, The Tokio Marine and Fire Insurance Co., Ltd and an FSA management group.

During 1999 White Mountains concluded the Mortgage Banking Sale and recorded an 11.6 million after tax gain on the sale.

During 1999 White Mountains concluded the VGI Sale and received net proceeds of \$139.0 million in cash after receiving a special dividend of assets and cash prior to the closing of \$76.6 million. In connection with the VGI Sale, White Mountains repaid \$15.0 million of indebtedness at VGI during 1999.

During 1999 the Company acquired IAG for \$86.7 million in cash.

During 1999 the Company issued a total of 1,137,495 common shares to its Chairman and its key employees in satisfaction of the Chairman's warrant exercise and various employee benefit plan obligations. In order to entice the Chairman to exercise his warrants to acquire Shares early, the Company paid the Chairman \$6.0 million to compensate him for the estimated interest cost of borrowing the strike price and the amounts required to prematurely pay his income taxes.

In connection with the Redomestication, White Mountains paid \$104.1 million in certain compensation benefits to its current and former employees and Directors in October 1999 at an incremental after tax cost of \$14.9 million. In connection with the compensation payments, White Mountains paid cash of \$89.8 million (primarily to its former employees) and issued \$14.3 million in Shares (primarily to its current employees, Directors and advisors). A significant portion of such compensation paid represented the acceleration of expenses that would have ordinarily been incurred in future periods which resulted in increased tax deductible expenses in 1999.

In connection with the Redomestication, the Company was treated as if it sold all of its directly owned assets in a fully taxable transaction in which gains, but not losses, were recognized. The Company incurred a United States income tax liability upon the Redomestication of approximately \$13.5 million.

In September 2000 White Mountains entered into a definitive agreement with CGNU to purchase its U.S. property and casualty insurance operations, CGU. The CGU purchase agreement, which was amended on October 15, 2000 and February 20, 2001, calls for a purchase price of \$2.17 billion, subject to certain adjustments, of which \$260.0 million will consist of a note payable (which must be paid in eighteen months in cash or Shares valued at \$245.00 per Share, at the Company's sole option). In addition, CGU will repay approximately \$.5 billion of debt outstanding to its parent at closing.

In connection with the CGU transaction, White Mountains has entered into an \$875.0 million Credit Agreement (the "Credit Agreement") with Lehman Brothers Inc. ("Lehman") as well as arranging for up to \$741.0 million of new equity commitments from a small group of outside investors. In connection with financing the transaction, White Mountains will contribute Folksamerica, PIC and MSA to CGU.

Completion of the CGU acquisition is subject to, among other matters, the receipt of regulatory approvals, the completion of financing and the satisfaction of other customary conditions. The stock purchase agreement dated September 25, 2000 and subsequent amendments thereto, terms of the note payable to CGNU (as amended on February 20, 2001), terms of the various equity commitments and the Credit Agreement are contained herein as Exhibits 10(a) through 10(f), and are incorporated by reference in their entirety.

On March 15, 2001 White Mountains commenced a cash tender offer (the "Note Offer") for its \$96.3 million aggregate principal amount of outstanding medium-term notes. In conjunction with the Note Offer, noteholder consents are being solicited to effect an amendment to the indenture governing these notes, which will facilitate White Mountains' pending acquisition of CGU. The Note Offer is scheduled to expire at 5:00 p.m., New York City time on April 16, 2001, unless extended or terminated.

The total consideration to be paid for each validly tendered and consented note due in 2003 will be based upon a fixed spread of 95 basis points over the yield to maturity on the applicable reference U.S. Treasury Note, and includes a consent payment of \$30.00 per \$1,000 principal amount of such notes. The total consideration to be paid for each validly tendered and consented note due 2008 will be based upon a fixed spread of 125 basis points over the yield to maturity on the applicable reference U.S. Treasury Note, and includes a consent payment of \$30.00 per \$1,000 principal amount of such notes. The Note Offer pricing date is expected to be April 11, 2001. Accrued and unpaid interest up to, but not including, the payment date will be paid for notes validly tendered and accepted for purchase.

The Note Offer is conditioned upon, among other things, (i) there being validly tendered and not withdrawn not less than a majority in aggregate principal amount of outstanding notes, (ii) the execution of the supplemental indenture governing the notes and (iii) the consummation of White Mountains' acquisition of CGU.

REINSURANCE OPERATIONS

Under the insurance laws of New York an insurer is restricted with respect to the amount of dividends it may pay without prior approval by state regulatory authorities. Accordingly, there is no assurance regarding the amount of such dividends that may be paid by Folksamerica Reinsurance Company in the future.

During the 2000 fourth quarter White Mountains provided \$259.6 million of capital to Folksamerica through the contribution of ACIC and BICC and through the issuance of a \$195.0 million inter-company note. Folksamerica subsequently contributed ACIC and \$80.0 million of such cash to Folksamerica Reinsurance Company in order to provide the statutory capital needed to support its recent acquisitions of PCA and Risk Capital. The remaining \$115.0 million was used by Folksamerica to repay its outstanding bank indebtedness. White Mountains intends to forgive the inter-company note at the closing of the CGU acquisition.

As part of the Folksamerica acquisition in 1998, White Mountains agreed to repay or refinance Folksamerica's existing long-term indebtedness by utilizing a six-year revolving credit agreement whereby Folksamerica could borrow up to \$120.0 million at market interest rates. This facility was repaid and terminated by Folksamerica during 2000 as described above.

In March 2000 Folksamerica acquired PCA for \$122.3 million in cash.

In May 2000 Folksamerica acquired Risk Capital for $20.3\ million$ in cash plus related expenses.

In 1999 Folksamerica acquired USF Re for total consideration of \$92.5 million. The purchase consideration included the issuance of a \$20.8 million, five-year note by Folksamerica (which has been reduced to zero at year-end 2000 due to adverse loss development at USF Re post acquisition) with the balance paid in cash.

MARKET RISK

White Mountains' consolidated balance sheet includes a substantial amount of assets and liabilities whose fair values are subject to market risk. The term market risk refers to the risk of loss arising from adverse changes in: interest rates, foreign currency exchange rates and other relevant market rates and prices such as prices for common equity securities. Due to White Mountains' sizable investments in fixed maturity investments and common equity securities and its use of medium-term and long-term debt financing, market risk can have a significant effect on White Mountains' consolidated financial position.

INTEREST RATE RISK

FIXED MATURITY PORTFOLIO. In connection with the Company's consolidated insurance and reinsurance subsidiaries, White Mountains invests in interest rate sensitive securities, primarily debt securities. White Mountains' strategy is to purchase fixed maturity investments that are attractively priced in relation to perceived credit risks. White Mountains' investments in fixed maturity investments are held as available for sale and, accordingly, White Mountains accepts that realized and unrealized losses on these instruments may occur. White Mountains does not use derivative securities to manage its interest rate risk associated with its fixed maturity investments, rather it manages the average duration of the fixed maturity portfolio in the anticipation of achieving an adequate yield without subjecting the portfolio to an unreasonable level of interest rate risk.

Increases and decreases in prevailing interest rates generally translate into decreases and increases in fair values of fixed maturity investments, respectively. Additionally, fair values of interest rate sensitive instruments may be affected by the credit worthiness of the issuer, prepayment options, relative values of alternative investments, the liquidity of the instrument and other general market conditions. These investments are carried at fair value on the balance sheet with unrealized gains reported net of tax in a separate component of shareholders equity.

The Company's fixed maturity portfolio is comprised of primarily investment grade corporate securities, U.S. government and agency securities, municipal obligations and mortgage-backed securities. Based on ratings by the NAIC, 100% of the Company's total fixed maturity portfolio at December 31, 2000 is considered to be investment grade (i.e., received a rating from the NAIC of 1 or 2).

INDEBTEDNESS. White Mountains utilizes debt financing at many levels of its businesses. Increases and decreases in prevailing interest rates generally translate into decreases and increases in fair values of fixed rate indebtedness, respectively, particularly long-term debt. Additionally, fair values of interest rate sensitive instruments may be affected by the credit worthiness of the issuer, prepayment options, relative values of alternative investments, the liquidity of the instrument and other general market conditions.

The table below summarizes the estimated effects of hypothetical increases and decreases in market interest rates on White Mountains' fixed maturity portfolio and long-term fixed rate indebtedness outstanding. Significant variations in market interest rates could produce changes in the timing of repayments due to prepayment options available to the issuer or the holder which are not reflected herein. It is assumed that the changes occur immediately and uniformly to each category of instrument containing interest rate risk.

Dollars in Millions	Fair Value at December 31, 2000	Assumed Change in Interest Rate	Estimated Fair Value after Change in Interest Rate	Percentage Increase (Decrease) to Shareholders' Equity
Fixed maturity investments	\$1,078.6	50 bp decrease 50 bp increase 100 bp increase 200 bp increase	\$ 1,104.2 1,053.9 1,029.8 983.6	1.6 % (1.5) (3.0) (5.9)
Fixed rate indebtedness	\$97.2	50 bp decrease 50 bp increase 100 bp increase 200 bp increase	\$ 98.3 96.2 95.1 93.0	(.1) % .1 .2 .4

FOREIGN CURRENCY EXCHANGE RATES

Folksamerica operates a branch office in Toronto, Canada to service its Canadian customers and a portion of BICC's premiums are denominated in a foreign currencies. Net unrealized foreign currency translation gains and losses associated with Folksamerica and BICC are reported, after tax, as a net amount in a separate component of shareholders' equity. Changes in the values of these operations due to currency fluctuations, after tax, are reported on the income statement as a component of other comprehensive net income. At December 31, 2000 and 1999, Folksamerica's and BICC's net assets denominated in foreign currency represented less than one percent of the Company's shareholders' equity, therefore, any significant change in foreign currency rates would not have a material impact on White Mountains' financial position.

EQUITY PRICE RISK

The carrying values of White Mountains' common equity securities and a significant portion of its other investments (primarily partnership interests invested in common equity securities) are based on quoted market prices or management's estimates of fair value (which is based, in part, on quoted market prices) as of the balance sheet date. Market prices of common equity securities are subject to fluctuations which could cause the amount to be realized upon sale of the investment to differ significantly from the current reported value. The fluctuations may result from perceived changes in the underlying economic characteristics of the investee, the relative price of alternative investments, general market conditions and supply and demand imbalances for a particular security.

The table below summarizes White Mountains' equity price risks as of December 31, 2000 and shows the effects of a hypothetical 20% increase and a 20% decrease in market prices as of that date.

Dollars in Millions	Fair Value at December 31, 2000	Assumed Price Change	Estimated Fair Value after Assumed Price Change	Percentage Increase (Decrease) to Shareholders' Equity
Common equity securities Other investments	\$144.8 \$142.9	20% increase 20% decrease 20% increase	\$ 173.8 \$ 115.9 \$ 171.5	2.0% (2.0)% 2.3%
=======================================	¥142.9	20% decrease	\$ 114.3	(2.3)%

OTHER MATTERS

ACCOUNTING FOR FSA OPTIONS AND FSA PREFERRED STOCK

White Mountains accounted for its investment in FSA Common Stock on the equity method of accounting and accounted for its unexercised stock options and convertible securities to acquire FSA Common Stock at fair value. Upon the exercises of FSA Options during 1999, the Company was required to write its investments in the FSA Options exercised to their original cost in order to transition the investment from fair value accounting to equity accounting. In connection with this accounting transition, the Company reduced its after tax net unrealized gains at the time of exercise by \$39.3 million and recorded a deferred credit of \$14.2 million that was to be amortized to income over a five-year period. In July 2000 White Mountains sold its investments in FSA as part of the Dexia Sale which resulted in the immediate recognition of the related unamortized deferred credit balance.

RETIREMENT OF COMMON SHARES HELD IN TREASURY

In conformance with Bermuda law, the Company retired all Shares held in its treasury at October 1999 and has retired all Shares repurchased thereafter. The retirement of treasury shares in 1999 resulted in a significant reclassification of several of the Company's various shareholders' equity accounts but did not affect total shareholders' equity.

YEAR 2000 UPDATE

Neither White Mountains nor any of its unconsolidated insurance affiliates experienced any significant Year 2000 disruptions to its business operations. White Mountains' total pretax cost of its Year 2000 remediation, excluding its unconsolidated insurance affiliates, was approximately \$3.0 million.

ITEM 7A. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

See "Market Risk Disclosures" contained in Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations.

ITEM 8. FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA

The financial statements and supplementary data have been filed as a part of this Annual Report on Form 10-K as indicated in the Index to Financial Statements and Financial Statement Schedules appearing on page 34 of this report.

ITEM 9. CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE

Effective with the 1999 audit, the Audit Committee of the Board appointed PricewaterhouseCoopers LLP ("PwC") as its independent auditors to succeed KPMG LLP ("KPMG"). PwC served as Folksamerica's independent auditors since 1981 and served as FSA's independent auditors since 1989. The Audit Committee recommended that PwC succeed KPMG as the Company's independent auditors in 1999 due to the significance of Folksamerica and FSA to the Company's financial position and results of operations. In connection with the 1998 audit, there were no disagreements with KPMG on any matter of accounting principles or practices, financial statement disclosure, or auditing scope and procedures which, if not resolved to their satisfaction, would have caused them to make reference in connection with their opinion to the subject matter of the disagreement. The Company requested KPMG to furnish a letter addressed to the SEC stating whether it agrees with the above statements. A copy of this letter, dated March 25, 1999, is contained herein as Exhibit 16.

PART III

ITEM 10. DIRECTORS AND EXECUTIVE OFFICERS

DIRECTORS (AS OF MARCH 26, 2001)

Name	Class	Term Ending	Age	Director Since
Raymond Barrette	III	2003	50	2000
John J. "Jack" Byrne	II	2002	68	1985
Patrick M. Byrne	I	2001	38	1997
Howard L. Clark, Jr.	III	2003	57	1986
Robert P. Cochran	III	2003	51	1994
Steven E. Fass	I	2001	55	2000
George J. Gillespie, III	II	2002	70	1986
John D. Gillespie	II	2002	42	1999
K. Thomas Kemp	I	2001	60	1994
Gordon S. Macklin	I	2001	72	1987
Frank A. Olson	II	2002	68	1996
Arthur Zankel	III	2003	69	1992

An outline of the principal occupation, business experience, recent business activities involving White Mountains and other affiliations of the Directors follows:

MR. BARRETTE was appointed President of the Company in January 2000 and became a Director in February 2000. He joined the Company in 1997 as its Executive Vice President and Chief Financial Officer. He was formerly a consultant with Tillinghast-Towers Perrin from 1994 to 1996 and was with Fireman's Fund from 1973 to 1993. Mr. Barrette is also a director of Folksamerica, PIC, ACIC and BICC.

MR. JACK BYRNE was re-appointed Chief Executive Officer of the Company in January 2000 after a brief retirement. He has served as Chairman of the Board of the Company since 1985 and formerly served as President and Chief Executive Officer from 1990 to 1997, and as Chief Executive Officer from 1985 to 1990. Mr. Byrne is also a director of Markel Corporation and Folksamerica.

MR. PATRICK BYRNE has been a director of the Company since 1997. Mr. Byrne serves as President and CEO of Overstock.com, an internet shopping service. Mr. Byrne formerly served as President and CEO of Fecheimer Bros. Co. (a wholly-owned subsidiary of Berkshire Hathaway Inc.), a manufacturer of uniforms and accessories, from 1997 to 1999 and President and CEO of Centricut, LLC, a manufacturer of industrial torch consumable parts, from 1994 to 1999. In addition, since 1991, Mr. Byrne has been the managing general partner of a number of limited partnerships investing in real estate, gaming, insurance and international trade. Mr. Byrne is the son of Mr. Jack Byrne.

MR. CLARK has been a director or advisor to the board since 1986. He is currently Vice Chairman of Lehman and was Chairman and CEO of Shearson Lehman Brothers Inc. from 1990 to 1993. Prior to joining Shearson Lehman Brothers Inc., Mr. Clark was Executive Vice President and Chief Financial Officer of American Express. He is also a director of Lehman, Maytag Corporation, H Power Corp. and Walter Industries, Inc. Lehman provided services to White Mountains during 2000. See "Item 13. Certain Relationships and Related Transactions".

MR. COCHRAN has been a director of the Company since 1994. Mr. Cochran was a founding principal of FSA and has served FSA in various capacities since 1985. He has been President and CEO and a director of FSA since 1990 and became Chairman in 1997. He is also Chairman of Financial Security Assurance Inc. and Financial Security Assurance (U.K.) Ltd.

MR. FASS was appointed to the Board in 2000. Mr. Fass has served as a Director, President and Chief Executive Officer of Folksamerica and its subsidiaries including Folksamerica Reinsurance Company since 1984. He joined Folksamerica as its Vice President, Treasurer and Chief Financial Officer in 1980. Mr. Fass is also a director of Esurance.

MR. GEORGE GILLESPIE has been a director of the Company since 1986. Mr. Gillespie has been a Partner in the law firm of Cravath, Swaine & Moore ("Cravath") since 1963. He is also a director of The Washington Post Company. Cravath provided legal services to White Mountains during 2000. See "Item 13. Certain Relationships and Related Transactions". Mr. Gillespie's son, Mr. John Gillespie, is also a director of the Company.

MR. JOHN GILLESPIE was appointed to the board in August 1999. He is the founder and Managing Partner of his own investment firm, Prospector Partners, LLC, in Hartford, Connecticut. Prior to forming Prospector Partners, Mr. Gillespie was President of the T. Rowe Price Growth Stock Fund and the New Age Media Fund, Inc. White Mountains owns limited partnership investment interests which were managed by Mr. Gillespie during 2000. See "Item 13. Certain Relationships and Related Transactions". Mr. Gillespie's father, Mr. George Gillespie, is also a director of the Company.

MR. KEMP has served as Deputy Chairman of the Company since 2000 and has been a director since 1994. Mr. Kemp served as the Company's President and CEO from 1997 to 2000 and served as Executive Vice President from 1993 to 1997, Vice President, Treasurer and Secretary from 1991 to 1993 and was formerly a Vice President of Fireman's Fund. Mr. Kemp is also a director of MSA, Eldorado Bancshares, Inc. and Amlin plc.

MR. MACKLIN has been a director of the Company since 1987. Mr. Macklin is currently a corporate financial advisor. Mr. Macklin formerly served as Chairman of White River Corporation, an information services company, from 1993 to 1998, as Chairman of Hambrecht and Quist Group, a venture capital and investment banking company, from 1987 until 1992, and as President of the National Association of Securities Dealers, Inc. from 1970 until 1987. He is a director of MCI Worldcom, Inc., Martek Biosciences Corporation, MedImmune Inc., Overstock.com and Spacehab, Inc., and is a trustee, director or managing general partner (as the case may be) of 47 of the investment companies in the Franklin Templeton Group of Funds.

MR. OLSON has been a director of the Company since 1996. He serves as Chairman of The Hertz Corporation ("Hertz"). Mr. Olson served as the CEO of Hertz from 1977 to 1999 and has been with that company since 1964. He is also a director of Becton Dickinson and Company, Amerada Hess Corporation and Warnaco Group, Inc. and was formerly Chairman and CEO of Allegis Corporation and United Airlines.

MR. ZANKEL has been a director or advisor to the board since 1992. He served as a General Partner of First Manhattan Co. from 1965 to 1999 and was Co-Managing Partner of First Manhattan from 1979 to 1997. Mr. Zankel is currently Managing Member of Zankel Capital Advisors, LLC which provided investment services to the Company during 2000. See "Item 13. Certain Relationships and Related Transactions". Mr. Zankel is also a director of Citigroup, Inc., Travelers Property Casualty Corp. and VICORP Restaurants, Inc.

EXECUTIVE OFFICERS (AS OF MARCH 26, 2001)

Name	Position	Age	Executive Officer Since
Raymond Barrette	President	50	1997
John J. Byrne	Chairman and Chief Executive Officer	68	1985
Reid T. Campbell	Vice President and Director of Finance	33	1996
K. Thomas Kemp	Deputy Chairman	60	1991
Michael S. Paquette	Senior Vice President and Controller	37	1993
James J. Ritchie	Chief Financial Officer of TACK Acquisition Corp.	46	2001
David G. Staples	Vice President	40	1997

All Executive Officers are elected by the Board for a term of one year or until their successors have been elected and have duly qualified. Information relating to Messrs. Barrette, Byrne and Kemp has been previously provided (see "Directors" above). An outline of the principal occupation and relevant business experiences of the other Executive Officers follows:

MR. CAMPBELL was elected Vice President and Director of Finance in 1998 and previously served as Assistant Controller from 1996 to 1998 and Director of Accounting from 1995 to 1996. Mr. Campbell has been with White Mountains since 1994. Mr. Campbell is a director of PIC.

MR. PAQUETTE was appointed Senior Vice President and Controller in 1997. Mr. Paquette previously served as Vice President and Controller since 1995 and as Vice President and Chief Accounting Officer from 1993 to 1995. Mr. Paquette has been with White Mountains since 1989.

MR. RITCHIE was appointed Chief Financial Officer of TACK Acquisition Corp. (the intended acquisition company for the CGU transaction) on March 1, 2001. Prior to joining White Mountains in 2001, Mr. Ritchie served as Senior Vice President and Chief Financial Officer of CIGNA Corporation's International Division. Mr. Ritchie was with CIGNA since 1986.

MR. STAPLES was elected Vice President in 1997 and has been with White Mountains since 1996. Prior to joining White Mountains, Mr. Staples served as Vice President for Crum & Forster Holdings, Inc. from 1993 to 1996.

FILINGS UNDER SECTION 16

Pursuant to SEC rules relating to the reporting of changes in beneficial ownership of Shares, the Company's Executive Officers, Directors and greater than 10% shareholders are believed to have filed all reports required under Section 16 on a timely basis during 2000.

COMPENSATION OF EXECUTIVE OFFICERS

The following tables set forth certain information regarding the salary, incentive compensation and benefits paid by White Mountains to its Chairman and CEO, its four most highly compensated Executive Officers and one former Executive Officer (collectively, the "Named Executive Officers").

					Long	-Term Compens	sation	
	-	Annua	l Compensati	on	Awa	rds	Payouts	
Name and Principal Position	Year	Salary(\$)	Bonus (\$)	Other Annual Compen- sation (\$)	Stock	Securities Underlying Options (#)	LTIP Payouts (\$)	All Other Compen- sation (\$)(a)
JACK BYRNE Chairman and CEO (b)	2000	\$ 282,692	\$ 350,000	\$0	\$0	Θ	\$ 0	\$ 9,100
RAYMOND BARRETTE President	2000 1999 1998	333,654 262,692 250,000	350,000 1,278,776 217,000	0 0 0	0 0 0	9,000 0 0	0 1,105,000 0	15,627 462,291 294,175
DAVID G. STAPLES Vice President	2000 1999 1998	147,308 135,077 128,769	150,000 1,081,313 106,000	0 0 0	0 0 0	9,000 0 0	0 520,000 399,000	5,169 19,489 15,330
REID T. CAMPBELL Vice President and Director of Finance	2000 1999 1998	108,077 99,077 91,769	150,000 129,918 84,000	0 0 0	0 0 0	9,000 0 0	0 520,000 399,000	6,177 14,816 12,326
K. THOMAS KEMP Deputy Chairman (former CEO)	2000 1999 1998	182,000 400,000 386,923	75,000 1,308,809 304,000	0 0 0	0 0 0	0 0 0	0 2,600,000 1,995,000	129,900 269,490 275,185
TERRY L. BAXTER Former Executive Officer (b)	2000 1999 1998	156,039 262,692 247,692	80,000 738,853 180,000	0 0 0	0 0 0	0 0 0	0 1,625,000 931,000	66,862 475,451 758,588

(a) Amounts include, when applicable, 401(k) Savings Plan matching contributions (which did not exceed \$10,200 per individual), principal credited to a former non-qualified deferred compensation plan, director fees and retainers paid by companies for which White Mountains is entitled to board representation and certain other compensation. The amounts for 2000, 1999 and 1998, respectively, relating to director fees and retainers of affiliates include: \$9,100, \$0, and \$0 for Mr. Byrne; \$9,685; \$22,450 and \$15,475 for Mr. Barrette; \$119,700, \$71,650 and \$75,100 for Mr. Kemp and \$57,500, \$41,342 and \$21,700 for Mr. Baxter. The 1999 and 1998 amounts for Mr. Barrette also include \$42,545 and \$249,646, respectively, in reimbursements principally associated with a Company-sponsored relocation. The 1999 amounts for Messrs. Barrette and Baxter also include \$351,917 in phantom stock awards resulting from the sale of WMSC. The 1998 amount for Mr. Baxter also includes \$665,000 in incentive compensation as interim Chairman of WMSC.

(b) In January 2000 Mr. Byrne replaced Mr. Kemp as CEO of the Company and Mr. Baxter retired from full-time service. The table above reflects Messrs. Byrne and Baxter's total compensation for 2000.

OPTION GRANTS IN LAST FISCAL YEAR

The following table summarizes, for the Named Executive Officers, options to acquire Shares ("Options") granted during 2000.

	Number of Securities Underlying Options	% of Total Options Granted to Employees in	Exercise Price of Base Price		Potential Realizable Assumed Annual Rates Price Appreciation for	of Stock
Name	Granted (#)	Fiscal Year	(\$/Sh)	Expiration Date	5% (\$)	10% (\$)
John J. Byrne	Θ	0%	\$ -	-	\$ 0	\$ 0
Raymond Barrette David G. Staples	9,000 9,000	11.1% 11.1%	106.19(a) 106.19(a)	December 2009 December 2009	0 0	704,567 704,567
Reid T. Campbell K. Thomas Kemp	9,000 0	11.1% 0%	106.19(a)	December 2009	0	704, 567
Terry L. Baxter	0	0%			0	0

(a) Represents the closing market value of Shares on the grant date of February 28, 2000. The exercise price of the Options increases by 6% annually on a pro rata basis. Options vest 10% per year through 2009. The Options are considered to be Incentive Stock Options for income tax purposes.

AGGREGATED OPTION EXERCISES IN LAST FISCAL YEAR AND FISCAL YEAR-END OPTION VALUES $% \left({{\left| {{{\rm{AG}}} \right|} \right|_{\rm{AG}}} \right)$

The following table summarizes, for the Named Executive Officers, Options exercised during the Company's latest fiscal year, and the number and in-the-money value of Options outstanding as of December 31, 2000.

				As of Decem	ber 31, 2000	
				ities Underlying ptions at Fiscal Year-End (#)	N	vercised In-the- Money Options at Cal Year-End (\$)
	Shares Acquired on Exercise (#)	Value Realized	Exercisable	Unexercisable	Exercisable	Unexercisable
John J. Byrne	Θ	\$ 0	0	Θ	\$ 0	\$ 0
Raymond Barrette	Θ	0	900	8,100	186,737	1,680,634
David G. Staples	Θ	0	900	8,100	186,737	1,680,634
Reid T. Campbell	Θ	Θ	900	8,100	186,737	1,680,634
K. Thomas Kemp	Θ	Θ	0	Θ	Θ	Θ
Terry L. Baxter	Θ	Θ	Θ	Θ	Θ	Θ
=======================================		================	=======================================	=======================================	==================	

The following table summarizes the Incentive Plan awards made to the Named Executive Officers during 2000. Such awards consisted of performance shares.

	Number of	Performance	Estimated Fu	uture Payo	outs (a)
Name	Performance Shares (#)	Period for Payout	Threshold (#) Ta	arget (#)	Maximum (#)
John J. Byrne	10,000	3 yrs.	Θ	10,000	20,000
Raymond Barrette	10,000	3 yrs.	Θ	10,000	20,000
David G. Staples	2,000	3 yrs.	Θ	2,000	4,000
Reid T. Campbell	2,000	3 yrs.	Θ	2,000	4,000
K. Thomas Kemp	2,000	3 yrs.	Θ	2,000	4,000
Terry L. Baxter	2,000	3 yrs.	0	2,000	4,000

(a) Such performance shares are payable upon completion of pre-defined business goals and are payable in cash based on the market value of Shares at the time of payment or Shares. The "Target" performance for the 2000 performance share award is the attainment of a corporate annualized return on equity ("ROE") of 13% after tax. The determination of ROE is generally based on the economic value of Shares with dividends reinvested. At an ROE of 6% or less ("Threshold") the percentage of performance shares payable will be 0% and at an ROE of 25% or more ("Maximum") the percentage of performance shares payable will become 200% of Target.

OTHER COMPENSATION ARRANGEMENTS

Pursuant to the Incentive Plan, under some circumstances such as a "Change in Control" followed by a termination without cause, constructive termination or an "Adverse Change" in the Incentive Plan, stock options will generally become fully exercisable and performance shares will become partially or fully payable. Such circumstances are more fully described in the Incentive Plan.

The Company does not provide pension benefits to its Executive Officers under a defined benefit or actuarial plan. The Company has previously provided non-qualified pension benefits to its Executive Officers under a deferred benefit plan but did not provide such benefits during 2000.

COMPENSATION OF DIRECTORS

Messrs. Patrick Byrne, Clark, Cochran, George Gillespie, John Gillespie, Kemp, Macklin, Olson and Zankel each received a retainer of \$50,000 during 2000 and fees of \$1,000 for each Board meeting and Committee meeting attended. The annual retainer relates to the twelve month period from May 2000 to May 2001. Messrs. Clark, John Gillespie and Macklin also received additional retainers of \$3,000, \$100,000 and \$6,000 during 2000 for their roles as Chairman of the Audit Committee, Chairman of the Investment Committee and Chairman of the Compensation and Human Resources Committees, respectively. Mr. Fass was granted 9,000 Options and 2,000 performance shares during 2000 on the same terms as Options and performance shares granted to other Executive Officers. Messrs. Jack Byrne and Barrette did not receive compensation for their role as a Director during 2000.

ITEM 12. SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

PRINCIPAL HOLDERS OF SHARES

To the knowledge of the Company, there was no person or entity beneficially owning more than 5% of Shares outstanding as of March 26, 2001, except as shown below:

Jack Byrne 80 South Main Street, Hanover, NH 03755 (b) 1,182, Franklin Mutual Advisers LLC 51 JFK Parkway, Short Hills, NJ 07078 (c) 750, Alliance Asset Accumulation Plan 777 San Marin Drive, Novato, CA 94998 (d) 367,	71 12.8%

- (a) Represents beneficial ownership of Shares as opposed to voting power.
- (b) Does not include 65,913 Shares donated to charitable foundations for which Mr. Byrne disclaims beneficial ownership, but for which his spouse retains voting power.
- (c) According to filings by such holders with the SEC, the Shares beneficially owned by Franklin Mutual Advisers LLC were acquired solely for investment purposes on behalf of client investment advisory accounts of such holders.
- (d) Represents Shares beneficially owned by employees of Fireman's Fund pursuant to an employee incentive savings plan. The trustee for such plan generally votes the Shares held by the plan in accordance with directions given by the participating Fireman's Fund employees to whose accounts Shares have been allocated.

BENEFICIAL STOCK OWNERSHIP OF DIRECTORS AND EXECUTIVE OFFICERS

The following table sets forth, as of March 26, 2001, beneficial ownership of Shares by each director of the Company, by each of the current Executive Officers, and by all Directors and Executive Officers as a group.

	Number of Sh	ares owned
Directors and Executive Officers	Beneficially (a)(b)	Economically (c)
Raymond Barrette	29,433	79,867
Jack Byrne (d)	1,182,959	1,207,959
Patrick M. Byrne	236,008	236,008
Reid T. Campbell	4,902	20,502
Howard L. Clark, Jr.	1,000	1,000
Robert P. Cochran	Θ	Θ
Steven E. Fass	4,415	16,515
George J. Gillespie, III	1,000	1,000
John D. Gillespie	1,676	1,676
K. Thomas Kemp	81,690	107,779
Gordon S. Macklin	15,000	15,000
Frank A. Olson	3,000	3,000
David G. Staples	5,183	21,283
Arthur Zankel	11,600	11,600
All Directors and Executive Officers as a group (16 persons) (d)	1,591,189	1,759,612

- (a) The Shares beneficially owned by Messrs Jack Byrne, Patrick Byrne, Kemp and all Directors and Executive Officers as a group represent 20.1%, 4.0%, 1.4% and 27.0% of the total Shares outstanding at March 26, 2001, respectively. All other Directors and Executive Officers beneficially owned less than 1% of the total Shares outstanding at that date. Represents beneficial ownership of Shares as opposed to voting power.
- (b) Includes vested Options to acquire 900 Shares for each of Messrs. Barrette, Fass, Campbell and Staples.
- (c) Incremental Shares shown as economically owned by Directors and Executive Officers represent unvested performance share awards, unvested Option awards, unvested restricted stock awards and earned phantom shares on compensation deferred.
- (d) Does not include 65,913 Shares donated to charitable foundations for which Mr. Byrne disclaims beneficial ownership, but for which his spouse retains voting power.

Mr. Clark is Vice Chairman of Lehman. Lehman has, for a number of years, provided investment banking services to White Mountains. Lehman is the arranger, the administrative agent and a lender under the Credit Facility.

 $\mbox{Mr.}$ George Gillespie is a Partner in Cravath. Cravath has, for many years, provided legal services to White Mountains.

White Mountains owns a limited partnership investment interest which was managed by Mr. John Gillespie during 2000.

White Mountains owns a limited partnership interest and an investment portfolio which was managed by Mr. Zankel during 2000.

White Mountains believes that all the above transactions were on terms that were reasonable and competitive. Additional transactions of this nature may be expected to take place in the ordinary course of business in the future.

PART IV

ITEM 14. EXHIBITS, FINANCIAL STATEMENT SCHEDULES, AND REPORTS ON FORM 8-K

a. DOCUMENTS FILED AS PART OF THE REPORT

The financial statements and financial statement schedules and reports of independent auditors have been filed as part of this Annual Report on Form 10-K as indicated in the Index to Financial Statements and Financial Statement Schedules appearing on page 34 of this report. A listing of exhibits filed as part of the report appear on pages 30 through 32 of this report.

b. REPORTS ON FORM 8-K

During the fourth quarter of 2000 the Company filed two Reports on Form 8-K. The first, dated October 19, 2000, announced that the Company amended its Stock Purchase Agreement with CGNU to purchase CGU. The second, dated October 20, 2000, announced that the Company had received a favorable tax ruling from the Internal Revenue Service regarding issues related to its 1991 sale of Fireman's Fund.

EXHIBIT NUN	IBER NAME
2	Plan of Reorganization (incorporated by reference herein to the Company's Registration Statement on S-4 (No. 333-87649) dated September 23, 1999)
3 (a)	Memorandum of Continuance of the Company (incorporated by reference herein to the Company's Registration Statement on S-4 (No. 333-87649) dated September 23, 1999)
3 (b)	Bye-Laws of the Company (incorporated by reference herein to the Company's Registration Statement on S-4 (No. 333-87649) dated September 23, 1999)
4	Indenture dated January 1, 1993, with The First National Bank of Chicago, as trustee, pursuant to the Company's offering of \$150 million of medium-term notes (incorporated by reference herein to the Company's Registration Statement on S-3 (No. 33-54006) dated October 30, 1992)
10 (a)	Stock Purchase Agreement among CGU International Holdings Luxembourg S.A., CGU Holdings LLC, CGNU PLC, the Company, Tack Holding Corp. and Tack Acquisition Corp. dated as of September 24, 2000 (incorporated by reference herein to Exhibit 99(a) of the Company's Report on Form 8-K dated September 24, 2000)
10 (b)	
10 (c)	
10 (d)	Convertible Preferred Stock Term Sheet relating to the Company's acquisition of CGU (incorporated by reference herein to Exhibit 99(e) of the Company's Report on Form 8-K dated October 19, 2000)
10 (e)	(incorporated by reference herein to Exhibit 99(f) of the Company's Report on Form 8-K dated October 19, 2000)
10 (f)	borrowers), Lehman (as arranger) and the several lenders as parties thereto relating to the Company's acquisition of CGU dated March 16, 2001 (*)
10 (g)	Stock Purchase and Indemnity Agreement by and among the Company and Dexia S.A. for all of the outstanding capital stock of White Mountains Holdings, Inc. and indirectly for certain of the outstanding capital stock of FSA (incorporated by reference herein to Exhibit 99.1 of the Company's Report on Form 8-K dated March 10, 2000)
10 (h)	
10 (i)	Stock Purchase Agreement as of December 30, 1999, by and among Humana Inc., Physician Corporation of America and Folksamerica Holding Company, Inc. (incorporated by reference herein to Exhibit 10(a) of the Company's Report on Form 8-K dated December 30, 1999)
10 (j)	Amended and Restated Management Contract by and between PCA and Humana Workers Compensation Services, Inc. (incorporated by reference herein to Exhibit 10(a) of the Company's Report on Form 8-K dated December 30, 1999)
10 (k)	

EXHIBIT NUMBER	NAME

- Stock Purchase Agreement dated March 31, 1999, by and Between the Centris Group, Inc. and Folksamerica 10 (1) Holding Company, Inc. (incorporated by reference herein to Exhibit 10(a) of the Company's Report on Form 8-K dated June 29, 1999)
- 10 (m) Source One Asset Purchase Agreement dated March 25, 1999 between the Company, Source One and Citicorp Mortgage Inc. (incorporated by reference herein to Exhibit 10(p) of the Company's 1998 Annual Report on Form 10-K)
- Transition Services Agreement dated March 25, 1999 between Source One and Citicorp Mortgage, Inc. (incorporated by reference herein to Exhibit 10(0) of the Company's 1998 Annual Report on Form 10-K) 10 (n)
- VGI Stock Acquisition Agreement dated February 10, 1999 between Unitrin, Inc. and the Company (incorporated by 10 (0) reference herein to Exhibit 10(n) of the Company's 1998 Annual Report on Form 10-K)
- Second Amended and Restated Credit Agreement dated February 24, 1999 among the Company, the Lenders (as named 10 (p) therein) and The First National Bank of Chicago (incorporated by reference herein to Exhibit 10(a) of the Company's 1999 Annual Report on Form 10-K)
- Amendment No. 1 dated March 23, 1999 to the Second Amended and Restated Credit Agreement dated February 24, 1999 among the Company, the Lenders (as named therein) and The First National Bank of Chicago (incorporated by reference herein to Exhibit 10(b) of the Company's 1999 Annual Report on Form 10-K) 10 (q)
- 10 (r) Amendment No. 2 dated July 30, 1999 to the Second Amended and Restated Credit Agreement dated February 24, 1999 among the Company, the Lenders (as named therein) and The First National Bank of Chicago (incorporated by reference herein to Exhibit 10(c) of the Company's 1999 Annual Report on Form 10-K)
- Amendment No. 3 dated October 29, 1999 to the Second Amended and Restated Credit Agreement dated February 24, 1999 10 (s) among the Company, the Lenders (as named therein) and The First National Bank of Chicago (incorporated by reference herein to Exhibit 10(d) of the Company's 1999 Annual Report on Form 10-K)
- Credit Agreement dated February 24, 1999 among Folksamerica Holding Company, Inc., the Lenders (as 10 (t) named therein) and The First National Bank of Chicago (incorporated by reference herein to Exhibit 10(e) of the Company's 1999 Annual Report on Form 10-K)
- Amendment No.1 dated June 29, 1999 to the Credit Agreement dated February 24, 1999 among Folksamerica Holding Company, Inc., the Lenders (as named therein) and The First National Bank of Chicago (incorporated by reference herein to Exhibit 10(f) of the Company's 1999 Annual Report on Form 10-K) Amendment No. 2 dated October 29, 1999 to the Credit Agreement dated February 24, 1999 among Folksamerica Holding 10 (u)
- 10 (v) Company, Inc., the Lenders (as named therein) and The First National Bank of Chicago (incorporated by reference herein to Exhibit 10(g) of the Company's 1999 Annual Report on Form 10-K)
- 10 (w)
- Guaranty, dated February 28, 1997, by the Company's 1999 Annual Report on Form 10-K) Guaranty, dated February 28, 1997, by the Company to and for the benefit of Chemical Mortgage Company (incorporated by reference herein to Exhibit 10(y) of the Company's 1996 Annual Report on Form 10-K) The Company's Long-Term Incentive Plan, as amended, (incorporated by reference to Appendix I of the Company's Notice of 1995 Annual Meeting of Shareholders and Proxy Statement) (**) Statement Re Computation of Per Share Earnings (***) 10 (x)
- 11

BIT NUMB	BER NAME	
16	Letter of KPMG LLP dated March 25, 1999, (incorporated by reference herein to Exhibit 16 of the Company's	
	1998 Annual Report on Form 10-K)	
21	Subsidiaries of the Registrant (*)	
23 (a)	Consent of PricewaterhouseCoopers dated March 26, 2001 (*)	
23 (b)	Consent of KPMG LLP dated March 26, 2001 (*)	
23 (c)	Consent of PricewaterhouseCoopers LLP dated March 26, 2001 relating to Folksamerica and FSA (*)	
24	Powers of Attorney (*)	
99 (a)	Report of PricewaterhouseCoopers LLP dated February 2, 1999	
	relating to Folksamerica (incorporated by reference herein to Exhibit 99(a) of the Company's 1998 Annual	

relating to Folksame Report on Form 10-K) erica (incorporated by reference herein to Exhibit 99(a) of the Company's 1998 Annual 99 (b) The Consolidated Financial Statements of FSA and the related Report of Independent Accountants as of

- December 31, 1999 and 1998 and for each of the three years in the period ended December 31, 1999 (incorporated by reference herein to Exhibit 99(b) of the Company's 1999 Annual Report on Form 10-K)
- Included herein. (*) (**)
- Management contracts or compensation plans/arrangements required to be filed as an exhibit pursuant to Item 14(a)3 of Form 10-K.
- (***) Not included herein as the information is contained elsewhere within report. See Note 1 of the Notes to Consolidated Financial Statements.
- d. FINANCIAL STATEMENT SCHEDULES The financial statement schedules and report of independent auditors have been filed as part of this Annual Report on Form 10-K as indicated in the Index to Financial Statements and Financial Statement Schedules appearing on page 34 of this report.

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the Company has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

Date: March 26, 2001

WHITE MOUNTAINS INSURANCE GROUP, LTD. By: /s/ MICHAEL S. PAQUETTE Senior Vice President and Controller

Pursuant to the requirements of the Securities Exchange Act of 1934, this report has been signed below by the following persons on behalf of the Company and in the capacities and on the dates indicated.

SIGNATURE	TITLE	DATE
/s/ RAYMOND BARRETTE	President and Director	March 26, 2001
Raymond Barrette	(Principal Financial Officer)	
/s/ JOHN J. BYRNE	Chairman and Chief Executive Officer (Principal Executive Officer)	March 26, 2001
John J. Byrne		
PATRICK M. BYRNE*	Director	March 26, 2001
Patrick M. Byrne		
HOWARD L. CLARK, JR.*	Director	March 26, 2001
Howard L. Clark, Jr.		
ROBERT P. COCHRAN*	Director	March 26, 2001
Robert P. Cochran		
STEVEN E. FASS*	Director	March 26, 2001
Steven E. Fass		
GEORGE J. GILLESPIE, III*	Director	March 26, 2001
George J. Gillespie, III		
JOHN D. GILLESPIE*	Director	March 26, 2001
John D. Gillespie		
/s/ K. THOMAS KEMP	Director	March 26, 2001
K. Thomas Kemp		
GORDON S. MACKLIN*	Director	March 26, 2001
Gordon S. Macklin		
FRANK A. OLSON*	Director	March 26, 2001
Frank A. Olson		
/s/ MICHAEL S. PAQUETTE	Senior Vice President and Controller	March 26, 2001
Michael S. Paquette	(Principal Accounting Officer)	
ARTHUR ZANKEL*	Director	March 26, 2001
Arthur Zankel		
*By: /s/ K. THOMAS KEMP		
K. Thomas Kemp, Attornev-in-Fact		

K. Thomas Kemp, Attorney-in-Fact

WHITE MOUNTAINS INSURANCE GROUP, LTD. INDEX TO CONSOLIDATED FINANCIAL STATEMENTS AND FINANCIAL STATEMENT SCHEDULES

CONSOLIDATED FINANCIAL STATEMENTS: F-1 Consolidated balance sheets as of December 31, 2000 and 1999. F-1 Consolidated statements of income and comprehensive income for each of the years ended F-2 December 31, 2000, 1999 and 1998. F-3 Consolidated statements of shareholders' equity for each of the years ended F-3 December 31, 2000, 1999 and 1998. F-3 Consolidated statements of cash flows for each of the years ended F-4 December 31, 2000, 1999 and 1998. F-4 Notes to consolidated financial statements. F-5 OTHER FINANCIAL INFORMATION: F-30 Report on management's responsibilities. F-31 Selected quarterly financial data (unaudited). F-31 Selected quarterly financial data (unaudited). F-33 FINANCIAL STATEMENT SCHEDULES: FS-1 I. Condensed financial information of the Registrant. FS-2 III. Supplementary insurance information. FS-4 IV. Reinsurance. FS-5 V. Valuation and qualifying accounts. FS-6 V. Supplemental information concerning property and casualty insurance underwriters. FS-7		Form 10-K	
Consolidated balance sheets as of December 31, 2000 and 1999.F-1Consolidated statements of income and comprehensive income for each of the years ended December 31, 2000, 1999 and 1998.F-2Consolidated statements of shareholders' equity for each of the years ended December 31, 2000, 1999 and 1998.F-3Consolidated statements of cash flows for each of the years ended December 31, 2000, 1999 and 1998.F-3Consolidated statements of cash flows for each of the years ended 			
Consolidated balance sheets as of December 31, 2000 and 1999. F-1 Consolidated statements of income and comprehensive income for each of the years ended F-2 December 31, 2000, 1999 and 1998. F-3 Consolidated statements of shareholders' equity for each of the years ended F-3 December 31, 2000, 1999 and 1998. F-3 Consolidated statements of cash flows for each of the years ended F-4 December 31, 2000, 1999 and 1998. F-4 Notes to consolidated financial statements. F-5 OTHER FINANCIAL INFORMATION: F-30 Report on management's responsibilities. F-31 Selected quarterly financial data (unaudited) F-33 FINANCIAL STATEMENT SCHEDULES: F-30 I. Summary of investments other than investments in related parties. FS-1 II. Condensed financial information of the Registrant. FS-2 III. Supplementary insurance information. FS-4 IV. Reinsurance. FS-5 V. Valuation and qualifying accounts. FS-6 V. Valuation and qualifying accounts. FS-6 V. Supplemental information concerning property and casualty insurance underwriters. FS-7	CONSOLIDATED ETNANCIAL STATEMENTS:		
December 31, 2000, 1999 and 1998.F-2Consolidated statements of shareholders' equity for each of the years ended December 31, 2000, 1999 and 1998.F-3Consolidated statements of cash flows for each of the years ended December 31, 2000, 1999 and 1998.F-4Notes to consolidated financial statements.F-5OTHER FINANCIAL INFORMATION: Report on management's responsibilities.F-30Reports of independent accountants.F-31Selected quarterly financial data (unaudited).F-33FINANCIAL STATEMENT SCHEDULES: I. Summary of investments of the Registrant.FS-1II. Condensed financial information of the Registrant.FS-2IV. Reinsurance.FS-4IV. Reinsurance.FS-5V. Valuation and qualifying accounts.FS-6VI. Supplemental information concerning property and casualty insurance underwriters.FS-7	Consolidated balance sheets as of December 31, 2000 and 1999	F-1	
December 31, 2000, 1999 and 1998.F-3Consolidated statements of cash flows for each of the years ended December 31, 2000, 1999 and 1998.F-4Notes to consolidated financial statements.F-5OTHER FINANCIAL INFORMATION: Report on management's responsibilities.F-30Reports of independent accountants.F-31Selected quarterly financial data (unaudited).F-33FINANCIAL STATEMENT SCHEDULES: I. Summary of investments other than investments in related parties.FS-1II. Condensed financial information of the Registrant.FS-2IV. Reinsurance.FS-4V. Valuation and qualifying accounts.FS-6VI. Supplemental information concerning property and casualty insurance underwriters.FS-7	December 31, 2000, 1999 and 1998	F-2	
December 31, 2000, 1999 and 1998.F-4Notes to consolidated financial statements.F-5OTHER FINANCIAL INFORMATION: Report on management's responsibilities.F-30Reports of independent accountants.F-31Selected quarterly financial data (unaudited).F-33FINANCIAL STATEMENT SCHEDULES: I. Summary of investments other than investments in related parties.FS-1II. Condensed financial information of the Registrant.FS-2III. Supplementary insurance information.FS-4IV. Reinsurance.FS-5V. Valuation and qualifying accounts.FS-6VI. Supplemental information concerning property and casualty insurance underwriters.FS-7	December 31, 2000, 1999 and 1998	F-3	
OTHER FINANCIAL INFORMATION: F-30 Report on management's responsibilities. F-30 Reports of independent accountants. F-31 Selected quarterly financial data (unaudited). F-33 FINANCIAL STATEMENT SCHEDULES: F-33 I. Summary of investments other than investments in related parties. FS-1 II. Condensed financial information of the Registrant. FS-2 III. Supplementary insurance information. FS-4 IV. Reinsurance. FS-5 V. Valuation and qualifying accounts. FS-6 VI. Supplemental information concerning property and casualty insurance underwriters. FS-7		F-4	
Report on management's responsibilitiesF-30Reports of independent accountantsF-31Selected quarterly financial data (unaudited)F-33FINANCIAL STATEMENT SCHEDULES:II. Summary of investments other than investments in related partiesFS-1II. Condensed financial information of the RegistrantFS-2III. Supplementary insurance informationFS-4IV. ReinsuranceFS-5V. Valuation and qualifying accountsFS-6VI. Supplemental information concerning property and casualty insurance underwritersFS-7	Notes to consolidated financial statements	F-5	
Reports of independent accountantsF-31Selected quarterly financial data (unaudited)F-33FINANCIAL STATEMENT SCHEDULES:F-33I. Summary of investments other than investments in related partiesFS-1II. Condensed financial information of the RegistrantFS-2III. Supplementary insurance informationFS-4IV. ReinsuranceFS-5V. Valuation and qualifying accountsFS-6VI. Supplemental information concerning property and casualty insurance underwritersFS-7			
Selected quarterly financial data (unaudited)F-33FINANCIAL STATEMENT SCHEDULES:II. Summary of investments other than investments in related partiesFS-1II. Condensed financial information of the RegistrantFS-2III. Supplementary insurance informationFS-4IV. Reinsurance.FS-5V. Valuation and qualifying accountsFS-6VI. Supplemental information concerning property and casualty insurance underwritersFS-7			
FINANCIAL STATEMENT SCHEDULES: FS-1 I. Summary of investments other than investments in related parties			
I. Summary of investments other than investments in related partiesFS-1II. Condensed financial information of the RegistrantFS-2III. Supplementary insurance informationFS-4IV. ReinsuranceFS-5V. Valuation and qualifying accountsFS-6VI. Supplemental information concerning property and casualty insurance underwritersFS-7	Selected quarterly financial data (unaudited)	F-33	
II. Condensed financial information of the RegistrantFS-2III. Supplementary insurance informationFS-4IV. ReinsuranceFS-5V. Valuation and qualifying accountsFS-6VI. Supplemental information concerning property and casualty insurance underwritersFS-7	FINANCIAL STATEMENT SCHEDULES:		
III. Supplementary insurance informationFS-4IV. ReinsuranceFS-5V. Valuation and qualifying accountsFS-6VI. Supplemental information concerning property and casualty insurance underwritersFS-7	I. Summary of investments other than investments in related parties	FS-1	
IV. ReinsuranceFS-5V. Valuation and qualifying accountsFS-6VI. Supplemental information concerning property and casualty insurance underwritersFS-7		FS-2	
V. Valuation and qualifying accounts FS-6 VI. Supplemental information concerning property and casualty insurance underwriters	III. Supplementary insurance information		
VI. Supplemental information concerning property and casualty insurance underwriters FS-7			

	December 31,			
Dollars in millions	2000	1999		
ASSETS				
Fixed maturity investments, at fair value (cost \$1,063.0 and \$957.9) Common equity securities, at fair value (cost \$127.5 and \$100.4)	\$ 1,078.6 144.8	\$ 924.5 108.4		
Short-term investments, at amortized cost (which approximated fair value)	735.9	117.5		
other investments (cost \$117.3 and \$59.1)	142.9	68.3		
Total investments	2,102.2	1,218.7		
Cash	4.4	3.9		
einsurance recoverable on paid and unpaid losses	777.2	193.7		
investment in unconsolidated insurance affiliate(s)	130.6	422.6		
insurance and reinsurance balances receivable	105.7	49.8		
beferred tax asset	105.1	52.5		
Deferred acquisition costs	27.2	22.2		
oodwill	25.4	3.6		
investment income accrued	20.1	15.0		
other assets	231.1	50.8		
let assets of discontinued mortgage banking operations	16.2			
Total assets	\$ 3,545.2	\$ 2,049.1		
IABILITIES				
.oss and loss adjustment expense reserves	\$ 1,556.3	\$ 851.0		
unds held under reinsurance treaties	420.0	31.4		
nearned insurance and reinsurance premiums	182.0	92.1		
hort-term debt	-	4.0		
ong-term debt	96.0	202.8		
Deferred credits	92.2	100.6		
ccounts payable and other liabilities	152.2	152.9		
Total liabilities		1,434.8		
HAREHOLDERS' EQUITY				
common Shares at \$1 par per Share- authorized 50,000,000 Shares,				
issued and outstanding 5,880,115 and 5,945,953 Shares	5.9	5.9		
aid-in surplus	66.2	67.0		
etained earnings	927.5	534.2		
accumulated other comprehensive income, after tax	46.9	7.2		
Total shareholders' equity	1,046.5	614.3		
Total liabilities and shareholders' equity	\$ 3,545.2	\$ 2,049.1		

See Notes to Consolidated Financial Statements including Note 16 for Commitments and Contingencies.

	Ye	ear Ended Decemb	er 31,
Millions, except per Share amounts	2000	1999	1998
REVENUES:			
Gains on sales of subsidiaries and insurance assets	\$ 385.8	\$ 103.9	\$-
Earned insurance and reinsurance premiums	334.4	283.2	246.0
Net investment income	85.9	61.9	36.8
Amortization of deferred credits and other benefits	41.4	25.8	2.7
Earnings (losses) from unconsolidated insurance affiliates	(2.6)	31.1	24.3
Net realized gains (losses) on sales of investments and other assets	(8.4)	69.6	71.0
Other revenue	11.7	3.7	9.5
Total revenues	848.2	579.2	390.3
EXPENSES :			
Losses and loss adjustment expenses	287.7	242.3	174.8
Insurance and reinsurance acquisition expenses	101.1	73.4	54.8
Compensation and benefits	59.5	67.8	51.5
General expenses	28.4	19.5	15.9
Interest expense	16.1	14.7	13.7
Total expenses	492.8	417.7	310.7
PRETAX EARNINGS	355.4	161.5 (53.1)	79.6
Tax provision		(53.1)	
NET INCOME FROM CONTINUING OPERATIONS	312.9	108.4	51.1
DISCONTINUED OPERATIONS, AFTER TAX:			
Gains from discontinued operations	95.0	11.6	-
Net income from discontinued operations	-	1.0	27.4
NET INCOME	407.9	121.0	78.5
THER COMPREHENSIVE INCOME (LOSS) ITEMS ARISING DURING THE YEAR, AFTER TAX:			
Net change in unrealized gains for investments held	56.3	(73.7)	38.1
Net change in foreign currency translation	(.7)	.9	(.9)
Recognition of net unrealized gains for investments sold	(15.9)	(45.2)	(46.1)
COMPREHENSIVE NET INCOME	\$ 447.6	\$ 3.0	\$ 69.6
BASIC EARNINGS PER SHARE:			
Net income from continuing operations	\$ 53.08	\$ 19.25	\$ 8.71
Net income	69.19	21.50	13.38
Comprehensive net income	75.93	.54	11.87
DILUTED EARNINGS PER SHARE:			
Net income from continuing operations	\$ 52.84	\$ 17.66	\$ 7.75
Net income	68.89	19.73	11.94
Comprehensive net income	75.60	. 39	10.58

See Notes to Consolidated Financial Statements.

Millions	SI Total	Common nares and paid-in surplus	Retained earnings			Foreign currency translation adjustment
Balances at January 1, 1998	\$ 658.9	\$386.9	\$1,008.9	\$(871.0)	\$134.1	\$ -
Net income	78.5	_	78.5	-	-	_
Dividends to shareholders	(9.4)	-	(9.4)	-	-	-
Shares repurchased and retired	· · ·	(1.8)	· · ·	-	-	-
Change in net unrealized investment	()	()	()			
gains and losses and other, after tax	(8.9)	-	-	-	(8.0)	(.9)
Other	3.2	-	3.2	-	-	-
Balances at December 31, 1998	702.5	385.1	1,063.2	(871.0)	126.1	(.9)
Net income	121.0	-	121.0	-	-	-
Dividends to shareholders	(8.8)	-	(8.8)		-	-
Issuances of Shares	57.1	-	(58.8)	115.9	-	-
Shares repurchased and retired	(139.5)	(312.2)	(582.4)	755.1	-	-
Change in net unrealized investment	(110.0)				(110.0)	0
gains and losses and other, after tax	(118.0)	-	-	-	(118.9)	.9
Balances at December 31, 1999	614.3	72.9	534.2	-	7.2	-
Note descent	407.0		107.0			
Net income	407.9	-	407.9	-	-	-
Dividends to shareholders	(7.1)	-	(7.1)	-	-	-
Shares repurchased and retired Change in net unrealized investment	(8.3)	(.8)	(7.5)	-	-	-
gains and losses and other, after tax	39.7	-	-	-	40.4	(.7)
BALANCES AT DECEMBER 31, 2000	\$1,046.5	\$ 72.1	\$ 927.5	\$	\$ 47.6	\$(.7)

See Notes to Consolidated Financial Statements.

		Year Ended December 31,			
Millions	2000	1999	1998		
Net income from continuing operations Reconciliation of net income to cash flows from operating activities:	\$ 312.9	\$ 108.4	\$ 51.1		
Gains on sales of subsidiaries and other insurance assets	(385.8)	(88.1)	-		
Undistributed loss (earnings) from unconsolidated insurance affiliates	2.6	(29.0)	(19.1)		
Net realized losses (gains) on sales of investments and other assets	8.4	(85.4)	(71.0)		
Amortization of deferred credits and other benefits	(41.4)	(25.8)	(2.7)		
Decrease (increase) in reinsurance recoverable	(367.2)	5.7	(2.7)		
Net increase in funds held and insurance and reinsurance premiums receiv	vable 432.8	28.9	2.3		
(Decrease) increase in insurance loss and loss adjustment expense reserv	ves (72.4)	(69.2)	13.7		
Net change in current and deferred income taxes receivable and payable	(12.6)	55.6	(7.1)		
Net change in other liabilities	31.6	(111.3)	11.8		
Net change in other assets	(33.0)	7.4	25.1		
Other, net	10.1	(5.5)	(5.3)		
Net cash used for operating activities	(114.0)	(208.3)	(3.9)		
Cash flows from investing activities:					
Net (increase) decrease in short-term investments	(614.6)	75.5	147.1		
Sales of common equity securities and other investments	204.0	256.4	137.5		
Sales of fixed maturity investments	315.1	237.7	116.7		
Maturities of fixed maturity investments	63.0	36.0	16.1		
Purchases of common equity securities and other investments	(205.6)	(71.1)	(56.1)		
Purchases of fixed maturity investments	(159.0)	(89.4)	(122.7)		
Purchases of consolidated and unconsolidated affiliates	60.1	(234.3)	(237.8)		
Proceeds from sales of consolidated and unconsolidated affiliates	570.4	144.5	-		
Net sales (purchases) of fixed assets	1.0	(1.0)	(1.1)		
Net cash provided from (used for) investing activities	234.4	354.3	(.3)		
Cash flows from financing activities:					
Net (decrease) increase of short-term debt	(4.0)	(51.6)	(.4)		
Issuances of long-term debt	15.0	100.0	50.0		
Repayments of long-term debt	(115.0)	100.0 (86.4) (139.4)	(1.1)		
Shares repurchased and retired	(8.8)	(· · · /	(19.5)		
Proceeds from exercises of warrants and options to acquire Shares	-	21.7	-		
Cash dividends paid to common shareholders	(7.1)	(8.8)	(9.4)		
Net cash (used for) provided from financing activities	(119.9)	(164.5)	19.6		
Net increase (decrease) in cash during year	.5	(18.5)	15.4		
Cash balance at beginning of year	3.9	22.4	7.0		
Cash balance at end of year	\$ 4.4	\$ 3.9	\$ 22.4		

See Notes to Consolidated Financial Statements.

NOTE 1. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

BASIS OF PRESENTATION

The accompanying consolidated financial statements include the accounts of the Company and its subsidiaries and have been prepared in accordance with accounting principles generally accepted in the United States ("GAAP"). The Company's consolidated operating subsidiaries at December 31, 2000 principally consisted of Folksamerica, PIC, ACIC, BICC and Waterford. The Company's unconsolidated affiliate at December 31, 2000 consisted of MSA. All significant intercompany transactions have been eliminated in consolidation. The financial statements include all adjustments considered necessary by management to fairly present the financial position, results of operations and cash flows of White Mountains. The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities as of the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates. Certain amounts in the prior period financial statements have been reclassified to conform with the current presentation, including the segregation of mortgage banking net assets and mortgage banking net income as discontinued operations which relates to a decision made by the Company during 1999 to exit from the mortgage banking business. See Note 2.

White Mountains has completed numerous significant transactions during the periods presented herein that have affected the comparability of the financial statement information presented herein.

INVESTMENT SECURITIES

White Mountains' portfolio of fixed maturity investments, common equity securities and other investments are classified as available for sale and are reported at fair value as of the balance sheet date. Net unrealized investment gains and losses, after tax, associated with such investments are reported as a net amount as a separate component of shareholders' equity. Changes in net unrealized investment gains and losses, after tax, are reported as a component of other comprehensive net income.

Premiums and discounts on fixed maturity investments are accreted to income over the anticipated life of the investment.

Other investments principally include investments in limited partnership interests which are recorded using the equity method of accounting.

Realized gains and losses resulting from sales of investment securities are accounted for using the specific identification method.

Short-term investments consist of money market funds, certificates of deposit and other securities which mature or become available for use within one year. Short-term investments are carried at amortized cost, which approximated fair value as of December 31, 2000 and 1999.

CASH

Cash includes amounts on hand and demand deposits with banks and other financial institutions. Amounts presented in the statement of cash flows are shown net of balances acquired and sold in the purchase or sale of the Company's consolidated subsidiaries.

INSURANCE AND REINSURANCE OPERATIONS

Premiums written are recognized as revenues and are earned ratably over the terms of the related policies or reinsurance treaties. Unearned premiums represent the portion of premiums collected that are applicable to future insurance or reinsurance coverage provided by policies or treaties in force.

Deferred acquisition costs represent commissions, premium taxes, brokerage expenses and other costs which are directly attributable to and vary with the production of new business. These costs are deferred and amortized over the applicable premium recognition period. Deferred acquisition costs are limited to the amount expected to be recovered from future earned premiums and anticipated investment income.

Losses and loss adjustment expenses are charged against income as incurred. Unpaid insurance losses and loss adjustment expenses are based on estimates (generally determined by claims adjusters, legal counsel and actuarial staff) of the ultimate costs of settling claims, including the effects of inflation and other societal and economic factors. Unpaid reinsurance losses and loss adjustment expenses are based primarily on reports received from ceding companies. Unpaid loss and loss adjustment expense reserves represent management's best estimate of ultimate losses and loss adjustment expenses, net of estimated salvage and subrogation recoveries, if applicable. Such estimates are regularly reviewed and updated and any adjustments resulting therefrom are reflected in current operations. The process of estimating loss and loss adjustment expenses involves a considerable degree of judgement by management and the ultimate amount of expense to be incurred could be considerably greater than or less than the amounts currently reflected in the financial statements.

Due to the nature of the policies written by White Mountains' insurance and reinsurance subsidiaries and the existence of reinsurance, the Company's exposure to liabilities related to damage or injury from environmental pollution or from asbestos injury claims arising out of product or general liability coverage is limited. White Mountains' insurance and reinsurance subsidiaries have estimated environmental and asbestos loss and loss adjustment expense liabilities based upon several factors including facts surrounding reported cases (such as policy limits and deductibles), current law, past and projected claim activity and past settlement values for similar claims. The Company's reserves for environmental and asbestos losses at December 31, 2000 represent management's best estimate of the Company's ultimate liability based on information currently available. However, as case law expands, White Mountains may be subject to environmental and asbestos loss and loss adjustment expense liabilities beyond that intended by policy coverage. Furthermore, in the event that current case law is expanded to include claims not contemplated in the establishment of White Mountains' recorded environmental and asbestos loss and loss adjustment expense reserves, the Company believes that it is unlikely that these claims will have a material adverse effect on its financial condition or liquidity. Nonetheless, due to the expansion of coverage and liability allowed under case law in the past and the possibilities of similar interpretations in the future, additional increases in environmental and asbestos loss reserves may emerge which would adversely affect the Company's financial position. Loss reserve additions arising from such future unfavorable case law interpretations cannot be reasonably estimated at the present time.

White Mountains' insurance and reinsurance subsidiaries enter into reinsurance contracts from time to time to protect their businesses from losses due to poor risk diversification, to manage their operating leverage ratios and to limit ultimate losses arising from catastrophic events. The majority of such reinsurance contracts are executed through excess of loss treaties and catastrophe contracts under which the reinsurer indemnifies for a specified part or all of certain types of losses over stipulated amounts arising from any one occurrence or event. To a lesser extent, White Mountains has entered into quota share treaties with reinsurers under which all risks meeting prescribed criteria are covered on a pro-rata basis. The amount of each risk retained by White Mountains is subject to maximum limits which vary by line of business and type of coverage. Retention limits are continually reviewed and are revised periodically as the capacity to underwrite risks changes. Amounts related to reinsurance contracts are recorded in accordance with SFAS No. 113, "Accounting and Reporting for Reinsurance of Short-Duration and Long-Duration Contracts".

Amounts recoverable from reinsurers are estimated in a manner consistent with the claim liability associated with the reinsured policies. The collectibility of reinsurance recoverables is subject to the solvency of the reinsurers. The Company is selective in regard to its reinsurers, placing reinsurance with only those reinsurers with a strong financial condition, industry ratings and underwriting ability. Management monitors the financial condition and ratings of its reinsurers on an ongoing basis.

Reinsurance premiums, commissions, expense reimbursements and reserves related to reinsured business are accounted for on a basis consistent with those used in accounting for the original policies issued and the terms of the reinsurance contracts. Premiums ceded to other companies are reported as a reduction of premiums written. Amounts applicable to reinsurance ceded for unearned premium reserves (e.g., prepaid reinsurance premiums) have been included as a component of other assets. Expense allowances received in connection with reinsurance ceded have been accounted for as a reduction of the related policy acquisition costs and are deferred and amortized accordingly. Fund held under reinsurance treaties represent contractual payments due to the reinsurer that White Mountains has retained to secure obligations of the freinsurer. Such amounts are recorded as liabilities in the consolidated financial statements.

DEFERRED CREDITS AND GOODWILL

As of December 31, 2000 and 1999, White Mountains had deferred credits of \$92.2 million and \$100.6 million, respectively and goodwill of \$25.4 million and \$3.6 million, respectively. The deferred credits and goodwill resulted principally from the acquisition activities outlined below.

In May 2000 White Mountains completed its acquisition of the reinsurance operations of Risk Capital for \$20.3 million in cash plus related expenses. Because the cost of Risk Capital was more than the fair value of its net identifiable assets at that date, White Mountains recorded \$24.9 million in goodwill at acquisition (\$23.3 million at December 31, 2000) which is being amortized to income over the estimated period of benefit of ten years.

In March 2000 White Mountains acquired PCA for \$122.3 million in cash. Because the cost of PCA was less than the fair value of its net identifiable assets acquired at that date, White Mountains recorded a \$37.9 million deferred credit at acquisition (\$33.0 million at December 31, 2000) which is being amortized to income over the estimated period of benefit of six years.

In October 1999 White Mountains acquired IAG (which consisted primarily of PIC, ACIC and BICC) for \$86.7 million in cash. Because the cost of acquiring PIC, ACIC and BICC was less than the value of their net identifiable assets, the Company recorded a \$62.0 million deferred credit at acquisition (\$37.0 million and \$57.7 million at December 31, 2000 and 1999, respectively) which is being amortized to income over the estimated period of benefit of three years.

During 1999 White Mountains exercised its FSA Options and acquired 2,560,607 shares of FSA Common Stock. Because the cost of White Mountains' investment in FSA Common Stock (resulting from the exercise of the FSA Options) was less than the incremental portion of FSA's net identifiable assets it acquired at the date of exercise, White Mountains recorded a \$14.2 million deferred credit upon exercise (\$13.0 million at December 31, 1999) that was being amortized to income ratably over the estimated period of benefit of five years. In July 2000 White Mountains sold its investments in FSA as part of the Dexia Sale which resulted in the immediate recognition of the related unamortized deferred credit balance.

In August 1998 White Mountains acquired all the outstanding common stock of Folksamerica thereby causing Folksamerica to become a consolidated subsidiary as of that date. Because the cost of White Mountains' investment in Folksamerica was less than the value of Folksamerica's net identifiable assets at that date, White Mountains recorded a \$39.8 million deferred credit (\$22.2 million and \$29.9 million at December 31, 2000 and 1999, respectively) which is being amortized to income ratably over the estimated period of benefit of five years.

FEDERAL AND FOREIGN INCOME AND WITHHOLDING TAXES

As a result of the Redomestication, income earned by the Offshore Companies will generally be subject to an effective overall tax rate lower than that imposed by the United States, however, no tax benefits will be attained in the event of net losses incurred by such companies. Onshore Companies continue to be subject to United States income taxes. Prior to the Redomestication, the Company filed a consolidated United States Federal income tax return with its subsidiaries. The Onshore Companies continue to file United States tax returns but may no longer do so on a group- wide consolidated basis. As a result, the aggregate United States income tax liability of the Onshore Companies may be higher than it otherwise would have been if part of a consolidated tax return.

The Company is no longer subject to United States income taxes on its direct earnings. The Company's Barbados subsidiaries are generally subject to a 5% United States withholding tax on dividends received from its subsidiaries as well as a 1% Barbados income tax on taxable earnings (which include dividends received from its subsidiaries). These taxes are recorded in addition to United States income taxes accrued by its Onshore Companies.

With regard to the Onshore Companies, deferred tax assets and liabilities are recorded when a difference between an asset or liability's financial statement value and its tax reporting value exists, and for other temporary differences as defined by SFAS No. 109, "Accounting for Income Taxes". The deferred tax asset or liability is recorded based on tax rates expected to be in effect when the difference reverses.

FOREIGN CURRENCY TRANSLATION

Folksamerica operates a branch office in Toronto, Canada to service its Canadian customers and a portion of BICC's premiums are denominated in foreign currencies. Net unrealized foreign currency translation gains and losses associated with Folksamerica and BICC are reported, after tax, as a net amount in a separate component of shareholders' equity. Changes in the values of these operations due to currency fluctuations, after tax, are reported on the income statement as a component of other comprehensive net income.

ACCOUNTING STANDARDS RECENTLY ADOPTED AND ISSUED

Effective January 1, 2001, Folksamerica Reinsurance Company, PIC and ACIC are required to adopt new regulations implementing a codification of statutory accounting principles for insurers. The purpose of the codification is to enhance the consistency of the accounting treatment of assets, liabilities, reserves, income and expenses of insurers, by setting forth the accounting practices and procedures to be followed in completing annual and quarterly financial statements required by state law.

Codification will not serve to materially change the aggregate policyholders' surplus of the Company's reinsurance and insurance operations as of December 31, 2000. However, New York (Folksamerica Reinsurance Company's state of domicile), has not yet adopted certain codification standards adopted by other states that conflict with New York law, particularly those standards relating to deferred taxes and goodwill. If New York were to adopt all proposed codification provisions, codification would serve to increase the aggregate policyholders' surplus of the Company's reinsurance and insurance operations as of December 31, 2000.

In October 1998, the American Institute of Certified Public Accountants (the "AICPA") issued Statement of Position ("SOP") 98-7 entitled "Deposit Accounting: Accounting for Insurance and Reinsurance Contracts That Do Not Transfer Risk". SOP 98-7 provides guidance on how to account for all insurance and reinsurance contracts that do not transfer insurance risk. SOP 98-7 is effective for periods beginning January 1, 2000, with early adoption permitted. The adoption of SOP 98-7 did not have a material impact on White Mountains' financial position or results of operations.

In June 1998, the Financial Accounting Standards Board (the "FASB") issued SFAS No. 133, "Accounting for Derivative Instruments and Hedging Activities" which requires companies to record all derivatives on the balance sheet as either assets or liabilities and measure those instruments at fair value. The manner in which companies are to record gains and losses resulting from changes in the values of those derivatives depends on the use of the derivative and whether it qualifies for hedge accounting. The Company is not currently invested in traditional derivative financial instruments for hedging or for any other purpose. However, under SFAS 133 derivatives may be deemed to be embedded in other financial instruments. If the embedded derivatives meet certain criteria, they must be bifurcated from the original contract and separately accounted for in a manner that is consistent with other derivative financial instruments. SFAS No. 133 is effective for fiscal years beginning after June 15, 2000. White Mountains has recently completed its evaluation of the impact of the adoption of SFAS 133 and has determined that SFAS No. 133 will not have a material impact on White Mountains' financial position or results of operations.

In March 1998, the AICPA issued SOP 98-1 entitled "Accounting For the Cost of Computer Software Developed or Obtained for Internal Use" which requires the capitalization of certain prospective costs in connection with developing or obtaining software for current use. The adoption of SOP 98-1 did not have a material impact on White Mountains' financial position or results of operations.

ACCOUNTING STANDARD RECENTLY PROPOSED

In February 2001 the FASB issued an exposure draft entitled "Business Combinations and Intangible Assets - Accounting for Goodwill". This exposure draft proposes, among other things, the prospective elimination of the pooling method of accounting for business combinations and sets forth new standards concerning accounting for deferred credits, goodwill and other intangible assets arising from such activities.

With respect to deferred credits, the proposal calls for the immediate recognition of all existing and prospective deferred credits through the income statement as an extraordinary gain. With respect to goodwill, the proposed exposure draft would require companies to amortize its existing and prospective goodwill when the asset is deemed to have been impaired rather than systematically over a perceived period of benefit.

The exposure draft, in its current form, would be effective for interim and annual periods beginning after a final accounting standard is issued. As of December 31, 2000 and 1999, White Mountains had deferred credits of \$92.2 million and \$100.6 million, respectively and goodwill of \$25.4 million and \$3.6 million, respectively.

EARNINGS PER SHARE

Basic earnings per Share amounts are based on the weighted average number of Shares outstanding. Diluted earnings per Share amounts are based on the weighted average number of Shares and the net effect of potential dilutive Shares outstanding. In 1999 and 1998 net income is reduced by an amount deemed to be reflective of the dilution to FSA's reported net income caused by its investment in FSA Preferred Stock. The following table outlines the Company's computation of earnings per Share for the years ended December 31, 2000, 1999 and 1998:

	Year Ended December 31,			
	2000	1999		
BASIC EARNINGS PER SHARE NUMERATORS (IN MILLIONS): Net income	¢407.0	* 101 0	¢ 70 5	
	===========	\$ 121.0		
Net income from continuing operations	\$312.9 =========			
Comprehensive net income	\$447.6	\$ 3.0		
DILUTED EARNINGS PER SHARE NUMERATORS (IN MILLIONS): Net income Dilution to earnings resulting from FSA Preferred Stock		\$ 121.0 (.6)		
Diluted net income	\$407.9	\$ 120.4		
Diluted net income from continuing operations	\$312.9		\$ 50.7	
Diluted comprehensive net income	\$447.6	\$ 2.4	\$ 69.2	
EARNINGS PER SHARE DENOMINATORS (IN THOUSANDS): Basic earnings per Share denominator (average Shares outstanding) Average outstanding dilutive options and warrants to acquire Shares (a)	5,895 26	5,630 472	5,866 669	
Diluted earnings per Share denominator	5,921	6,102	6,535	
BASIC EARNINGS PER SHARE (IN DOLLARS): Net income	\$69.19	\$ 21.50	\$ 13.38	
Net income from continuing operations	\$53.08	\$ 19.25	\$ 8.71	
Comprehensive net income	\$75.93		\$ 11.87	
DILUTED EARNINGS PER SHARE (IN DOLLARS): Net income	\$68.89	\$ 19.73	\$ 11.94	
Net income from continuing operations	\$52.84	\$ 17.66	\$ 7.75	
Comprehensive net income	\$75.60	\$.39	\$ 10.58	

(a) See Note 10 for detailed information concerning outstanding dilutive options and warrants to acquire Shares.

FORMER INSURANCE OPERATIONS.

In 1991, White Mountains sold Fireman's Fund to Allianz of America, Inc. Since 1991 the Company had carried a reserve related to various outstanding tax issues involving the sale. In September 2000, the Company was informed that the Internal Revenue Service agreed with the position taken by White Mountains in its 1991 tax return and, on October 19, 2000, the Company received the Technical Advice Memorandum from the Internal Revenue Service's National Office confirming this conclusion. As a result, the Company released a \$95.0 million reserve during 2000 to income which is presented as a gain from discontinued insurance operations.

MORTGAGE BANKING OPERATIONS

On May 1, 1999, White Mountains concluded the Mortgage Banking Sale which encompassed substantially all the mortgage banking assets of WMSC and received net proceeds totalling \$180.6 million. Mortgage banking assets and liabilities that were not part of the Citibank sale principally included WMSC's investments in derivative instruments (which were fully liquidated in 1999), various residual mortgage assets and preferred stock, each of which were substantially liquidated during 1999. White Mountains recorded a \$17.9 million pretax (\$11.6 million after tax) gain on the sale of its mortgage banking net assets (which is net of anticipated future liabilities) during 1999. As a result of White Mountains' decision to dispose of its net mortgage banking assets, these activities are shown as discontinued operations herein.

Net cash provided from (used for) discontinued mortgage banking operations totalled \$(.1) million, \$(16.4) million and \$13.3 million for the years ended December 31, 2000, 1999 and 1998, respectively.

Summary condensed financial results of discontinued mortgage banking operations follow:

CONDENSED STATEMENTS OF NET ASSETS

Millions		Dece 2000	ember 3	1, 1999
ASSETS:				
Cash and investments Residual mortgage assets, net Other assets	\$	17.7 7.6 10.4	\$	13.8 26.9 17.8
Total assets	\$	35.7	\$	58.5
LIABILITIES:				
Accounts payable and other liabilities	\$	19.5	\$	42.2
Net assets of discontinued mortgage banking operations	\$	16.2	\$	16.3

Millions	2000	Year Ended Decemb 1999	er 31, 1998
REVENUES:			
Net investment income and investment gains	\$ 4.2	\$ 27.9	\$ 81.6
Net gain on sales of mortgages	-	25.4	86.8
Net mortgage servicing revenue	-	10.9	43.3
Other mortgage operations revenue	-	12.0	47.1
Total revenues	4.2	76.2	258.8
EXPENSES:			
Compensation and benefits	-	28.5	78.7
Interest expense	-	24.5	70.2
General expenses	4.2	19.3	59.5
Total expenses	4.2	72.3	208.4
Pretax earnings		3.9	50.4
Income tax provision	-	(1.7)	(19.3)
Net income before preferred stock dividends		2.2	31.1
Preferred stock dividends	-	(1.2)	(3.7)
Net income from discontinued mortgage banking operations	 \$ -	\$ 1.0	\$ 27.4

NOTE 3. REINSURANCE OPERATIONS

On August 18, 1998, White Mountains acquired all of the outstanding Folksamerica Common Stock for \$169.1 million thereby causing Folksamerica to become a consolidated subsidiary of White Mountains as of that date. Prior to that date, White Mountains owned a 50% non-consolidated economic interest in Folksamerica.

Supplemental condensed unaudited pro forma financial information for the year ended December 31, 1998, which assumes that White Mountains' acquisition of Folksamerica had occurred as of January 1, 1998, follows:

	(Unaudited)
Millions, except per Share amounts	Pro Forma Year Ended December 31, 1998
Total revenues	\$ 576.3
Net income from continuing operations	\$ 95.0
Comprehensive net income	\$ 95.8
BASIC EARNINGS PER SHARE: Net income Comprehensive net income	\$ 16.19 \$ 16.33
DILUTED EARNINGS PER SHARE: Net income Comprehensive net income	\$ 14.46 \$ 14.59

The unaudited pro forma information presented does not purport to represent what White Mountains' results of operations actually would have been had White Mountains acquired all the outstanding capital stock of Folksamerica as of January 1, 1998, or to project White Mountains' results of operations for any future date or period.

LOSS AND LOSS ADJUSTMENT EXPENSE RESERVE ACTIVITY

The following table summarizes White Mountains' loss and loss adjustment expense reserve activity relating to Folksamerica for the years ended December 31, 2000 and 1999 and the interim period from August 18, 1998 to December 31, 1998:

Millions		iod Ended Dece	mber 31,
	2000	1999	1998
Gross beginning balance Less beginning reinsurance recoverable on unpaid losses		\$ 723.2 (129.0)	\$ 726.1 (124.1)
Net loss and loss adjustment expense reserves	645.9	594.2	602.0
Loss and loss adjustment expense reserves acquired - USF Re (1) Loss and loss adjustment expense reserves acquired - PCA (1) Loss and loss adjustment expense reserves acquired - Risk Capital (1) Loss and loss adjustment expense reserves transferred and assumed (2)		106.5 - - -	- - -
Losses and loss adjustment expenses incurred relating to: Current year losses Prior year losses	247.9 22.9	152.9 29.3	
Total incurred losses and loss adjustment expenses	270.8	182.2	59.7
Loss and loss adjustment expenses paid relating to: Current year losses Prior year losses		(55.4) (181.6)	(13.0) (54.5)
Total loss and loss adjustment expense payments	(415.0)	(237.0)	(67.5)
Net ending balance Plus ending reinsurance recoverable on unpaid losses	797.9 702.8		
Gross ending balance	\$1,500.7	\$ 782.1	\$ 723.2

 Reinsurance recoverables on unpaid losses acquired in the Risk Capital, PCA and USF Re acquisitions were \$59.1 million, \$153.3 million and \$21.8 million, respectively.

(2) Includes \$270.6 million of loss reserves ceded to Imagine Re during 2000. See Note 6.

Prior year reserve strengthening of \$22.9 million affected most of Folksamerica's segments including the portfolios acquired with USF Re and Risk Capital.

During 1999 Folksamerica acquired USF Re. Significant assets and liabilities acquired through USF Re included \$204.1 million of cash and investments and \$106.5 million of net loss and loss adjustment expense reserves. The purchase consideration included the issuance of a \$20.8 million, five-year note by Folksamerica (which could be reduced by adverse loss development at USF Re post acquisition). Incurred losses for the years ended December 31, 2000 and 1999 related to prior accident years include adverse loss development on USF Re acquired reserves which resulted in a reduction of the USF Re seller note of \$6.8 million and \$14.0 million during those periods, respectively.

On March 31, 2000, Folksamerica acquired PCA. Significant assets and liabilities acquired through PCA included \$339.8 million of cash and investments and \$253.8 million of net loss and loss adjustment expense reserves.

On May 5, 2000, Folksamerica acquired the reinsurance operations of Risk Capital. Significant assets and liabilities acquired through the Risk Capital transaction included \$249.9 million of cash and investments, \$108.6 million of premiums receivable, \$312.6 million of net loss and loss adjustment expense reserves and \$82.0 million of net unearned reinsurance premiums. In addition, the Risk Capital acquisition provided White Mountains with two specialty underwriting units (Accident & Health and Marine) and several significant new treaty clients.

ASBESTOS AND ENVIRONMENTAL LOSS AND LOSS ADJUSTMENT EXPENSE RESERVE ACTIVITY

The following table summarizes Folksamerica's reported environmental and asbestos loss and loss adjustment expense reserve activities gross and net of reinsurance for the years ended December 31, 2000, 1999 and 1998:

Millions		Per	iod Ended D	ecember 31,		
	2000		199	9	199	8
	Gross	Net	Gross	Net	Gross	Net
Asbestos: Beginning balance	\$ 34.5	\$ 22.3	\$ 29.5	\$ 17.9	\$ 18.7	\$ 14.5
Incurred losses and loss adjustment expenses Paid losses and loss adjustment expenses	3.8	2.7	13.0	9.3	\$ 10.7 14.7 (3.9)	6.6
Ending balance	33.6	21.0	34.5	22.3	29.5	17.9
Environmental: Beginning balance Incurred losses and loss adjustment expenses Paid losses and loss adjustment expenses	9.8 1.4 (2.3)		14.9 .5 (5.6)			10.3 2.3 (1.7)
Ending balance	8.9	7.9	9.8	8.4	14.9	10.9
Total asbestos and environmental: Beginning balance Incurred losses and loss adjustment expenses Paid losses and loss adjustment expenses	44.3 5.2 (7.0)	30.7 3.5 (5.3)	44.4 13.5 (13.6)	28.8 9.1 (7.2)		24.8 8.9 (4.9)
Ending balance	\$ 42.5	\$ 28.9	\$ 44.3	\$ 30.7	\$ 44.4	\$ 28.8

ADDITIONAL REINSURANCE OPERATIONS INFORMATION

Folksamerica's policyholders' surplus, as reported to various regulatory authorities as of December 31, 2000 and 1999, was \$443.9 million and \$338.5 million, respectively. The large increase in surplus from 1999 to 2000 resulted principally from the contribution of ACIC and \$80.0 million in cash to Folksamerica Reinsurance Company during the 2000 fourth quarter. Folksamerica's statutory net income (loss) for the years ended December 31, 2000 and 1999 and for the period from August 18, 1998 to December 31, 1998 was \$(20.0) million, \$48.6 million and \$9.0 million, respectively. The principal differences between Folksamerica's statutory amounts and the amounts reported in accordance with GAAP include deferred taxes, deferred acquisition costs, gains recognized under retroactive reinsurance contracts and market value adjustments for debt securities. Folksamerica's statutory policyholders' surplus at December 31, 2000 was in excess of the minimum requirements of relevant state insurance regulations.

Under the insurance laws of the state of New York, Folksamerica may pay dividends only from earned surplus as determined on a statutory basis. Generally, the maximum amount of cash dividends that a New York-domiciled company may pay out of its statutory earned surplus without prior regulatory approval in any twelve month period is the lesser of the company's statutory net investment income or 10% of statutory surplus. Accordingly, there is no assurance that dividends may be paid by Folksamerica in the future. At December 31, 2000, Folksamerica had the ability to pay a dividend to its shareholder of \$44.4 million without prior approval of regulatory authorities.

During the year ended December 31, 2000, Folksamerica received approximately 56.4% of its gross reinsurance premiums written from three major reinsurance brokers as follows: (i) E.W. Blanch - 21.6%; (ii) Guy Carpenter and affiliates - 17.6%; and (iii) AON Re, Inc. - 17.2%. During the year ended December 31, 2000, Folksamerica received no more than 10% of its gross reinsurance premiums from any individual ceding company.

NOTE 4. CONSOLIDATED INSURANCE OPERATIONS

On October 15, 1999, White Mountains completed its acquisition of IAG for \$86.7 million in cash. IAG's principal operating subsidiaries are PIC, a commercial and personal lines writer, and ACIC and BICC, both of which are in run- off.

On June 17, 1999, White Mountains completed the VGI Sale and received net proceeds of \$139.0 million in cash after receiving a special dividend prior to the closing of \$76.6 million (net of related tax liabilities) consisting of cash, investment securities and the common stock of Waterford. The VGI Sale resulted in a pretax gain of \$88.1 million (\$53.8 million after tax). As part of the VGI Sale, White Mountains has provided Unitrin, Inc. with certain adverse loss development protections that will be settled as of December 31, 2002. During 2000 White Mountains provided \$5.4 million in reserves for such adverse loss development protections. Waterford was sold to a third party on January 5, 2001 for cash proceeds of \$23.8 million.

For the years ended December 31,1999 and 1998 VGI's contribution to the Company's consolidated net income was not material.

LOSS AND LOSS ADJUSTMENT EXPENSE RESERVE ACTIVITY

The following table summarizes loss and loss adjustment expense reserve activity for White Mountains' consolidated property and casualty insurance operations for the years ended December 31, 2000, 1999 and 1998:

Millions		Ended Decembe	er 31,
	2000	1999	1998
Gross beginning balance Less beginning reinsurance recoverable on unpaid losses	\$ 68.9 (32.8)	\$ 88.5 (8.9)	\$ 71.9 (8.7)
Net loss and loss adjustment expense reserves	36.1	79.6	63.2
Loss and loss adjustment expense reserves sold - VGI Loss and loss adjustment expense reserves acquired - IAG Losses and loss adjustment expenses incurred relating to:	-	(87.8) 22.5	-
Current year losses Prior year losses	16.2 .7	57.5 2.6	108.4 6.7
Total incurred losses and loss adjustment expenses	16.9	60.1	115.1
Loss and loss adjustment expense payments	(21.1)	(38.3)	(98.7)
Net ending balance Plus ending reinsurance recoverable on unpaid losses	31.9 23.7	36.1 32.8	79.6 8.9
Gross ending balance	\$ 55.6	\$ 68.9	\$ 88.5

Gross ending loss and loss adjustment expense balances for 2000 and 1999 include approximately \$27.2 million and \$29.3 million of environmental and asbestos reserves, respectively, relating to the run-off activities of ACIC and BICC. Until 1983, ACIC wrote (and subsequently ceded a portion to BICC) primary and excess liability insurance for many large national and international chemical, manufacturing and pharmaceutical companies, as well as facultative and treaty reinsurance for the same types of risks. Since 1983, management has concentrated its run-off efforts on commuting its loss exposures with its insureds. Due to the limited number of remaining policies under which a claim could be filed and the length of time since the policies were written, management expects any future development related to these reserves to be immaterial.

The aggregate policyholders' surplus of PIC and ACIC at December 31, 2000 and 1999, as reported to regulatory authorities, was \$60.5 million and \$70.6 million, respectively. Statutory net loss for the year ended December 31, 2000 and the period from October 16, 1999 to December 31, 1999 for PIC and ACIC totalled \$1.7 million and \$3.9 million, respectively. The principal differences between PIC and ACIC's statutory amounts and the amounts reported in accordance with GAAP (PIC and ACIC's aggregate shareholder's equity was \$67.6 million and \$73.0 million at December 31, 2000 and 1999, respectively, and its net income (loss) was \$1.0 million and \$(3.3) million for the year ended December 31, 2000 and the period from October 16, 1999 to December 31, 1999, respectively) include deferred taxes, deferred acquisition costs and market value adjustments for debt securities. PIC and ACIC's statutory policyholders' surplus at December 31, 2000 was in excess of the minimum requirements of relevant state insurance regulations.

Under the insurance laws of the states under which PIC and ACIC are domiciled, PIC and ACIC may pay dividends only from earned surplus as determined on a statutory basis. Generally, the maximum amount of cash dividends that PIC and ACIC may pay out of their statutory earned surplus without prior regulatory approval in any twelve month period is the greater of the company's statutory net income or 10% of statutory surplus. Accordingly, there is no assurance that dividends may be paid by PIC and ACIC in the future. At December 31, 2000, \$4.1 million of PIC and ACIC's total statutory surplus was available for the payment of dividends to its shareholder without prior approval of regulatory authorities.

NOTE 5. INVESTMENT SECURITIES

White Mountains' net investment income is comprised primarily of interest income associated with the fixed maturity investments of its consolidated insurance and reinsurance operations, dividend income from its equity investments and interest income from its short-term investments. Net investment income for 2000, 1999 and 1998 consisted of the following:

Millions	Yea 2000	r Ended Decembe 1999	er 31, 199
Investment income:			
Fixed maturity investments	\$ 63.1	\$ 53.6	\$ 28.
Common equity securities	2.8	3.2	3.
Short-term investments	19.9	6.1	3.
Other	.7	.3	2.
Total investment income	86.5	63.2	
Less investment expenses and other charges	(.6)	(1.3)	(.8
Net investment income, before tax	\$ 85.9	\$ 61.9	\$ 36.

The composition of pretax realized investment gains (losses) for investments sold consisted of the following:

Millions	2000	Year Ended	December 1999	31,	1998
Fixed maturity investments Common equity securities Other investments	\$(9.2) (4.7) 5.1	\$.7 61.3 7.6	\$	1.6 22.6 46.8
Net realized investment gains (losses)	\$(8.8)	 \$	69.6	 \$	71.0

The components of White Mountains' change in unrealized investment gains, after tax, as recorded on the statements of income and comprehensive income were as follows:

	Vos	r Ended Decembe	
Millions	2000	1999	1998
Gross realized gains from investment securities	\$ 22.3	\$ 90.4	\$ 74.0
Gross realized losses from investment securities	(31.1)	(20.8)	(3.0)
Net realized gains (losses) from investment securities sold	(8.8)	69.6	71.0
Recognition of unrealized gains from investment in unconsolidated affiliate sold	32.5	-	
Recognition of net unrealized gains for investments sold, before tax	23.7	69.6	71.0
Income taxes attributable to investments sold	(7.8)	(24.4)	(24.9)
Recognition of net unrealized gains for investments sold, after tax	\$ 15.9	\$ 45.2	\$ 46.1
Net change in unrealized gains for investments securities held	\$ 65.1	\$ (11.7)	\$ 25.0
Net change in unrealized gains from investments in unconsolidated affiliates held	6.2	(98.9)	33.7
Net change in unrealized investment gains for investments held, before tax	71.3	(110.6)	58.7
Income taxes attributable to investments held	(15.0)	36.9	(20.6)
Net change in unrealized gains for investments held, after tax	56.3	(73.7)	38.1
Recognition of net unrealized gains for investments sold, after tax	(15.9)	(45.2)	(46.1)
Change in net unrealized investment gains, after tax	\$ 40.4	\$(118.9)	\$ (8.0)

The components of White Mountains' ending net unrealized investment gains and losses on its investment portfolio and its investments in unconsolidated insurance affiliate(s) were as follows:

	December				
Millions	2000	1999			
Investment securities:					
Gross unrealized investment gains	\$ 70.5	\$ 26.5			
Gross unrealized investment losses	(12.0)	(42.7)			
Net unrealized gains (losses) from investment securities	58.5	(16.2)			
Net unrealized gains from investments in unconsolidated insurance affiliates	2.9	30.0			
Total net unrealized investment gains, before tax	61.4	13.8			
Income taxes attributable to such gains	(13.8)	(6.6)			
Total net unrealized investment gains, after tax	\$ 47.6	\$ 7.2			

The cost or amortized cost, gross unrealized investment gains and losses, and carrying values of White Mountains' fixed maturity investments as of December 31, 2000 and 1999, were as follows:

	DECEMBER 31, 2000							
Millions	Cost or amortized cost	Gross unrealized gains	Gross unrealized losses	Carrying value				
U. S. Government and agency obligations Debt securities issued by industrial corporations Municipal obligations	\$ 309.4 310.6 265.7	\$ 9.5 3.9 3.0	\$ (.7) (1.6) (.2)	\$ 318.2 312.9 268.5				
Mortgage-backed securities Foreign government obligations Preferred stocks	102.4 52.2 22.7	1.3 .3 .6	(.4) (.1) -	103.3 52.4 23.3				
Total fixed maturity investments	\$1,063.0	\$18.6	\$(3.0)	\$1,078.6				

		December	r 31, 1999	
Millions	Cost or amortized cost	Gross unrealized gains	Gross unrealized losses	Carrying value
U. S. Government and agency obligations Debt securities issued by industrial corporations Municipal obligations Mortgage-backed securities Foreign government obligations	\$351.5 330.8 132.0 93.1 50.5	\$.6 .4 .7 .1 .1	(14.5) (11.8) (4.0) (3.9) (1.1)	\$337.6 319.4 128.7 89.3 49.5
Total fixed maturity investments	\$957.9	\$1.9	\$(35.3)	\$924.5

The cost or amortized cost and carrying value of White Mountains' fixed maturity investments at December 31, 2000 is presented below by contractual maturity. Actual maturities could differ from contractual maturities because borrowers may have the right to call or prepay certain obligations with or without call or prepayment penalties.

	DECEMBER 31, 20						
Millions Due in one year or less Due after one year through five years Due after five years through ten years	Millions	Cost or amortized cost	Carrying value				
	\$ 70.7 470.4 287.2	\$ 70.9 478.1 295.8					
Due after ten years Mortgage-backed securities Preferred stocks	109.6 102.4 22.7	107.2 103.3 23.3					
Total	\$1,063.0	\$1,078.6					

The cost or amortized cost, gross unrealized investment gains and losses, and carrying values of White Mountains' common equity securities and other investments as of December 31, 2000 and 1999, were as follows:

	DECEMBER 31, 2000						
Millions	Cost or amortized cost	Gross unrealized gains	Gross unrealized losses	Carrying value			
Common equity securities	\$127.5	\$21.1	\$(3.8)	\$144.8			
Other investments	\$117.3	\$30.8	\$(5.2)	\$142.9			

		December 31, 1999					
Millions	Cost or amortized cost	Gross unrealized gains	Gross unrealized losses	Carrying value			
Common equity securities	\$100.4	\$14.8	\$(6.8)	\$108.4			
Other investments	======================================	\$ 9.8	\$ (.6)	\$ 68.3			

White Mountains' consolidated insurance and reinsurance operations are required to maintain deposits with certain insurance regulatory agencies in order to maintain their insurance licenses. The fair value of such deposits totalled \$69.1 million and \$59.5 million as of December 31, 2000 and 1999, respectively.

Sales and maturities of investments, excluding short-term investments, totalled \$582.1 million, \$530.1 million and \$270.3 million for the years ended December 31, 2000, 1999 and 1998, respectively. There were no non-cash exchanges or involuntary sales of investment securities during 2000, 1999 and 1998.

NOTE 6. THIRD PARTY REINSURANCE

In the normal course of business, White Mountains' insurance and reinsurance subsidiaries seek to limit losses that may arise from catastrophes or other events that may cause unfavorable underwriting results by reinsuring certain levels of risk in various areas of exposure with other insurance enterprises or reinsurers. White Mountains remains liable for risks reinsured with third parties to the extent that the reinsurer is unable to honor its obligations under reinsurance contracts at the time of loss.

REINSURANCE OPERATIONS

Folksamerica has exposure to losses caused by hurricanes, earthquakes, winter storms, windstorms and other catastrophic events. Folksamerica continually assesses its concentration of underwriting exposures in catastrophe prone areas and develops strategies to manage this exposure, primarily through the purchase of catastrophe reinsurance. Folksamerica's current catastrophe reinsurance program provides coverage for losses incurred as a result of individual catastrophic events between \$7.5 million and \$17.5 million.

In the fourth quarter of 2000 Folksamerica purchased reinsurance coverage from Imagine Re which was designed to reduce its statutory operating leverage and to provide the following protections: (i) adverse development protection of up to \$65.0 million on the run-off of loss and unearned premium reserves acquired in the Risk Capital transaction and any additional premium reported for the 2000 underwriting year; and (ii) \$50.0 million of adverse development protection from the reserves remaining from the USF Re acquisition and remaining environmental and asbestos exposures. Amounts related to reserves transferred to Imagine Re for liabilities incurred as a result of past insurable events have been accounted for as retroactive reinsurance in accordance with SFAS No. 113. Reinsurance recoverables of \$321.7 million were recorded at December 31, 2000, to reflect the loss reserves transferred (\$250.0 million), fourth quarter incurred losses attached to the premium run-off (\$51.1 million) and adverse development on loss reserves transferred (\$20.6 million). At December 31, 2000, Folksamerica has recorded a \$20.6 million deferred gain related to the adverse development on loss reserves transferred which will be recognized in income over the expected settlement period of the underlying claims. Contractual premiums due to Imagine Re of \$315.0 million have been recorded as funds withheld under reinsurance treaties at December 31, 2000. Folksamerica holds a letter of credit and funds held as collateral for amounts due from Imagine Re.

At December 31, 2000 and 1999, Folksamerica had reinsurance recoverables with a carrying value of \$116.7 million and \$46.7 million, respectively, associated with London Life and Casualty Reinsurance Corporation. Folksamerica holds a letter of credit and funds held as collateral for amounts due from London Life and Casualty.

INSURANCE OPERATIONS

ACIC is a party to an aggregate excess of loss contract with Gerling Global International Reinsurance Company, Ltd. ("Gerling") to reinsure direct excess liability policies written prior to December 31, 1985. At December 31, 2000 and 1999, ACIC had reinsurance recoverables with a carrying value of \$17.0 million and \$23.3 million, respectively, with Gerling under the contract. ACIC holds a letter of credit and assets held in trust as collateral for amounts due under the Gerling contract.

The effects of reinsurance on White Mountains' written and earned premiums and on loss and loss adjustment expenses were as follows:

		Insurance (Operations	
Millions	Reinsurance Operations	ACIC, PIC and BICC (b)	Waterford	Total
YEAR ENDED DECEMBER 31, 2000				
Gross written premiums: Direct Assumed	\$	\$ 26.4 .1		\$ 32.2 484.7
Ceded				(161.7)
Net written premiums	\$ 332.6	\$ 22.7	\$ (.1)	\$ 355.2
Gross earned premiums: Direct Assumed Ceded	\$ 4.1 476.1 (167.7)	\$ 26.1 .1 (4.6)	\$ 2.3 .3 (2.3)	\$ 32.5 476.5 (174.6)
Net earned premiums	\$ 312.5	\$ 21.6	\$.3	\$ 334.4
Losses and loss adjustment expenses: Direct Assumed Ceded	\$ (.6) 468.5 (197.1)	\$ 18.3 (1.7)	\$ 2.3 .3 (2.3)	\$20.0 468.8 (201.1)
Net losses and loss adjustment expenses	\$ 270.8	\$ 16.6	\$.3	\$ 287.7

		Insuranc	e Operations		
Millions	Operations	ACIC, PIC and BICC (b)	VGI	Waterford	Total
Year ended December 31, 1999					
Gross written premiums:					
Direct Assumed	\$ 2.2 236.2	\$ 4.9	\$ 48.2 30.1		\$ 64.8 270.1
Ceded	(36.7)		(10.9)		
Net written premiums	\$ 201.7	\$ 4.2	\$ 67.4	\$ 1.4	\$ 274.7
Gross earned premiums:					
Direct	\$ 2.4	\$ 5.9			\$ 63.8
Assumed	241.0		35.1		280.6
Ceded	(32.4)	(4.1)	(13.2)	(11.5)	(61.2)
Net earned premiums	\$ 211.0	\$ 4.5	\$ 66.0	\$ 1.7	\$ 283.2
Losses and loss adjustment expenses:					
Direct	\$ 7.8	\$ 4.8			\$ 55.3
Assumed	205.4		14.6		
Ceded	(31.0)	(.2)	(6.6)	(9.7)	(47.5)
Net losses and loss adjustment expenses	\$ 182.2	\$ 6.2	\$ 52.5	\$ 1.4	\$ 242.3

					eratior	IS		
Millions	Operat.	surance ions (a)	ACIC, PI BICC	C and		Wa	terford	Tota
Year ended December 31, 1998					 			
Gross written premiums:								
Direct	\$	1.9	\$	-	\$ 		9.6	\$ 107.8
Assumed		81.2						152.6
Ceded		(9.4)		-	(2.7)		(9.7)	(21.8)
Net written premiums	\$	73.7	\$	-	\$ 163.5	\$	1.4	\$ 238.6
Gross earned premiums:					 			
Direct	\$	2.0	\$	-	\$ 98.0	\$	6.4	\$ 106.4
Assumed		92.2		-	68.3		1.0	161.5
Ceded		(8.8)		-	(6.6)		(6.5)	(21.9)
Net earned premiums	\$	85.4	\$		\$ 159.7	\$.9	\$ 246.0
Losses and loss adjustment expenses:					 			
Direct	\$	4.4	\$	-	\$ 79.5	\$	3.6	\$ 87.5
Assumed		74.3		-	37.9			112.9
Ceded		(19.0)		-	(2.8)		(3.8)	(25.6)
Net losses and loss adjustment expenses	\$	59.7	\$		\$ 114.6	\$.5	\$ 174.8

(a) Excludes premiums and loss and loss adjustment expenses from January 1, 1998 to August 17, 1998 during which time Folksamerica was not a consolidated subsidiary of White Mountains. See Note 3.

(b) Excludes premiums and loss and loss adjustment expenses from January 1, 1998 to October 14, 1999 during which time ACIC, PIC and BICC were not consolidated subsidiaries of White Mountains. See Note 4.

NOTE 7. DEBT

SHORT-TERM DEBT

At December 31, 2000 the Company had no short-term debt outstanding. At December 31, 1999 the Company had short-term debt outstanding of \$4.0 million consisting of 7.39% medium-term notes which were repaid on their maturity date in February 2000.

In addition, the Company has from time-to-time entered into revolving credit agreements whereby it may borrow funds at short-term market interest rates. As of December 31, 1999 the Company could borrow up to \$50.0 million under a short-term debt facility which the Company let expire in 2000. There were no borrowings outstanding under this facility at December 31, 1999.

LONG-TERM DEBT

Long-term debt outstanding as of December 31, 2000 and 1999 consisted of the following:

Nilliona		2000	December	31,	, 1999
Millions	2				
The Company:					
Medium-term notes	\$	96.3		\$	96.4
Less net discounts		(.3)			(.4)
Total		96.0			96.0
Folksamerica:					
Revolving credit agreement		-		1	100.0
USF Re seller note		-			6.8
Total		-		-	106.8
Total long-term debt	\$	96.0		\$2	202.8

At December 31, 2000 the Company had \$96.3 million in principal amount of outstanding medium-term notes with an average maturity of 2.6 years and a yield to maturity of 7.87%. During 2000 and 1999 the Company repurchased \$.1 million and \$15.9 million, respectively, in principal amount of its medium-term notes due in February 2003.

As part of the Folksamerica acquisition in 1998, White Mountains agreed to repay or refinance Folksamerica's existing long-term indebtedness by utilizing a six-year revolving credit agreement whereby Folksamerica could borrow up to \$120.0 million at market interest rates. Outstanding borrowings of \$115.0 million were repaid in December 2000 and the facility was terminated.

As part of its 1999 acquisition of USF Re, Folksamerica issued a \$20.8 million, five-year note which could be reduced by adverse loss development at USF Re post acquisition. During 2000 and 1999, Folksamerica reduced the principal amount of the USF Re note by \$6.8 million and \$14.0 million, respectively, in response to post acquisition adverse loss development experienced on its loss reserves assumed from USF Re.

Total interest paid by White Mountains for its short-term and long-term indebtedness was \$16.1 million, \$15.6 million and \$13.3 million in 2000, 1999 and 1998, respectively.

Of the \$96.3 million in medium-term notes outstanding at December 31, 2000, \$86.3 million of such notes mature in 2003 with the balance of the notes maturing in 2008.

NOTE 8. INCOME TAXES

In connection with the Redomestication, the Company and certain of its subsidiaries changed their domicile to either Bermuda or Barbados while certain other subsidiaries remained domiciled in the United States. As a result, income earned by the Offshore Companies will generally be subject to an effective overall tax rate lower than that imposed by the United States, however, no tax benefits will be attained in the event of net losses incurred by such companies. These factors may serve to increase or decrease White Mountains' effective tax rate for 1999 and beyond, depending on the events and circumstances occurring during such periods.

In connection with the Redomestication, the Company was treated as if it sold all of its directly owned assets in a fully taxable transaction in which gains, but not losses, were recognized. The Company incurred a tax liability upon the Redomestication of approximately \$13.5 million.

The total income tax provision consisted of the following:

	Year Ended December 31,						
Millions	2000	1999	1998				
United States income tax provision State and local income tax provision United States withholding tax and foreign income tax provision	\$ 38.0 1.7 2.8	\$ 47.0 6.0 .1	\$ 26.9 1.6				
Total income tax provision	\$ 42.5	\$ 53.1	\$ 28.5				
Net income tax payments	\$ 54.5	\$ 14.1	\$ 35.7				
Income taxes recorded directly to shareholders' equity related to: Changes in net unrealized investment gains and losses Changes in net foreign currency translation gains and losses	\$ 7.2 \$ (.4)	\$(61.3) \$.5	\$(4.3) \$ (.5)				

The components of the income tax provision (benefit) on pretax earnings follow:

	Ye	Year Ended December 31,			
Millions	2000	1999	1998		
Current Deferred	\$ 58.2 (15.7)	\$14.3 38.8	\$ 30.3 (1.8)		
Total income tax provision on pretax earnings	\$ 42.5	\$53.1	\$ 28.5		

Deferred income taxes reflect the net tax effects of temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts received for tax purposes. Deferred income tax assets and liabilities are shown net in circumstances where a consolidated income tax return is filed. An outline of the significant components of White Mountains' deferred tax assets and liabilities follows:

	Decer	mber 31,
Millions	2000	1999
Deferred income tax assets related to:		
Discounting of loss reserves	\$ 60.3	\$ 45.3
Net operating loss carryforwards (which principally expire in 2010)	29.3	.9
Unearned insurance and reinsurance premiums	7.2	5.7
Deferred gain on reinsurance contract	7.1	-
Compensation and benefit accruals	4.1	4.4
Other items	(2.9)	(3.8)
Total deferred income tax assets	\$ 105.1	\$ 52.5

		Dece	mber 3	1,	
Millions		2000	1999		
Deferred income tax liabilities related to: Earnings from unconsolidated insurance affiliates Net unrealized investment gains Other items	\$.2 2.6	\$	18.1 13.9 5.5	
Total deferred income tax liabilities (recorded in accounts payable and other liabilities)	\$	2.8	\$	37.5	

The Company believes that it is more likely than not that results of future operations will generate sufficient taxable income to realize the deferred tax asset balances carried as of December 31, 2000 and 1999.

A reconciliation of taxes calculated using the 35% United States statutory rate (the tax rate at which the majority of the Company's worldwide operations are subject to) to the income tax provision on pretax earnings follows:

	Yea	r Ended December	31,
Millions	2000	1999	1998
Tax provision at the United States statutory rate	\$ 124.4	\$56.5	\$27.9
Differences in taxes resulting from:	(2, 2)		
Deferred credit amortization and purchase price adjustments	(6.3)	(7.9)	(.9)
Tax reserve adjustments	5.5	6.1	5.4
State income taxes, net	1.2	3.9	1.0
Non-United States net earnings	(88.6)	(3.6)	-
United States income tax incurred upon the Redomestication	11.0	2.5	-
Tax exempt interest and dividends	(3.9)	(3.6)	(3.8)
Foreign and withholding taxes	2.8	.1	-
Other, net	(3.6)	(.9)	(1.1)
Total income tax provision on pretax earnings	\$ 42.5	\$53.1	\$28.5

The non-United States component of pretax earnings was \$395.9 million and \$9.0 million for the year ended December 31, 2000 and 1999, respectively.

During 2000, the Company released a $95.0\ million$ tax reserve relating to its 1991 sale of Fireman's Fund. See Note 2.

NOTE 9. RETIREMENT AND POSTRETIREMENT PLANS

Prior to the Redomestication, White Mountains terminated two non-qualified compensation plans and paid participants a total of \$88.6 million in full satisfaction of White Mountains' long-term obligations. This payment resulted in an incremental pretax cost to the Company of \$15.2 million, however, this action provided the Company with increased tax deductible expenses during 1999 and served to reduce future compensation and benefit expenses.

During 1999 the Board mandated deferrals of compensation earned by certain of its executive officers that was payable in connection with the Redomestication. Deferred balances are invested in phantom shares which are valued based on the market value of Shares. As of December 31, 2000 and 1999, these balances totalled \$7.7 million and \$2.9 million, respectively.

White Mountains has various defined contribution employee savings plans for the benefit of substantially all its employees. The costs of these plans are not material to White Mountains' financial statements.

Folksamerica has a defined benefit pension plan for the benefit of its employees and for former employees of the Company's discontinued mortgage banking operations. Benefits under this plan are based on years of service and each employee's highest average eligible compensation over the last five consecutive years of employment. The fair value of the assets in Folksamerica's defined benefit plan exceed its current and future employee obligations under the plan, therefore, the cost of this plan is not material to White Mountains' financial statements.

White Mountains' does not have any significant ongoing postretirement benefit plan obligations.

NOTE 10. EMPLOYEE SHARE-BASED COMPENSATION PLANS

White Mountains' Incentive Plan provides for granting to participants of the Company (and certain of its subsidiaries) various types of share-based incentive awards including performance shares and options to acquire Shares. At December 31, 2000, 206,400 Shares remained available for grant under the Incentive Plan.

Performance shares are conditional grants of a specified maximum number of Shares or an equivalent amount of cash. The grants are generally payable (subject to the attainment of a specified after tax return on equity) at the end of a three year period or as otherwise determined by the Compensation Committee of the Board. The financial goal for full payment of the performance shares is the achievement of a 13% annual after tax return on equity (as specifically defined by the Compensation Committee) as measured over the applicable performance periods. Performance share payments can double if returns significantly exceed 13%. The Compensation Committee consists solely of disinterested, non-management Directors.

For the three year performance periods beginning 2000, 1999 and 1998 the Company granted a total of 34,000, 31,300 and 47,800 performance shares, respectively, to its employees. No performance shares were paid in 2000. During 1999 and 1998 the Company paid a total of 141,650 and 47,129 performance shares, respectively, to its participants in cash and Shares. Performance shares paid during 1999 included 58,100 performance shares relating to the period from 1996 to 1998, 40,300 performance shares paid early relating to the period from 1997 to 1999 and 43,250 performance shares paid early relating to the period from 1998 to 2000.

White Mountains expenses performance shares ratably over the performance period assuming full vesting at the current market values of Shares. During 2000, 1999 and 1998, White Mountains recorded \$25.8 million, \$6.1 million and \$7.2 million, of pretax performance share expense, respectively. The Company's performance share expense for 2000 was significantly higher than that of prior years as a result of: (i) a significant increase in the market value of Shares during the year; and (ii) the recording of \$12.0 million of additional performance share expense at an amount greater than 100% in light of the achievement of above plan results for 2000.

Folksamerica's defined contribution plan (the "Folksamerica 401(k) Plan") offers its participants the ability to invest their balances in several different investment options including Shares. As of December 31, 2000 and 1999 the Folksamerica 401(k) Plan owned less than 1% of the total Shares outstanding.

As of December 31, 2000 there were 81,000 options to acquire Shares outstanding which had an effective exercise price of \$111.59 per Share at that date. Of the 81,000 options granted during 2000, 8,100 options were vested and exercisable at December 31, 2000. There were no options to acquire Shares outstanding at December 31, 1999. The Company's Chairman formerly held warrants to acquire Shares outstanding entitling him to buy 1,000,000 Shares for \$21.66 per Share through January 2, 2002. During 1999 the Chairman exercised the warrants early in exchange for a one-time payment of \$6.0 million. This one-time payment compensated the Chairman for the estimated interest cost of borrowing the strike price and the amounts required to prematurely pay his income taxes. The 1999 exercise of warrants provided the Company with increased tax deductible expenses.

SFAS No. 123, "Accounting for Stock Based Compensation," requires disclosure regarding significant employee compensation involving the issuance of Shares and encourages companies to recognize compensation expense based on the fair value of such awards on the date of grant. White Mountains has not adopted the recognition and measurement criteria of SFAS No. 123 and alternatively has chosen to disclose the pro forma effects of SFAS No. 123 as follows:

Hillions, except per Share amounts	Year Ended December 31, 1998
let income:	
As reported	\$ 78.5
Pro forma	77.4
asic net income per Share:	
As reported	\$ 13.38
Pro forma	12.56
iluted net income per Share:	
As reported	\$ 11.94
Pro forma	\$ 11.94 11.20

The SFAS No. 123 disclosure above presents the expense of warrants to acquire Shares over the life of the award using the Black Scholes option pricing model. Significant assumptions used in the presentation include a 5.0% risk-free interest rate, an expected Share volatility of .167 and an expected life of five years. The pro forma net income figures disclosed above may not be representative of the effects on net income to be reported in future years.

SFAS No. 123 disclosures are not presented for the years ended December 31, 2000 and 1999 as the pro forma effect of vested options to acquire Shares at December 31, 2000 would have been \$.3 million or \$.06 per diluted Share (and would therefore not be material to the above presentation) and there were no options to acquire Shares outstanding at December 31, 1999.

NOTE 11. SHAREHOLDERS' EQUITY

SHARE REPURCHASES AND RETIREMENT

During 2000, 1999 and 1998 the Company repurchased 65,838 Shares for \$8.3 million, 1,020,150 Shares for \$139.5 million and 151,916 Shares for \$19.8 million, respectively. In conformance with Bermuda law, the Company retires all Shares it repurchases.

SHARES ISSUED

During 1999 the Company issued a total of 1,137,495 Shares to its employees and Directors in satisfaction of the Chairman's warrant exercise, option exercises and various employee benefit plan obligations. No Shares were issued during 2000.

DIVIDENDS ON SHARES

During 2000, 1999 and 1998 the Company declared and paid cash dividends totalling \$1.20, \$1.60 and \$1.60 per Share, respectively.

NOTE 12. SEGMENT INFORMATION

White Mountains has determined that its reportable segments include Reinsurance, Property and Casualty Insurance, Investments in Unconsolidated Insurance Affiliates and Holding Company (primarily the operations of the Company and certain of its intermediate subsidiary holding companies). Investment results are included within the segment to which the investments relate. The Company has made this determination based on consideration of the following criteria: (i) the nature of the business activities of each of the Company's subsidiaries and affiliates; (ii) the manner in which the Company's subsidiaries and affiliates are organized; (iii) the existence of primary managers responsible for specific subsidiaries and affiliates; and (iv) the organization of information provided to the Board. Management and the Board does not currently review its operating results on a geographic basis. There are no significant intercompany transactions among White Mountains' segments other than occasional intercompany sales and transfers of investment securities (gains and losses resulting from such transfers have been eliminated herein).

Certain amounts in the prior periods have been reclassified to conform with the current presentation which involved the segregation of the mortgage banking net assets, revenues and pretax earnings as discontinued operations.

Selected financial information for White Mountains' segments follows:

Millions	P Reinsurance	roperty and Casualty Insurance	Investments in Unconsolidated Affiliate(s)	Holding Company	Total
YEAR ENDED DECEMBER 31, 2000					
	• • • • • •	• • • • -	•	•	• • • • • •
Revenues from external customers Gains on sales of subsidiaries and insurance assets	\$ 313.4	\$ 32.7	\$-	\$- 385.8	\$ 346.1 385.8
Net realized losses on sales of investments and other assets	6 (11.8)	(.6)	_	4.0	(8.4)
Net investment income	57.5	8.3	-	20.1	85.9
Losses from unconsolidated affiliate(s)	-	-	(2.6)	-	(2.6)
Amortization of deferred credits and other benefits	19.3	-	1.4	20.7	41.4
Total revenues	378.4	40.4	(1.2)	430.6	848.2
Pretax earnings before interest expense	(19.1)	5.2	(1.2)	386.6	371.5
Interest expense	(8.2)		(=-=)	(7.9)	(16.1)
Income tax provision	16.8	(.6)	-	(58.7)	(42.5)
Net income (loss) from continuing operations	\$ (10.5)	\$ 4.6	\$ (1.2)	\$ 320.0	\$ 312.9
Year ended December 31, 1999					
Revenues from external customers	\$ 211.0	\$ 75.6	\$ -	\$.3	\$ 286.9
Gains on sales of subsidiaries and insurance assets	+	-	÷ -	88.1	88.1
Net realized gains on sales of investments and other assets	43.8	10.0	-	31.6	85.4
Net investment income	49.1	4.9	2.6	5.3	61.9
Earnings from unconsolidated affiliates	-	-	31.1	-	31.1
Amortization of deferred credits and other benefits	20.3	-	1.2	4.3	25.8
Total revenues	324.2	90.5	34.9(a)	129.6	579.2
Pretax earnings before interest expense	60.6	.2	34.9	80.5	176.2
Interest expense	(5.9)	(.4)	-	(8.4)	(14.7)
Income tax provision	(9.8)	-	(1.1)	(42.2)	(53.1)
Net income (loss) from continuing operations	\$ 44.9 \$	(.2)	\$ 33.8	\$ 29.9	\$ 108.4

Year ended December 31, 1998	 		 				
Revenues from external customers	\$ 85.4	\$ 170.0	\$ -	\$.1	\$ 255	5.5
Net realized gains on sales of investments and other assets	(.8)	5.4	-		66.4	71	1.0
Net investment income	18.1	8.2	3.8		6.7	36	6.8
Earnings from unconsolidated affiliates	-	-	24.3		-	24	4.3
Amortization of deferred credits and other benefits	2.7	-	-		-	2	2.7
Total revenues	105.4	183.6	28.1(a)	73.2	390	9.3
 Pretax earnings before interest expense	 8.1	7.7	 28.1		49.4	93	3.3
Interest expense	(1.4)	(1.1)	-		(11.2)	(13.	.7)
Income tax provision	(1.2)	(1.8)	(7.6)		(17.9)	(28.	.5)
Net income (loss) from continuing operations	\$ 5.5	\$ 4.8	\$ 20.5	•••••	20.3	\$ 51	1.1

 (a) Includes interest income on White Mountains' investment in MediaOne preferred stock (considered to be related to White Mountains' investment in FSA) of \$2.6 million and \$3.8 million for 1999 and 1998, respectively.

Ending assets	Р	roperty and Casualty	Investments in Unconsolidated	Holding	Net assets of Discontinued	
Millions	Reinsurance	Insurance	Affiliates	Company	Operations	Total
December 31, 2000	\$2,681.4	156.6	130.6	560.4	16.2	\$3,545.2
December 31, 1999	\$1,294.3	198.6	422.6	117.3	16.3	\$2,049.1

NOTE 13. INVESTMENTS IN UNCONSOLIDATED AFFILIATES

INVESTMENT IN FSA

On July 5, 2000 White Mountains concluded the sale of its indirect, wholly-owned subsidiary, White Mountains Holdings, Inc. (which controlled a substantial amount of its holdings of FSA) and all its other holdings of FSA common stock to Dexia for proceeds of \$620.4 million which represented a gain of \$391.2 million.

White Mountains owned 6,943,316 and 3,460,200 shares of FSA Common Stock at December 31, 1999 and 1998, respectively. This represented approximately 21.2% and 11.6%, respectively, of the total shares of FSA Common Stock outstanding at those times. At December 31, 1999 and 1998, White Mountains also owned FSA Preferred Stock which provided White Mountains the right to acquire 2,000,000 additional shares of FSA Common Stock for net consideration of \$59.3 million. At December 31, 1998, White Mountains also owned FSA Options which, in total, gave White Mountains the right to acquire 2,560,607 additional shares of FSA Common Stock for net aggregate consideration of \$115.7 million.

Prior to the Dexia Sale, White Mountains' accounted for its investment in FSA Common Stock using the equity method. White Mountains' accounted for its investment in FSA Preferred Stock and FSA Options under the provisions SFAS No. 115 whereby the investment was reported at fair value as of the balance sheet date, with related unrealized investment gains and losses, after tax, reported as a net amount in a separate component of shareholders' equity and reported on the income statement as a component of other comprehensive net income.

Millions		2000		1999		1998	
AMOUNTS RECORDED BY WHITE MOUNTAINS:							
Investment in FSA Common Stock	\$	-	\$	262.2	\$	119.7	
Investment in FSA Options and Preferred Stock		-		41.1		114.4	
Total investment in FSA	\$	-	\$	303.3	\$	234.1	
	===		=====	========	:====	======	
Equity in earnings (loss) from FSA Common Stock (a)	\$	(3.6)	\$	19.5	\$	13.8	
Dividends received from FSA Common Stock		1.4		2.1		1.5	
Equity in net unrealized investment gains (losses) from FSA's investment portfoli	o (t) -		(14.0)		3.1	
Unrealized investment gains (losses) on FSA Options and Preferred Stock (b)		-		(4.1)		26.6	
Write-down from fair value to equity value upon exercise of FSA Options, before t	ax	-		(45.8)		-	
	====			=======	====	======	

(a) Recorded net of related amortization of goodwill.

At December 31, 1999 and 1998, White Mountains' consolidated retained earnings included \$53.3 million and \$35.9 million, respectively, of accumulated undistributed earnings of FSA (net of related amortization of goodwill).

INVESTMENT IN MSA

At December 31, 2000, 1999 and 1998, White Mountains owned 222,093 shares of MSA Common Stock. This represented 50.0% of the total shares of MSA Common Stock outstanding at those times. White Mountains' investment in MSA is accounted for using the equity method. The following tables provides summary financial information for MSA as wells as amounts recorded by White Mountains relating to its investment in MSA:

Millions	2000	1999	1998
MSA BALANCE SHEET DATA: Total investments	\$ 449.0	\$ 466.3	\$ 465.9
Total assets	608.7	582.3	\$ 405.9 581.6
Unearned premium reserve	129.6	118.3	113.0
Loss and loss adjustment expense reserves	202.1	198.4	212.2
Shareholders' equity MSA INCOME STATEMENT DATA:	253.8	233.4	232.5
Net premiums written	\$ 265.4	\$ 242.7	\$ 258.5
Net premiums earned	254.1	237.4	226.3
Net investment income	23.5	24.9	22.1
Net income	3.8	25.8	13.4
Comprehensive net income	16.4	. 9	21.2
AMOUNTS RECORDED BY WHITE MOUNTAINS:			
Investment in MSA Common Stock (a)	\$ 130.6	\$ 119.3	\$ 120.2
Equity in earnings from MSA Common Stock (b) Equity in net unrealized investment gains (losses) from MSA's	1.0	11.6	4.9
investment portfolio (c)	6.2	(12.5)	4.1

(a) Includes related goodwill of \$3.7 million, \$2.6 million and \$4.0 million at December 31, 2000, 1999 and 1998, respectively.

(b) Recorded net of related amortization of goodwill.

(c) Recorded directly to shareholders' equity (after tax) as a component of other comprehensive net income.

At December 31, 2000 and 1999, White Mountains' consolidated retained earnings included \$26.7 million and \$25.7 million, respectively, of accumulated undistributed earnings of MSA (net of related amortization of goodwill).

⁽b) Recorded directly to shareholders' equity (after tax) as a component of other comprehensive net income.

INVESTMENT IN FOLKSAMERICA

In August 1998 White Mountains acquired all of the remaining outstanding shares of the common stock of Folksamerica which resulted in Folksamerica becoming a consolidated subsidiary of White Mountains at that time. Prior to August 1998, Folksamerica was an unconsolidated insurance affiliate of White Mountains whereby White Mountains' investment in Folksamerica consisted of investments in Folksamerica common stock and Folksamerica preferred stock and warrants. White Mountains investment in the common stock of Folksamerica was accounted for using the equity method and its investment in the preferred stock and warrants of Folksamerica was accounted for under the provisions of SFAS No. 115. For the year ended December 31, 1998, White Mountains recorded \$5.6 million of pretax earnings from its unconsolidated investment in Folksamerica.

NOTE 14. FAIR VALUE OF FINANCIAL INSTRUMENTS

White Mountains carries all its financial instruments on its balance sheet at fair value with the exception of long- term indebtedness. At December 31, 2000 and 1999, the market value of White Mountains' long-term indebtedness was \$97.2 million and \$206.7 million, respectively, which compared to a carrying value of \$96.0 million and \$202.8 million, respectively. The fair value of long-term indebtedness is estimated by discounting future cash flows using incremental borrowing rates for similar types of borrowing arrangements or quoted market prices. Considerable judgement is required to develop such estimates of fair value. Therefore, the estimates provided herein are not necessarily indicative of the amounts that could be realized in a current market exchange.

NOTE 15. RELATED PARTY TRANSACTIONS

White Mountains formerly owned two short-range aircraft jointly with Haverford Utah, LLC ("Haverford"). Messrs. Jack Byrne, Patrick Byrne and Kemp (each Directors of the Company) were principals of Haverford. Both aircraft were acquired from unaffiliated third parties during 1996. In exchange for Haverford's 20% ownership interest in the aircraft, Haverford contributed capital equal to 20% of the total initial cost of the aircraft and paid its pro rata share of all fixed costs plus the direct operating costs for its use pursuant to a Joint Ownership Agreement. The aircraft were sold to a third party during 2000 which resulted in Haverford receiving 20% of the cash proceeds.

During 1998 White Mountains sold its 25% joint ownership interest in a private jet operated by a third party to Haverford for cash proceeds of \$500,000. The purchase price received from Haverford represented a payment of \$437,500 for White Mountains' joint ownership interest (which resulted in White Mountains recognizing a pretax gain on sale of approximately \$75,000) and \$62,500 for reimbursement of prepaid aircraft expenses which were required to be paid to the operator prior to the sale to Haverford.

Mr. Clark, a Director of the Company, is Vice Chairman of Lehman. Lehman has, from time to time, provided various services to White Mountains including investment banking services, brokerage services, underwriting of debt and equity securities and financial consulting services. Lehman is the arranger, the administrative agent and a lender under the Credit Facility. See Note 17.

 ${\rm Mr.}$ George Gillespie, a Director of the Company, is a Partner in Cravath, which has been retained by White Mountains from time to time to perform legal services.

White Mountains owns a limited partnership investment interest which was managed by Mr. John Gillespie, a Director of the Company.

White Mountains owns a limited partnership interest and an investment portfolio which was managed by Mr. Zankel, a Director of the Company.

White Mountains believes that all the above transactions were on terms that were reasonable and competitive. Additional transactions of this nature may be expected to take place in the ordinary course of business in the future.

NOTE 16. COMMITMENTS AND CONTINGENCIES

Folksamerica leases its office space under noncancellable leases expiring at various dates through 2010. Rental expense for all of Folksamerica's locations was approximately \$2.1 million, \$1.6 million and \$1.6 million for the years ended December 31, 2000, 1999 and 1998, respectively. Folksamerica's future annual minimum rental payments required under noncancellable leases for office space are \$2.9 million for 2001, \$2.8 million for both 2002 and 2003, \$3.0 million for 2004, and \$16.0 million for years thereafter. White Mountains also has various other lease obligations which are immaterial in the aggregate.

Various claims have been made against White Mountains in the normal course of its business. Based on all information available at the date of this report, management believes that the outcome of such claims will not, in the aggregate, have a material effect on White Mountains' financial position or results of operations.

NOTE 17. SUBSEQUENT EVENTS

PENDING ACQUISITION OF CGU

In September 2000 White Mountains entered into a definitive agreement with CGNU to purchase its U.S. property and casualty insurance operations, CGU. The CGU purchase agreement, which was amended on October 15, 2000 and February 20, 2001, calls for a purchase price of \$2.17 billion, subject to certain adjustments, of which \$260.0 million will consist of a note payable (which must be paid in eighteen months in cash or Shares valued at \$245.00 per Share, at the Company's sole option). In addition, CGU will repay approximately \$.5 billion of debt outstanding to its parent at closing.

In connection with the CGU transaction, White Mountains has arranged for up to \$741.0 million of new equity commitments from a small group of outside investors and, on March 16, 2001, finalized its \$875.0 million Credit Agreement with Lehman. In connection with financing the transaction, White Mountains will contribute Folksamerica, PIC and MSA to CGU.

Completion of the CGU acquisition is subject to, among other matters, the receipt of regulatory approvals, the completion of financing and the satisfaction of other customary conditions.

CASH TENDER OFFER AND CONSENT SOLICITATION FOR OUTSTANDING MEDIUM-TERM NOTES

On March 15, 2001 White Mountains commenced the Note Offer providing for the repurchase of its \$96.3 million aggregate principal amount of outstanding medium-term notes. In conjunction with the Note Offer, noteholder consents are being solicited to effect an amendment to the indenture governing these notes, which will facilitate White Mountains' pending acquisition of CGU. The Note Offer is scheduled to expire at 5:00 p.m., New York City time on April 16, 2001, unless extended or terminated.

The total consideration to be paid for each validly tendered and consented note due in 2003 will be based upon a fixed spread of 95 basis points over the yield to maturity on the applicable reference U.S. Treasury Note, and includes a consent payment of \$30.00 per \$1,000 principal amount of such notes. The total consideration to be paid for each validly tendered and consented note due 2008 will be based upon a fixed spread of 125 basis points over the yield to maturity on the applicable reference U.S. Treasury Note, and includes a consent payment of \$30.00 per \$1,000 principal amount of such notes. The Note Offer pricing date is expected to be April 11, 2001. Accrued and unpaid interest up to, but not including, the payment date will be paid for notes validly tendered and accepted for purchase.

The Note Offer is conditioned upon, among other things, (i) there being validly tendered and not withdrawn a majority in aggregate principal amount of outstanding notes, (ii) the execution of the supplemental indenture governing the notes and (iii) the consummation of White Mountains' acquisition of CGU.

The financial information included in this report, including the audited consolidated financial statements, has been prepared by the management of White Mountains. The consolidated financial statements have been prepared in accordance with GAAP and, where necessary, include amounts based on informed estimates and judgments. In those instances where there is no single specified accounting principle or standard, management makes a choice from reasonable, accepted alternatives which are believed to be most appropriate under the circumstances. Financial information presented elsewhere in this report is consistent with that shown in the financial statements.

White Mountains maintains internal financial and accounting controls designed to provide reasonable and cost effective assurance that assets are safeguarded from loss or unauthorized use, that transactions are recorded in accordance with management's policies and that financial records are reliable for preparing financial statements. The internal controls structure is documented by written policies and procedures which are communicated to all appropriate personnel and is updated as necessary. White Mountains' business ethics policies require adherence to ethical standards in the conduct of its business. Compliance with these controls, policies and procedures is continuously maintained and monitored by management.

PricewaterhouseCoopers has audited the consolidated financial statements of White Mountains as of December 31, 2000 and 1999 and for each of the two years then ended and KPMG LLP has audited the consolidated financial statements of White Mountains for the year ended December 31, 1998. These firms have issued their unqualified reports thereon, which appear on pages F-31 and F-32.

In connection with their audits, the independent auditors provide an objective, independent review and evaluation of the structure of internal controls to the extent they consider necessary. Management reviews all recommendations of the independent auditors concerning the structure of internal controls and responds to such recommendations with corrective actions, as appropriate.

The Audit Committee of the Board, which is comprised solely of independent, qualified directors, has general responsibility for the oversight and surveillance of the accounting, reporting and financial control practices of White Mountains as well as establishing and maintaining the Company's Audit Committee Charter. The Audit Committee, which reports to the full Board, annually reviews the overall quality and effectiveness of the independent auditors and management with respect to the financial reporting process and the adequacy of internal controls. The independent auditors have free access to the Audit Committee, without members of management present, to discuss the results of their audits, the adequacy of internal controls and any other matter that they believe should be brought to the attention of the Audit Committee.

/s/ John J. Byrne Chairman and Chief Executive Officer /s/ Raymond Barrette President /s/ Michael S. Paquette Senior Vice President and Controller

To the Board of Directors and Shareholders of White Mountains Insurance Group, Ltd.:

In our opinion, the 2000 and 1999 consolidated financial statements listed in the index referenced under Item 14(a) on page 34 present fairly, in all material respects, the financial position of White Mountains Insurance Group, Ltd. and subsidiaries at December 31, 2000 and 1999, and the results of their operations and their cash flows for each of the two years in the period ended December 31, $2000\ in\ conformity\ with\ accounting\ principles\ generally\ accepted\ in\ the\ United$ States. In addition, in our opinion, the 2000 and 1999 financial statement schedules listed in the index referenced under Item 14(a) on page 34 present fairly, in all material respects, the information set forth therein when read in conjunction with the related 2000 and 1999 consolidated financial statements. These financial statements and financial statement schedules are the responsibility of the Company's management; our responsibility is to express an opinion on these financial statements and financial statement schedules based on our audits. We conducted our audits of these statements in accordance with auditing standards generally accepted in the United States, which require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for the opinion expressed above.

/s/ PricewaterhouseCoopers

Hamilton, Bermuda February 9, 2001, except for Note 17, which is as of March 16, 2001

Board of Directors and Shareholders White Mountains Insurance Group, Ltd.

We have audited the accompanying consolidated statements of income and comprehensive income, shareholders' equity, and cash flows of White Mountains Insurance Group, Ltd. and Subsidiaries (the "Company") for the year ended December 31, 1998 (collectively, the "consolidated financial statements"). In connection with our audit of the consolidated financial statements, we also have audited the 1998 financial information in Schedule II Condensed financial information of the registrant, Schedule III Supplementary insurance information, Schedule IV Reinsurance, Schedule V Valuation and qualifying accounts, and Schedule VI Supplemental information for property-casualty insurance underwriters (collectively, the "financial statement schedules"). These consolidated financial statements and financial statement schedules are the responsibility of the Company's management. Our responsibility is to express an opinion on these consolidated financial statements and financial statement schedules based on our audit. We did not audit the consolidated financial statements of Folksamerica Holding Company, Inc. ("Folksamerica"), a wholly-owned subsidiary and Financial Security Assurance Holdings, Ltd. ("FSA"), an 11.6 percent owned equity investee company. The financial statements of Folksamerica reflect total revenues of 27.0 percent in 1998 of the related consolidated totals. The Company's equity in earnings of FSA was \$13.8 million for the year ended December 31, 1998. The financial statements of Folksamerica and FSA were audited by other auditors, PricewaterhouseCoopers LLP, whose reports were furnished to us, and our opinion, insofar as it relates to the amounts included for Folksamerica and FSA, is based solely on the reports of other auditors.

We conducted our audit in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audit and the reports other auditors provide a reasonable basis for our opinion.

In our opinion, based on our audit and the reports of the other auditors, the consolidated financial statements referred to above present fairly, in all material respects, the results of operations and cash flows of White Mountains Insurance Group, Ltd. and Subsidiaries for the year ended December 31, 1998 in conformity with accounting principles generally accepted in the United States of America. Also in our opinion, based on our audit and the reports of other auditors, the 1998 information in the related financial statement schedules, when considered in relation to the basic financial statements taken as a whole, presents fairly, in all material respects, the information set forth therein.

/s/ KPMG LLP

February 12, 1999 Providence, Rhode Island

Selected quarterly financial data for 2000 and 1999 is shown in the following table. The quarterly financial data includes, in the opinion of management, all recurring adjustments necessary for a fair presentation of the results of operations for the interim periods.

	2000	Three Mo	nths Ende	d (a)	1999	Three Mon	ths Ended	(b)
Millions, except per share amounts	Dec. 31 S	Sept. 30	June 30	Mar. 31	Dec. 31	Sept. 30	June 30	Mar. 31
Revenues Expenses	\$ 55.1 79.3	\$546.1 161.9	\$ 163.9 161.0	\$ 83.1 90.6	\$146.3 129.3	\$106.5 83.4	\$199.5 94.7	\$ 126.9 110.3
Pretax earnings (loss) Income tax benefit (provision)	(24.2) 8.4	384.2 (49.7)	2.9 (3.1)	(7.5) 1.9	17.0 2.7	23.1 (9.0)	104.8 (41.0)	16.6 (5.8)
Net income (loss) from continuing operations Net income (loss) from discontinued operations	(15.8)	334.5 95.0	(.2)	(5.6)	19.7 (3.3)	14.1	63.8 12.9	10.8 3.0
Net income (loss)	\$(15.8) ======	\$429.5	\$ (.2)	\$ (5.6)	\$ 16.4	\$ 14.1	\$ 76.7	\$ 13.8
Net income (loss) from continuing operations po Basic Diluted	er Share: \$(2.68) (2.68)	\$56.86 56.59	\$ (.03) (.03)	\$ (.94) (.94)	\$ 2.73 2.69	\$ 2.72 2.43	\$11.58 11.56	\$ 1.84 1.64
Net income (loss) per Share: Basic Diluted	\$(2.68) (2.68)	\$73.02 72.67	\$ (.03) (.03)	\$ (.94) (.94)	\$ 2.73 2.69	\$ 2.72 2.43	\$13.94 12.41	\$ 2.36 2.10

(a) The quarterly amounts subsequent to March 31, 2000 reflect the acquisitions of PCA and Risk Capital. The quarterly amounts for the three month period ended September 30, 2000 reflect the Dexia Sale and a \$95.0 million reserve release relating to Fireman's Fund. The quarterly amounts for the three month period ended December 31, 2000 have been impacted by a significant reinsurance contract with a third party.

(b) The quarterly amounts for the three month period ended June 30, 1999 reflect the VGI Sale and the Mortgage Banking Sale. The Mortgage Banking Sale represented a decision by the Company to exit the mortgage banking business. As a result, all mortgage banking activities are presented herein as discontinued operations.

SUMMARY OF INVESTMENTS -- OTHER THAN INVESTMENTS IN RELATED PARTIES AT DECEMBER 31, 2000

lillions	Cost	FAIR VALUE
ixed maturities:		
Bonds :		
United States Government and government agencies and authorities	\$ 411.8	\$ 421.5
Corporate bonds	310.6	312.9
States, municipalities and political subdivisions	265.7	268.5
Foreign governments	52.2	52.4
Redeemable preferred stocks	22.7	23.3
Total fixed maturities	1,063.0	1,078.6
Common equity securities:		
Banks, trust and insurance companies	47.1	63.0
Industrial, miscellaneous and other	80.4	81.8
Total common equity securities	127.5	
other investments	117.3	142.9
Short-term investments	735.9	735.9
Total investments	\$ 2,043.7	\$ 2,102.2

NOTE - fair value was equal to carrying value at December 31, 2000.

FS-1

WHITE MOUNTAINS INSURANCE GROUP, LTD. (REGISTRANT ONLY)

CONDENSED BALANCE SHEETS

	Dec	ember 3	ber 31,	
Millions	2000		1999	
Assets:				
Fixed maturities	\$ -	\$	39.6	
Common equity securities and other investments	30.5		26.3	
Short-term investments, at amortized cost	96.4		5.7	
Other assets	56.8		1.2	
Investments in consolidated affiliates	1,040.3		770.9	
Total assets	\$ 1,224.0	\$	843.7	
iabilities:				
Short-term debt	\$ -	\$	4.0	
Long-term debt	96.0		96.0	
Deferred credits	37.0		57.8	
Accounts payable and other liabilities	44.5		71.6	
Total liabilities	177.5		229.4	
Shareholders' equity	1,046.5		614.3	
Total liabilities and shareholders' equity	\$ 1,224.0	\$	843.7	

CONDENSED STATEMENTS OF INCOME AND COMPREHENSIVE INCOME

	Ye	ar Ended December	31,	
Millions	2000	1999	1998	
Revenues	\$ 26.8	\$ 29.1	\$28.9	
Expenses	43.5	46.6	23.2	
Pretax earnings (loss)	(16.7)	(17.5)	5.7	
Income tax provision	(26.6)	(6.0)	(8.3)	
Net loss	(43.3)	(23.5)	(2.6)	
Earnings from consolidated affiliates	356.2	131.9	53.7	
Earnings from discontinued operations, after tax	95.0	12.6	27.4	
Consolidated net income	407.9	121.0	78.5	
Other comprehensive net income (loss) items, after tax	39.7	(118.0)	(8.9)	
Consolidated comprehensive net income	\$ 447.6	\$ 3.0	\$69.6	

WHITE MOUNTAINS INSURANCE GROUP, LTD. (REGISTRANT ONLY)

CONDENSED STATEMENTS OF CASH FLOWS

	Ye	ear Ended Decemb	er 31,
Millions	2000	1999	1998
Net income	\$ 407.9	\$ 121.0	\$ 78.5
Reconciliation of net income to net cash from operating activities: Net realized gains on sales of investments and insurance assets Distributions from consolidated subsidiaries in excess of current earnings	(2.7)	(21.7) 213.0	(26.7)
Undistributed current earnings from consolidated subsidiaries Amortization of deferred credits	(250.1) (20.8)	-	(78.0)
Deferred income tax provision Release of tax reserve on sale of former insurance subsidiary	(95.0)	39.7	.1
(Increase) decrease in other assets Increase (decrease) in accounts payable and other liabilities	(24.3) 37.3	2.5 (112.5)	(4.2) 18.1
Net cash provided from (used for) operating activities	52.3	242.0	(12.2)
Cash flows from investing activities: Net (increase) decrease in short-term investments, net of balances acquired Sales of investment securities Purchases of investment securities	(84.8) 53.8 (1.4)	18.3 5.4	(8.5)
Investments in consolidated affiliates, net of balances acquired Investments in unconsolidated affiliates Sale of securities carried in other assets	-	(73.5) (50.0) -	- - 26.8
Net cash (used for) provided from investing activities	(32.4)	(99.8)	18.3
Cash flows from financing activities: Purchases of common shares retired Proceeds from exercises of warrants and options to acquire Shares	(8.8)	(139.4) 21.7	(19.5)
Repayment of long-term debt Intercompany borrowings from subsidiaries Cash dividends paid to common shareholders	(4.0) - (7.1)	(15.9) - (8.8)	- 23.0 (9.4)
Net cash used for financing activities	(19.9)	(142.4)	(5.9)
Net (decrease) increase in cash during year Cash balance at beginning of year		(.2)	.2
Cash balance at end of year	\$ -	\$ -	\$.2

WHITE MOUNTAINS INSURANCE GROUP, LTD.

SUPPLEMENTARY INSURANCE INFORMATION (MILLIONS)

Column A	Column B	Column C	Column D	Column E	Column F	Column G	Column H
Segment	Deferred acquisition costs	Future policy benefits, losses, claims and loss expenses	Unearned premiums	Other policy claims and benefits payable	Premiums earned	Net investment income (b)	Benefits, claims, losses, and settlement expenses
Years ended:							
December 31, 2000:							
Reinsurance	\$ 26.2	\$1,500.7	\$171.9	\$ -	\$ 312.5	\$ 57.4	\$ 270.8
Property casualty insurance	1.0	55.6	10.1	-	21.9	8.3	16.9
December 31, 1999:							
Reinsurance	\$ 21.3	\$ 782.1	\$ 79.0	\$ -	\$ 211.0	\$ 49.1	\$ 182.2
Property casualty insurance	.9	68.9	13.1	-	72.2	4.9	60.1
December 31, 1998:							
Reinsurance (a)	\$ 20.7	\$ 723.2	\$ 71.2	\$ -	\$ 85.4	\$ 18.1	\$ 59.7
Property casualty insurance	14.7	88.5	81.9	-	160.6	8.3	115.1

Column A	Column I	Column J	Column K
Segment	costs	operating expenses	written
Years ended:			
December 31, 2000: Reinsurance Property casualty insurance	\$ 98.0 3.1	\$ - -	\$ 332.6 22.6
December 31, 1999: Reinsurance Property casualty insurance	\$ 64.3 9.1	\$ - -	\$ 201.7 73.0
December 31, 1998: Reinsurance (a) Property casualty insurance	\$ 29.1 25.7	\$ - -	\$ 73.7 164.9

(a) The amounts shown for Reinsurance in columns F through K represent activity for Folksamerica from August 18, 1998 through December 31, 1998.

(b) The amounts shown exclude net investment income relating to non-insurance operations of \$20.2 million, \$7.9 million and \$10.4 million for the twelve months ended December 31, 2000, 1999 and 1998, respectively.

WHITE MOUNTAINS INSURANCE GROUP, LTD.

REINSURANCE

Column A	Column B	Column C	Column D	Column E	Column F
Premiums earned (Dollars in millions)	Gross amount	Ceded to other companies	Assumed from other companies	Net amount	Percentage of amount assumed to net
Years ended:					
December 31, 2000: Property and casualty insurance Reinsurance	\$ 28.4 4.1	\$ (6.9) (167.7)	\$.4 476.1	\$ 21.9 312.5	1.8% 152.4%
December 31, 1999: Property and casualty insurance Reinsurance	\$ 61.4 2.4	\$ (28.8) (32.4)	\$ 39.6 241.0	\$ 72.2 211.0	54.8% 114.2%
December 31, 1998: Property and casualty insurance Reinsurance (a)	\$104.4 2.0	\$ (13.1) (8.8)	\$ 69.3 92.2	\$ 160.6 85.4	43.2% 108.0%

(a) Amounts shown in columns B through F represent activity for Folksamerica from August 18, 1998 through December 31, 1998.

WHITE MOUNTAINS INSURANCE GROUP, LTD.

VALUATION AND QUALIFYING ACCOUNTS

Column A	Column B	Colum	n C	Column D	Column E	
	Additions		ons			
Millions	Balance at beginning of period	Charged to costs and expenses	to other	Deductions described	Balance at end of period	
Years ended:						
December 31, 2000: Reinsurance recoverable: Allowance for reinsurance balances Property and casualty insurance: Allowance for uncollectible accounts	\$ 1.2 1.1	\$-	\$-	\$-	\$ 1.2 1.1	
December 31, 1999: Reinsurance recoverable: Allowance for reinsurance balances Property and casualty insurance: Allowance for uncollectible accounts	\$ 1.2 .4	\$-	\$-	\$- .7(a)	\$ 1.2 1.1	
December 31, 1998: Reinsurance recoverable: Allowance for reinsurance balances (b) Property and casualty insurance: Allowance for uncollectible accounts	\$ - .3	\$ - 1.9	\$-	\$ 1.2(a) (1.8)(b)	\$ 1.2 .4	

(a) Represents allowances acquired from insurance and reinsurance operations

(b) Represents charge-offs of balances receivable

WHITE MOUNTAINS INSURANCE GROUP, LTD. SUPPLEMENTAL INFORMATION FOR PROPERTY AND CASUALTY INSURANCE UNDERWRITERS (MILLIONS)

Column A	Column B	Column C	Column D	Column E	Column F	Column G
Affiliation with registrant		Adjustment	any, deducted in	Unearned Premiums	Earned Premiums	Net investment income
Consolidated reinsurance						
operations:						
2000	\$ 26.2	\$ 1,500.7	\$ -	\$ 171.9	\$ 312.5	\$ 57.4
1999	21.3	782.1	÷ -	79.0	211.0	49.1
1998(a		723.2	-	71.2	85.4	18.1
Consolidated property and						
casualty insurance operations:						
2000	\$ 1.0	\$ 55.6	\$ -	\$ 10.1	\$ 21.9	\$ 8.3
1999	.9	68.9	-	13.1	72.2	4.9
1998	14.7	88.5	-	81.9	160.6	8.3
50%-or-less owned property						
and casualty investees (b):						
2000	\$ 18.8	\$ 101.1	\$ -	\$ 64.8	\$ 127.1	\$ 11.8
1999	17.5	99.2	-	59.2	118.7	12.5
1998	16.0	106.1	-	56.5	103.6	10.1

Column A	Colu	ımn H	Column I	Column J	Column K
Affiliation with registrant	Adjustmer Incurred (1) Current	Prior	Amortization of deferred policy acquisition costs	and Claims Adjustment	Premiums written
Consolidated reinsurance operations: 2000 1999 1998(a)	\$ 247.9 152.9 58.6	\$ 22.9 29.3 1.1	\$ 98.0 64.3 29.1	\$ 415.0 237.0 67.5	\$ 332.6 201.7 73.7
Consolidated property and casualty insurance operations: 2000 1999 1998	\$ 16.2 57.5 108.4	\$.7 2.6 6.7	\$ 3.1 9.1 25.7	\$ 21.1 38.3 98.7	\$ 22.6 73.0 164.9
50%-or-less owned property and casualty investees (b): 2000 1999 1998	\$ 91.9 89.3 82.5	\$(3.3) (7.0) (5.1)	\$ 37.2 34.4 24.2	\$ 89.0 84.3 48.1	\$ 132.7 121.3 118.3

(a) The amounts shown for Reinsurance in columns F through K represent activity for Folksamerica from August 18, 1998 through December 31, 1998.

⁽b) The amounts shown represent White Mountains' share of its property and casualty affiliate, MSA, which was 50.0% owned during the three year period ended December 31, 2000.

EXHIBIT 10(f)

EXECUTION COPY

\$875,000,000

CREDIT AGREEMENT

among

TACK HOLDING CORP.,

TACK ACQUISITION CORP.,

as Primary Borrower,

WHITE MOUNTAINS INSURANCE GROUP, LTD., as Additional Borrower,

The Several Lenders from Time to Time Parties Hereto,

> LEHMAN BROTHERS INC., as Arranger,

> FLEET NATIONAL BANK, as Syndication Agent,

BANK OF AMERICA, N.A. as Documentation Agent

and

LEHMAN COMMERCIAL PAPER INC., as Administrative Agent

Dated as of March 16, 2001

PAGE

Section 1 DE	FINITIONS
1.1	Defined Terms2
1.2	Other Definitional Provisions
Section 2 AM	10UNT AND TERMS OF COMMITMENTS
2.1	Term Loan Commitments
2.2	Procedure for Term Loan Borrowing
2.3	Repayment of Term Loans
2.4	Revolving Credit Commitments
2.5	Procedure for Revolving Credit Borrowing
2.6	Swing Line Commitment
2.7	Procedure for Swing Line Borrowing; Refunding of Swing Line Loans
2.8	Repayment of Loans; Evidence of Debt
2.9	Commitment Fees, etc
2.10	Termination or Reduction of Revolving Credit Commitments
2.10	Optional Prepayments
2.12	Mandatory Prepayments and Commitment Reductions
2.12	Conversion and Continuation Options
2.13	Minimum Amounts and Maximum Number of Eurodollar Tranches
2.14	Interest Rates and Payment Dates
2.15	Computation of Interest and Fees
2.10	Inability to Determine Interest Rate
2.17	Pro Rata Treatment and Payments
2.18	Requirements of Law
	requirements of Law
2.20	
2.21	Indemnity
2.22	Illegality
2.23	Change of Lending Office
2.24	Replacement of Lenders under Certain Circumstances
2.25	Optional Commitment Increase
	TTERS OF CREDIT
3.1	L/C Commitment
3.2	Procedure for Issuance of Letter of Credit48
3.3	Fees and Other Charges
3.4	L/C Participations
3.5	Reimbursement Obligation of the Borrowers
3.6	Obligations Absolute
3.7	Letter of Credit Payments
3.8	Applications
Section 4 RE	PRESENTATIONS AND WARRANTIES
4.1	Financial Condition

i

4.2	No Change
4.3	Corporate Existence; Compliance with Law
4.4	Corporate Power; Authorization; Enforceable Obligations
4.5	No Legal Bar
4.6	No Material Litigation
4.7	No Default
4.8	Ownership of Property; Liens
4.9	Intellectual Property
4.10	Taxes
4.11	Federal Regulations
4.12	ERISA
4.13	Investment Company Act; Other Regulations
4.14	Subsidiaries
4.15	Use of Proceeds
4.16	Environmental Matters
4.17	Accuracy of Information, etc
4.18	Security Documents
4.19	Solvencý
4.20	Insurance Regulatory Matters
Section 5 CO	NDITIONS PRECEDENT
5.1	Conditions to Initial Extension of Credit
5.2	Conditions to Each Extension of Credit
Section 6 AF	FIRMATIVE COVENANTS
6.1	Financial Statements
6.2	Certificates; Other Information
6.3	Payment of Obligations
6.4	Conduct of Business and Maintenance of Existence, etc
6.5	Maintenance of Property; Insurance
6.6	Inspection of Property; Books and Records; Discussions
6.7	Notices
6.8	Environmental Laws
6.9	Interest Rate Protection
6.10	Additional Collateral, etc
6.11	Subordinated Seller Note
6.12	Tax Sharing Agreement
Section 7 NE	GATIVE COVENANTS
7.1	Financial Condition Covenants
7.2	Limitation on Indebtedness and Issuance of Preferred Stock
7.3	Limitation on Liens
7.4	Limitation on Fundamental Changes
7.5	Limitation on Disposition of Property
7.6	Limitation on Restricted Payments
7.7	Limitation on Investments.

ii

	7.8	Limitation on Optional Payments, Redemptions and Modifications of Debt Instruments
		and Preferred Stock, etc
	7.9	Limitation on Transactions with Affiliates
	7.10	Limitation on Sales and Leasebacks
	7.11	Limitation on Changes in Fiscal Periods
	7.12	Limitation on Negative Pledge Clauses
	7.13	Limitation on Lines of Business
Section	8 EVE	ENTS OF DEFAULT
Section	9 THE	E ADMINISTRATIVE AGENT
	9.1	Appointment
	9.2	Delegation of Duties
	9.3	Exculpatory Provisions
	9.4	Reliance by Administrative Agent
	9.5	Notice of Default
	9.6	Non-Reliance on Administrative Agent and Other Lenders
	9.7	Indemnification
	9.8	Administrative Agent in Its Individual Capacity
	9.9	Successor Administrative Agent
	9.10	Authorization to Release Liens and Guarantees
	9.11	The Arranger, the Syndication Agent and the Documentation Agents
Section	10 MI	ISCELLANEOUS
	10.1	Amendments and Waivers
	10.2	Notices
	10.3	No Waiver; Cumulative Remedies
	10.4	Survival of Representations and Warranties91
	10.5	Payment of Expenses
	10.6	Successors and Assigns; Participations and Assignments92
	10.7	Adjustments; Set-off95
	10.8	Counterparts
		Severability
	10.10	Integration
		GOVERNING LAW
	10.12	Submission To Jurisdiction; Waivers
		Acknowledgments
	10.14	Confidentiality
	10.15	Release of Collateral and Guarantee Obligations
	10.16	Accounting Changes
		Delivery of Lender Addenda
	10.18	WAIVERS OF JURY TRIAL

iii

4.4	Consents, Authorizations, Filings and Notices
4.14	Subsidiaries
4.18-1	UCC Filing Jurisdictions
4.18-2	UCC Financing Statements to Remain on File
4.18-3	UCC Financing Statements to be Terminated
7.2(e)	Existing Indebtedness
7.3(f)	Existing Liens

EXHIBITS:

A B C D F-1 F-2 G-1 G-2 G-3 H I J K L

Form of Guarantee and Collateral Agreement
Form of Compliance Certificate
Form of Closing Certificate
[Reserved]
Form of Assignment and Acceptance
Form of Legal Opinion of Cravath Swaine & Moore
Form of Legal Opinion of Roger Singer
Form of Term Note
Form of Revolving Credit Note
Form of Swing Line Note
Form of Prepayment Option Notice
Form of Exemption Certificate
Form of Lender Addendum
Form of Revolving Commitment Increase Supplement
Form of New Lender Supplement

iv

CREDIT AGREEMENT, dated as of March 16, 2001, among TACK HOLDING CORP., a Delaware corporation (the "PARENT"), TACK ACQUISITION CORP., a Delaware corporation (the "PRIMARY BORROWER"), WHITE MOUNTAINS INSURANCE GROUP, LTD., a company existing under the laws of Bermuda ("WHITE MOUNTAINS" or the "ADDITIONAL BORROWER"), the several banks and other financial institutions or entities from time to time parties to this Agreement (the "LENDERS"), LEHMAN BROTHERS INC., as sole advisor, sole lead arranger and sole book manager (in such capacity, the "ARRANGER"), FLEET NATIONAL BANK, as syndication agent (in such capacity, the "SYNDICATION AGENT"), BANK OF AMERICA, N.A.. as documentation agent (in such capacity, the "DOCUMENTATION AGENT") and LEHMAN COMMERCIAL PAPER INC., as administrative agent (in such capacity, the "ADMINISTRATIVE Agent").

WITNESSETH:

WHEREAS White Mountains, the Parent, the Primary Borrower, CGU International Holdings Luxembourg S.A., a Luxembourg corporation ("CGU LUXEMBOURG"), CGU Holdings LLC, a Delaware limited liability company ("CGU DELAWARE") and CGNU plc, a company incorporated under the laws of England and wales ("CGNU plc") have entered into the Stock Purchase Agreement, dated as of September 24, 2000, as amended by Amendment No.1 to the Stock Purchase Agreement, dated as of October 15, 2000, and Amendment No.2 to the Stock Purchase Agreement, dated as of February 20, 2001 (the "STOCK PURCHASE AGREEMENT"; together with all schedules, exhibits, annexes and amendments thereto, all other agreements referred to therein and all side letters and agreements affecting the terms thereof or entered into in connection therewith, in each case, as amended, supplemented or otherwise modified from time to time, the "ACQUISITION DOCUMENTATION");

WHEREAS, pursuant to the Stock Purchase Agreement, the Primary Borrower will acquire (the "ACQUISITION") all of the outstanding capital stock of CGU Corporation, a Delaware corporation ("CGU");

WHEREAS, CGU Luxembourg and CGU Delaware are both indirect wholly owned subsidiaries of CGNU plc, and CGU Luxembourg owns approximately 54.1% of the outstanding capital stock of CGU, and CGU Delaware owns approximately 45.9% of such capital stock;

WHEREAS, the Parent and the Primary Borrower were both formed by White Mountains on September 24, 2000 for the purpose of effecting the Acquisition;

WHEREAS, to provide a portion of the financing for the purchase price to be paid by the Primary Borrower in connection with the Acquisition, for the payment of fees and expenses relating to the Acquisition, for the repayment of certain existing indebtedness of CGU and its subsidiaries and for working capital needs and general corporate purposes of the Primary Borrower and its Subsidiaries, the Primary Borrower has requested the Lenders to provide credit facilities in an aggregate amount of up to \$875,000,000; and

WHEREAS, the Lenders are willing to make such credit facilities available upon and subject to the terms and conditions hereinafter set forth;

NOW, THEREFORE, in consideration of the premises and the agreements hereinafter set forth, the parties hereto hereby agree as follows:

SECTION 1 DEFINITIONS

1.1 DEFINED TERMS. As used in this Agreement, the terms listed in this Section 1.1 shall have the respective meanings set forth in this Section 1.1.

"ACQUIRED ASSETS": all of the Capital Stock of CGU and all of its property and casualty insurance company subsidiaries (other than property and subsidiaries transferred to CGNU plc and its affiliates in the Preclosing Transactions).

"ACQUIRED ASSETS FINANCIAL STATEMENTS": as defined in Section 4.1(b).

"ACQUISITION": as defined in the recitals hereto.

"ACQUISITION DOCUMENTATION": as defined in the recitals hereto.

"ADDITIONAL BORROWER": as defined in the preamble hereto.

"ADMINISTRATIVE AGENT": as defined in the preamble hereto.

"AFFILIATE": as to any Person, any other Person that, directly or indirectly, is in control of, is controlled by, or is under common control with, such Person. For purposes of this definition, "control" of a Person means the power, directly or indirectly, either to (a) vote 10% or more of the securities having ordinary voting power for the election of directors (or persons performing similar functions) of such Person or (b) direct or cause the direction of the management and policies of such Person, whether by contract or otherwise.

"AGGREGATE EXPOSURE": with respect to any Lender at any time, an amount equal to (a) until the Closing Date, the aggregate amount of such Lender's Commitments at such time and (b) thereafter, the sum of (i) the aggregate then unpaid principal amount of such Lender's Term Loans and (ii) the amount of such Lender's Revolving Credit Commitment then in effect or, if the Revolving Credit Commitments have been terminated, the amount of such Lender's Revolving Extensions of Credit then outstanding.

"AGGREGATE EXPOSURE PERCENTAGE": with respect to any Lender at any time, the ratio (expressed as a percentage) of such Lender's Aggregate Exposure at such time to the sum of the Aggregate Exposures of all Lenders at such time.

"AGREEMENT": this Credit Agreement, as amended, supplemented or otherwise modified from time to time.

"A.M. BEST": A.M. Best Company, together with any Person succeeding thereto by merger, consolidation or acquisition of all or substantially all of its assets, including substantially all of its business of rating insurance companies. "ANNUAL STATEMENT": the annual statutory financial statement of any Insurance Subsidiary required to be filed with the Department of its jurisdiction of incorporation or organization, which statement shall be in the form required by such Insurance Subsidiary's jurisdiction of incorporation or organization or, if no specific form is so required, in the form of financial statements permitted by such Department to be used for filing annual statutory financial statements and shall contain the type of information permitted or required by such Department to be disclosed therein, together with all exhibits or schedules filed therewith.

"APPLICABLE MARGIN": for each Type of Loan, the rate per annum set forth under the relevant column heading below which corresponds with the most current rating of the Facilities issued by S&P and Moody's respectively:

Level		Loans and Tranche A	for Tranche B Term Loans which are	Applicable Margin for Revolving Credit Loans and Tranche A Term Loans which are Base Rate Loans	for Tranche B Term Loans which are Base
1	(greater than or equal to) BBB+ and Baa1		2.50%	0.50%	1.50%
2	BBB and Baa2	1.75%	2.75%	0.75%	1.75%
3	BBB- and Baa3	2.125%	2.875%	1.125%	1.875%
4	BBB- or Baa3 and BB+ or Ba1	2.25%	3.00%	1.25%	2.00%
5	BB+ and Ba1	2.50%	3.25%	1.50%	2.25%
6	(less than or equal to) BB and Ba2	2.75%	3.50%	1.75%	2.50%

Changes in the Applicable Margin shall become effective on the date on which S&P and/or Moody's changes the rating it has issued for the Facilities. In the event of split ratings, the lower rating shall apply, PROVIDED that if the ratings are split by two or more levels, the rating level next higher than the lower of the two ratings will apply. Level 4 applies when there is either a BBB- or a Baa3 rating and a BB+ and/or a Ba1 rating. In the event that, at any time, the Facilities are not rated by one of such rating agencies, the rating of such rating agency shall be deemed to be the rating level next lower than the rating issued by the other of such rating agencies. In the event that, at any time, the Facilities are rated by neither such rating agency, the Applicable Margin shall be as for Level 6. Notwithstanding the foregoing, if the Facilities receive a rating of BBB or higher from S&P and a rating of Baa2 or higher from Moody's, the Applicable Margins shall be as for Level 3 until the first anniversary of the Closing Date.

"APPLICATION": an application, in such form as the Issuing Lender may specify from time to time, requesting the Issuing Lender to issue a Letter of Credit.

"ARRANGER": as defined in the preamble hereto.

"ASSET SALE": any Disposition of Property or series of related Dispositions of Property (excluding any such Disposition permitted by clause (a), (b), (c), (d), (e) or (f) of Section 7.5) which yields gross proceeds to the Parent, the Primary Borrower or any of its Subsidiaries (valued at the initial principal amount thereof in the case of non-cash proceeds consisting of notes or other debt securities and valued at fair market value in the case of other non-cash proceeds) in excess of \$5,000,000. Any Disposition of Property by an Insurance Subsidiary shall not constitute an Asset Sale.

"ASSIGNEE": as defined in Section 10.6(c).

"ASSIGNOR": as defined in Section 10.6(c).

"AVAILABLE REVOLVING CREDIT COMMITMENT": with respect to any Revolving Credit Lender at any time, an amount equal to the excess, if any, of (a) such Lender's Revolving Credit Commitment then in effect OVER (b) such Lender's Revolving Extensions of Credit then outstanding; PROVIDED, that in calculating any Lender's Revolving Extensions of Credit for the purpose of determining such Lender's Available Revolving Credit Commitment pursuant to Section 2.9(a), the aggregate principal amount of Swing Line Loans then outstanding shall be deemed to be zero.

"BASE RATE": for any day, a rate of interest per annum (rounded upwards, if necessary, to the next 1/16 of 1%) equal to the greater of (a) the Prime Rate in effect on such day and (b) the Federal Funds Effective Rate in effect on such day plus 1/2 of 1%. For purposes hereof: "PRIME RATE" shall mean the prime lending rate as set forth on the British Banking Association Telerate page 5 (or such other comparable page as may, in the opinion of the Administrative Agent, replace such page for the purpose of displaying such rate), as in effect from time to time. The Prime Rate is a reference rate and does not necessarily represent the lowest or best rate actually available. The Administrative Agent or any Lender may make commercial loans or other loans at rates of interest at, above or below the Prime Rate. Any change in the Base Rate due to a change in the Prime Rate or the Federal Funds Effective Rate shall be effective as of the opening of business on the effective Rate, respectively.

"BASE RATE LOANS": Loans for which the applicable rate of interest is based upon the Base Rate.

"BENEFITTED LENDER": as defined in Section 10.7.

"BERKSHIRE HATHAWAY": Berkshire Hathaway Inc., or an affiliate.

"BERKSHIRE PREFERRED STOCK": the \$300,000,000 aggregate liquidation preference amount of non-voting preferred stock to be issued by the Primary Borrower to Berkshire Hathaway on the Closing Date, for a purchase price of \$225,000,000, having terms and conditions set forth in the term sheet attached to the letter, dated October 13, 2000, from Warren E. Buffett to John J. Byrne.

"BOARD": the Board of Governors of the Federal Reserve System of the United States (or any successor).

"BORROWER": either the Primary Borrower or the Additional Borrower, as the context requires (the Primary Borrower and the Additional Borrower, collectively, the "BORROWERS").

"BORROWING DATE": any Business Day specified by the relevant Borrower as a date on which such Borrower requests the relevant Lenders to make Loans hereunder.

"BUSINESS DAY": means (i) with respect to any borrowing, payment or rate selection of Eurodollar Loans, a day (other than a Saturday or Sunday) on which banks generally are open in New York City for the conduct of substantially all of their commercial lending activities, interbank wire transfers can be made on the Fedwire system and dealings in Dollars are carried on in the London interbank market and (ii) for all other purposes, a day (other than a Saturday or Sunday) on which banks generally are open in New York for the conduct of substantially all of the commercial lending activities, and interbank wire transfers can be made on the Fedwire system.

"CAPITAL EXPENDITURES": for any period, with respect to any Person, the aggregate of all expenditures by such Person for the acquisition or leasing (pursuant to a capital lease) of fixed or capital assets or additions to equipment (including replacements, capitalized repairs and improvements during such period) which are required to be capitalized under GAAP on a balance sheet of such Person.

"CAPITAL LEASE OBLIGATIONS": with respect to any Person, the obligations of such Person to pay rent or other amounts under any lease of (or other arrangement conveying the right to use) real or personal property, or a combination thereof, which obligations are required to be classified and accounted for as capital leases on a balance sheet of such Person under GAAP; and, for the purposes of this Agreement, the amount of such obligations at any time shall be the capitalized amount thereof at such time determined in accordance with GAAP.

"CAPITAL AND SURPLUS": as to any Insurance Subsidiary, as of any date, the total amount shown on line 27, page 3, column 1 of the Annual Statement of such Insurance Subsidiary, or an amount determined in a consistent manner for any date other than one as of which an Annual Statement is prepared (or any successor line, page or column that contains the same information).

"CAPITAL STOCK": any and all shares, interests, participations or other equivalents (however designated) of capital stock of a corporation, any and all equivalent ownership interests in a Person (other than a corporation) and any and all warrants, rights or options to purchase any of the foregoing.

"CASH EQUIVALENTS": (a) marketable direct obligations issued by, or unconditionally guaranteed by, the United States Government or issued by any agency thereof and backed by the full faith and credit of the United States, in each case maturing within one year from the date of acquisition; (b) certificates of deposit, time deposits, eurodollar time deposits or overnight bank deposits having maturities of six months or less from the date of acquisition issued by any Lender or by any commercial bank organized under the laws of the United States of America or any state thereof having combined capital and surplus of not less than \$500,000,000; (c) commercial paper of an issuer rated at least A-2 by Standard & Poor's Ratings Services ("S&P") or P-2 by Moody's Investors Service, Inc. ("MOODY'S"), or carrying an equivalent rating by a nationally recognized rating agency, if both of the two named rating agencies cease publishing ratings of commercial paper issuers generally, and maturing within six months from the date of acquisition; (d) repurchase obligations of any Lender or of any commercial bank satisfying the requirements of clause (b) of this definition, having a term of not more than 30 days with respect to securities issued or fully guaranteed or insured by the United States government; (e) securities with maturities of one year or less from the date of acquisition issued or fully guaranteed by any state, commonwealth or territory of the United States, by any political subdivision or taxing authority of any such state, commonwealth or territory or by any foreign government, the securities of which state, commonwealth, territory, political subdivision, taxing authority or foreign government (as the case may be) are rated at least A by S&P or A by Moody's; (f) securities with maturities of six months or less from the date of acquisition backed by standby letters of credit issued by any Lender or any commercial bank satisfying the requirements of clause (b) of this definition; and (g) shares of money market mutual or similar funds which invest primarily in assets satisfying the requirements of clauses (a) through (f) of this definition.

"CASH FLOW": of the Primary Borrower for any period, the sum, without duplication, of (a) dividends actually paid to the Primary Borrower or any Subsidiary Guarantor by any Subsidiary in cash or Cash Equivalents during such period plus (b) additional amounts which would be permitted to be paid to the Primary Borrower or any Subsidiary Guarantor during such period by any Wholly Owned Subsidiary of the Primary Borrower that is a first tier Insurance Subsidiary without prior approval by any Governmental Authority in accordance with the terms of any order, statute, rule or governmental regulation pertaining to such first-tier Insurance Subsidiary (it being understood that for any period in 2001, such additional amounts shall be calculated on a pro forma basis as if the Acquisition had closed on December 31, 2000) plus or minus (c) net cash tax receipts/payments of the Primary Borrower during such period pursuant to the Tax Sharing Agreement (other than any taxes or credits in respect of asset sales). "CASH INTEREST EXPENSE": of any Person for any period, total cash interest expense (including that attributable to Capital Lease Obligations) of such Person for such period with respect to all outstanding Indebtedness of such Person (including, without limitation, all commissions, discounts and other fees and charges owed by such Person with respect to letters of credit and bankers' acceptance financing and net costs of such Person under Hedge Agreements in respect of interest rates to the extent such net costs are allocable to such period in accordance with GAAP).

"CGNU PLC": as defined in the recitals hereto.

"CGU": as defined in the recitals hereto.

"CGU DELAWARE": as defined in the recitals hereto.

"CGU LUXEMBOURG": as defined in the recitals hereto.

"CLOSING DATE": the date on which the conditions precedent set forth in Section 5.1 shall have been satisfied, which date shall be not later than April 13, 2001; PROVIDED, however, that, if on April 13, 2001 receipt of all approvals of Governmental Authorities required for the consummation of the Acquisition as set forth in the Stock Purchase Agreement shall not have been obtained but all other conditions to the consummation of the Acquisition as set forth in the Stock Purchase Agreement shall have been fulfilled or shall be capable of being fulfilled if April 13, 2001 were to be the Closing Date, then the last date upon which the Closing Date may occur shall be June 30, 2001.

"CODE": the Internal Revenue Code of 1986, as amended from time to time.

"COLLATERAL": all Property of the Loan Parties, now owned or hereafter acquired, upon which a Lien is purported to be created by any Security Document.

"COMMITMENT": with respect to any Lender, each of the Tranche A Term Loan Commitment, the Tranche B Term Loan Commitment, and the Revolving Credit Commitment of such Lender.

"COMMITMENT FEE RATE": the rate per annum set forth below which corresponds with the most current rating of the Facilities issued by S&P and/or Moody's respectively.

Level	Rating	Commitment Fee

1	(g	reater than or equal to) BBB+ and Baa1	0.25%
2	BB	B and Baa2	0.25%
3	BB	B- and Baa3	0.375%
4	BB	B- or Baa3 and BB+ or Ba1	0.375%
5	BB	+ and Ba1	0.375%
6	(1	ess than or equal to) BB and Ba2	0.375%

Changes in the Commitment Fee shall become effective on the date on which S&P and/or Moody's changes the rating it has issued for the Facilities. In the event of split ratings, the lower rating shall apply, PROVIDED that if the ratings are split by two or more levels, the rating level next higher than the lower of the two ratings will apply. Level 4 applies when there is either a BBB- or a Baa3 rating and a BB+ and/or a Ba1 rating. In the event that, at any time, the Facilities are not rated by one of such rating agencies, the rating of such rating agency shall be deemed to be the rating level next lower than the rating agencies. In the event that, at any time, the rating issued by the other of such rating agencies. In the event that, at any time, the Facilities are rated by neither such rating agency, the Commitment Fee Rate shall be as for Level 6.

"COMMONLY CONTROLLED ENTITY": an entity, whether or not incorporated, that is under common control with the Primary Borrower within the meaning of Section 4001 of ERISA or that is treated as a single employer with the Primary Borrower under Section 414 of the Code.

"COMPLIANCE CERTIFICATE": a certificate duly executed by a Responsible Officer of the Primary Borrower substantially in the form of Exhibit B.

"CONFIDENTIAL INFORMATION MEMORANDUM": the Confidential Information Memorandum dated November 2000 and furnished to the initial Lenders in connection with the syndication of the Facilities, as supplemented and modified by the Confidential Information Memorandum Supplement.

"CONFIDENTIAL INFORMATION MEMORANDUM SUPPLEMENT": the Confidential Information Memorandum Supplement dated February 2001 and furnished to the Lenders in connection with the syndication of the Facilities.

"CONSOLIDATED NET INCOME": of any Person for any period, the consolidated net income (or loss) of such Person and its Subsidiaries for such period, determined on a consolidated basis in accordance with GAAP; PROVIDED, that in calculating Consolidated Net Income of the Primary Borrower and its consolidated Subsidiaries for any period, there shall be excluded the income (or deficit) of any Person accrued prior to the date it becomes a Subsidiary of the Primary Borrower or is merged into or consolidated with the Primary Borrower or any of its Subsidiaries.

"CONSOLIDATED NET WORTH": at any date, the sum of all amounts that would, in conformity with GAAP, but excluding the effects of SFAS 115, be included on a consolidated balance sheet of the Primary Borrower and its Subsidiaries under stockholders' equity at such date. Consolidated Net Worth shall in any event include the amount of the Berkshire Preferred Stock so long as it is outstanding.

"CONTRACTUAL OBLIGATION": as to any Person, any provision of any security issued by such Person or of any agreement, instrument or other undertaking to which such Person is a party or by which it or any of its Property is bound. "CONTROL INVESTMENT AFFILIATE": as to any Person, any other Person that (a) directly or indirectly, is in control of, is controlled by, or is under common control with, such Person and (b) is organized by such Person primarily for the purpose of making equity or debt investments in one or more companies. For purposes of this definition, "control" of a Person means the power, directly or indirectly, to direct or cause the direction of the management and policies of such Person, whether by contract or otherwise. Control Investment Affiliates of any Lender shall include any Related Fund of such Lender.

"DEFAULT": any of the events specified in Section 8, whether or not any requirement for the giving of notice, the lapse of time, or both, has been satisfied.

"DEFAULTING LENDER": any Lender that defaults in its obligation to make any Loan hereunder, so long as such default is continuing.

"DEPARTMENT": with respect to any Insurance Subsidiary, the insurance commissioner or other Governmental Authority of such Insurance Subsidiary's state of domicile with which such Insurance Subsidiary is required to file its Annual Statement.

"DERIVATIVES COUNTERPARTY": as defined in Section 7.6.

"DISPOSITION": with respect to any Property, any sale, lease, sale and leaseback, assignment, conveyance, transfer or other disposition thereof; and the terms "DISPOSE" and "DISPOSED OF" shall have correlative meanings.

"DOLLARS" and "\$": lawful currency of the United States of America.

"DOMESTIC SUBSIDIARY": any Subsidiary of the Primary Borrower organized under the laws of any jurisdiction within the United States of America.

"ENVIRONMENTAL LAWS": any and all laws, rules, orders, regulations, statutes, ordinances, guidelines, codes, decrees, or other legally enforceable requirements (including, without limitation, common law) of any international authority, foreign government, the United States, or any state, local, municipal or other governmental authority, regulating, relating to or imposing liability or standards of conduct concerning protection of the environment or of human health, as has been, is now, or may at any time hereafter be, in effect.

"ENVIRONMENTAL PERMITS": any and all permits, licenses, approvals, registrations, notifications, exemptions and other authorizations required under any Environmental Law.

"ERISA": the Employee Retirement Income Security Act of 1974, as amended from time to time.

"EUROCURRENCY RESERVE REQUIREMENTS": for any day, the aggregate (without duplication) of the maximum rates (expressed as a decimal fraction) of reserve requirements in effect on such day (including, without limitation, basic, supplemental, marginal and emergency reserves) under any regulations of the Board or other Governmental Authority having jurisdiction with respect thereto dealing with reserve requirements prescribed for eurocurrency funding (currently referred to as Eurocurrency Liabilities" in Regulation D of the Board) maintained by a member bank of the Federal Reserve System.

"EURODOLLAR RATE": with respect to each day during each Interest Period, the rate per annum determined on the basis of the rate for deposits in Dollars for a period equal to such Interest Period commencing on the first day of such Interest Period appearing on Page 3750 of the Telerate screen as of 11:00 A.M., London time, two Business Days prior to the beginning of such Interest Period. In the event that such rate does not appear on Page 3750 of the Telerate screen (or otherwise on such screen), the "EURODOLLAR RATE" for purposes of this definition shall be determined by reference to such other comparable publicly available service for displaying eurodollar rates as may be selected by the Administrative Agent.

"EURODOLLAR LOANS": Loans for which the applicable rate of interest is based upon the Eurodollar Rate.

"EURODOLLAR TRANCHE": the collective reference to Eurodollar Loans under any Facility the then current Interest Periods with respect to all of which begin on the same date and end on the same later date (whether or not such Loans shall originally have been made on the same day).

"EVENT OF DEFAULT": any of the events specified in Section 8, PROVIDED that any requirement for the giving of notice, the lapse of time, or both, has been satisfied.

"EXCESS CASH": cash of the Primary Borrower on the Closing Date in excess of \$80,000,000, after giving effect to the making of the Loans on the Closing Date and the application of the proceeds thereof (and assuming each of the Facilities is fully drawn).

"EXCLUDED FOREIGN SUBSIDIARIES": any Foreign Subsidiary in respect of which either (i) the pledge of all of the Capital Stock of such Subsidiary as Collateral or (ii) the guaranteeing by such Subsidiary of the Obligations, would, in the good faith judgment of the Primary Borrower, result in materially adverse tax consequences to the Primary Borrower.

"FACILITY": each of (a) the Tranche A Term Loan Commitments and the Tranche A Term Loans made thereunder (the "TRANCHE A TERM LOAN FACILITY"), (b) the Tranche B Term Loan Commitments and the Tranche B Term Loans made thereunder (the "TRANCHE B TERM LOAN FACILITY"), and (c) the Revolving Credit Commitments and the extensions of credit made thereunder (the "REVOLVING CREDIT FACILITY").

"FEDERAL FUNDS EFFECTIVE RATE": for any day, the weighted average of the rates on overnight federal funds transactions with members of the Federal Reserve System arranged by federal funds brokers, as published on the next succeeding Business Day by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations for the day of such transactions received by the Administrative Agent from three federal funds brokers of recognized standing selected by it.

"FIXED CHARGE COVERAGE RATIO": for any period, the ratio of Cash Flow of the Primary Borrower for such period to Fixed Charges of the Primary Borrower for such period. For purposes of calculating the Fixed Charge Coverage Ratio, (a) with respect to any principal installment payments in respect of the Term Loans made annually (under the circumstances described in the provisos to Section 2.3(a) and Section 2.3(b)) for the first two years following the Closing Date, one-fourth of each such annual installment payment shall be deemed to have been paid at the end of each complete fiscal quarter commencing after the Closing Date and (b) with respect to such calculation for the first three complete fiscal quarters of the Primary Borrower commencing after the Closing Date, Fixed Charges of the Primary Borrower for the relevant period shall be deemed to equal Fixed Charges of the Primary Borrower for such fiscal quarter (and, in the case of the latter two such determinations, each previous fiscal quarter commencing after the Closing Date) MULTIPLIED BY 4, 2 and 4/3, respectively.

"FIXED CHARGES": of the Primary Borrower for any period, the sum (without duplication) of (a) Cash Interest Expense of the Primary Borrower for such period, (b) scheduled payments made by the Primary Borrower during such period on account of principal of Indebtedness (including scheduled principal payments in respect of the Term Loans but excluding termination of the Revolving Credit Facility), (c) Capital Expenditures of the Primary Borrower for such period and (d) any dividend required to be paid by the Primary Borrower on, or any payment required to be made by the Primary Borrower on account of, the purchase, redemption, defeasance, retirement or other acquisition of, any Capital Stock of the Primary Borrower, whether now or hereafter outstanding, or any other distribution required to be made by the Primary Borrower in respect thereof, either directly or indirectly, whether in cash or property or in obligations of the Primary Borrower or any Subsidiary (including the dividends to the extent paid to service the Berkshire Preferred Stock).

"FOLKSAMERICA": Folksamerica Holding Company, Inc.

"FOLKSAMERICA PROMISSORY NOTE": the promissory note, dated as of June 29, 1999, issued by Folksamerica to Centris Group, the original principal amount of which was \$20,750,000 carried on the GAAP financial statements of Folksamerica as of December 31, 2000 at zero.

"FOREIGN SUBSIDIARY": any Subsidiary of the Primary Borrower that is not a Domestic Subsidiary.

"FUNDING OFFICE": the office specified from time to time by the Administrative Agent as its funding office by notice to the Borrowers and the Lenders.

"GAAP": generally accepted accounting principles in the United States of America as in effect from time to time, except that for purposes of Section 7.1, GAAP shall be determined on the basis of such principles in effect on the date hereof and consistent with those used in the preparation of the most recent audited financial statements referred to in Section 4.1(b).

"GOVERNMENTAL AUTHORITY": any nation or government, any state or other political subdivision thereof and any entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, and any corporation or other entity owned or controlled, through stock or capital ownership or otherwise, by any of the foregoing, including any board of insurance, insurance department or insurance commissioner.

"GUARANTEE AND COLLATERAL AGREEMENT": the Guarantee and Collateral Agreement to be executed and delivered by the Parent, the Primary Borrower and each Subsidiary Guarantor, substantially in the form of Exhibit A, as the same may be amended, supplemented or otherwise modified from time to time.

"GUARANTEE OBLIGATION": as to any Person (the "GUARANTEEING PERSON"), any obligation of (a) the guaranteeing person or (b) another Person (including, without limitation, any bank under any letter of credit) to induce the creation of which the guaranteeing person has issued a reimbursement, counterindemnity or similar obligation, in either case guaranteeing or in effect guaranteeing any Indebtedness, leases, dividends or other obligations (the "PRIMARY OBLIGATIONS") of any other third Person (the "PRIMARY OBLIGOR") in any manner, whether directly or indirectly, including, without limitation, any obligation of the guaranteeing person, whether or not contingent, (i) to purchase any such primary obligation or any Property constituting direct or indirect security therefor, (ii) to advance or supply funds (1) for the purchase or payment of any such primary obligation or (2) to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor, (iii) to purchase Property, securities or services primarily for the purpose of assuring the owner of any such primary obligation of the ability of the primary obligation or (iv) otherwise to assure or hold harmless the owner of any such primary obligation against loss in respect thereof; PROVIDED, HOWEVER, that the term Guarantee Obligation shall not include endorsements of instruments for deposit or collection in the ordinary course of business. The amount of any Guarantee Obligation of any guaranteeing person shall be deemed to be the lower of (a) an amount equal to the stated or determinable amount of the primary obligation in respect of which such Guarantee Obligation is made and (b) the maximum amount for which such guaranteeing person may be liable pursuant to the terms of the instrument embodying such Guarantee Obligation, unless such primary obligation and the maximum amount for which such guaranteeing person may be liable are not stated or determinable, in which case the amount of such Guarantee Obligation shall be such guaranteeing person's maximum reasonably anticipated liability in respect thereof as determined by the Primary Borrower in good faith.

"GUARANTORS": the collective reference to the Parent, the Primary Borrower (in respect of borrowings by the Additional Borrower pursuant to Section 2.25) and the Subsidiary Guarantors. "HEDGE AGREEMENTS": all interest rate swaps, caps or collar agreements or similar arrangements entered into by the Primary Borrower or its Subsidiaries providing for protection against fluctuations in interest rates or currency exchange rates or the exchange of nominal interest obligations, either generally or under specific contingencies.

"INDEBTEDNESS": of any Person at any date, without duplication, (a) all indebtedness of such Person for borrowed money, (b) all obligations of such Person for the deferred purchase price of Property or services (other than trade payables incurred in the ordinary course of such Person's business), (c) all obligations of such Person evidenced by notes, bonds, debentures or other similar instruments, (d) all indebtedness created or arising under any conditional sale or other title retention agreement with respect to Property acquired by such Person (even though the rights and remedies of the seller or lender under such agreement in the event of default are limited to repossession or sale of such Property), (e) all Capital Lease Obligations of such Person, (f) all obligations of such Person, contingent or otherwise, as an account party or applicant under acceptance, letter of credit or similar facilities, (g) all obligations of such Person, contingent or otherwise, to purchase, redeem, retire or otherwise acquire for value any Capital Stock of such Person, (h) all Guarantee Obligations of such Person in respect of obligations of the kind referred to in clauses (a) through (g) above, (i) all obligations of the kind referred to in clauses (a) through (h) above secured by (or for which the holder of such obligation has an existing right, contingent or otherwise, to be secured by) any Lien on Property (including, without limitation, accounts and contract rights) owned by such Person, whether or not such Person has assumed or become liable for the payment of such obligation and (j) for the purposes of Section 8(e) only, all obligations of such Person in respect of Hedge Agreements.

"INDEMNIFIED LIABILITIES": as defined in Section 10.5.

"INDEMNITEE": as defined in Section 10.5.

"INSOLVENCY": with respect to any Multiemployer Plan, the condition that such Plan is insolvent within the meaning of Section 4245 of ERISA.

"INSOLVENT": pertaining to a condition of Insolvency.

"INSURANCE REGULATIONS": any law, regulation, rule, directive or order applicable to an insurance company.

"INSURANCE REGULATOR": any Person charged with the administration, oversight or enforcement of any Insurance Regulation.

"INSURANCE SUBSIDIARY": any Subsidiary which is required to be licensed by any Department as an insurer or reinsurer and each direct or indirect Subsidiary of such Subsidiary.

"INTELLECTUAL PROPERTY": the collective reference to all rights, priorities and privileges relating to intellectual property, whether arising under United States, multinational or foreign laws or otherwise, including, without limitation, copyrights, copyright licenses, patents, patent licenses, trademarks, trademark licenses, technology, know-how and processes, and all rights to sue at law or in equity for any infringement or other impairment thereof, including the right to receive all proceeds and damages therefrom.

"INTEREST COVERAGE RATIO": for any period, the ratio of Cash Flow of the Primary Borrower for such period to Cash Interest Expense of the Primary Borrower for such period; PROVIDED, that, for the purposes of determining the Interest Coverage Ratio for the first three complete fiscal quarters of the Primary Borrower commencing after the Closing Date, Cash Interest Expense of the Primary Borrower for the relevant period shall be deemed to equal Cash Interest Expense of the Primary Borrower for such fiscal quarter (and, in the case of the latter two such determinations, each previous fiscal quarter commencing after the Closing Date) MULTIPLIED BY 4, 2 and 4/3, respectively.

"INTEREST PAYMENT DATE": (a) as to any Base Rate Loan, the last day of each March, June, September and December to occur while such Loan is outstanding and the final maturity date of such Loan, (b) as to any Eurodollar Loan having an Interest Period of three months or shorter, the last day of such Interest Period, (c) as to any Eurodollar Loan having an Interest Period longer than three months, each day that is three months, or a whole multiple thereof, after the first day of such Interest Period and the last day of such Interest Period and (d) as to any Loan (other than any Revolving Credit Loan that is a Base Rate Loan and any Swing Line Loan), the date of any repayment or prepayment made in respect thereof.

"INTEREST PERIOD": as to any Eurodollar Loan, (a) initially, the period commencing on the borrowing or conversion date, as the case may be, with respect to such Eurodollar Loan and ending one, two, three or six months (or, if consented to by each relevant Lender, nine or twelve months) thereafter, as selected by the relevant Borrower in its notice of borrowing or notice of conversion, as the case may be, given with respect thereto; and (b) thereafter, each period commencing on the last day of the next preceding Interest Period applicable to such Eurodollar Loan and ending one, two, three or six months (or, if consented to by each relevant Lender, nine or twelve months) thereafter, as selected by such Borrower by irrevocable notice to the Administrative Agent not less than three Business Days prior to the last day of the then current Interest Period with respect thereto; PROVIDED that, all of the foregoing provisions relating to Interest Periods are subject to the following:

(i) if any Interest Period would otherwise end on a day that is not a Business Day, such Interest Period shall be extended to the next succeeding Business Day unless the result of such extension would be to carry such Interest Period into another calendar month in which event such Interest Period shall end on the immediately preceding Business Day;

(ii) any Interest Period in respect of Revolving Credit Loans that would otherwise extend beyond the Revolving Credit Termination Date shall end on the Revolving Credit Termination Date, and any other Interest Period in respect of Term Loans under any Term Loan Facility that would otherwise extend beyond the date final payment is due under such Term Loan Facility shall end on such due date; and

(iii) any Interest Period that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Business Day of the calendar month at the end of such Interest Period.

"INVESTOR GROUP": Franklin Mutual Advisers, LLC, Leucadia National Corporation, Banc One Capital Corporation, Fairfax Financial Holdings Limited, Gotham Partners, Markel Corporation, DLJ Merchant Banking, Prospector Partners, and certain management of White Mountains and its Subsidiaries, GCU and its Subsidiaries, and certain other parties.

"INVESTMENTS": as defined in Section 7.7.

"ISSUING LENDER": Fleet National Bank and any other Revolving Credit Lender from time to time designated by the Primary Borrower as an Issuing Lender with the consent of such Revolving Credit Lender and the Administrative Agent.

"L/C COMMITMENT": \$25,000,000.

"L/C FEE PAYMENT DATE": the last day of each March, June, September and December and the last day of the Revolving Credit Commitment Period.

"L/C OBLIGATIONS": at any time, an amount equal to the sum of (a) the aggregate then undrawn and unexpired amount of the then outstanding Letters of Credit and (b) the aggregate amount of drawings under Letters of Credit that have not then been reimbursed pursuant to Section 3.5.

"L/C PARTICIPANTS": with respect to any Letter of Credit, the collective reference to all the Revolving Credit Lenders other than the Issuing Lender that issued such Letter of Credit.

"LEHMAN ENTITY": any of Lehman Commercial Paper Inc. or any of its affiliates (including Syndicated Loan Funding Trust).

"LENDER ADDENDUM": with respect to any initial Lender, a Lender Addendum, substantially in the form of Exhibit J, to be executed and delivered by such Lender on the Closing Date as provided in Section 10.17.

"LENDERS": as defined in the preamble hereto.

"LETTERS OF CREDIT": as defined in Section 3.1(a).

"LEVERAGE RATIO": as at the last day of each fiscal quarter of the Primary Borrower, the ratio of (a) Total Debt on such day to (b) the sum of (i) Total Debt on such day plus (ii) Consolidated Net Worth on such day. "LICENSE": any license, certificate of authority, permit or other authorization which is required to be obtained from any Governmental Authority in connection with the operation, ownership or transaction of insurance or reinsurance business.

"LIEN": any mortgage, pledge, hypothecation, assignment, deposit arrangement, encumbrance, lien (statutory or other), charge or other security interest or any preference, priority or other security agreement or preferential arrangement of any kind or nature whatsoever (including, without limitation, any conditional sale or other title retention agreement and any capital lease having substantially the same economic effect as any of the foregoing).

"LOAN": any loan made by any Lender pursuant to this Agreement.

"LOAN DOCUMENTS": this Agreement, the Security Documents, the Syndication Letter Agreement, the Applications and the Notes.

"LOAN PARTIES": the Parent, the Borrowers, and each Subsidiary of the Primary Borrower that is a party to a Loan Document.

"MAJORITY FACILITY LENDERS": with respect to any Facility, the holders of more than 50% of the aggregate unpaid principal amount of the Term Loans or the Total Revolving Extensions of Credit, as the case may be, outstanding under such Facility (or, in the case of the Revolving Credit Facility, prior to any termination of the Revolving Credit Commitments, the holders of more than 50% of the Total Revolving Credit Commitments).

"MAJORITY REVOLVING CREDIT FACILITY LENDERS": the Majority Facility Lenders in respect of the Revolving Credit Facility.

"MANDATORY PREPAYMENT AMOUNT": as defined in Section 2.18(d).

"MATERIAL ADVERSE EFFECT": (a) when such term is used in the representation and warranty to be made on the Closing Date in Section 4.2, any event or development that has had or would reasonably be expected to have a material adverse effect on (i) the business, operations, property or financial condition of the Primary Borrower and its subsidiaries, taken as a whole, the Acquired Assets or White Mountains and its Subsidiaries, taken as a whole, or (ii) the validity or enforceability of any of this Agreement or any of the other Loan Documents or the rights and remedies of the Administrative Agent and the Lenders thereunder, other than any such effect resulting from (A) changes in general economic or securities or financial market conditions (including changes in interest rates), (B) any write-off of assets or establishment of liabilities identified in Section 1.1(a) of the Disclosure Schedule to the Stock Purchase Agreement, in each case in amounts materially consistent with the amounts reflected in the updated projections set forth in the Confidential Information Memorandum or (C) any ratings downgrade of any Material Insurance Subsidiary by A.M. Best to a level not below A-; and (b) when such term is used in any context other than as set forth in the foregoing clause (a), any event, development or circumstance that has had or could

reasonably be expected to have a material adverse effect on (i) the Acquisition or the other transactions contemplated hereby, (ii) the business, assets, property or financial condition of the Primary Borrower and its Subsidiaries taken as a whole, or (iii) the validity or enforceability of this Agreement or any of the other Loan Documents or the rights and remedies of the Administrative Agent and the Lenders hereunder or thereunder. "MATERIAL INSURANCE SUBSIDIARY": an Insurance Subsidiary having Capital and Surplus of \$50,000,000 or more.

"MATERIALS OF ENVIRONMENTAL CONCERN": any gasoline or petroleum (including crude oil or any fraction thereof) or petroleum products, polychlorinated biphenyls, urea-formaldehyde insulation, asbestos, pollutants, contaminants, radioactivity, and any other substances or forces of any kind, whether or not any such substance or force is defined as hazardous or toxic under any Environmental Law, that is in relevant form or concentration regulated pursuant to or could reasonably be expected to give rise to liability under any Environmental Law.

"MOODY'S": Moody's Investors Service, Inc. (or any successor thereto).

"MULTIEMPLOYER PLAN": a Plan that is a multiemployer plan as defined in Section 4001(a)(3) of ERISA.

"NAIC": the National Association of Insurance Commissioners or any successor thereto, or in the absence of the National Association of Insurance Commissioners or such successor, any other association, agency or other organization performing advisory, coordination or other like functions among insurance departments, insurance commissioners and similar Governmental Authorities of the various states of the United States towards the promotion of uniformity in the practices of such Governmental Authorities.

"NET PROCEEDS": (a) in connection with any Asset Sale or any Recovery Event, the proceeds thereof in the form of cash and Cash Equivalents (including any such proceeds received by way of deferred payment of principal pursuant to a note or installment receivable or purchase price adjustment receivable or otherwise, but only as and when received) of such Asset Sale or Recovery Event, net of (i) attorneys' fees, accountants' fees, investment banking fees, amounts required to be applied to the repayment of Indebtedness secured by a Lien expressly permitted hereunder on any asset which is the subject of such Asset Sale or Recovery Event (other than any Lien pursuant to a Security Document) and other customary fees and expenses actually incurred in connection therewith, (ii) taxes paid or reasonably estimated to be payable as a result thereof (after taking into account any available tax credits or deductions and any tax sharing arrangements) and (iii) appropriate amounts to be provided by the Parent, the Primary Borrower or any Subsidiary, as the case may be, as a reserve required in accordance with GAAP against any liabilities associated with such Asset Sale and retained by the Parent, the Primary Borrower or any Subsidiary, as the case may be, after such Asset Sale, including, without limitation, reserves for pension and other postemployment benefit liabilities and liabilities under any indemnification obligations associated with such Asset Sale; and (b) in connection with any issuance or sale of equity securities or debt securities or instruments or the incurrence of loans, the cash proceeds received from such issuance or incurrence, and cash received in respect of any non-cash proceeds received from such issuance or incurrence, net of attorneys' fees, investment banking fees, accountants' fees, underwriting discounts and commissions and other customary fees and expenses actually incurred in connection therewith.

"NEW LENDER SUPPLEMENT": as defined in Section 2.25.

"NON-EXCLUDED TAXES": as defined in Section 2.20(a).

"NON-INSURANCE SUBSIDIARY": any Subsidiary other than an Insurance Subsidiary.

"NON-U.S. LENDER": as defined in Section 2.20(d).

"NOTE": any promissory note evidencing any Loan.

"OPTIONAL PREPAYMENT AMOUNT": as defined in Section 2.18(d).

"OTHER TAXES": any and all present or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies arising from any payment made hereunder or from the execution, delivery or enforcement of, or otherwise with respect to, this Agreement or any other Loan Document.

"PARTICIPANT": as defined in Section 10.6(b).

"PAYMENT OFFICE": the office specified from time to time by the Administrative Agent as its payment office by notice to each Borrower and the Lenders.

"PBGC": the Pension Benefit Guaranty Corporation established pursuant to Subtitle A of Title IV of ERISA (or any successor).

"PERSON": an individual, partnership, corporation, limited liability company, business trust, joint stock company, trust, unincorporated association, joint venture, Governmental Authority or other entity of whatever nature.

"PLAN": at a particular time, any employee benefit plan that is covered by ERISA and in respect of which the Primary Borrower or a Commonly Controlled Entity is (or, if such plan were terminated at such time, would under Section 4069 of ERISA be deemed to be) an "employer" as defined in Section 3(5) of ERISA.

"PRECLOSING TRANSACTIONS": as defined in the Stock Purchase Agreement.

"PREPAYMENT OPTION NOTICE": as defined in Section 2.18(d).

"PRIMARY BORROWER": as defined in the recitals hereto.

"PRO FORMA BALANCE SHEET": as defined in Section 4.1(a).

"PROJECTIONS": as defined in Section 6.2(c).

"PROPERTY": any right or interest in or to property of any kind whatsoever, whether real, personal or mixed and whether tangible or intangible, including, without limitation, Capital Stock.

"QUARTERLY STATEMENT": the quarterly statutory financial statement of any Insurance Subsidiary required to be filed with the Department of its jurisdiction of incorporation, which statement shall be in the form required by such Insurance Subsidiary's jurisdiction of incorporation or, if no specific form is so required, in the form of financial statements permitted by such Department to be used for filing quarterly statutory financial statements and shall contain the type of information permitted or required by such Department to be disclosed therein, together with all exhibits or schedules filed therewith.

"RECOVERY EVENT": any settlement of or payment in respect of any property or casualty insurance claim or any condemnation proceeding relating to any asset of the Parent, the Primary Borrower or any of its Subsidiaries. It is understood and agreed that "Recovery Event" shall not include any claim paid pursuant to a policy issued as part of the insurance business of the Primary Borrower and its Insurance Subsidiaries or in settlement of any such claim.

"REFUNDED SWING LINE LOANS": as defined in Section 2.7.

"REFUNDING DATE": as defined in Section 2.7.

"REGISTER": as defined in Section 10.6(d).

"REGULATION H": Regulation H of the Board as in effect from time to time.

"REGULATION U": Regulation U of the Board as in effect from time to time.

"REIMBURSEMENT OBLIGATION": the obligation of any Borrower to reimburse each Issuing Lender pursuant to Section 3.5 for amounts drawn under Letters of Credit issued by such Issuing Lender for the account of such Borrower.

"REINSURANCE AGREEMENT": any agreement, contract, treaty, certificate or other arrangement by which any Insurance Subsidiary agrees to transfer or cede to another insurer all or part of the liability assumed or assets held by it under one or more insurance, annuity, reinsurance or retrocession policies, agreements, contracts, treaties, certificates or similar arrangements. Reinsurance Agreements shall include, but not be limited to, any agreement, contract, treaty, certificate or other arrangement which is treated as such by the applicable Department.

"REINVESTMENT DEFERRED AMOUNT": with respect to any Reinvestment Event, the aggregate Net Proceeds received by the Parent, the Primary Borrower or any of its Subsidiaries in connection therewith that are not applied to prepay the Term Loans or reduce the Revolving Credit Commitments pursuant to Section 2.12(b) as a result of the delivery of a Reinvestment Notice.

"REINVESTMENT EVENT": any Asset Sale or Recovery Event in respect of which the Primary Borrower has delivered a Reinvestment Notice.

"REINVESTMENT NOTICE": a written notice executed by a Responsible Officer of the Primary Borrower stating that no Default or Event of Default has occurred and is continuing and that the Primary Borrower (directly or indirectly through a Subsidiary) intends and expects to use all or a specified portion of the Net Proceeds of an Asset Sale or Recovery Event to acquire assets useful in its business.

"REINVESTMENT PREPAYMENT AMOUNT": with respect to any Reinvestment Event, the Reinvestment Deferred Amount relating thereto LESS any amount expended prior to the relevant Reinvestment Prepayment Date to acquire assets useful in the Primary Borrower's business.

"REINVESTMENT PREPAYMENT DATE": with respect to any Reinvestment Event, the earlier of (a) the date occurring twelve months after such Reinvestment Event and (b) the date on which the Primary Borrower shall have determined not to, or shall have otherwise ceased to, acquire assets useful in the Primary Borrower's business with all or any portion of the relevant Reinvestment Deferred Amount.

"RELATED FUND": with respect to any Lender that is a fund that invests in commercial loans, any other fund that invests in commercial loans and is managed or advised by the same investment advisor as such Lender or by an Affiliate of such investment advisor.

"REORGANIZATION": with respect to any Multiemployer Plan, the condition that such plan is in reorganization within the meaning of Section 4241 of ERISA.

"REPORTABLE EVENT": any of the events set forth in Section 4043(c) of ERISA, other than those events as to which the thirty day notice period is waived.

"REQUIRED LENDERS": at any time, the holders of more than 50% of (a) until the Closing Date, the Commitments and (b) thereafter, the sum of (i) the aggregate unpaid principal amount of the Term Loans then outstanding and (ii) the Total Revolving Credit Commitments then in effect or, if the Revolving Credit Commitments have been terminated, the Total Revolving Extensions of Credit then outstanding. The Term Loans outstanding and the Total Revolving Commitments in effect (or, when applicable, Total Revolving Extensions of Credit outstanding) of any Defaulting Lender shall be excluded for purposes of any vote of Required Lenders.

"REQUIRED PREPAYMENT LENDERS": the Majority Facility Lenders in respect of each Facility.

"REQUIREMENT OF LAW": as to any Person, the Certificate of Incorporation and By-Laws or other organizational or governing documents of such Person (excluding, in the case of Section 2.19(a)(i), any of the foregoing relating to the Administrative Agent or any Lender), and any law, treaty, rule or regulation or determination of an arbitrator or a court or other Governmental Authority, in each case applicable to or binding upon such Person or any of its Property or to which such Person or any of its Property is subject.

"RESPONSIBLE OFFICER": the chief executive officer, president, chief financial officer, Vice President and Director of Finance (or equivalent responsible party), chief accounting officer or treasurer of the Primary Borrower or any Insurance Subsidiary, as the context requires, but in any event, with respect to financial matters, the chief financial officer or Vice President and Director of Finance (or equivalent responsible party) of the Primary Borrower or any such Insurance Subsidiary.

"RESTRICTED PAYMENTS": as defined in Section 7.6.

"REVOLVING COMMITMENT INCREASE DATE": as defined in Section 2.25.

"REVOLVING COMMITMENT INCREASE SUPPLEMENT": as defined in Section 2.25.

"REVOLVING CREDIT COMMITMENT": as to any Lender, the obligation of such Lender, if any, to make Revolving Credit Loans and participate in Swing Line Loans and Letters of Credit, in an aggregate principal and/or face amount not to exceed the amount set forth under the heading "Revolving Credit Commitment" opposite such Lender's name on Schedule 1 to the Lender Addendum delivered by such Lender, or, as the case may be, in the Assignment and Acceptance pursuant to which such Lender became a party hereto, as the same may be changed from time to time pursuant to the terms hereof. The aggregate amount of the Total Revolving Credit Commitments on the date of this Agreement is \$175,000,000.

"REVOLVING CREDIT COMMITMENT PERIOD": the period from and including the Closing Date to the Revolving Credit Termination Date.

"REVOLVING CREDIT LENDER": each Lender that has a Revolving Credit Commitment or that is the holder of Revolving Credit Loans.

"REVOLVING CREDIT LOANS": as defined in Section 2.4.

"REVOLVING CREDIT PERCENTAGE": as to any Revolving Credit Lender at any time, the percentage which such Lender's Revolving Credit Commitment then constitutes of the Total Revolving Credit Commitments (or, at any time after the Revolving Credit Commitments shall have expired or terminated, the percentage which the aggregate amount of such Lender's Revolving Extensions of Credit then outstanding constitutes the amount of the Total Revolving Extensions of Credit then outstanding).

"REVOLVING CREDIT TERMINATION DATE": the fifth anniversary of the Closing Date.

"REVOLVING EXTENSIONS OF CREDIT": as to any Revolving Credit Lender at any time, an amount equal to the sum of (a) the aggregate principal amount of all Revolving Credit Loans made by such Lender then outstanding, (b) the principal amount equal to such Lender's Revolving Credit Percentage of the L/C Obligations then outstanding and (c) the principal amount equal to such Lender's Revolving Credit Percentage of the aggregate principal amount of Swing Line Loans then outstanding.

"RISK BASED CAPITAL RATIO": with respect to any Insurance Subsidiary on any date of determination thereof, the ratio (expressed as a percentage) of (a) the Total Adjusted Capital (as defined by the NAIC) of such Insurance Subsidiary to (b) the Authorized Control Level Risk-Based Capital (as defined by the NAIC) of such Insurance Subsidiary. It is understood and agreed that the Risk Based Capital Ratio shall not be applied to Potomac Insurance Company of Pennsylvania.

 $"\ensuremath{\mathsf{S\&P}}"\ensuremath{:}$ Standard & Poor's Rating Services (or any successor thereto).

"SAP": with respect to any Insurance Subsidiary, the statutory accounting practices prescribed or permitted by the Department in the jurisdiction of such Insurance Subsidiary for the preparation of annual statements and other financial reports by insurance companies of the same type as such Insurance Subsidiary, which are applicable to the circumstances as of the date of determination.

"SEC": the Securities and Exchange Commission (or successors thereto or an analogous Governmental Authority).

"SECURITY DOCUMENTS": the collective reference to the Guarantee and Collateral Agreement and all other security documents hereafter delivered to the Administrative Agent granting a Lien on any Property of any Person to secure the obligations and liabilities of any Loan Party under any Loan Document.

"SFAS": Statements of Financial Accounting Standards adopted by the Financial Accounting Standards Board.

"SINGLE EMPLOYER PLAN": any Plan that is covered by Title IV of ERISA, but which is not a Multiemployer Plan.

"SOLVENT": with respect to any Person, as of any date of determination, (a) the amount of the "present fair saleable value" of the assets of such Person will, as of such date, exceed the amount of all "liabilities of such Person, contingent or otherwise" after giving effect to the expected value of rights of indemnity, contribution and subrogation, as of such date, as such quoted terms are determined in accordance with applicable federal and state laws governing determinations of the insolvency of debtors, (b) the present fair saleable value of the assets of such Person will, as of such date, be greater than the amount that will be required to pay the liability of such Person on its debts as such debts become absolute and matured after giving effect to the expected value of rights of indemnity, contribution and subrogation, (c) such Person will not have, as of such date, an unreasonably small amount of capital with which to conduct its business, and (d) such Person will be able to pay its debts as they mature after giving effect to the expected value of rights of indemnity, contribution and subrogation. For purposes of this definition, (i) "debt" means liability on a "claim", and (ii) "claim" means any (x) right to payment, whether or not such a right is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, secured or unsecured or (y) right to an equitable remedy for breach of performance if such breach gives rise to a right to payment, whether or not such right to an equitable remedy is reduced to judgment, fixed, contingent, matured or unmatured, disputed, undisputed, secured or unsecured.

"SPECIFIED EXCEPTIONS": (a) any write-off of assets or establishment of liabilities and related adjustments for items identified in Section 1.1(a) of the Disclosure Schedule to the Stock Purchase Agreement, (b) adjustments in amounts materially consistent with the amounts reflected in the updated projections set forth in the Confidential Information Memorandum, (c) adjustments to reserves and reinsurance recoverables as set forth in independent consultants' reports made available to the Lenders in conjunction with the Confidential Information Memorandum, and (d) any adjustments related to discontinued operations of CGU.

"SPECIFIED HEDGE AGREEMENT": any Hedge Agreement entered into by (a) the Primary Borrower or any of its Subsidiaries and (b) any Lender or any affiliate thereof, as counterparty.

"STOCK PURCHASE AGREEMENT": as defined in the recitals hereto.

"SUBORDINATED SELLER NOTE": the promissory note in the principal amount of \$260,000,000, maturing on the date which is eighteen months after the Closing Date, to be issued by the Parent to Subsidiaries of CGNU plc on the Closing Date pursuant to the Acquisition Documentation.

"SUBSIDIARY": as to any Person, a corporation, partnership, limited liability company or other entity of which shares of stock or other ownership interests having ordinary voting power (other than stock or such other ownership interests having such power only by reason of the happening of a contingency) to elect a majority of the board of directors or other managers of such corporation, partnership or other entity are at the time owned, or the management of which is otherwise controlled, directly or indirectly through one or more intermediaries, or both, by such Person. Unless otherwise qualified, all references to a "Subsidiary" or to "Subsidiaries" in this Agreement shall refer to a Subsidiary or Subsidiaries of the Primary Borrower.

"SUBSIDIARY GUARANTOR": each Subsidiary of the Primary Borrower other than A.W.G. Dewar, Inc., any Insurance Subsidiary and any Excluded Foreign Subsidiary.

"SURPLUS DEBENTURES": as to any Insurance Subsidiary, debt securities of such Insurance Subsidiary the proceeds of which are permitted to be included, in whole or in part, as Capital and Surplus of such Insurance Subsidiary as approved and permitted by the applicable Department. "SWING LINE COMMITMENT": the obligation of the Swing Line Lender to make Swing Line Loans pursuant to Section 2.6 in an aggregate principal amount at any one time outstanding not to exceed \$10,000,000.

"SWING LINE LENDER": Lehman Commercial Paper Inc., in its capacity as the Lender of Swing Line Loans.

"SWING LINE LOANS": as defined in Section 2.6.

"SWING LINE PARTICIPATION AMOUNT": as defined in Section 2.7.

"SYNDICATION LETTER AGREEMENT": the Syndication Letter Agreement, dated as of the date of this Agreement, between the Primary Borrower and the Administrative Agent relating to the syndication of the Facilities.

"TAX SHARING AGREEMENT": as defined in Section 6.12.

"TERM LOAN FACILITIES": the collective reference to the Tranche A Term Loan Facility and the Tranche B Term Loan Facility.

"TERM LOAN LENDERS": the collective reference to the Tranche A Term Loan Lenders and the Tranche B Term Loan Lenders.

"TERM LOANS": the collective reference to the Tranche A Term Loans and the Tranche B Term Loans.

"THIRD PARTY PREFERRED STOCK": as defined in Section 5.1(b).

"TOTAL DEBT": at any date, the aggregate principal amount of all Indebtedness of the Primary Borrower at such date, determined on a stand alone basis in accordance with GAAP.

"TOTAL REVOLVING CREDIT COMMITMENTS": at any time, the aggregate amount of the Revolving Credit Commitments then in effect.

"TOTAL REVOLVING EXTENSIONS OF CREDIT": at any time, the aggregate amount of the Revolving Extensions of Credit of the Revolving Credit Lenders outstanding at such time.

"TRANCHE A TERM LOAN": as defined in Section 2.1.

"TRANCHE A TERM LOAN COMMITMENT": as to any Lender, the obligation of such Lender, if any, to make a Tranche A Term Loan to the Primary Borrower hereunder in a principal amount not to exceed the amount set forth under the heading "Tranche A Term Loan Commitment" opposite such Lender's name on Schedule 1 to the Lender Addendum delivered by such Lender, or, as the case may be, in the Assignment and Acceptance pursuant to which such Lender became a party hereto, as the same may be changed from time to time pursuant to the terms hereof. The aggregate amount of the Tranche A Term Loan Commitments on the date of this Agreement is \$300,000,000.

"TRANCHE A TERM LOAN FACILITY": as defined in the definition of "Facility" in this Section 1.1.

"TRANCHE A TERM LOAN LENDER": each Lender that has a Tranche A Term Loan Commitment or is the holder a Tranche A Term Loan.

"TRANCHE A TERM LOAN PERCENTAGE": as to any Lender at any time, the percentage which such Lender's Tranche A Term Loan Commitment then constitutes of the aggregate Tranche A Term Loan Commitments (or, at any time after the Closing Date, the percentage which the aggregate principal amount of such Lender's Tranche A Term Loans then outstanding constitutes of the aggregate principal amount of the Tranche A Term Loans then outstanding).

"TRANCHE B TERM LOAN": as defined in Section 2.1.

"TRANCHE B TERM LOAN COMMITMENT": as to any Lender, the obligation of such Lender, if any, to make a Tranche B Term Loan to the Primary Borrower hereunder in a principal amount not to exceed the amount set forth under the heading "Tranche B Term Loan Commitment" opposite such Lender's name on Schedule 1 to the Lender Addendum delivered by such Lender, or, as the case may be, in the Assignment and Acceptance pursuant to which such Lender became a party hereto, as the same may be changed from time to time pursuant to the terms hereof. The aggregate amount of the Tranche B Term Loan Commitments on the date of this Agreement is \$400,000,000.

"TRANCHE B TERM LOAN FACILITY": as defined in the definition of "Facility" in this Section 1.1.

"TRANCHE B TERM LOAN LENDER": each Lender that has a Tranche B Term Loan Commitment or is the holder of a Tranche B Term Loan.

"TRANCHE B TERM LOAN PERCENTAGE": as to any Lender at any time, the percentage which such Lender's Tranche B Term Loan Commitment then constitutes of the aggregate Tranche B Term Loan Commitments (or, at any time after the Closing Date, the percentage which the aggregate principal amount of such Lender's Tranche B Term Loans then outstanding constitutes of the aggregate principal amount of the Tranche B Term Loans then outstanding).

"TRANSFEREE": as defined in Section 10.14.

"TYPE": as to any Loan, its nature as a Base Rate Loan or a Eurodollar Loan.

"UCC": as defined in Section 3.6.

"UCP": as defined in Section 3.6.

"WARRANT AGREEMENT": the warrant agreement to be entered into among White Mountains, the Primary Borrower and Berkshire Hathaway, pursuant to which Berkshire Hathaway will purchase, for aggregate cash consideration of \$75,000,000, warrants to purchase common stock of White Mountains, having terms and conditions set forth in the term sheet attached to the letter, dated October 13, 2000, from Warren E. Buffett to John J. Byrne.

"WHITE MOUNTAINS": as defined in the recitals hereto.

"WHOLLY OWNED SUBSIDIARY": as to any Person, any other Person all of the Capital Stock of which (other than directors' qualifying shares required by law) is owned by such Person directly and/or through other Wholly Owned Subsidiaries.

"WHOLLY OWNED SUBSIDIARY GUARANTOR": any Subsidiary Guarantor that is a Wholly Owned Subsidiary of the Primary Borrower.

1.2 OTHER DEFINITIONAL PROVISIONS. Unless otherwise specified therein, all terms defined in this Agreement shall have the defined meanings when used in the other Loan Documents or any certificate or other document made or delivered pursuant hereto or thereto.

(a) As used herein and in the other Loan Documents, and any certificate or other document made or delivered pursuant hereto or thereto, accounting terms relating to the Parent, any Borrower and the Subsidiaries of the Primary Borrower not defined in Section 1.1 and accounting terms partly defined in Section 1.1, to the extent not defined, shall have the respective meanings given to them under GAAP or SAP, as the case may be.

(b) References herein to particular pages, columns, lines or sections of any Person's Annual Statement shall be deemed, where appropriate, to be references to the corresponding page, column, line or section of such Person's Quarterly Statement, or if no such corresponding page, column, line or section exists or if any report form changes, then to the corresponding item referenced thereby. In the event the pages, columns, lines or sections of the Annual Statement referenced herein are changed or renumbered from the pages, columns, lines and sections applicable to the Annual Statement of CGU for 1999, all such references shall be deemed references to such page, column, line or section as so renumbered or changed.

(c) The words "hereof", "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement, and Section, Schedule and Exhibit references are to this Agreement unless otherwise specified.

(d) The meanings given to terms defined herein shall be equally applicable to both the singular and plural forms of such terms.

SECTION 2 AMOUNT AND TERMS OF COMMITMENTS

2.1 TERM LOAN COMMITMENTS. Subject to the terms and conditions hereof, (a) the Tranche A Term Loan Lenders severally agree to make term loans (each, a "TRANCHE A TERM

LOAN") to the Primary Borrower on the Closing Date in an amount for each Tranche A Term Loan Lender not to exceed the amount of the Tranche A Term Loan Commitment of such Lender and (b) the Tranche B Term Loan Lenders severally agree to make term loans (each, a "TRANCHE B TERM LOAN") to the Primary Borrower on the Closing Date in an amount for each Tranche B Term Loan Lender not to exceed the amount of the Tranche B Term Loan Commitment of such Lender. The Term Loans may from time to time be Eurodollar Loans or Base Rate Loans, as determined by the Primary Borrower and notified to the Administrative Agent in accordance with Sections 2.2 and 2.13.

2.2 PROCEDURE FOR TERM LOAN BORROWING. The Primary Borrower shall give the Administrative Agent irrevocable notice (which notice must be received by the Administrative Agent prior to 10:00 A.M., New York City time, one Business Day (in the case of Base Rate Loans) or three Business Days (in the case of Eurodollar Loans) prior to the anticipated Closing Date) requesting that the Term Loan Lenders make the Term Loans on the Closing Date and specifying the amount to be borrowed. The Tranche B Term Loans made on the Closing Date shall initially be Base Rate Loans. Upon receipt of such notice the Administrative Agent shall promptly notify each Term Loan Lender thereof. Not later than 12:00 Noon, New York City time, on the Closing Date each Term Loan Lender shall make available to the Administrative Agent at the Funding Office an amount in immediately available funds equal to the Term Loan or Term Loans to be made by such Lender. The Administrative Agent shall make available to the Primary Borrower the aggregate of the amounts made available to the Administrative Agent.

2.3 REPAYMENT OF TERM LOANS. (a) The Tranche A Term Loan of each Tranche A Lender shall mature in 17 consecutive quarterly installments, commencing on March 31, 2002, each of which shall be in an amount equal to such Lender's Tranche A Term Loan Percentage multiplied by the percentage set forth below opposite such installment of the aggregate principal amount of Tranche A Term Loans made on the Closing Date:

INSTALLMENT	PERCENTAGE
March 31, 2002 June 30, 2002 September 30, 2002 December 31, 2002 March 31, 2003 September 30, 2003 December 31, 2003 March 31, 2004 June 31, 2004 September 30, 2004 December 31, 2005 June 31, 2005 September 30, 2005 December 31, 2005 March 31, 2005	$\begin{array}{c} 15.00\% \\ 4.375\% \\ 4.375\% \\ 4.375\% \\ 4.375\% \\ 5.00\% \\ 5.00\% \\ 5.00\% \\ 5.00\% \\ 5.00\% \\ 5.00\% \\ 5.625\% \\ 5.625\% \\ 5.625\% \\ 5.625\% \\ 5.625\% \\ 6.25\% \\ 6.25\% \\ 6.25\% \\ 6.25\% \\ 6.25\% \end{array}$
	0120/0

PROVIDED, that if, having used its best efforts to obtain regulatory approvals required to permit quarterly payment of dividends by its Insurance Subsidiaries (the Primary Borrower hereby covenanting to use its best efforts to obtain such regulatory approvals), the Primary Borrower does not obtain such approvals for payments prior to June 30, 2002, principal payments in respect of the Tranche A Term Loans scheduled to be made on June 30, 2002, September 30, 2002, December 31, 2002 and March 31, 2003 shall be made on March 31, 2003.

(b) The Tranche B Term Loan of each Tranche B Lender shall mature in 21 consecutive quarterly installments, commencing on March 31, 2002, each of which shall be in an amount equal to such Lender's Tranche B Term Loan Percentage multiplied by the percentage set forth below opposite such installment of the aggregate principal amount of Tranche B Term Loans made on the Closing Date:

INSTALLMENT	PERCENTAG	Ξ-
March 31, 2002 June 30, 2002 September 30, 2002 December 31, 2002 March 31, 2003 June 30, 2003 September 30, 2003 December 31, 2004 June 30, 2004 September 30, 2004 December 31, 2004 March 31, 2005 September 30, 2005 December 31, 2005 March 31, 2006 June 31, 2006 September 30, 2006 December 31, 2006 March 31, 2006	$\begin{array}{c} 1.00\% \\ 0.25\% \\$	

PROVIDED, that if, having used its best efforts to obtain regulatory approvals required to permit quarterly payment of dividends by its Insurance Subsidiaries (the Primary Borrower hereby covenanting to use its best efforts to obtain such regulatory approvals), the Primary Borrower does not obtain such approvals for payments prior to June 30, 2002, principal payments in respect of the Tranche B Term Loans scheduled to be made on June 30, 2002, September 30, 2002, December 31, 2002 and March 31, 2003 shall be made on March 31, 2003.

2.4 REVOLVING CREDIT COMMITMENTS. (a) Subject to the terms and conditions hereof, the Revolving Credit Lenders severally agree to make revolving credit loans ("REVOLVING CREDIT LOANS") to the Primary Borrower (and, from and after the Revolving Commitment Increase Date and subject to the conditions set forth in Section 2.25, the Additional Borrower) from time to time during the Revolving Credit Commitment Period in an aggregate principal amount at any one time outstanding for each Revolving Credit Lender which, when added to such Lender's Revolving Credit Percentage of the sum of (i) the L/C Obligations then outstanding and (ii) the aggregate principal amount of the Swing Line Loans then outstanding, does not exceed the amount of such Lender's Revolving Credit Commitments by borrowing Credit Commitment Period the Primary Borrower (and, from and after the Revolving Commitment Increase Date and subject to the conditions set forth in Section 2.25, the Additional Borrower) may use the Revolving Credit Commitments by borrowing, prepaying the Revolving Credit Loans in whole or in part, and reborrowing, all in accordance with the terms and conditions hereof. The Revolving Credit Loans may from time to time be Eurodollar Loans or Base Rate Loans, as determined by the relevant Borrower and notified to the Administrative Agent in accordance with Sections 2.5 and 2.13, provided that no Revolving Credit Loan shall be made as a Eurodollar Loan after the day that is one month prior to the Revolving Credit Termination Date.

(b) Each Borrower shall repay all outstanding Revolving Credit Loans made to such Borrower on the Revolving Credit Termination Date.

2.5 PROCEDURE FOR REVOLVING CREDIT BORROWING. The Primary Borrower (and, from and after the Revolving Commitment Increase Date and subject to the conditions set forth in Section 2.25, the Additional Borrower) may borrow under the Revolving Credit Commitments on any Business Day during the Revolving Credit Commitment Period, PROVIDED that the relevant Borrower shall give the Administrative Agent irrevocable notice (which notice must be received by the Administrative Agent prior to 12:00 Noon, New York City time, (a) three Business Days prior to the requested Borrowing Date, in the case of Eurodollar Loans, or (b) one Business Day prior to the requested Borrowing Date, in the case of Base Rate Loans), specifying

(i) the amount and Type of Revolving Credit Loans to be borrowed, (ii) the requested Borrowing Date and (iii) in the case of Eurodollar Loans, the length of the initial Interest Period therefor. Each borrowing of Revolving Credit Loans under the Revolving Credit Commitments shall be in an amount equal to (x) in the case of Base Rate Loans, \$1,000,000 or a whole multiple thereof (or, if the then aggregate Available Revolving Credit Commitments are less than \$1,000,000, such lesser amount) and (y) in the case of Eurodollar Loans, \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof; PROVIDED, that the Swing Line Lender may request, on behalf of the Primary Borrower, borrowings of Base Rate Loans under the Revolving Credit Commitments in other amounts pursuant to Section 2.7. Upon receipt of any such notice from any Borrower, the Administrative Agent shall promptly notify each Revolving Credit Lender thereof. Each Revolving Credit Lender will make its Revolving Credit Percentage of the amount of each borrowing of Revolving Credit Loans available to the Administrative Agent for the account of such Borrower at the Funding Office prior to 12:00 Noon, New York City time, on the Borrowing Date requested by such Borrower in funds immediately available to the Administrative Agent. Such borrowing will then be made available to such Borrower by the Administrative Agent in like funds as received by the Administrative Agent.

2.6 SWING LINE COMMITMENT. (a) Subject to the terms and conditions hereof, the Swing Line Lender agrees that, during the Revolving Credit Commitment Period, it will make available to the Primary Borrower in the form of swing line loans ("SWING LINE LOANS") a portion of the credit otherwise available to the Primary Borrower under the Revolving Credit Commitments; PROVIDED that (i) the aggregate principal amount of Swing Line Loans outstanding at any time shall not exceed the Swing Line Commitment then in effect (notwithstanding that the Swing Line Loans outstanding Revolving Credit Loans hereunder, may exceed the Swing Line Commitment then in effect or such Swing Line Lender's Revolving Credit Commitment then in effect) and (ii) the Primary Borrower shall not request, and the Swing Line Lender shall not make, any Swing Line Loan if, after giving effect to the making of such Swing Line Loan, the aggregate amount of the Available Revolving Credit Commitment Swing Line Loan, the aggregate amount of the Revolving Credit Commitment Period, the Primary Borrower may use the Swing Line Commitment by borrowing, repaying and reborrowing, all in accordance with the terms and conditions hereof. Swing Line Loans shall be Base Rate Loans only.

(b) The Primary Borrower shall repay all outstanding Swing Line Loans on the Revolving Credit Termination Date. Each payment in respect of Swing Line Loans shall be made to the Swing Line Lender.

2.7 PROCEDURE FOR SWING LINE BORROWING; REFUNDING OF SWING LINE LOANS. (a) The Primary Borrower may borrow under the Swing Line Commitment on any Business Day during the Revolving Credit Commitment Period, PROVIDED, the Primary Borrower shall give the Swing Line Lender irrevocable telephonic notice confirmed promptly in writing (which telephonic notice must be received by the Swing Line Lender not later than 1:00 P.M., New York City time, on the proposed Borrowing Date), specifying (i) the amount to be borrowed and (ii) the requested Borrowing Date. Each borrowing under the Swing Line Commitment shall be in an amount equal to \$500,000 or a whole multiple of \$100,000 in excess thereof. Not later than 3:00 P.M., New York City time, on the Borrowing Date specified in the borrowing notice in respect of any Swing Line Loan, the Swing Line Lender shall make available to the Administrative Agent at the Funding Office an amount in immediately available funds equal to the amount of such Swing Line Loan. The Administrative Agent shall make the proceeds of such Swing Line Loan available to the Primary Borrower on such Borrowing Date in like funds as received by the Administrative Agent.

(b) The Swing Line Lender, at any time and from time to time in its sole and absolute discretion may, on behalf of the Primary Borrower (which hereby irrevocably directs the Swing Line Lender to act on its behalf), on one Business Day's notice given by the Swing Line Lender no later than 12:00 Noon, New York City time, request each Revolving Credit Lender to make, and each Revolving Credit Lender hereby agrees to make, a Revolving Credit Loan, in an amount equal to such Revolving Credit Lender's Revolving Credit Percentage of the aggregate amount of the Swing Line Loans (the "REFUNDED SWING LINE LOANS") outstanding on the date of such notice, to repay the Swing Line Lender. Each Revolving Credit Lender shall make the amount of such Revolving Credit Loan available to the Administrative Agent at the Funding Office in immediately available funds, not later than 10:00 A.M., New York City time, one Business Day after the date of such notice. The proceeds of such Revolving Credit Loans shall be made immediately available by the Administrative Agent to the Swing Line Lender for application by the Swing Line Lender to the repayment of the Refunded Swing Line Loans. Upon the written request of any Revolving Credit Lender, the Administrative Agent will, within three Business Days of such request, inform such Revolving Credit Lender of the aggregate amount of Swing Line Loans outstanding on the date of such request.

(c) If prior to the time a Revolving Credit Loan would have otherwise been made pursuant to Section 2.7(b), one of the events described in Section 8(f) shall have occurred and be continuing with respect to the Primary Borrower, or if for any other reason, as determined by the Swing Line Lender in its sole discretion, Revolving Credit Loans may not be made as contemplated by Section 2.7(b), each Revolving Credit Lender shall, on the date such Revolving Credit Loan was to have been made pursuant to the notice referred to in Section 2.7(b) (the "REFUNDING DATE"), purchase for cash an undivided participating interest in the then outstanding Swing Line Loans by paying to the Swing Line Lender an amount (the "SWING LINE PARTICIPATION AMOUNT") equal to (i) such Revolving Credit Lender's Revolving Credit Percentage TIMES (ii) the sum of the aggregate principal amount of Swing Line Loans then outstanding which were to have been repaid with such Revolving Credit Loans.

(d) Whenever, at any time after the Swing Line Lender has received from any Revolving Credit Lender such Lender's Swing Line Participation Amount, the Swing Line Lender receives any payment on account of the Swing Line Loans, the Swing Line Lender will distribute to such Lender its Swing Line Participation Amount (appropriately adjusted, in the case of interest payments, to reflect the period of time during which such Lender's participating interest was outstanding and funded and, in the case of principal and interest payments, to reflect such Lender's PRO RATA portion of such payment if such payment is not sufficient to pay the principal of and interest on all Swing Line Loans then due); PROVIDED, HOWEVER, that in the event that such payment received by the Swing Line Lender is required to be returned, such Revolving Credit Lender will return to the Swing Line Lender any portion thereof previously distributed to it by the Swing Line Lender. (e) Each Revolving Credit Lender's obligation to make the Loans referred to in Section 2.7(b) and to purchase participating interests pursuant to Section 2.7(c) shall be absolute and unconditional and shall not be affected by any circumstance, including, without limitation, (i) any setoff, counterclaim, recoupment, defense or other right which such Revolving Credit Lender or the Primary Borrower may have against the Swing Line Lender, the Primary Borrower or any other Person for any reason whatsoever; (ii) the occurrence or continuance of a Default or an Event of Default or the failure to satisfy any of the other conditions specified in Section 5; (iii) any adverse change in the condition (financial or otherwise) of the Primary Borrower; (iv) any breach of this Agreement or any other Loan Document by the Primary Borrower, any other Loan Party or any other Revolving Credit Lender; or (v) any other circumstance, happening or event whatsoever, whether or not similar to any of the foregoing.

2.8 REPAYMENT OF LOANS; EVIDENCE OF DEBT. (a) Each Borrower hereby unconditionally promises to pay to the Administrative Agent for the account of the appropriate Revolving Credit Lender the then unpaid principal amount of each Revolving Credit Loan of such Revolving Credit Lender made to such Borrower on the Revolving Credit Termination Date (or on such earlier date on which the Loans become due and payable pursuant to Section 8). The Primary Borrower hereby unconditionally promises to pay to the Administrative Agent for the account of the appropriate Revolving Credit Lender or Term Loan Lender, as the case may be, (i) the then unpaid principal amount of each Swing Line Loan of such Swing Line Lender on the Revolving Credit Termination Date (or on such earlier date on which the Loans become due and payable pursuant to Section 8) and (iii) the principal amount of each Term Loan of such Term Loan Lender in installments according to the amortization schedule set forth in Section 2.3 (or on such earlier date on which the Loans become due and payable pursuant to Section 8). Each Borrower hereby further agrees to pay interest to the Administrative Agent for the account of the appropriate Lender on the unpaid principal amount of the Loans made to such Borrower from time to time outstanding from the date hereof until payment in full thereof at the rates per annum, and on the dates, set forth in Section 2.15.

(b) Each Lender shall maintain in accordance with its usual practice an account or accounts evidencing indebtedness of each Borrower to such Lender resulting from each Loan of such Lender from time to time, including the amounts of principal and interest payable and paid to such Lender from time to time under this Agreement.

(c) The Administrative Agent, on behalf of the Borrowers, shall maintain the Register pursuant to Section 10.6(d), and a subaccount therein for each Lender, in which shall be recorded (i) the amount of each Loan made hereunder and any Note evidencing such Loan, the Type of such Loan and each Interest Period applicable thereto, (ii) the amount of any principal or interest due and payable or to become due and payable from each Borrower to each Lender hereunder and (iii) both the amount of any sum received by the Administrative Agent hereunder from any Borrower and each Lender's share thereof.

(d) The entries made in the Register and the accounts of each Lender maintained pursuant to Section 2.8(b) shall, to the extent permitted by applicable law, be PRIMA FACIE evidence of the existence and amounts of the obligations of each Borrower therein recorded; PROVIDED, HOWEVER, that the failure of any Lender or the Administrative Agent to maintain the Register or any such account, or any error therein, shall not in any manner affect the obligation

of each Borrower to repay (with applicable interest) the Loans made to such Borrower by such Lender in accordance with the terms of this Agreement.

(e) Each Borrower agrees that, upon the request to the Administrative Agent by any Lender, such Borrower will execute and deliver to such Lender a promissory note of such Borrower evidencing any Term Loans, Revolving Credit Loans or Swing Line Loans, as the case may be, made by such Lender to such Borrower, substantially in the forms of Exhibit G-1, G-2 or G-3, respectively, with appropriate insertions as to date and principal amount.

2.9 COMMITMENT FEES, ETC. (a) The Primary Borrower agrees to pay to the Administrative Agent for the account of each Revolving Credit Lender a commitment fee for the period from and including the Closing Date to the last day of the Revolving Credit Commitment Period, computed at the Commitment Fee Rate on the average daily amount of the Available Revolving Credit Commitment of such Lender during the period for which payment is made, payable quarterly in arrears on the last day of each March, June, September and December and on the Revolving Credit Termination Date, commencing on the first of such dates to occur after the date hereof.

(b) The Primary Borrower agrees to pay to the Administrative Agent on the Closing Date for the account of each Revolving Credit Lender and Tranche A Term Loan Lender, a commitment fee for the period from the date of this Agreement to the Closing Date, calculated on the aggregate amount of such Lender's Revolving Credit Commitment and Tranche A Term Loan Commitment at a rate per annum equal to (i) for the period from the date of this Agreement to April 13, 2001, .375% and (ii) from and after April 13, 2001, the Applicable Margin for Eurodollar Loans under the Revolving Credit Facility and Tranche A Term Loan Facility.

(c) The Primary Borrower agrees to pay to the Administrative Agent on the Closing Date for the account of each Tranche B Term Loan Lender, a commitment fee for the period from the date of this Agreement to the Closing Date, calculated on the aggregate amount of such Lender's Tranche B Term Loan Commitment at a rate per annum equal to (i) for the period from the date of this Agreement to April 13, 2001, .375% and (ii) from and after April 13, 2001, the Applicable Margin for Eurodollar Loans under the Tranche B Term Loan Facility.

(d) The Primary Borrower agrees to pay to the Administrative Agent the fees in the amounts and on the dates from time to time agreed to in writing by the Primary Borrower and the Administrative Agent.

(e) The Primary Borrower agrees to pay to the Arranger the fees in the amounts and on the dates from time to time agreed to in writing by the Primary Borrower and the Arranger.

2.10 TERMINATION OR REDUCTION OF REVOLVING CREDIT COMMITMENTS. The Primary Borrower shall have the right, upon not less than three Business Days' notice to the Administrative Agent, to terminate the Revolving Credit Commitments or, from time to time, to reduce the aggregate amount of the Revolving Credit Commitments; PROVIDED that no such termination or reduction of Revolving Credit Commitments shall be permitted if, after giving effect thereto and to any prepayments of the Revolving Credit Loans and Swing Line Loans made on the effective date thereof, the Total Revolving Extensions of Credit would exceed the Total Revolving Credit Commitments. Any such reduction shall be in an amount equal to \$1,000,000, or a whole multiple thereof, and shall reduce permanently the Revolving Credit Commitments then in effect.

2.11 OPTIONAL PREPAYMENTS. (a) Each Borrower may at any time and from time to time prepay the Loans made to such Borrower, in whole or in part, without premium or penalty, upon irrevocable notice delivered to the Administrative Agent at least three Business Days prior thereto in the case of Eurodollar Loans and at least one Business Day prior thereto in the case of Base Rate Loans, which notice shall specify the date and amount of prepayment and whether the prepayment is of Eurodollar Loans or Base Rate Loans; provided, that (i) if a Eurodollar Loan is prepaid on any day other than the last day of the Interest Period applicable thereto, the relevant Borrower shall also pay any amounts owing pursuant to Section 2.21 and (ii) no prior notice is required for the prepayment of Swing Line Loans. Upon receipt of any such notice the Administrative Agent shall promptly notify each relevant Lender thereof. If any such notice is given, the amount specified in such notice shall be due and payable on the date specified therein, together with (except in the case of Revolving Credit Loans that are Base Rate Loans and Swing Line Loans) accrued interest to such date on the amount prepaid. Partial prepayments of Term Loans and Revolving Credit Loans shall be in an aggregate principal amount of \$1,000,000 or a whole multiple thereof. Partial prepayments of Swing Line Loans shall be in an aggregate principal amount of \$100,000 or a whole multiple thereof

(b) Each optional prepayment in respect of the Tranche B Term Loans on or prior to the date that is eighteen months after the Closing Date shall be accompanied by a prepayment premium equal to (i) if such prepayment is made prior to the first anniversary of the Closing Date, 1.5% of the principal amount of such prepayment and (ii) if such prepayment is made on or after the first anniversary of the Closing Date and prior to the date that is eighteen months after the Closing Date, 0.5% of the principal amount of such prepayment. Any prepayment of the Loans upon the refinancing thereof (whether with proceeds of equity or Indebtedness) or upon the occurrence of a Change of Control shall be deemed to be an optional prepayment.

2.12 MANDATORY PREPAYMENTS AND COMMITMENT REDUCTIONS. (a) Unless the Required Prepayment Lenders and the Primary Borrower shall otherwise agree, if any Capital Stock shall be issued or sold by the Parent (except (i) any Capital Stock of the Parent the proceeds of which are used solely to purchase or prepay the Subordinated Seller Note and/or the Third Party Preferred Stock and (ii) any Capital Stock of the Parent issued to White Mountains as a part of the transaction in which the Berkshire Preferred Stock is redeemed with proceeds received from the exercise of warrants under the Warrant Agreement), on the date of such issuance or sale, an amount equal to 50% of the Net Proceeds thereof in excess of \$5,000,000 shall be contributed to the common equity of the Primary Borrower and applied by the Primary Borrower toward the prepayment of the Term Loans and the reduction of the Revolving Credit Commitments as set forth in Section 2.12(d); PROVIDED, that \$20,000,000 of the proceeds of preferred stock issued by the Parent and sold to a third party in connection with the transactions contemplated by this Agreement shall be exempted from the requirements of this Section 2.12(a).

(b) Unless the Required Prepayment Lenders and the Primary Borrower shall otherwise agree, if on any date the Primary Borrower or any of its Non-Insurance Subsidiaries shall receive Net Proceeds from any Asset Sale or Recovery Event, then, on such date, the Term Loans shall be prepaid, and/or the Revolving Credit Commitments shall be reduced, in each case in accordance with Section 2.12(d), by the Required Portion of such Net Proceeds. For purposes of this Section 2.12(b), the "REQUIRED PORTION" of the Net Cash Proceeds of any Asset Sale or Recovery Event shall be (i) if no Reinvestment Notice is delivered in respect of such Net Cash Proceeds on or prior to the date of receipt of such Net Cash Proceeds, 100% of such Net Cash Proceeds in excess of \$10,000,000, and (ii) if the assets subject to such Disposition are insurance-related businesses or assets and a Reinvestment Notice is delivered in respect of such Net Cash Proceeds on or prior to the date of receipt of such Net Cash Proceeds, (A) with respect to Net Cash Proceeds of any Asset Sale or Recovery Event consummated in any fiscal year which, together with all other Asset Sales and Recovery Events Surplus in an amount not exceeding 10% of combined Capital and Surplus of all Insurance Subsidiaries as of the last day of the immediately preceding fiscal year, 0% of such Net Cash Proceeds and (B) with respect to the Net Cash Proceeds of any Asset Sale or Recovery Event consummated in any fiscal year to the extent such Net Cash Proceeds are not covered by the foregoing sub-clause (A), 100% of such Net Cash Proceeds and (iii) if the assets subject to such Disposition are not insurance-related businesses or assets and a Reinvestment Notice is delivered in respect of such Net Cash Proceeds on or prior to the date of receipt or such Net Cash Proceeds, 0% of such Net Cash Proceeds. For purposes of determining the amount of Net Cash Proceeds of any Disposition to be allocated to sub-clause (A) and sub-clause (B), respectively, of clause (ii) of the preceding sentence in the case of any Disposition to which both such sub-clauses apply, the amount of Net Cash Proceeds allocated to such sub-clause (B) shall be the aggregate amount of such Net Cash Proceeds multiplied by a fraction (not to exceed 1.0), the numerator of which is the amount of Capital and Surplus Disposed of in such Disposition and all prior Dispositions during such fiscal year in excess of 10% of combined Capital and Surplus of all Insurance Subsidiaries as of the last day of the immediately preceding fiscal year, and the denominator of which is the aggregate amount of Capital and Surplus Disposed of in such Disposition. In addition, in the event any Reinvestment Notice is delivered in respect of any Reinvestment Event, then, on the Reinvestment Prepayment Date applicable to such Reinvestment Event, the Term Loans shall be prepaid, and/or the Revolving Credit Commitments shall be reduced, by an amount equal to the Reinvestment Prepayment Amount with respect to the relevant Reinvestment Event, as set forth in Section 2.12(d).

(c) Unless the Required Prepayment Lenders and the Primary Borrower shall otherwise agree, if, on the Closing Date, the aggregate unassigned surplus of dividend-paying first-tier Insurance Subsidiaries is less than \$950,000,000 (it being understood that such aggregate unassigned surplus shall not be less than \$900,000,000) and the Primary Borrower has received (in accordance with Section 5.1(b)(iv)(A)), as the proceeds of equity issued to the Parent, the amount by which such aggregate unassigned surplus is less than \$950,000,000, the Commitments shall be reduced by such amount immediately prior to the making of Loans on the Closing Date as set forth in Section 2.12(d).

(d) Amounts to be applied in connection with prepayments and Commitment reductions made pursuant to paragraphs (a) and (b) of this Section shall be applied, FIRST, to the

prepayment of the Term Loans and, SECOND, to reduce permanently the Revolving Credit Commitments; PROVIDED, that the Revolving Credit Commitments shall not be reduced to less that \$100,000,000 pursuant to this Section 2.12. Amounts to be applied in connection with Commitment reductions made pursuant to paragraph (c) of this Section shall be applied to reduce the Commitments pro rata according to the respective principal amounts of the Tranche A Term Loan Commitment, the Tranche B Term Loan Commitment and the Revolving Credit Commitment. Any reduction of the Revolving Credit Commitments pursuant to this Section shall be accompanied by prepayment of the Revolving Credit Loans and/or Swing Line Loans to the extent, if any, that the Total Revolving Extensions of Credit exceed the amount of the Total Revolving Credit Commitments as so reduced, PROVIDED that if the aggregate principal amount of Revolving Credit Loans and Swing Line Loans then outstanding is less than the amount of such excess (because L/C Obligations constitute a portion thereof), the Primary Borrower shall, to the extent of the balance of such excess, replace outstanding Letters of Credit and/or deposit an amount in cash in a cash collateral account established with the Administrative Agent for the benefit of the Lenders on terms and conditions satisfactory to the Administrative Agent.

2.13 CONVERSION AND CONTINUATION OPTIONS. (a) Each Borrower may elect from time to time to convert Eurodollar Loans made to such Borrower to Base Rate Loans by giving the Administrative Agent at least two Business Days' prior irrevocable notice of such election, provided that any such conversion of Eurodollar Loans may be made only on the last day of an Interest Period with respect thereto. Each Borrower may elect from time to time to convert Base Rate Loans made to such Borrower to Eurodollar Loans by giving the Administrative Agent at least three Business Days' prior irrevocable notice of such election (which notice shall specify the length of the initial Interest Period therefor), PROVIDED that no Base Rate Loan under a particular Facility may be converted into a Eurodollar Loan (i) when any Event of Default has occurred and is continuing and the Administrative Agent has, or the Majority Facility Lenders in respect of such Facility have, determined in its or their sole discretion not to permit such conversions or (ii) after the date that is one month prior to the final scheduled termination or maturity date of such Facility. Upon receipt of any such notice the Administrative Agent shall promptly notify each relevant Lender thereof.

(b) Each Borrower may elect to continue any Eurodollar Loan made to such Borrower as such upon the expiration of the then current Interest Period with respect thereto by giving irrevocable notice to the Administrative Agent, in accordance with the applicable provisions of the term "Interest Period" set forth in Section 1.1, of the length of the next Interest Period to be applicable to such Loans, PROVIDED that no Eurodollar Loan under a particular Facility may be continued as such (i) when any Event of Default has occurred and is continuing and the Administrative Agent has, or the Majority Facility Lenders in respect of such Facility have, determined in its or their sole discretion not to permit such continuations or (ii) after the date that is one month prior to the final scheduled termination or maturity date of such Facility, and PROVIDED, FURTHER, that if the relevant Borrower shall fail to give any required notice as described above in this paragraph or if such continuation is not permitted pursuant to the preceding proviso, such Loans shall be converted automatically to Base Rate Loans on the last day of such then expiring Interest Period. Upon receipt of any such notice the Administrative Agent shall promptly notify each relevant Lender thereof.

2.14 MINIMUM AMOUNTS AND MAXIMUM NUMBER OF EURODOLLAR TRANCHES. Notwithstanding anything to the contrary in this Agreement, all borrowings, conversions, continuations and optional prepayments of Eurodollar Loans and all selections of Interest Periods shall be in such amounts and be made pursuant to such elections so that, (a) after giving effect thereto, the aggregate principal amount of the Eurodollar Loans comprising each Eurodollar Tranche shall be equal to \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof and (b) no more than ten Eurodollar Tranches shall be outstanding at any one time.

2.15 INTEREST RATES AND PAYMENT DATES. (a) Each Eurodollar Loan shall bear interest for each day during each Interest Period with respect thereto at a rate per annum equal to the Eurodollar Rate determined for such day plus the Applicable Margin.

(b) Each Base Rate Loan shall bear interest at a rate per annum equal to the Base Rate plus the Applicable Margin.

(c) (i) If all or a portion of the principal amount of any Loan or Reimbursement Obligation shall not be paid when due (whether at the stated maturity, by acceleration or otherwise), such overdue amount shall bear interest at a rate per annum that is equal to (x) in the case of the Loans, the rate that would otherwise be applicable thereto pursuant to the foregoing provisions of this Section PLUS 2% or (y) in the case of Reimbursement Obligations, the rate applicable to Base Rate Loans under the Revolving Credit Facility PLUS 2%, and (ii) if all or a portion of any interest payable on any Loan or Reimbursement Obligation or any commitment fee or other amount payable hereunder shall not be paid when due (whether at the stated maturity, by acceleration or otherwise), such overdue amount shall bear interest at a rate per annum equal to the rate then applicable to Base Rate Loans under the relevant Facility PLUS 2% (or, in the case of any such other amounts that do not relate to a particular Facility, the rate then applicable to Base Rate Loans under the Revolving Credit Facility PLUS 2%), in each case, with respect to clauses (i) and (ii) above, from the date of such non-payment until such amount is paid in full (after as well as before judgment).

(d) Interest shall be payable in arrears on each Interest Payment Date, PROVIDED that interest accruing pursuant to paragraph (c) of this Section shall be payable from time to time on demand.

2.16 COMPUTATION OF INTEREST AND FEES. (a) Interest, fees, commissions payable pursuant hereto shall be calculated on the basis of a 360-day year for the actual days elapsed, except that, with respect to Base Rate Loans on which interest is calculated on the basis of the Prime Rate, the interest thereon shall be calculated on the basis of a 365-day (or 366-day, as the case may be) year for the actual days elapsed. The Administrative Agent shall as soon as practicable notify the relevant Borrower and the relevant Lenders of each determination of a Eurodollar Rate. Any change in the interest rate on a Loan resulting from a change in the Base Rate or the Eurocurrency Reserve Requirements shall become effective as of the opening of business on the day on which such change becomes effective. The Administrative Agent shall as soon as practicable notify the relevant Borrower and the relevant Lenders of the effective date and the amount of each such change in interest rate. (b) Each determination of an interest rate by the Administrative Agent pursuant to any provision of this Agreement shall be conclusive and binding on the Borrowers and the Lenders in the absence of manifest error. The Administrative Agent shall, at the request of the any Borrower, deliver to such Borrower a statement showing the quotations used by the Administrative Agent in determining any interest rate pursuant to Section 2.15(a) with respect to Loans made to such Borrower.

2.17 INABILITY TO DETERMINE INTEREST RATE. If prior to the first day of any Interest Period:

(a) the Administrative Agent shall have determined (which determination shall be conclusive and binding upon the relevant Borrower) that, by reason of circumstances affecting the relevant market, adequate and reasonable means do not exist for ascertaining the Eurodollar Rate for such Interest Period, or

(b) the Administrative Agent shall have received notice from the Majority Facility Lenders in respect of the relevant Facility that the Eurodollar Rate determined or to be determined for such Interest Period will not adequately and fairly reflect the cost to such Lenders (as conclusively certified by such Lenders) of making or maintaining their affected Loans during such Interest Period,

the Administrative Agent shall give telecopy or telephonic notice thereof to the relevant Borrower and the relevant Lenders as soon as practicable thereafter. If such notice is given (x) any Eurodollar Loans under the relevant Facility requested to be made on the first day of such Interest Period shall be made as Base Rate Loans, (y) any Loans under the relevant Facility that were to have been converted on the first day of such Interest Period to Eurodollar Loans shall be continued as Base Rate Loans and (z) any outstanding Eurodollar Loans under the relevant Facility shall be converted, on the last day of the then current Interest Period with respect thereto, to Base Rate Loans. Until such notice has been withdrawn by the Administrative Agent, no further Eurodollar Loans under the relevant Facility shall be made or continued as such, nor shall the relevant Borrower have the right to convert Loans under the relevant Facility to Eurodollar Loans.

2.18 PRO RATA TREATMENT AND PAYMENTS. (a) Each borrowing by any Borrower from the Lenders hereunder, each payment by any Borrower on account of any commitment fee or Letter of Credit fee, and any reduction of the Commitments of the Lenders, shall be made pro rata according to the respective Tranche A Term Loan Percentages, Tranche B Term Loan Percentages or Revolving Credit Percentages, as the case may be, of the relevant Lenders. Each payment (other than prepayments) in respect of principal or interest in respect of the Term Loans and each payment in respect of fees payable hereunder shall be applied to the amounts of such obligations owing to the Lenders.

(b) Subject to Section 2.18(d), each mandatory prepayment required by Section 2.12 to be applied to Term Loans shall be allocated between the Term Loan Facilities PRO RATA according to the respective outstanding principal amounts of Term Loans under such Facilities. Subject to Section 2.18(d), each optional prepayment in respect of the Term Loans shall be allocated between the Term Loan Facilities PRO RATA according to the respective outstanding principal amounts of Term Loans under such Facilities. Each payment (including each prepayment) of the Term Loans outstanding under any Term Loan Facility shall be allocated among the Term Loan Lenders holding such Term Loans PRO RATA based on the principal amount of such Term Loans held by such Term Loan Lenders, and shall be applied (i) in the case of mandatory prepayments pursuant to Section 2.12(b), to the installments thereof PRO RATA based on the remaining outstanding principal amount of such installments, and (ii) in the case of all other prepayments, FIRST, to the next scheduled installment thereof to be due after the date of such prepayment, and, SECOND, to the remaining installments PRO RATA based on the remaining outstanding principal amount of such installments. Amount of such installments. Amount of such remaining new prepayments.

(c) Each payment (including each prepayment) by any Borrower on account of principal of and interest on the Revolving Credit Loans of such Borrower shall be made PRO RATA according to the respective outstanding principal amounts of the Revolving Credit Loans of such Borrower then held by the Revolving Credit Lenders. Each payment in respect of Reimbursement Obligations in respect of any Letter of Credit shall be made to the relevant Issuing Lender.

(d) Notwithstanding anything to the contrary in Sections 2.11, 2.12 or 2.18(b), so long as any Tranche A Term Loans are outstanding, each Tranche B Term Loan Lender may, at its option, decline up to 100% of the portion of any optional prepayment or mandatory payment applicable to the Tranche B Term Loans of such Lender; accordingly, with respect to the amount of any optional prepayment described in Section 2.11 or mandatory prepayment described in Section 2.12 that is allocated to Tranche B Term Loans (such amounts, respectively, the "OPTIONAL PREPAYMENT AMOUNT" and the "MANDATORY PREPAYMENT AMOUNT"), at any time when Tranche A Term Loans remain outstanding, the Primary Borrower will:

(i) in the case of any optional prepayment of the Tranche B Term Loans which the Primary Borrower wishes to make, give the Administrative Agent telephonic notice (promptly confirmed in writing) requesting that the Administrative Agent prepare and provide to each Tranche B Term Loan Lender a notice (each, a "PREPAYMENT OPTION NOTICE") as described below; and

(ii) in the case of any mandatory prepayment required to be made pursuant to Section 2.12, on the date specified in Section 2.12 for such prepayment, (A) give the Administrative Agent telephonic notice (promptly confirmed in writing) requesting that the Administrative Agent prepare and provide to each Tranche B Term Loan Lender a Prepayment Option Notice as described below and (B) deposit with the Administrative Agent the Mandatory Prepayment Amount.

As promptly as practicable after receiving such notice from the Primary Borrower, the Administrative Agent will send to each Tranche B Term Loan Lender a Prepayment Option Notice, which shall be substantially in the form of Exhibit H, and shall include an offer by the Primary Borrower to prepay on the Prepayment Date the Tranche B Term Loans of such Lender by an amount equal to the portion of the Optional Prepayment Amount or Mandatory Prepayment Amount, as the case may be, indicated in such Lender's Prepayment Option Notice as being applicable to such Lender's Tranche B Term Loans. The "PREPAYMENT DATE" in respect of any Prepayment Option Notice shall be the date which is (i) in the case of a Prepayment Option Notice relating to an optional prepayment, the later of (A) three Business Days after the date of such Prepayment Option Notice and (B) the date on which the Primary Borrower has advised the Administrative Agent that it wishes to make such optional prepayment and (ii) in the case of a Prepayment Option Notice relating to a mandatory prepayment, the date which is three Business Days after the date of such Prepayment Option Notice.

On the Prepayment Date:

(i) in the case of any optional prepayment, the Primary Borrower shall pay to the Administrative Agent the Optional Prepayment Amount, and the Administrative Agent shall (A) apply the Optional Prepayment toward prepayment of the outstanding Tranche B Term Loans in respect of which Lenders have accepted optional prepayment as described above and (B) apply the remaining portion of the Optional Prepayment Amount not accepted by the Tranche B Term Loan Lenders toward prepayment of the Tranche A Term Loans; and

(ii) in the case of any mandatory prepayment, the Administrative Agent shall (A) apply the Mandatory Prepayment Amount toward prepayment of the outstanding Tranche B Term Loans in respect of which Lenders have accepted mandatory prepayment as described above and (B) apply the remaining portion of the Mandatory Prepayment Amount not accepted by the Tranche B Term Loan Lenders toward prepayment of the Tranche A Term Loans.

The procedures described above in this paragraph (d) shall not be applicable in the case of a prepayment in full of all Term Loans.

(e) The application of any payment of Loans under any Facility (including optional and mandatory prepayments) shall be made, FIRST, to Base Rate Loans under such Facility and, SECOND, to Eurodollar Loans under such Facility. Each payment of the Loans (except in the case of Swing Line Loans and Revolving Credit Loans that are Base Rate Loans) shall be accompanied by accrued interest to the date of such payment on the amount paid.

(f) All payments (including prepayments) to be made by any Borrower hereunder, whether on account of principal, interest, fees or otherwise, shall be made without setoff or counterclaim and shall be made prior to 12:00 Noon, New York City time, on the due date thereof to the Administrative Agent, for the account of the relevant Lenders, at the Payment Office, in Dollars and in immediately available funds. Any payment made by any Borrower after 12:00 Noon, New York City time, on any Business Day shall be deemed to have been made on the next following Business Day. The Administrative Agent shall distribute such payments to the Lenders promptly upon receipt in like funds as received. If any payable on a day other than a Business Day, such payment shall be extended to the next succeeding Business Day. If any payment on a Eurodollar Loan becomes due and payable on a day other than a Business Day unless the result of such extension would be to extend such payment into another calendar month, in which event such such agament shall be made on the immediately preceding Business Day. In the case of any extension of any payment of principal pursuant to the preceding two sentences, interest thereon shall be payable at the then applicable rate during such extension.

(g) Unless the Administrative Agent shall have been notified in writing by any Lender prior to a borrowing that such Lender will not make the amount that would constitute its share of such borrowing available to the Administrative Agent, the Administrative Agent may assume that such Lender is making such amount available to the Administrative Agent, and the Administrative Agent may, in reliance upon such assumption, make available to the relevant Borrower a corresponding amount. If such amount is not made available to the Administrative Agent by the required time on the Borrowing Date therefor, such Lender shall pay to the Administrative Agent, on demand, such amount with interest thereon at a rate equal to the daily average Federal Funds Effective Rate for the period until such Lender makes such amount immediately available to the Administrative Agent. A certificate of the Administrative Agent submitted to any Lender with respect to any amounts owing under this paragraph shall be conclusive in the absence of manifest error. If such Lender's share of such borrowing is not made available to the Administrative Agent by such Lender within three Business Days after such Borrowing Date, the Administrative Agent shall also be entitled to recover such amount with interest thereon at the rate per annum applicable to Base Rate Loans under the relevant Facility, on demand, from the relevant Borrower.

(h) Unless the Administrative Agent shall have been notified in writing by a Borrower prior to the date of any payment due to be made by such Borrower hereunder that such Borrower will not make such payment to the Administrative Agent, the Administrative Agent may assume that such Borrower is making such payment, and the Administrative Agent may, but shall not be required to, in reliance upon such assumption, make available to the Lenders their respective pro rata shares of a corresponding amount. If such payment is not made to the Administrative Agent by such Borrower within three Business Days after such due date, the Administrative Agent shall be entitled to recover, on demand, from each Lender to which any amount which was made available pursuant to the preceding sentence, such amount with interest thereon at the rate per annum equal to the daily average Federal Funds Effective Rate. Nothing herein shall be deemed to limit the rights of the Administrative Agent or any Lender against any Borrower.

2.19 REQUIREMENTS OF LAW. (a) If the adoption of or any change in any Requirement of Law or in the interpretation or application thereof or compliance by any Lender with any request or directive (whether or not having the force of law) from any central bank or other Governmental Authority made subsequent to the date hereof:

> (i) shall subject any Lender to any tax of any kind whatsoever with respect to this Agreement, any Letter of Credit, any Application or any Eurodollar Loan made by it, or change the basis of taxation of payments to such Lender in respect thereof (except for Non-Excluded Taxes covered by Section 2.20 and changes in the rate of tax on the overall net income of such Lender);

 (ii) shall impose, modify or hold applicable any reserve, special deposit, compulsory loan or similar requirement against assets held by, deposits or other liabilities in or for the account of, advances, loans or other extensions of credit by, or any other acquisition of funds by, any office of such Lender that is not otherwise included in the determination of the Eurodollar Rate hereunder; or

(iii) shall impose on such Lender any other condition;

and the result of any of the foregoing is to increase the cost to such Lender, by an amount which such Lender deems to be material, of making, converting into, continuing or maintaining Eurodollar Loans or issuing or participating in Letters of Credit, or to reduce any amount receivable hereunder in respect thereof, then, in any such case, the Primary Borrower shall promptly pay such Lender, upon its demand, any additional amounts necessary to compensate such Lender for such increased cost or reduced amount receivable. If any Lender becomes entitled to claim any additional amounts pursuant to this Section, it shall promptly notify the Primary Borrower (with a copy to the Administrative Agent) of the event by reason of which it has become so entitled.

(b) If any Lender shall have determined that the adoption of or any change in any Requirement of Law regarding capital adequacy or in the interpretation or application thereof or compliance by such Lender or any corporation controlling such Lender with any request or directive regarding capital adequacy (whether or not having the force of law) from any Governmental Authority made subsequent to the date hereof shall have the effect of reducing the rate of return on such Lender's or such corporation's capital as a consequence of its obligations hereunder or under or in respect of any Letter of Credit to a level below that which such Lender or such corporation consideration such Lender's or such corporation's policies with respect to capital adequacy) by an amount deemed by such Lender to be material, then from time to time, after submission by such Lender to the Primary Borrower (with a copy to the Administrative Agent) of a written request therefor, the Primary Borrower shall pay to such Lender such additional amount or amounts as will compensate such Lender or such corporation for such reduction.

(c) In addition to, and without duplication of, amounts which may become payable from time to time pursuant to paragraphs (a) and (b) of this Section 2.19, the Primary Borrower agrees to pay to each Lender which requests compensation under this paragraph (c) by notice to the Primary Borrower, on the last day of each Interest Period with respect to any Eurodollar Loan made by such Lender, at any time when such Lender shall be required to maintain reserves against "Eurocurrency liabilities" under Regulation D of the Board of Governors of the Federal Reserve System (or, at any time when such Lender may be required by the Board of Governors of the Federal Reserve System or by any other Governmental Authority, whether within the United States or in another relevant jurisdiction, to maintain reserves against any other category of liabilities which includes deposits by reference to which the Eurodollar Rate is determined as provided in this Agreement or against any category of extensions of credit or other assets of such Lender which includes any such Eurodollar Loans), an additional amount (determined by such Lender's calculation or, if an accurate calculation is impracticable, reasonable estimate using such reasonable means of allocation as such Lender shall determine) equal to the actual costs, if any, incurred by such Lender during such Interest Period as a result of the applicability of the foregoing reserves to such Eurodollar Loans.

(d) A certificate as to any additional amounts payable pursuant to this Section submitted by any Lender to the Primary Borrower (with a copy to the Administrative Agent) shall be conclusive in the absence of manifest error. The obligations of the Primary Borrower pursuant to this Section shall survive the termination of this Agreement and the payment of the Loans and all other amounts payable hereunder.

 $2.20\ \textsc{TAXES}.$ (a) Except as required by law, all payments made by each Borrower under this Agreement shall be made free and clear of, and without deduction or withholding for or on account of, any present or future income, stamp or other taxes, levies, imposts, duties, charges, fees, deductions or withholdings, now or hereafter imposed, levied, collected, withheld or assessed by any Governmental Authority, excluding net income taxes and franchise and doing business taxes (imposed in lieu of net income taxes) imposed on the Administrative Agent or any Lender as a result of a present or former connection between the Administrative Agent or such Lender and the jurisdiction of the Governmental Authority imposing such tax or any political subdivision or taxing authority thereof or therein (other than any such connection arising solely from the Administrative Agent's or such Lender's having executed, delivered or performed its obligations or received a payment under, or enforced, this Agreement or any other Loan Document). If any such non-excluded taxes, levies, imposts, duties, charges, fees, deductions or withholdings ("NON-EXCLUDED TAXES") or any Other Taxes are required to be withheld from any amounts payable to the Administrative Agent or any Lender hereunder, the amounts so payable to the Administrative Agent or such Lender shall be increased to the extent necessary to yield to the Administrative Agent or such Lender (after payment of all Non-Excluded Taxes and Other Taxes) interest or any such other amounts payable hereunder at the rates or in the amounts specified in this Agreement; PROVIDED, HOWEVER, that no Borrower shall be required to increase any such amounts payable to any Lender with respect to any Non-Excluded Taxes (i) that are attributable to such Lender's failure to comply with the requirements of paragraph (d) or (e) of this Section or (ii) that are withholding taxes imposed on amounts payable to such Lender at the time such Lender becomes a party to this Agreement, except to the extent that such Lender's assignor (if any) was entitled, at the time of assignment, to receive additional amounts from such Borrower with respect to such Non-Excluded Taxes pursuant to Section 2.20(a).

(b) In addition, the Primary Borrower shall pay any Other Taxes to the relevant Governmental Authority in accordance with applicable law.

(c) Whenever any Non-Excluded Taxes or Other Taxes are payable by any Borrower, as soon as practicable thereafter such Borrower shall send to the Administrative Agent for its own account or for the account of the relevant Lender, as the case may be, a certified copy of an official receipt received by such Borrower showing payment thereof (or other evidence of such payment reasonably satisfactory to the Administrative Agent). If any Borrower fails to pay any Non-Excluded Taxes or Other Taxes when due to the appropriate taxing authority or fails to remit to the Administrative Agent the required receipts or other required documentary evidence, such Borrower shall indemnify the Administrative Agent and the Lenders for any incremental taxes, interest or penalties that may become payable by the Administrative Agent or any Lender as a result of any such failure. The agreements in this Section 2.20 shall survive the termination of this Agreement and the payment of the Loans and all other amounts payable hereunder.

(d) Each Lender (or Transferee) that is not a citizen or resident of the United States of America, a corporation, partnership or other entity created or organized in or under the laws of the United States of America (or any jurisdiction thereof), or any estate or trust that is subject to federal income taxation regardless of the source of its income (a "NON-U.S. LENDER") shall deliver to the Borrowers and the Administrative Agent (or, in the case of a Participant, to the Lender from which the related participation shall have been purchased) two copies of either U.S. Internal Revenue Service Form W-8BEN or Form W-8ECI (or other applicable form), or, in the case of a Non-U.S. Lender claiming exemption from U.S. federal withholding tax under Section 871(h) or 881(c) of the Code with respect to payments of "portfolio interest" a statement substantially in the form of Exhibit I and a Form W-8BEN (or other applicable form), or any subsequent versions thereof or successors thereto properly completed and duly executed by such Non-U.S. Lender claiming complete exemption from, or a reduced rate of, U.S. federal withholding tax on all payments by any Borrower under this Agreement and the other Loan Documents. Such forms shall be delivered by each Non-U.S. Lender on or before the date it becomes a party to this Agreement (or, in the case of any Participant, on or before the date such Participant purchases the related participation). In addition, each Non-U.S. Lender shall deliver such forms promptly upon the obsolescence or invalidity of any form previously delivered by such Non-U.S. Lender. Each Non-U.S. Lender shall promptly notify the each Borrower at any time it determines that it is no longer in a position to provide any previously delivered certificate to such Borrower (or any other form of certification adopted by the U.S. taxing authorities for such purpose). Notwithstanding any other provision of this paragraph, a Non-U.S. Lender shall not be required to deliver any form pursuant to this paragraph that such Non-U.S. Lender is not legally able to deliver.

(e) A Lender that is entitled to an exemption from or reduction of non-U.S. withholding tax under the law of the jurisdiction in which a Borrower is located, or any treaty to which such jurisdiction is a party, with respect to payments under this Agreement shall deliver to such Borrower (with a copy to the Administrative Agent), at the time or times prescribed by applicable law or reasonably requested by such Borrower, such properly completed and executed documentation prescribed by applicable law as will permit such payments to be made without withholding or at a reduced rate, PROVIDED that such Lender is legally entitled to complete, execute and deliver such documentation and in such Lender's reasonable judgment such completion, execution or submission would not materially prejudice the legal position of such Lender.

2.21 INDEMNITY. Each Borrower agrees to indemnify each Lender for, and to hold each Lender harmless from, any loss or expense that such Lender sustains or incurs as a consequence of (a) default by such Borrower in making a borrowing of, conversion into or continuation of Eurodollar Loans after such Borrower has given a notice requesting the same in accordance with the provisions of this Agreement, (b) default by such Borrower in making any prepayment after such Borrower has given a notice thereof in accordance with the provisions of this Agreement or (c) the making by such Borrower of a prepayment or conversion of Eurodollar Loans on a day that is not the last day of an Interest Period with respect thereto; PROVIDED that any request for indemnification made by a Lender pursuant to this Section 2.21 shall be made within six months of the incurrence of the loss or expense requested to be indemnified. Such indemnification may include an amount equal to the excess, if any, of (i) the amount of interest that would have accrued on the amount so prepaid, or not so borrowed, converted or continued, for the period from the date of such prepayment or of such failure to borrow, convert or continue to the last day of such Interest Period (or, in the case of a failure to borrow, convert or continue, the Interest Period that would have commenced on the date of such failure) in each case at the applicable rate of interest for such Loans provided for herein (excluding, however, the Applicable Margin included therein, if any) over (ii) the amount of interest (as reasonably determined by such Lender) that would have accrued to such Lender on such amount by placing such amount on deposit for a comparable period with leading banks in the interbank Eurodollar market. A certificate as to any amounts payable pursuant to this Section submitted to any Borrower by any Lender shall be conclusive in the absence of manifest error. This covenant shall survive the termination of this Agreement and the payment of the Loans and all other amounts payable hereunder.

2.22 ILLEGALITY. Notwithstanding any other provision herein, if the adoption of or any change in any Requirement of Law or in the interpretation or application thereof shall make it unlawful for any Lender to make or maintain Eurodollar Loans as contemplated by this Agreement, (a) the commitment of such Lender hereunder to make Eurodollar Loans, continue Eurodollar Loans as such and convert Base Rate Loans to Eurodollar Loans shall forthwith be canceled and (b) such Lender's Loans then outstanding as Eurodollar Loans, if any, shall be converted automatically to Base Rate Loans on the respective last days of the then current Interest Periods with respect to such Loans or within such earlier period as required by law. If any such conversion of a Eurodollar Loan occurs on a day which is not the last day of the then current Interest Period with respect thereto, the relevant Borrower shall pay to such Lender such amounts, if any, as may be required pursuant to Section 2.21.

2.23 CHANGE OF LENDING OFFICE. Each Lender agrees that, upon the occurrence of any event that it knows to give rise to the operation of Section 2.19, 2.20(a) or 2.22 with respect to such Lender, it will use all commercially reasonable efforts (subject to overall policy considerations of such Lender) to designate another lending office for any Loans affected by such event, or to assign its rights and obligations hereunder with respect to such Loans to another of its offices, branches or affiliates with the object of avoiding the consequences of such event, or to assign its rights and obligations to another of its offices, branches or affiliates, with the object of avoiding the consequences of such event, or to assign its rights and obligations hereunder with respect to such Loans to another of its offices, branches or affiliates, with the object of avoiding the consequences of such event; PROVIDED, that such designation is made on terms that, in the reasonable sole judgment of such Lender, cause such Lender and its lending office(s) to suffer no economic, legal or regulatory disadvantage, and provided, further, that nothing in this Section shall affect or postpone any of the obligations of any Borrower or the rights of any Lender pursuant to Section 2.19, 2.20(a) or 2.22.

2.24 REPLACEMENT OF LENDERS UNDER CERTAIN CIRCUMSTANCES. The Primary Borrower shall be permitted to replace any Lender (a) that requests reimbursement for amounts owing pursuant to Section 2.19, (b) with respect to which any Borrower is required to pay any amounts under Section 2.20 or (c) that defaults in its obligation to make Loans hereunder, with a replacement financial institution; PROVIDED that (i) such replacement does not conflict with any Requirement of Law, (ii) no Event of Default shall have occurred and be continuing at the time of such replacement, (iii) prior to any such replacement, such Lender shall have taken no action under Section 2.22 so as to eliminate the continued need for payment of amounts owing pursuant to Section 2.19 or 2.20, (iv) the replacement financial institution shall purchase, at par, all Loans and other amounts owing to such replaced Lender on or prior to the date of replacement, (v) such Borrower shall be liable to such replaced Lender under Section 2.21 (as though Section 2.21 were applicable) if any Eurodollar Loan owing to such replaced Lender shall be purchased other than on the last day of the Interest Period relating thereto, (vi) the replacement financial institution, if not already a Lender, shall be reasonably satisfactory to the Administrative Agent, (vii) the replaced Lender and replacement Lender shall be obligated to make such replacement in accordance with the provisions of Section 10.6 (including, without limitation, obtaining the consents provided for therein) (provided that the relevant Borrower shall be obligated to pay the registration and processing fee referred to therein), (viii) the relevant Borrower shall pay all additional amounts (if any) required pursuant to Section 2.19 or 2.20, as the case may be, in respect of any period prior to the date on which such replacement shall be consummated, and (ix) any such replacement shall not be deemed to be a waiver of any rights that the relevant Borrower, the Administrative Agent or any other Lender shall have against the replaced Lender.

2.25 OPTIONAL COMMITMENT INCREASE. (a) The Primary Borrower shall have the right at any one time during the period commencing on the Closing Date and ending on the first anniversary thereof to increase the Total Revolving Credit Commitments to an aggregate amount not to exceed \$200,000,000 (i) by requesting in writing that any Lender or Lenders increase the amount of each such Lender's Revolving Credit Commitment or (ii) if, within 21 days of receipt of such request Lenders have not indicated their willingness to increase the Total Revolving Credit Commitments to the requested amount, by requesting that one or more banks or other financial institutions not a party to this Agreement become a Lender or Lenders hereunder; provided, that (x) the addition of any bank or financial institution becoming a Lender pursuant to clause (ii) above shall be subject to the consent of the Administrative Agent (which consent shall not be unreasonably withheld), (y) the Revolving Credit Commitment of any bank or other financial institution becoming a Lender pursuant to clause (ii) above shall be at least \$10,000,000 (except that in the event that the aggregate amount of the increase of Revolving Credit Commitments pursuant to clause (i) above shall be greater than \$40,000,000 and less than \$50,000,000, the Revolving Credit Commitment of one bank or financial institution becoming a Lender pursuant to clause (ii) above may be less than \$10,000,000) and (z) the amount of the increase of any Lender's Revolving Credit Commitment pursuant to clause (i) above shall be at least \$5,000,000.

(b) Any increase in the Total Revolving Credit Commitments pursuant to Section 2.25(a)(i) shall be effective only upon the execution and delivery to the Primary Borrower and the Administrative Agent of a commitment increase supplement in substantially the form of Exhibit K (a "REVOLVING COMMITMENT INCREASE SUPPLEMENT"), which Revolving Commitment Increase Supplement shall be delivered to the Administrative Agent not less than five Business Days prior to the date on which the increase provided for therein is to become effective (the "REVOLVING COMMITMENT INCREASE DATE") and shall specify (i) the amount of any increase in the Revolving Credit Commitment of any Lender and (ii) the Revolving Commitment Increase Date.

(c) Any additional bank, financial institution or other entity which elects to become a party to this Agreement and provide a Revolving Credit Commitment pursuant to Section 2.25 (a)(ii) shall execute a New Lender Supplement (each, a "NEW LENDER SUPPLEMENT") with the Borrowers and the Administrative Agent, substantially in the form of Exhibit L. Upon its receipt of a New Lender Supplement executed by an additional bank, financial institution or other entity which elects to become a party to this Agreement and provide a Revolving Credit Commitment pursuant to Section 2.25(a)(ii) (and which has been approved by the Administrative Agent as provided in Section 2.25(a)), the Administrative Agent shall (i) promptly accept such New Lender Supplement and (ii) record the information and Revolving Credit Commitment contained therein in the Register on the effective date determined pursuant thereto, whereupon such bank, financial institution or other entity shall become a Lender for all purposes and to the same extent as if originally a party hereto and shall be bound by and entitled to the benefits of this Agreement.

(d) If the Total Revolving Credit Commitments are increased pursuant to this Section 2.25, the amount of such increase shall, from and after the Revolving Commitment Increase Date, be available for usage by the Additional Borrower as Revolving Credit Loans and Letters of Credit (it being understood that such amount shall not be available for usage by the Additional Borrower as Swingline Loans); PROVIDED that not more than \$10,000,000 of the L/C Commitment shall be available to the Additional Borrower for the issuance of Letters of Credit in accordance with Section 3.

(e) Any increase in the Total Revolving Credit Commitments pursuant to this Section 2.25 shall not be effective unless:

(i) no Default or Event of Default shall have occurred and be continuing on the Revolving Commitment Increase Date;

(ii) each of the representations and warranties made by each Borrower in Section 4, or in any certificate delivered pursuant hereto, shall be true and correct in all material respects on the Revolving Commitment Increase Date with the same effect as though made on and as of such date, except to the extent such representations and warranties expressly relate to an earlier date in which case such representations and warranties shall be true and correct in all material respects as of such earlier date; and

(iii) the Administrative Agent shall have received such corporate resolutions, legal opinions and other documents which it shall reasonably request with respect to the obligations of the Additional Borrower hereunder.

Each notice requesting an increase in the Total Revolving Credit Commitments pursuant to this Section 2.25 shall constitute a certification to the effect set forth in clauses (i) and (ii) of this Section 2.25(e).

(f) No Lender shall at any time be required to agree to a request of the Primary Borrower to increase its Revolving Credit Commitment or obligations hereunder.

SECTION 3 LETTERS OF CREDIT

3.1 L/C COMMITMENT. (a) Subject to the terms and conditions hereof, each Issuing Lender, in reliance on the agreements of the other Revolving Credit Lenders set forth in Section 3.4(a), agrees to issue letters of credit ("LETTERS OF CREDIT") for the account of the Primary Borrower (or, from and after the Revolving Commitment Increase Date and subject to the limitations set forth in Section 2.25, the Additional Borrower) on any Business Day during the Revolving Credit Commitment Period in such form as may be approved from time to time by such Issuing Lender; PROVIDED, that no Issuing Lender shall have any obligation to issue any Letter of Credit if, after giving effect to such issuance, (i) the L/C Obligations would exceed the L/C Commitment or (ii) the aggregate amount of the Available Revolving Credit Commitments would be less than zero. Each Letter of Credit shall (i) be denominated in Dollars and (ii) expire no later than the earlier of (x) the first anniversary of its date of issuance and (y) the date which is five Business Days prior to the Revolving Credit Termination Date, provided that any Letter of Credit with a one-year term may provide for the renewal thereof for additional one-year periods (which shall in no event extend beyond the date referred to in clause (y) above).

(b) No Issuing Lender shall at any time be obligated to issue any Letter of Credit hereunder if such issuance would conflict with, or cause such Issuing Lender or any L/C Participant to exceed any limits imposed by, any applicable Requirement of Law.

3.2 PROCEDURE FOR ISSUANCE OF LETTER OF CREDIT. The Primary Borrower (or, from and after the Revolving Commitment Increase Date and subject to the limitations set forth in Section 2.25, the Additional Borrower) may from time to time request that an Issuing Lender issue a Letter of Credit by delivering to such Issuing Lender (with a copy to the Administrative Agent) at its address for notices specified herein an Application therefor, completed to the satisfaction of such Issuing Lender, and such other certificates, documents and other papers and information as such Issuing Lender may request. Upon receipt of any Application, an Issuing Lender will process such Application and the certificates, documents and other papers and information delivered to it in connection therewith in accordance with its customary procedures and shall promptly issue the Letter of Credit requested thereby by issuing the original of such Letter of Credit to the beneficiary thereof or as otherwise may be agreed to by such Issuing Lender and the relevant Borrower (but in no event shall any Issuing Lender be required to issue any Letter of Credit earlier than three Business Days after its receipt of the Application therefor and all such other certificates, documents and other papers and information relating thereto). Promptly after issuance by an Issuing Lender of a Letter of Credit, such Issuing Lender shall furnish a copy of such Letter of Credit to the relevant Borrower. Each Issuing Lender shall, within three days of such issuance, give notice to the Administrative Agent of the issuance of each Letter of Credit issued by such Issuing Lender (including the amount thereof). Upon the written request of any Revolving Credit Lender, the Administrative Agent will, within three Business Days of such request, inform such Revolving Credit Lender of the aggregate drawable amount of all outstanding Letters of Credit issued to the Borrowers on the date of such request.

3.3 FEES AND OTHER CHARGES. (a) Each Borrower will pay to the Administrative Agent, for the account of the Revolving Credit Lenders, a fee on the aggregate drawable amount of all outstanding Letters of Credit issued for such Borrower's account at a per annum rate equal to the Applicable Margin then in effect with respect to Eurodollar Loans under the Revolving Credit Facility, to be shared ratably among the Revolving Credit Lenders in accordance with their respective Revolving Credit Percentages and payable quarterly in arrears on each L/C Fee Payment Date after the issuance date. In addition, each Borrower shall pay to the relevant Issuing Lender for its own account a fronting fee on the aggregate drawable amount of all outstanding Letters of Credit issued by such Issuing Lender for such Borrower's account at a rate to be agreed upon by such Borrower and such Issuing Lender, payable quarterly in arrears on each L/C Fee Payment Date after the date of issuance of such Letter of Credit.

(b) In addition to the foregoing fees, the relevant Borrower shall pay or reimburse each Issuing Lender for such normal and customary costs and expenses as are incurred or charged by such Issuing Lender in issuing, negotiating, effecting payment under, amending or otherwise administering any Letter of Credit.

 $3.4\ \text{L/C}$ PARTICIPATIONS. (a) Each Issuing Lender irrevocably agrees to grant and hereby grants to each L/C Participant, and, to induce each Issuing Lender to issue Letters of Credit hereunder, each L/C Participant irrevocably agrees to accept and purchase and hereby accepts and purchases from each Issuing Lender, on the terms and conditions hereinafter stated, for such L/C Participant's own account and risk, an undivided interest equal to such L/C Participant's Revolving Credit Percentage in each Issuing Lender's obligations and rights under each Letter of Credit issued by such Issuing Lender hereunder and the amount of each draft paid by such Issuing Lender thereunder. Each L/C Participant unconditionally and irrevocably agrees with each Issuing Lender that, if a draft is paid under any Letter of Credit issued by such Issuing Lender for which such Issuing Lender is not reimbursed in full by the relevant Borrower in accordance with the terms of this Agreement, such L/C Participant shall pay to such Issuing Lender upon demand at such Issuing Lender's address for notices specified herein an amount equal to such L/C Participant's Revolving Credit Percentage of the amount of such draft, or any part thereof, that is not so reimbursed.

(b) If any amount required to be paid by any L/C Participant to an Issuing Lender pursuant to Section 3.4(a) in respect of any unreimbursed portion of any payment made by such Issuing Lender under any Letter of Credit is paid to such Issuing Lender within three Business Days after the date such payment is due, such L/C Participant shall pay to such $\ensuremath{\bar{\text{Issuing}}}$ Lender on demand an amount equal to the product of (i) such amount, times (ii) the daily average Federal Funds Effective Rate during the period from and including the date such payment is required to the date on which such payment is immediately available to such Issuing Lender, times (iii) a fraction the numerator of which is the number of days that elapse during such period and the denominator of which is 360. If any such amount required to be paid by any L/C Participant pursuant to Section 3.4(a) is not made available to such Issuing Lender by such L/C Participant within three Business Days after the date such payment is due, such Issuing Lender shall be entitled to recover from such L/C Participant, on demand, such amount with interest thereon calculated from such due date at the rate per annum applicable to Base Rate Loans under the Revolving Credit Facility. A certificate such Issuing Lender submitted to any L/C Participant with respect to any such amounts owing under this Section shall be conclusive in the absence of manifest error.

(c) Whenever, at any time after an Issuing Lender has made payment under any Letter of Credit and has received from any L/C Participant its PRO RATA share of such payment in accordance with Section 3.4(a), such Issuing Lender receives any payment related to such Letter of Credit (whether directly from the relevant Borrower or otherwise, including proceeds of collateral applied thereto by such Issuing Lender), or any payment of interest on account thereof, such Issuing Lender will distribute to such L/C Participant its PRO RATA share thereof; PROVIDED, HOWEVER, that in the event that any such payment received by such Issuing Lender shall be required to be returned by such Issuing Lender, such L/C Participant shall return to such Issuing Lender the portion thereof previously distributed by such Issuing Lender to it.

3.5 REIMBURSEMENT OBLIGATION OF THE BORROWERS. Each Borrower agrees to reimburse each Issuing Lender, on the next Business Day after each date on which such Issuing Lender notifies such Borrower of the date and amount of a draft presented under any Letter of Credit issued for such Borrower's account and paid by such Issuing Lender, for the amount of (a) such draft so paid and (b) any taxes, fees, charges or other costs or expenses incurred by such Issuing Lender in connection with such payment (the amounts described in the foregoing clauses (a) and (b) in respect of any drawing, collectively, the "PAYMENT AMOUNT"). Each such payment shall be made to such Issuing Lender at its address for notices specified herein in lawful money of the United States of America and in immediately available funds. Interest shall be payable on each Payment Amount from the date of the applicable drawing until payment in full at the rate set forth in (i) until the second Business Day following the date of the applicable drawing, Section 2.15(b) and (ii) thereafter, Section 2.15(c). Each drawing under any Letter of Credit issued for either Borrower's account shall (unless an event of the type described in clause (i) or (ii) of Section 8(f) shall have occurred and be continuing with respect to such Borrower, in which case the procedures specified in Section 3.4 for funding by L/C Participants shall apply) constitute a request by such Borrower to the Administrative Agent for a borrowing pursuant to Section 2.5 of Base Rate Loans (or, in the case of the Primary Borrower, at the option of the Administrative Agent and the Swing Line Lender in their sole discretion, a borrowing pursuant to Section 2.7 of Swing Line Loans) in the amount of such drawing. The Borrowing Date with respect to such borrowing shall be the first date on which a borrowing of Revolving Credit Loans (or, if applicable, Swing Line Loans) could be made, pursuant to Section 2.5 (or, if applicable, Section 2.7), if the Administrative Agent had received a notice of such borrowing at the time the Administrative Agent receives notice from the relevant Issuing Lender of such drawing under such Letter of Credit.

3.6 OBLIGATIONS ABSOLUTE. Each Borrower's obligations under this Section 3 shall be absolute and unconditional under any and all circumstances and irrespective of any setoff, counterclaim or defense to payment that any Borrower may have or have had against any Issuing Lender, any beneficiary of a Letter of Credit or any other Person. Each Borrower also agrees with each Issuing Lender that such Issuing Lender shall not be responsible for, and such Borrower's Reimbursement Obligations under Section 3.5 shall not be affected by, among other things, the validity or genuineness of documents or of any endorsements thereon, even though such documents shall in fact prove to be invalid, fraudulent or forged, or any dispute between or among such Borrower and any beneficiary of any Letter of Credit or any other party to which such Letter of Credit may be transferred or any claims whatsoever of such Borrower against any beneficiary of such Letter of Credit or any such transferee. No Issuing Lender shall be liable for any error, omission, interruption or delay in transmission, dispatch or delivery of any message or advice, however transmitted, in connection with any Letter of Credit, except for errors or omissions found by a final and nonappealable decision of a court of competent jurisdiction to have resulted from the gross negligence or willful misconduct of such Issuing Lender. Each Borrower agrees that any action taken or omitted by an Issuing Lender under or in connection with any Letter of Credit issued by it for the account of such Borrower or the related drafts or documents, if done in the absence of gross negligence or willful misconduct and in accordance with the standards of care specified in the Uniform Commercial Code in effect in the State of

New York (the "UCC") or, if applicable to such Letter of Credit, the Uniform Customs and Practice for Documentary Credits as published by the International Chamber of Commerce most recently at the time of issuance of any Letter of Credit (the "UCP"), shall be binding on the Borrower and shall not result in any liability of such Issuing Lender to such Borrower.

3.7 LETTER OF CREDIT PAYMENTS. If any draft shall be presented for payment under any Letter of Credit, the relevant Issuing Lender shall promptly notify the relevant Borrower of the date and amount thereof. The responsibility of the relevant Issuing Lender to the relevant Borrower in connection with any draft presented for payment under any Letter of Credit, in addition to any payment obligation expressly provided for in such Letter of Credit issued by such Issuing Lender, shall be limited to determining that the documents (including each draft) delivered under such Letter of Credit in connection with such presentment appear on their face to be in conformity with such Letter of Credit.

3.8 APPLICATIONS. To the extent that any provision of any Application related to any Letter of Credit is inconsistent with the provisions of this Section 3, the provisions of this Section 3 shall apply.

SECTION 4 REPRESENTATIONS AND WARRANTIES

To induce the Administrative Agent and the Lenders to enter into this Agreement and to make the Loans and issue or participate in the Letters of Credit, (i) the Parent and the Primary Borrower hereby jointly and severally represent and warrant to the Administrative Agent and each Lender and (ii) the Additional Borrower, upon becoming a Borrower hereunder, shall be deemed to represent and warrant to the Administrative Agent and each Lender that:

4.1 FINANCIAL CONDITION. (a) The estimated pro forma consolidated balance sheet of the Primary Borrower and its consolidated Subsidiaries as at September 30, 2000 (including the notes thereto) (the "PRO FORMA BALANCE SHEET"), copies of which have heretofore been furnished to each Lender, has been prepared giving effect (as if such events had occurred on such date) to (i) the consummation of the Acquisition, (ii) the Loans to be made and the use of proceeds thereof, (iii) the other financing transactions to be consummated in connection with the Acquisition, (iv) an increase in reserves of at least \$800,000,000 at September 30, 2000 for 1999 and prior accident years, (v) an aggregate unassigned surplus of dividend paying first-tier Insurance Subsidiaries of at least \$950,000,000, (vi) the net cost of the Reinsurance Agreements described in Section 5.1(m) and (vii) the payment of fees and expenses in connection with the foregoing. The Pro Forma Balance Sheet has been prepared based on the best information available to the Primary Borrower as of the date of delivery thereof, and presents fairly on a pro forma basis the estimated financial position of Primary Borrower and its consolidated Subsidiaries as at September 30, 2000, assuming that the events specified in the preceding sentence had actually occurred at such date, and the PRO FORMA balance sheet referred to in Section 5.1(c)(i)(B) shall be, if required to be delivered pursuant to such Section, prepared based on the best information available to the Primary Borrower as of the date of delivery thereof, and shall present fairly on a pro forma basis the estimated financial position of Primary Borrower and its consolidated Subsidiaries as at December 31, 2000, assuming that the events specified in the preceding sentence had actually occurred at such date.

(b) The unaudited pro forma consolidated balance sheets and consolidated income statements of the entities comprising the Acquired Assets as at December 31, 1999 (including the notes thereto), (collectively, the "ACQUIRED ASSETS FINANCIAL STATEMENTS"), copies of which have heretofore been furnished to each Lender, present fairly the consolidated financial condition of such entities as at such date, and the consolidated results of their operations for the respective fiscal years then ended. All such financial statements, including the schedules and notes thereto, have been prepared in accordance with GAAP applied consistently throughout the periods involved. The entities comprising the Acquired Assets do not have any material Guarantee Obligations, contingent liabilities and liabilities for taxes, or any long-term leases or unusual forward or long-term commitments, including, without limitation, any interest rate or foreign currency swap or exchange transaction or other obligation in respect of derivatives, that are not reflected in the most recent financial statements referred to in this paragraph. During the period from December 31, 1999 to and including the date hereof there has been no Disposition by the entities comprising the Acquired Assets of any material part of their business or Property, except for the Preclosing Transactions.

(c) The audited consolidated balance sheets of White Mountains as at December 31, 1998 and December 31, 1999, and the related consolidated statements of income and of cash flows for the fiscal years ended on such dates, reported on by and accompanied by unqualified reports from PricewaterhouseCoopers and KPMG LLP with respect to the fiscal years ended on December 31, 1998 and December 31, 1999 respectively, present fairly the consolidated financial condition of White Mountains as at such date, and the consolidated results of its operations and its consolidated cash flows for the respective fiscal years then ended. The unaudited consolidated balance sheet of White Mountains as at September 30, 2000, and the related unaudited consolidated statements of income and cash flows for the nine-month period ended on such date, present fairly the consolidated financial condition of White Mountains as at such date, and the consolidated results of its operations and its consolidated cash flows for the nine-month period then ended (subject to normal year-end audit adjustments). All such financial statements, including the related schedules and notes thereto, have been prepared in accordance with GAAP applied consistently throughout the periods involved (except as approved by the aforementioned firms of accountants and disclosed therein). White Mountains and its Subsidiaries do not have any material Guarantee Obligations, contingent liabilities and liabilities for taxes, or any long-term leases or unusual forward or long-term commitments, including, without limitation, any interest rate or foreign currency swap or exchange transaction or other obligation in respect of derivatives, that are not reflected in the most recent financial statements referred to in this paragraph or filings made by White Mountains with the SEC after December 31, 1999 and prior to March 1, 2001. During the period from December 31, 1999 to and including the date hereof there has been no Disposition by White Mountains of any material part of its business or Property, except as disclosed in filings made by White Mountains with the SEC after December 31, 1999 and prior to March 1, 2001.

(d) The audited consolidated balance sheets of CGU as at December 31, 1999, and the related consolidated statements of income and of cash flows for the fiscal years ended on such dates, reported on by and accompanied by unqualified reports from PricewaterhouseCoopers, present fairly the consolidated financial condition of CGU as at such date, and the consolidated results of its operations and its consolidated cash flows for the respective fiscal years then ended. The unaudited consolidated balance sheet of CGU as at

September 30, 2000, and the related unaudited consolidated statements of income and cash flows for the nine-month period ended on such date, present fairly the consolidated financial condition of CGU as at such date, and the consolidated results of its operations and its consolidated cash flows for the nine-month period then ended (subject to normal year-end audit adjustments). All such financial statements, including the related schedules and notes thereto, have been prepared in accordance with GAAP applied consistently throughout the periods involved (except as approved by the aforementioned firms of accountants and disclosed therein). CGU and its Subsidiaries do not have any material Guarantee Obligations, contingent liabilities and liabilities for taxes, or any long-term leases or unusual forward or long-term commitments, including, without limitation, any interest rate or foreign currency swap or exchange transaction or other obligation in respect of derivatives, that are not reflected in the most recent financial statements referred to in this paragraph. During the period from September 30, 2000 to and including the date hereof there has been no Disposition by CGU of any material part of its business or Property other than the Preclosing Transactions.

(e) The audited combined financial statements of admitted assets, liabilities and Capital and Surplus of the Insurance Subsidiaries of CGU as at December 31, 1999, and the related combined statements of income, Capital and Surplus and cash flows for the fiscal years ended on such date, reported on by and accompanied by unqualified reports from PricewaterhouseCoopers, present fairly the combined financial condition of the Insurance Subsidiaries of CGU as at such date, and the combined results of their operations and their combined cash flows for the respective fiscal years then ended. All such financial statements, including the related schedules and notes thereto, have been prepared in accordance with SAP applied consistently throughout the periods involved (except as approved by the aforementioned firms of accountants and disclosed therein). The Insurance Subsidiaries of CGU do not have any material Guarantee Obligations, contingent liabilities and liabilities for taxes, or any long-term leases or unusual forward or long-term commitments, including, without limitation, any interest rate or foreign currency swap or exchange transaction or other obligation in respect of derivatives, that are not reflected in the most recent financial statements referred to in this paragraph. During the period from December 31, 1999 to and including the date hereof there has been no Disposition by CGU of any material part of the business or Property of its Insurance Subsidiaries, except for the Preclosing Transactions.

The representations and warranties set forth in paragraphs (a), (b), (d) and (e) above are qualified in their entirety by the Specified Exceptions.

4.2 NO CHANGE. (a) Up to and including the Closing Date (prior to giving effect to the consummation of the Acquisition), since December 31, 1999 and (b) from and including the Closing Date (after giving effect to the Acquisition), since the Closing Date, there has been no development or event that has or could reasonably be expected to have a Material Adverse Effect.

4.3 CORPORATE EXISTENCE; COMPLIANCE WITH LAW. Each of the Parent, each Borrower and its Subsidiaries (a) is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization, (b) has the corporate or other power and authority, and the legal right, to own and operate its Property, to lease the Property it operates as lessee and to conduct the business in which it is currently engaged, (c) is duly qualified as a foreign

corporation and in good standing under the laws of each jurisdiction where its ownership, lease or operation of Property or the conduct of its business requires such qualification, except to the extent failure to so qualify or be in good standing could not, in the aggregate, reasonably be expected to have a Material Adverse Effect, and (d) is in compliance with all Requirements of Law except to the extent that the failure to comply therewith could not, in the aggregate, reasonably be expected to have a Material Adverse Effect. It is understood that this provision shall not imply a representation with respect to environmental matters, which representation is set forth in Section 4.16.

4.4 CORPORATE POWER; AUTHORIZATION; ENFORCEABLE OBLIGATIONS. Each Loan Party has the corporate or other power and authority, and the legal right, to make, deliver and perform the Loan Documents to which it is a party and, in the case of each Borrower, to borrow hereunder. Each Loan Party has taken all necessary corporate or other action to authorize the execution, delivery and performance of the Loan Documents to which it is a party and, in the case of each Borrower, to authorize the borrowings on the terms and conditions of this Agreement. No consent or authorization of, filing with, notice to or other act by or in respect of, any Governmental Authority or any other Person is required in connection with the Acquisition, the borrowings hereunder or the execution, delivery, performance, validity or enforceability of this Agreement or any of the other Loan Documents, except (i) consents, authorizations, filings and notices described in Schedule 4.4, which consents, authorizations, filings and notices have been obtained or made and are in full force and effect and (ii) the filings referred to in Section 4.18, except to the extent failure to obtain any consents, authorizations, filings and notices could not, in the aggregate, reasonably be expected to have a Material Adverse Effect. Each Loan Document has been duly executed and delivered on behalf of each Loan Party that is a party thereto. This Agreement constitutes, and each other Loan Document upon execution will constitute, a legal, valid and binding obligation of each Loan Party that is a party thereto, enforceable against each such Loan Party in accordance with its terms, except as enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting the enforcement of creditors' rights generally and by general equitable principles (whether enforcement is sought by proceedings in equity or at law).

4.5 NO LEGAL BAR. The execution, delivery and performance of this Agreement and the other Loan Documents, the issuance of Letters of Credit, the borrowings hereunder and the use of the proceeds thereof will not violate any Requirement of Law or any Contractual Obligation of the Parent, any Borrower or any of its Subsidiaries and will not result in, or require, the creation or imposition of any Lien on any of their respective properties or revenues pursuant to any Requirement of Law or any such Contractual Obligation (other than the Liens created by the Security Documents), except to the extent such violation or Lien could not, in the aggregate, reasonably be expected to have a Material Adverse Effect.

4.6 NO MATERIAL LITIGATION. No litigation, investigation or proceeding of or before any arbitrator or Governmental Authority is pending or, to the knowledge of the Parent or any Borrower, threatened by or against the Parent, any Borrower or any of its Subsidiaries or against any of their respective properties or revenues (a) with respect to any of the Loan Documents or any of the transactions contemplated hereby or thereby, or (b) that could reasonably be expected to have a Material Adverse Effect. 4.7 NO DEFAULT. Neither the Parent, any Borrower or any of its Subsidiaries is in default under or with respect to any of its Contractual Obligations in any respect that could reasonably be expected to have a Material Adverse Effect. No Default or Event of Default has occurred and is continuing. It is understood that CGU and/or its Subsidiaries have lost the right to remarket Industrial Revenue Bonds in an aggregate principal amount not greater than \$30,750,000 as a result of the loss or withdrawal of a credit rating and are being required to repurchase the bonds as remarket dates occur.

4.8 OWNERSHIP OF PROPERTY; LIENS. Each of the Parent, each Borrower and its Subsidiaries has title in fee simple to, or a valid leasehold interest in, all its real property, and good title to, or a valid leasehold interest in, all its other Property, and none of such Property is subject to any Lien except as permitted by Section 7.3, except to the extent such defects in title could not, in the aggregate, reasonably be expected to have a Material Adverse Effect.

4.9 INTELLECTUAL PROPERTY. The Parent, each Borrower and each of its Subsidiaries owns, or is licensed to use, all Intellectual Property necessary for the conduct of its business as currently conducted. No material claim has been asserted and is pending by any Person challenging or questioning the use of any Intellectual Property or the validity or effectiveness of any Intellectual Property, nor does the Parent or any Borrower know of any valid basis for any such claim. The use of Intellectual Property by the Parent, each Borrower and its Subsidiaries does not infringe on the rights of any Person in any material respect. It is understood that the Primary Borrower and its Subsidiaries will be permitted to use the CGU name and related tradenames for (i) with respect to Insurance Subsidiaries, a period of eighteen months after the Closing Date (with conditional six-month extensions through September 24, 2005 in jurisdictions where regulatory approval for name changes has not been received) but not thereafter and (ii) with respect to Non-Insurance Subsidiaries, a period of 180 days after the Closing Date but not thereafter.

4.10 TAXES. Each of the Parent, each Borrower and each of its Subsidiaries has filed or caused to be filed all material Federal, state and other tax returns that are required to be filed (taking into account any applicable extensions) and has paid all taxes shown to be due and payable on said returns or on any assessments made against it or any of its Property and all other material taxes, fees or other charges imposed on it or any of its Property by any Governmental Authority (other than any the amount or validity of which are currently being contested in good faith by appropriate proceedings and with respect to which reserves in conformity with SAP or GAAP, as applicable, have been provided on the books of the Parent, any Borrower or its Subsidiaries, as the case may be); and, to the knowledge of the Parent and each Borrower, no tax Lien has been filed, and no claim is being asserted, with respect to any such tax, fee or other charge.

4.11 FEDERAL REGULATIONS. No part of the proceeds of any Loans will be used for "purchasing" or "carrying" any "margin stock" within the respective meanings of each of the quoted terms under Regulation U as now and from time to time hereafter in effect or for any purpose that violates the provisions of the Regulations of the Board. If requested by any Lender or the Administrative Agent, the relevant Borrower will furnish to the Administrative Agent and each Lender a statement to the foregoing effect in conformity with the requirements of FR Form G-3 or FR Form U-1 referred to in Regulation U.

4.12 ERISA. Neither a Reportable Event nor an "accumulated funding deficiency" (within the meaning of Section 412 of the Code or Section 302 m ofERISA) has occurred during the five-year period prior to the date on which this representation is made or deemed made with respect to any Plan, and each Plan has complied in all material respects with the applicable provisions of ERISA and the Code. No termination of a Single Employer Plan has occurred, and no Lien in favor of the PBGC or a Plan has arisen, during such five-year period, other than in connection with acquisitions of Houston General Insurance Group, York Insurance Company and A.W.G. Dewar, Inc. The present value of all accrued benefits under each Single Employer Plan (based on those assumptions used to fund such Plans) did not, as of the last annual valuation date prior to the date on which this representation is made or deemed made, exceed the value of the assets of such Plan allocable to such accrued benefits by an amount which, if the Plan then terminated, could reasonably be expected to result in a Material Adverse Effect. Neither Borrower nor any Commonly Controlled Entity of any Borrower has had a complete or partial withdrawal from any Multiemployer Plan that has resulted or could reasonably be expected to result in a material liability under ERISA, and neither Borrower nor any Commonly Controlled Entity of any Borrower would become subject to any liability under ERISA which could reasonably be expected to result in a Material Adverse Effect if such Borrower or any such Commonly Controlled Entity were to withdraw completely from all Multiemployer Plans as of the valuation date most closely preceding the date on which this representation is made or deemed made. No such Multiemployer Plan is in Reorganization or Insolvent.

4.13 INVESTMENT COMPANY ACT; OTHER REGULATIONS. No Loan Party is an "investment company", or a company "controlled" by an "investment company", within the meaning of the Investment Company Act of 1940, as amended. No Loan Party is subject to regulation under any Requirement of Law (other than Regulation X of the Board) which limits its ability to incur Indebtedness.

4.14 SUBSIDIARIES. (a) The Subsidiaries listed on Schedule 4.14 constitute all the Subsidiaries of the Primary Borrower on the Closing Date after giving effect to the Acquisition and the Preclosing Transactions. Schedule 4.14 sets forth as of the date of this Agreement the name and jurisdiction of incorporation of each such Subsidiary and, as to each such Subsidiary, the percentage of each class of Capital Stock owned by each Loan Party.

(b) There are no outstanding subscriptions, options, warrants, calls, rights or other agreements or commitments (other than stock options granted to employees or directors and directors' qualifying shares) of any nature relating to any Capital Stock of the Parent, the Primary Borrower or any Subsidiary, except as disclosed on Schedule 4.14.

4.15 USE OF PROCEEDS. The proceeds of the Loans, and the Letters of Credit, shall be used to finance a portion of the Acquisition, to repay certain existing indebtedness of CGU and its subsidiaries, to pay related fees and expenses and for working capital and general corporate purposes of the relevant Borrower and its Subsidiaries in the ordinary course of business. 4.16 ENVIRONMENTAL MATTERS. Other than exceptions to any of the following that could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect:

(a) Each Borrower and its Subsidiaries: (i) are, and within the period of all applicable statutes of limitation have been, in compliance with all applicable Environmental Laws; (ii) hold all Environmental Permits (each of which is in full force and effect) required for any of their current operations or for any property owned, leased, or otherwise operated by any of them; and (iii) are, and within the period of all applicable statutes of limitation have been, in compliance with all of their Environmental Permits.

(b) Materials of Environmental Concern are not present at, on, under, in, or about any real property now and were not present at, on, under, in or about any real property formerly owned, leased or operated by any of them during the period of such ownership, lease or operation, owned, leased or operated by any Borrower or any of its Subsidiaries, and neither any Borrower nor any of its Subsidiaries (nor any entity for the actions of which any of them is responsible) has released or discharged any Materials of Environmental Concern at any property formerly owned, leased or operated by any of them or at any other location (including, without limitation, any location to which Materials of Environmental Concern have been sent for re-use or recycling or for treatment, storage, or disposal) which could reasonably be expected to (i) give rise to liability of any Borrower or any of its Subsidiaries under any applicable Environmental Law, or (ii) adversely interfere with any Borrower's or any of its Subsidiaries' continued operations.

(c) There is no judicial, administrative, or arbitral proceeding (including any notice of violation or alleged violation) under or relating to any Environmental Law to which any Borrower or any of its Subsidiaries is, or to the knowledge of the Parent or any Borrower will be, named as a party that is pending or, to the knowledge of the Parent or any Borrower threatened, nor has any Borrower nor any of its Subsidiaries received any request for information the subject of which could reasonably be expected to result in any of the foregoing.

(d) Neither any Borrower nor any of its Subsidiaries has received any written notice that it is a potentially responsible party under or relating to the federal Comprehensive Environmental Response, Compensation, and Liability Act or any similar Environmental Law, or with respect to any Materials of Environmental Concern.

(e) Neither any Borrower nor any of its Subsidiaries has entered into or agreed to any consent decree, order, or settlement or other agreement, or is subject to any judgment, decree, or order or other agreement, in any judicial, administrative, arbitral, or other forum for dispute resolution, relating to compliance with or liability under any Environmental Law.

(f) Neither any Borrower nor any of its Subsidiaries has assumed or retained, by contract or operation of law, any liabilities of any kind, fixed or contingent, reasonably likely to form the basis for any claim under any ${\rm Environmental}$ Law or with respect to any Material of ${\rm Environmental}$ Concern.

(g) It is understood that Section 4.16 does not apply to claims made pursuant to policies for which any Borrower or any of its Subsidiaries is or is asserted to be responsible, issued pursuant to the insurance business of the Borrower and/or its Insurance Subsidiaries or to the settlement of any such claims.

4.17 ACCURACY OF INFORMATION, ETC. No statement or information contained in this Agreement, any other Loan Document, the Confidential Information Memorandum or any other document, certificate or statement furnished to the Administrative Agent or the Lenders or any of them, by or on behalf of any Loan Party for use in connection with the transactions contemplated by this Agreement or the other Loan Documents, contained as of the date such statement, information, document or certificate was so furnished (or, in the case of the Confidential Information Memorandum, as of the date of this Agreement), any untrue statement of a material fact or omitted to state a material fact necessary in order to make the statements contained herein or therein not misleading. The projections and pro forma financial information contained in the materials referenced above are based upon good faith estimates and assumptions believed by management of the Primary Borrower to be reasonable at the time made, it being recognized by the Lenders that such financial information as it relates to future events is not to be viewed as fact and that actual results during the period or periods covered by such financial information may differ from the projected results set forth therein by a material amount. As of the date hereof, the representations and warranties contained in the Acquisition Documentation are true and correct in all material respects. This Section 4.17 is qualified in its entirety by the Specified Exceptions.

4.18 SECURITY DOCUMENTS. The Guarantee and Collateral Agreement is effective to create in favor of the Administrative Agent, for the benefit of the Lenders, a legal, valid and enforceable security interest in the Collateral described therein and proceeds thereof. In the case of the Pledged Stock described in the Guarantee and Collateral Agreement, when any stock certificates representing such Pledged Stock are delivered to the Administrative Agent, and in the case of the other Collateral described in the Guarantee and Collateral Agreement, when financing statements in appropriate form are filed in the offices specified on Schedule 4.18-1 (which financing statements have been duly completed and executed and delivered to the Administrative Agent) and such other filings as are specified on Schedule 3 to the Guarantee and Collateral Agreement have been completed (all documents required for which filings have been duly completed and executed and delivered to the Administrative Agent), the Guarantee and Collateral Agreement shall constitute a perfected Lien on, and security interest in, all right, title and interest of the Loan Parties in such Collateral and the proceeds thereof, as security for the Obligations (as defined in the Guarantee and Collateral Agreement), in each case to the extent that such security interest can be perfected by the delivery of stock certificates and other instruments with effective indorsements, the filing of UCC financing statements, and filings in respect of Intellectual Property in the United States Patent and Trademarks Office. Schedule 4.18-2 lists each UCC Financing Statement that (i) names any Loan Party (other than the Additional Borrower) as debtor and (ii) will remain on file after the Closing Date. Schedule 4.18-3 lists each UCC Financing Statement that (i) names any Loan Party (other than the Additional Borrower) as debtor and (ii) will be terminated on or prior to the Closing Date; and

on or prior to the Closing Date, the Primary Borrower will have delivered to the Administrative Agent, or caused to be filed, duly completed UCC termination statements, signed by the relevant secured party, in respect of each UCC Financing Statement listed in Schedule 4.18-3.

4.19 SOLVENCY. Each Loan Party is, and after giving effect to the Acquisition and the incurrence of all Indebtedness and obligations being incurred in connection herewith and therewith will be and will continue to be, Solvent.

4.20 INSURANCE REGULATORY MATTERS. No License of any Insurance Subsidiary, the loss of which could reasonably be expected to have a Material Adverse Effect, is the subject of a proceeding for suspension or revocation. To the knowledge of the Parent or the Primary Borrower, there is no sustainable basis for such suspension or revocation, and no such suspension or revocation has been threatened by any Governmental Authority.

SECTION 5 CONDITIONS PRECEDENT

5.1 CONDITIONS TO INITIAL EXTENSION OF CREDIT. The agreement of each Lender to make the initial extension of credit requested to be made by it hereunder is subject to the satisfaction, prior to or concurrently with the making of such extension of credit on the Closing Date, of the following conditions precedent:

(a) LOAN DOCUMENTS. The Administrative Agent shall have received (i) this Agreement, executed and delivered by a duly authorized officer of the Parent and each Borrower, and (ii) the Guarantee and Collateral Agreement, executed and delivered by a duly authorized officer of the Parent, the Primary Borrower and each Subsidiary Guarantor.

(b) ACQUISITION, ETC. The following transactions shall have been consummated, or shall be consummated concurrently with the making of the initial Loans on the Closing Date and the Administrative Agent shall have received satisfactory evidence thereof (it being understood that any shortfall in the amounts required to be received by White Mountains, the Parent, the Primary Borrower or CGU pursuant to paragraphs (ii), (iii), (iv) and (v) below may be replaced with Excess Cash):

(i) The Acquisition shall have been consummated pursuant to the Acquisition Documentation; and the Acquisition Documentation shall not have been amended, supplemented, waived or otherwise modified in any material respect without the prior written consent of the Administrative Agent.

(ii) White Mountains shall have received commitments from (i) the Investor Group to purchase for cash at least \$441,000,000 of equity securities of White Mountains and (ii) Berkshire Hathaway to purchase (A) the Berkshire Preferred Stock for cash in an amount not less than \$225,000,000, and (B) for cash in an amount not less than \$75,000,000 a warrant to purchase common stock of White Mountains pursuant to the Warrant Agreement. (iii) The Parent shall have received at least \$875,000,000 PLUS the amount, if any, by which the aggregate unassigned surplus of dividend-paying first-tier Insurance Subsidiaries is less than \$950,000,000 (it being understood that such aggregate unassigned surplus shall not be less than \$900,000,000) from the proceeds of equity issued by the Parent to White Mountains and/or the Investor Group, including \$20,000,000 from the issuance of preferred stock which shall be sold to a third party investor (the "THIRD PARTY PREFERRED STOCK").

(iv) The Primary Borrower shall have received (A) at least \$875,000,000 PLUS the amount, if any, by which the aggregate unassigned surplus of dividend-paying first-tier Insurance Subsidiaries is less than \$950,000,000 (it being understood that such aggregate unassigned surplus shall not be less than \$900,000,000) as the proceeds of common stock issued to the Parent and (B) at least \$225,000,000 as the proceeds of the Berkshire Preferred Stock.

(v) CGU shall have received (A) as dividends from its Subsidiaries cash in an amount not less than \$313,000,000 and (B) cash in the amount of \$235,000,000 as proceeds of the Preclosing Transactions, and all such cash described in the foregoing clauses (A) and (B) shall be available for use to pay a portion of the purchase price of the Acquisition.

 $({\rm vi})$ The Preclosing Transactions shall have been consummated in accordance with the Stock Purchase Agreement on or prior to the Closing Date.

(c) PRO FORMA BALANCE SHEET; FINANCIAL STATEMENTS. The Lenders and the Administrative Agent shall have received (i) (A) the $\ensuremath{\mathsf{Pro}}$ Forma Balance Sheet and (B) if the Closing Date occurs after March 31, 2001, an estimated PRO FORMA balance sheet as at December 31, 2000, reflecting the events and transactions described in clauses (i) to (vi) of Section 4.1(c) (except that the amount set forth in clause (v) of Section 4.1(a) may be reduced by not more than \$50,000,000 if the Commitments have been correspondingly reduced pursuant to Section 2.12(d)), as if such events and transactions had occurred on December 31, 2000, (ii) the Acquired Assets Financial Statements, (iii) audited consolidated financial statements of White Mountains for the 1998, 1999 and, if the Closing Date occurs after March 31, 2001, 2000 fiscal years, (iv) if the Closing Date occurs prior to March 31, 2001, unaudited interim consolidated financial statements of White Mountains for the nine-month period ended on September 30, 2000, (v) audited consolidated financial statements of CGU for the 1999 and, if the Closing Date occurs after March 31, 2001, 2000 fiscal years, (vi) if the Closing Date occurs prior to March 31, 2001, unaudited interim consolidated financial statements of CGU for the nine-month period ended on September 30, 2000 and (vii) audited combined financial statements of admitted assets, liabilities and Capital and Surplus of the Insurance Subsidiaries of CGU for the 1999 and, if the Closing Date occurs after March 31, 2001, 2000 fiscal years. All such financial statements shall be prepared in accordance with GAAP or SAP, as appropriate, and shall be certified by an appropriate financial officer or accountants, as applicable. The financial statements referred to in clauses (i), (ii), (v), (vi) and (vii) above are qualified in their entirety by the Specified Exceptions.

(d) APPROVALS. All governmental and third party approvals necessary in connection with the Acquisition, the continuing operations of the Parent, each Borrower and its Subsidiaries and the transactions contemplated hereby shall have been obtained and be in full force and effect, and all applicable waiting periods shall have expired without any action being taken or threatened by any competent authority which would restrain, prevent or otherwise impose adverse conditions on the Acquisition or the financing contemplated hereby.

(e) RELATED AGREEMENTS. The Administrative Agent shall have received, with a copy for each Lender, true and correct copies, certified as to authenticity by the Primary Borrower, of the Subordinated Seller Note, the Tax Sharing Agreement and such other documents or instruments as may be reasonably requested by the Administrative Agent.

(f) FEES. The Lenders, the Arranger and the Administrative Agent shall have received all fees required to be paid, and all expenses for which invoices have been presented (including reasonable fees, disbursements and other charges of counsel to the Administrative Agent), on or before the Closing Date. All such amounts will be paid with proceeds of Loans made on the Closing Date and will be reflected in the funding instructions given by the Primary Borrower to the Administrative Agent on or before the Closing Date.

(g) BUSINESS PLAN. The Lenders and the Administrative Agent shall have received a satisfactory business plan for fiscal years 2001-2007 and a satisfactory written analysis of the business and prospects of the Primary Borrower and its Subsidiaries for the period from the Closing Date through December 31, 2007 (which business plan has been included in the Confidential Information Memorandum and is deemed satisfactory to the Administrative Agent and the Lenders).

(h) LIEN SEARCHES. The Administrative Agent shall have received the results of a recent lien search in each of the jurisdictions in which Uniform Commercial Code financing statement or other filings or recordations should be made to evidence or perfect security interests in all assets of the Loan Parties, and such search shall reveal no liens on any of the assets of the Loan Party, except for Liens permitted by Section 7.3.

(i) CLOSING CERTIFICATE. The Administrative Agent shall have received a certificate of each Loan Party, dated the Closing Date, substantially in the form of Exhibit C, with appropriate insertions and attachments.

(j) LEGAL OPINIONS. The Administrative Agent shall have received the following executed legal opinions:

(i) the legal opinion of Cravath Swaine & Moore, counsel to White Mountains, the Parent, the Primary Borrower and its Subsidiaries, and the Additional Borrower, substantially in the form of Exhibit F-1;

(ii) the legal opinion of Roger Singer, general counsel of CGU and its Subsidiaries, substantially in the form of Exhibit F-2; and

(iii) to the extent consented to by the relevant counsel, each legal opinion, if any, delivered in connection with the Acquisition Documentation, accompanied by a reliance letter in favor of the Lenders.

Each such legal opinion shall cover such other matters incident to the transactions contemplated by this Agreement as the Administrative Agent may reasonably require.

(k) PLEDGED STOCK; STOCK POWERS; ACKNOWLEDGMENT AND CONSENT. With respect to the shares of Capital Stock pledged pursuant to the Guarantee and Collateral Agreement, (i) if such shares constitute "certificated securities" within the meaning of Article 8 of the UCC, the Administrative Agent shall have received the stock certificates representing such shares, indorsed in blank by an effective indorsement or accompanied by appropriate stock powers duly executed in blank and (ii) if such shares constitute "uncertificated securities" within the meaning of Article 8 of the UCC, all actions shall have been taken that are required by Article 8 of the UCC to cause the Administrative Agent to acquire "control" of such shares within the meaning of Article 8 of the Uniform Commercial Code in effect in the State of New York. The Administrative Agent shall have received an Acknowledgment, substantially in the form of Annex II to the Guarantee and Collateral Agreement, duly executed by any issuer of Capital Stock pledged pursuant to the Guarantee and Collateral Agreement that is not itself a party to the Guarantee and Collateral Agreement.

(1) FILINGS, REGISTRATIONS AND RECORDINGS. Each document (including, without limitation, any Uniform Commercial Code financing statement) required by the Security Documents or under law or reasonably requested by the Administrative Agent to be filed, registered or recorded in order to create in favor of the Administrative Agent, for the benefit of the Lenders, a perfected Lien on the Collateral described therein shall have been filed, registered or recorded or shall have been delivered to the Administrative Agent be in proper form for filing, registration or recordation.

(m) REINSURANCE AGREEMENTS. Insurance Subsidiaries of CGU shall have entered into (i) a Reinsurance Agreement with Berkshire Hathaway (materially consistent with that described in the letter, dated July 18, 2000, from Ajit Jain of Berkshire Hathaway Group to Ray Barrette of White Mountains, together with the Reinsurance Agreement Terms attached to such letter) for not less than \$2,500,000,000 (relating to certain discontinued operations) and (ii) a Reinsurance Agreement (the terms of which shall be reasonably acceptable to the Arranger) with one of the third party reinsurers previously identified to the Arranger or a comparably rated or more highly rated third party reinsurer providing excess of loss coverage for accident year 2000 and prior with a limit of between \$550,000,000 and \$600,000,000 (relating to certain continuing operations) at a cost not to exceed \$350,000,000.

(n) STATUTORY NET WORTH. The aggregate Capital and Surplus of the Primary Borrower's Insurance Subsidiaries shall be not less than \$2,300,000,000 as of the end of the fiscal quarter most recently ended prior to the Closing Date, after giving pro forma effect to the consummation of the Acquisition and related transactions, including the Reinsurance Agreements referenced in Section 5.1(m), the increase in reserves described in Section 5.1(c) and the contribution of Folksamerica and Main Street America as a part of the Preclosing Transactions.

(o) INSURANCE SUBSIDIARY RATING. All Material Insurance Subsidiaries shall have received a financial strength rating of at least A- from A.M. Best.

5.2 CONDITIONS TO EACH EXTENSION OF CREDIT. The agreement of each Lender to make any extension of credit requested to be made by it hereunder on any date (including, without limitation, its initial extension of credit) is subject to the satisfaction of the following conditions precedent:

(a) REPRESENTATIONS AND WARRANTIES. Each of the representations and warranties made by any Loan Party in or pursuant to the Loan Documents shall be true and correct on and as of such date as if made on and as of such date, except to the extent that they expressly relate to an earlier date, in which case they shall be true and correct as of such earlier date.

(b) NO DEFAULT. No Default or Event of Default shall have occurred and be continuing on such date or after giving effect to the extensions of credit requested to be made on such date.

Each borrowing by and issuance of a Letter of Credit on behalf of any Borrower hereunder shall constitute a representation and warranty by such Borrower as of the date of such extension of credit that the conditions contained in this Section 5.2 have been satisfied.

SECTION 6 AFFIRMATIVE COVENANTS

The Parent and the Primary Borrower hereby jointly and severally agree that, from and after the Closing Date and so long as the Commitments remain in effect, any Letter of Credit remains outstanding or any Loan or other amount is owing to any Lender or the Administrative Agent hereunder, each of the Parent and the Primary Borrower shall and shall cause each of its Subsidiaries to:

6.1 FINANCIAL STATEMENTS. Furnish to the Administrative Agent (with sufficient copies for distribution by the Administrative Agent to each Lender):

(a) (i) as soon as available, but in any event within 95 days after the end of each fiscal year of the Primary Borrower subsequent to the Closing Date, a copy of the audited consolidated balance sheet of the Primary Borrower and its consolidated Subsidiaries as at the end of such year and the related audited consolidated statements of income and of cash flows for such year, setting forth in each case in comparative form the figures as of the end of and for the previous year, reported on without a "going concern" or like qualification or exception, or qualification arising out of the scope of the audit, by PricewaterhouseCoopers LLP or other independent certified public accountants of nationally recognized standing; and (ii) as soon as available, but in any event not later than 50 days after the end of each of the first three fiscal quarters of each fiscal year of the Primary Borrower subsequent to the Closing Date, the unaudited consolidated balance sheet of the Primary Borrower and its consolidated Subsidiaries as at the end of such fiscal quarter and the related unaudited consolidated statements of income and of cash flows for such fiscal quarter and the portion of the fiscal year through the end of such fiscal quarter, setting forth in each case in comparative form the figures as of the end of and for the corresponding period in the previous year, certified by a Responsible Officer of the Primary Borrower as being fairly stated in all material respects (subject to normal year-end audit adjustments);

all such financial statements, together with notes to such financial statements, to fairly present in all material respects the financial condition and income and cash flows of the subject thereof as at the dates and for the periods covered thereby, and to be prepared in accordance with GAAP applied consistently throughout the periods reflected therein and with prior periods (except as approved by such accountants or officer, as the case may be, and disclosed therein).

(b) (i) to the extent such a report is required by law to be prepared, as soon as available but not later than 85 days after the end of each fiscal year of each Material Insurance Subsidiary subsequent to the Closing Date, copies of the unaudited Annual Statement of such Material Insurance Subsidiary, certified by a Responsible Officer of such Material Insurance Subsidiary, all such statements to be prepared in accordance with SAP consistently applied throughout the periods reflected therein and, if required by the applicable Governmental Authority, audited and certified by independent certified public accountants of recognized national standing (it being understood that 150 days shall be allowed for delivery of audited statements);

(ii) as soon as available but not later than 70 days after the end of each of the first three fiscal quarters of each fiscal year of each Material Insurance Subsidiary subsequent to the Closing Date, copies of the Quarterly Statement of each of the Material Insurance Subsidiaries, certified by a Responsible Officer of such Material Insurance Subsidiary, all such statements to be prepared in accordance with SAP consistently applied through the period reflected herein;

(iii) within 15 days after being delivered to any Material Insurance Subsidiary subsequent to the Closing Date, any final Report on Examination issued by the applicable Department or the NAIC that results in material adjustments to the financial statements referred to in paragraphs (b)(i) or (b)(ii) above; and

(iv) within 100 days after the end of each fiscal year of each Material Insurance Subsidiary subsequent to the Closing Date, a copy of the "Statement of Actuarial Opinion" and "Management Discussion and Analysis" for each such Material Insurance Subsidiary which is provided to the applicable Department (or equivalent information should such Department no longer require such a statement) as to the adequacy of loss reserves of such Material Insurance Subsidiary, such opinion to be in the format prescribed by the insurance code of the state of domicile of such Material Insurance Subsidiary.

6.2 CERTIFICATES; OTHER INFORMATION. Furnish to the Administrative Agent (with sufficient copies for distribution by the Administrative Agent to each Lender), or, in the case of clause (f), to the relevant Lender:

(a) concurrently with the delivery of the financial statements referred to in Section 6.1(a)(i) (and, if such financial statements are audited, Section 6.1(b)(i)), a certificate of the independent certified public accountants reporting on such financial statements stating that in making the examination necessary therefor no knowledge was obtained of any Default or Event of Default (it being understood that such certificate may be limited in scope and qualified in accordance with customary practices of the accounting profession), except as specified in such certificate;

(b) concurrently with the delivery of any financial statements pursuant to Section 6.1, (i) a certificate of a Responsible Officer stating that, to the best of such Responsible Officer's knowledge, each Loan Party during such period has observed or performed in all material respects all of its covenants and other agreements, and satisfied every condition, contained in this Agreement and the other Loan Documents to which it is a party to be observed, performed or satisfied by it, and that such Responsible Officer has obtained no knowledge of any Default or Event of Default except as specified in such certificate and (ii) in the case of quarterly or annual financial statements, (x) a Compliance Certificate containing all information and calculations necessary for determining compliance by the Parent, the Primary Borrower and its Subsidiaries with the provisions of this Agreement referred to therein as of the last day of the fiscal quarter or fiscal year of the Primary Borrower, as the case may be, and (y) to the extent not previously disclosed to the Administrative Agent, a listing of any Intellectual Property acquired by any Loan Party since the date of the most recent list delivered pursuant to this clause (y) (or, in the case of the first such list so delivered, since the Closing Date);

(c) as soon as available, and in any event no later than 95 days after the end of each fiscal year of the Primary Borrower subsequent to the Closing Date, a consolidated budget for the following fiscal year (including a projected consolidated balance sheet of the Primary Borrower and its Subsidiaries as of the end of the following fiscal year, and the related consolidated statements of projected cash flow (cash flow to be prepared on a Parent only basis), projected changes in financial position and projected income), and, as soon as available, significant revisions, if any, of such budget and projections with respect to such fiscal year (collectively, the "PROJECTIONS"), which Projections shall in each case be accompanied by a certificate of a Responsible Officer of the Primary Borrower stating that such Projections have been prepared in good faith and are based on good faith estimates and assumptions believed by the Primary Borrower to be reasonable at the time made (it being recognized by the Lenders that such opinions, projections and forecasts as to any future event or state of affairs are not to be viewed as factual information and that actual results during the period or periods covered by any such opinion, projection or forecast may differ from the opinions and projected and forecast results); (d) within 50 days after the end of each first, second and third fiscal quarter of the Primary Borrower subsequent to the Closing Date, and within 95 days after the end of each fourth fiscal quarter of the Primary Borrower subsequent to the Closing Date, a narrative discussion and analysis of the financial condition and results of operations of the Primary Borrower and its Subsidiaries for such fiscal quarter and for the period from the beginning of the then current fiscal year to the end of such fiscal quarter, as compared to the portion of the Projections covering such periods and to the comparable periods of the previous year;

(e) within 10 days after the same are sent, copies of all financial statements and reports that the Parent or any Borrower (it being understood that White Mountains shall be required to furnish such financial statements and reports in its capacity as an Additional Borrower only after becoming an Additional Borrower hereunder) sends to the holders of any class of its debt securities or public equity securities and, within 10 days after the same are filed, copies of all financial statements and reports that the Parent or such Borrower may make to, or file with, the SEC;

(f) promptly, such additional financial and other information as any Lender may from time to time reasonably request.

6.3 PAYMENT OF OBLIGATIONS. Pay, discharge or otherwise satisfy at or before maturity or before they become delinquent, as the case may be, all its material obligations of whatever nature, except where the amount or validity thereof is currently being contested in good faith by appropriate proceedings and reserves in conformity with GAAP with respect thereto have been provided on the books of the Parent, the Primary Borrower or its Subsidiaries, as the case may be; PROVIDED, that the Borrowers may, in the ordinary course of business, extend payments on those payables if beneficial to the operation of their businesses.

6.4 CONDUCT OF BUSINESS AND MAINTENANCE OF EXISTENCE, ETC. (a) (i) Preserve, renew and keep in full force and effect its corporate existence and (ii) take all reasonable action to maintain all licenses, rights, privileges and franchises necessary or desirable in the normal conduct of its business, except, in each case, as otherwise permitted by Section 7.4 and except, in the case of clause (i) above (with respect to Subsidiaries) and clause (ii) above, to the extent that failure to do so could not reasonably be expected to have a Material Adverse Effect; and (b) comply with all Contractual Obligations and Requirements of Law, except to the extent that failure to comply therewith could not, in the aggregate, reasonably be expected to have a Material Adverse Effect.

6.5 MAINTENANCE OF PROPERTY; INSURANCE. (a) Keep all Property and systems useful and necessary in its business in good working order and condition, ordinary wear and tear excepted and (b) maintain with financially sound and reputable insurance companies (not affiliated with the Parent) insurance on all its Property in at least such amounts and against at least such risks (but including in any event public liability, product liability and business interruption) as are usually insured against in the same general area by companies engaged in the same or a similar business (it being understood that, to the extent consistent with prudent business practices of Persons carrying on a similar business in a similar location, a program of self-insurance for first and other loss layers may be utilized).

6.6 INSPECTION OF PROPERTY; BOOKS AND RECORDS; DISCUSSIONS. (a) Keep proper books of records and account in which full, true and correct entries in conformity with GAAP (or SAP as applicable) and all Requirements of Law shall be made of all dealings and transactions in relation to its business and activities and (b) upon reasonable prior notice, permit representatives of any Lender to visit and inspect any of its properties and examine and make abstracts from any of its books and records at any reasonable time and as often as may reasonably be desired and to discuss the business, operations, properties and financial and other condition of the Parent, the Primary Borrower and its Subsidiaries with officers and employees of the Parent, the Primary Borrower and its Subsidiaries and, so long as a representative of the Primary Borrower is present during such discussions (unless a Default has occurred and is continuing), with its independent certified public accountants. It is understood that (i) any information obtained by any Lender in any visit or inspection pursuant to this Section shall be subject to the confidentiality requirements of Section 10.14, (ii) the Parent and the Borrowers may impose, with respect to any Lender or any Affiliate of any Lender reasonably deemed by the Parent to be engaged significantly in a business which is directly competitive with any material business of the Parent, the Borrowers and their Subsidiaries, reasonable restrictions on access to proprietary information of the Parent, the Borrowers and their Subsidiaries and (iii) so long as no Default shall be continuing, the Lenders will coordinate their visits through the Administrative Agent with a view to preventing the visits provided for by this Section from becoming unreasonably burdensome to the Parent, the Borrowers and their Subsidiaries

6.7 NOTICES. Promptly give notice to the Administrative Agent (it being agreed that the Administrative Agent shall, upon receipt of such notice, notify each Lender thereof) of:

(a) the occurrence of any Default or Event of Default;

(b) any (i) default or event of default under any Contractual Obligation of the Parent, the Primary Borrower or any of its Subsidiaries or (ii) litigation, investigation or proceeding which may exist at any time between the Parent, the Primary Borrower or any of its Subsidiaries and any Governmental Authority, that in either case, if not cured or if adversely determined, as the case may be, could reasonably be expected to have a Material Adverse Effect;

(c) (i) any litigation or proceeding affecting the Parent, the Primary Borrower or any of its Subsidiaries (other than claims-related litigation involving an Insurance Subsidiary) in which the amount involved is \$20,000,000 or more and not covered by insurance or in which material injunctive or similar relief is sought and (ii) any claims-related litigation affecting any Insurance Subsidiary in which the exposure (net of reinsurance arrangements) of any Insurance Subsidiary is \$20,000,000 or more and for which it reasonably could be expected that liability would result;

(d) the following events, as soon as possible and, in any event, within 30 days after a Responsible Officer of the Primary Borrower knows thereof: (i) the occurrence of any Reportable Event with respect to any Plan, a failure to make any required contribution to a Plan, the creation of any Lien in favor of the PBGC or a Plan or any withdrawal from, or the termination, Reorganization or Insolvency of, any Multiemployer Plan or (ii) the institution of proceedings or the taking of any other action by the PBGC or the Primary Borrower or any Commonly Controlled Entity or any Multiemployer Plan with respect to the withdrawal from, or the termination, Reorganization or Insolvency of, any Plan; and

(e) any development or event that has had a Material Adverse $\mathsf{Effect.}$

Each notice pursuant to this Section shall be accompanied by a statement of a Responsible Officer setting forth details of the occurrence referred to therein and stating what action the Parent, the Primary Borrower or the relevant Subsidiary proposes to take with respect thereto.

6.8 ENVIRONMENTAL LAWS. (a) Comply in all material respects with, and use reasonable efforts to ensure compliance in all material respects by all tenants and subtenants, if any, with, all applicable Environmental Laws, and obtain and comply in all material respects with and maintain, and use reasonable efforts to ensure that all tenants and subtenants obtain and comply in all material respects with and maintain, any and all licenses, approvals, notifications, registrations or permits required by applicable Environmental Laws. For purposes of this section 6.8(a), noncompliance by the Parent, the Primary Borrower and its Subsidiaries with any applicable Environmental Law shall be deemed not to constitute a breach of this section provided that, (i) upon learning of any actual or suspected noncompliance, each of them shall promptly undertake reasonable efforts to achieve compliance, and (ii) in any case, such non-compliance and any other noncompliance with Environmental Law , individually or in the aggregate could not reasonably be expected to have a Material Adverse Effect.

(b) Conduct and complete all investigations, studies, sampling and testing, and all remedial, removal and other actions required under Environmental Laws and promptly comply in all material respects with all orders and directives of all Governmental Authorities regarding Environmental Laws, other than such orders or directions as have been properly challenged in good faith; PROVIDED, that in any case such orders and directives could not reasonably be expected to have a Material Adverse Effect.

(c) It is understood that Section 6.8 shall not apply to claims under policies issued as part of the insurance business of the Borrower or its Insurance Subsidiaries.

6.9 INTEREST RATE PROTECTION. In the case of the Primary Borrower, within 150 days after the Closing Date, enter into Hedge Agreements to the extent necessary to provide that at least 50% of the aggregate principal amount of the Term Loans is subject to either a fixed interest rate or interest rate protection for a period of not less than three years, which Hedge Agreements shall have terms and conditions reasonably satisfactory to the Administrative Agent.

6.10 ADDITIONAL COLLATERAL, ETC. (a) With respect to any assets of the type constituting Collateral under the Guarantee and Collateral Agreement acquired after the Closing Date by the Parent, the Primary Borrower or any of its Non-Insurance Subsidiaries (other than assets acquired by an Excluded Foreign Subsidiary) as to which the Administrative Agent, for the benefit of the Lenders, does not have a perfected Lien, promptly (i) execute and deliver to the Administrative Agent such amendments to the Guarantee and Collateral Agreement or such other documents as the Administrative Agent reasonably deems necessary to grant to the Administrative Agent, for the benefit of the Lenders, a security interest in such assets and (ii)

take all actions necessary to grant to the Administrative Agent, for the benefit of the Lenders, a perfected security interest in such assets, including, without limitation, the filing of Uniform Commercial Code financing statements in such jurisdictions as may be required by the Guarantee and Collateral Agreement or by law.

(b) With respect to any new Subsidiary (other than an Excluded Foreign Subsidiary) created or acquired after the Closing Date (which, for the purposes of this paragraph, shall include any existing Subsidiary that ceases to be an Excluded Foreign Subsidiary), by the Parent, the Primary Borrower or any of its Non-Insurance Subsidiaries, promptly (i) execute and deliver to the Administrative Agent such amendments to the Guarantee and Collateral Agreement as the Administrative Agent reasonably deems necessary to grant to the Administrative Agent, for the benefit of the Lenders, a perfected security interest in all shares of Capital Stock of such new Subsidiary that are owned by the Parent, the Primary Borrower or any of its Subsidiaries, (ii) take all actions required by Article 8 of the UCC to cause the Administrative Agent to acquire "control" of such shares of Capital Stock within the meaning of Article 8 of the UCC, (iii) cause such new Subsidiary (unless such Subsidiary is an Insurance Subsidiary) (A) to become a party to the Guarantee and Collateral Agreement and (B) to take such actions necessary to grant to the Administrative Agent for the benefit of the Lenders a perfected security interest in the Collateral described in the Guarantee and Collateral Agreement with respect to such new Subsidiary, and (iv) if requested by the Administrative Agent, deliver to the Administrative Agent legal opinions relating to the matters described above, which opinions shall be in form and substance, and from counsel, reasonably satisfactory to the Administrative Agent.

(c) With respect to any new Excluded Foreign Subsidiary created or acquired after the Closing Date by the Parent, the Primary Borrower or any of its Non-Insurance Subsidiaries (other than any Excluded Foreign Subsidiaries), promptly (i) execute and deliver to the Administrative Agent such amendments to the Guarantee and Collateral Agreement or such other documents as the Administrative Agent reasonably deems necessary in order to grant to the Administrative Agent, for the benefit of the Lenders, a perfected security interest in the Capital Stock of such new Subsidiary that is owned by the Parent, the Primary Borrower or any of its Subsidiaries (other than any Excluded Foreign Subsidiaries), (provided that in no event shall more than 66% of the total outstanding Capital Stock of any such new Excluded Foreign Subsidiary be required to be so pledged), (ii) take all actions required by Article 8 of the UCC to cause the Administrative Agent to acquire "control" of such shares of Capital Stock within the meaning of Article 8 of the UCC or take such other action as may be necessary or, in the reasonable opinion of the Administrative Agent, desirable to perfect the Lien of the Administrative Agent thereon, and (iii) if requested by the Administrative Agent, deliver to the Administrative Agent legal opinions relating to the matters described above, which opinions shall be in form and substance, and from counsel, reasonably satisfactory to the Administrative Agent.

(d) For avoidance of doubt, it is understood that no Insurance Subsidiary shall be required to become a guarantor under the Guarantee and Collateral Agreement or to create a security interest in any of its assets to secure the obligations of the Borrowers hereunder. It is understood that perfection actions with respect to Property becoming Collateral after the Closing Date will be consistent with actions required by Section 5.1 with respect to Collateral of the same type on the Closing Date, unless any change in law or customary practices require any additional or different actions.

6.11 SUBORDINATED SELLER NOTE. In the case of the Parent or the Primary Borrower, on or before the date which is eighteen months after the Closing Date, either (i) purchase the Subordinated Seller Note (to the extent not prepaid pursuant to Section 7.6(d)) for common or preferred stock of White Mountains or the Parent or (ii) refinance the Subordinated Seller Note (to the extent not prepaid pursuant to Section 7.6(d)) with (a) the proceeds of the issuance of common or preferred stock of the Parent or White Mountains or (b) subordinated Indebtedness of the Parent or the Primary Borrower permitted pursuant to Section 7.2, the maturity date of which shall be later than the date which is six months after the final maturity date of the Term Loans and the terms and conditions of which shall be reasonably satisfactory to the Required Lenders.

6.12 TAX SHARING AGREEMENT. In the case of the Primary Borrower, within one month following the Closing Date, enter into with the Parent, and each Subsidiary of the Primary Borrower, and file with each appropriate Department, a tax sharing agreement (the "TAX SHARING AGREEMENT") reasonably satisfactory in form and substance to the Administrative Agent.

SECTION 7 NEGATIVE COVENANTS

The Parent and the Primary Borrower hereby jointly and severally agree that, from and after the Closing Date and so long as the Commitments remain in effect, any Letter of Credit remains outstanding or any Loan or other amount is owing to any Lender or the Administrative Agent hereunder, each of the Parent and the Primary Borrower shall not, and shall not permit any of its Subsidiaries to, directly or indirectly:

7.1 FINANCIAL CONDITION COVENANTS.

(a) INTEREST COVERAGE RATIO. Permit the Interest Coverage Ratio for any period of four consecutive fiscal quarters of the Primary Borrower ending on any date set forth below to be less than the ratio set forth below opposite such date:

DATE	INTEREST COVERAGE RATIO
June 30, 2001	2.25:1.00
September 30, 2001	2.25:1.00
December 31, 2001	2.25:1.00
March 31, 2002	2,25;1,00
June 30, 2002	2.50:1.00
September 30, 2002	2.50:1.00
December 31, 2002	2.50:1.00
March 31, 2003	2.50:1.00
June 30, 2003	2.75:1.00
September 30, 2003	2.75:1.00
December 31, 2003	2.75:1.00
March 31, 2004	2.75:1.00
June 30, 2004 and thereafter	3.00:1.00

(b) FIXED CHARGE COVERAGE RATIO. Permit the Fixed Charge Coverage Ratio for any period of four consecutive fiscal quarters of the Primary Borrower ending on any date set forth below to be less than the ratio set forth below opposite such date:

DATE	FIXED CHARGE COVERAGE RATIO
June 30, 2001	1.20:1.00
September 30, 2001	1.20:1.00
December 31, 2001	1.20:1.00
March 31, 2002	1.20:1.00
June 30, 2002	1.20:1.00
September 30, 2002	1.20:1.00
December 31, 2002	1.20:1.00
March 31, 2003	1.20:1.00
June 30, 2003	1.20:1.00
September 30, 2003	1.20:1.00
December 31, 2003	1.20:1.00
March 31, 2004 June 30, 2004 and thereafter	1.20:1.00

(c) LEVERAGE RATIO. Permit the Leverage Ratio as at the end of any fiscal quarter of the Primary Borrower set forth below to exceed the ratio set forth below opposite such fiscal quarter:

FISCAL QUARTER	LEVERAGE RATIO
June 30, 2001	39%
September 30, 2001	39%
December 31, 2001	39%
March 31, 2002	39%
June 30, 2002	35%
September 30, 2002	35%
December 31, 2002	35%
March 31, 2003	35%
June 30, 2003	30%
September 30, 2003	30%
December 31, 2003	30%
March 31, 2004 June 30, 2004 and thereafter	30% 25%

(d) MAINTENANCE OF NET WORTH. Permit Consolidated Net Worth on any date of determination thereof to be less than the sum of (without duplication) (i) \$1,630,000,000, (ii) 80% of the Net Proceeds of the issuance or sale of equity securities by the Primary Borrower consummated after the Closing Date or any capital contribution made to the Primary Borrower after the Closing Date by any holder of its common stock and (iii) 50% of positive Consolidated Net Income of the Primary Borrower for each fiscal quarter ending after the Closing Date and on or prior to such date of determination, MINUS 100% of cash distributions and dividends paid, or shares repurchased, by the Primary Borrower after the Closing Date and on or prior to such date of determination, as permitted by Section 7.6.

(e) MAINTENANCE OF CAPITAL AND SURPLUS. Permit the combined Capital and Surplus of the Insurance Subsidiaries on any date of determination thereof to be less than the sum of (without duplication) (i) \$2,140,000,000, (ii) 80% of the Net Proceeds of the issuance or sale of equity securities by the Insurance Subsidiaries consummated after the Closing Date or any capital contribution made to the Insurance Subsidiaries after the Closing Date by any holder of common stock thereof and (iii) 50% of positive combined Consolidated Net Income of the Insurance Subsidiaries for each fiscal quarter ending after the Closing Date and on or prior to such date of determination, MINUS (x) 100% of cash distributions and dividends paid by the Insurance Subsidiaries after the Closing Date and on or prior to such date of determination and (y) the Capital and Surplus of any Insurance Subsidiary Disposed of after the Closing Date and prior to the date of such determination.

(f) RISK BASED CAPITAL RATIO. Permit the Risk Based Capital Ratio at the end of any fiscal year (i) of any of any Material Insurance Subsidiary except Folksamerica Reinsurance Company to be less than 300% or (ii) of Folksamerica Reinsurance Company to be less than (x) 250% with respect to the fiscal years ending December 31, 2001 and December 31, 2002 and (y) 300% with respect to the fiscal year ending December 31, 2003 and any fiscal year thereafter.

7.2 LIMITATION ON INDEBTEDNESS AND ISSUANCE OF PREFERRED STOCK. Create, incur, assume or suffer to exist any Indebtedness or, in the case of any Subsidiary, issue any preferred stock, except:

(a) Indebtedness of any Loan Party pursuant to any Loan Document;

(b) Indebtedness of the Parent or the Primary Borrower to any Subsidiary and of any Subsidiary Guarantor to the Parent, the Primary Borrower or any other Subsidiary;

(c) Indebtedness (including, without limitation, Capital Lease Obligations) secured by Liens permitted by Section 7.3(g) in an aggregate principal amount not to exceed \$20,000,000 at any one time outstanding; (d) Indebtedness of CGU Insurance Company outstanding on the date hereof (except for intercompany indebtedness to be repaid in connection with the consummation of the Acquisition) in an aggregate principal amount not to exceed \$30,750,000;

(e) Indebtedness outstanding on the date hereof and listed on Schedule 7.2(e) (pro forma for the Acquisition) and any refinancings, refundings, renewals or extensions thereof (without any increase in the principal amount thereof or any shortening of the maturity of any principal amount thereof);

(f) Indebtedness of Folksamerica in respect of the Folksamerica Promissory Note;

(g) Guarantee Obligations made in the ordinary course of business by the Parent, the Primary Borrower or any of its Subsidiaries of obligations of the Primary Borrower or any Subsidiary Guarantor;

(h) Indebtedness of the Parent in respect of the Subordinated Seller Note in an aggregate principal amount not to exceed \$260,000,000 and any refinancing thereof permitted pursuant to Section 6.11, PROVIDED, the maturity date of such refinancing shall be no earlier than the date that is six months after the final maturity date of the Term Loans;

(i) Indebtedness of the Primary Borrower in respect of any refinancing of the Subordinated Seller Note permitted pursuant to Section 6.11, PROVIDED, that (i) the maturity date of such Indebtedness shall be no earlier than the date that is six months after the final maturity date of the Term Loans, (ii) the interest rate with respect to such Indebtedness shall not be greater than 13% per annum and (iii) the terms and conditions of such Indebtedness shall be reasonably satisfactory to the Required Lenders (it being understood and agreed that any Indebtedness permitted by this Section 7.2(i) shall not be considered as equity for the purposes of calculating compliance with the covenants set forth in Section 7.1);

(j) Guarantee Obligations of the Primary Borrower in respect of loans to officers and directors of the Parent, the Primary Borrower, White Mountains or any of their respective Subsidiaries in an aggregate principal amount not exceeding \$15,000,000;

(k) Indebtedness in respect of Surplus Debentures in an aggregate principal amount not to exceed \$200,000,000 during the term of this Agreement issued by any Insurance Subsidiaries;

(1) unsecured Indebtedness of the Parent or the Primary Borrower in an aggregate principal amount not to exceed \$400,000,000, the maturity date of which shall be later than the date that is six months after the final maturity date of the Term Loans and the interest rate with respect to which shall not be greater than 13% per annum, PROVIDED that, unless the Required Prepayment Lenders and the Primary Borrower shall otherwise agree, the Net Proceeds of such Indebtedness are used immediately upon receipt thereof to prepay the Term Loans (and after the Term Loans are repaid in full, to permanently reduce the Revolving Credit Commitments); (m) Hedge Agreements in respect of Indebtedness otherwise permitted hereby that bears interest at a floating rate, so long as such Hedge Agreements are not entered into for speculative purposes;

(n) short-term Indebtedness of Insurance Subsidiaries to provide short-term liquidity to facilitate claims payments in the event of catastrophes, in an aggregate principal amount not to exceed \$400,000,000, which is secured by the assets of Insurance Subsidiaries;

(o) Indebtedness of any Insurance Subsidiary incurred in the ordinary course of its business and letters of credit issued for the account of any Insurance Subsidiary in the ordinary course of its business;

(p) Indebtedness of Folksamerica in an aggregate principal amount of up to \$25,000,000, the proceeds of which are used to finance acquisitions;

(q) Indebtedness of the Parent, the Primary Borrower or any Subsidiary Guarantor to White Mountains in an aggregate principal amount not to exceed 10,000,000; and

(r) additional unsecured Indebtedness of the Primary Borrower or any of its Subsidiaries in an aggregate principal amount (for the Primary Borrower and all Subsidiaries) not to exceed \$110,000,000 at any one time outstanding, not more than \$15,000,000 of which may mature prior to the final maturity of the Term Loans.

 $7.3\ \rm LIMITATION\ \rm ON\ \rm LIENS.$ Create, incur, assume or suffer to exist any Lien upon any of its Property, whether now owned or hereafter acquired, except for:

(a) Liens for taxes not yet due or which are being contested in good faith by appropriate proceedings, PROVIDED that adequate reserves with respect thereto are maintained on the books of the Primary Borrower or its Subsidiaries, as the case may be, in conformity with GAAP;

(b) carriers', warehousemen's, mechanics', materialmen's, repairmen's or other like Liens arising in the ordinary course of business that secure payments that are not more than 60 days delinquent in accordance with their terms or that are being contested in good faith by appropriate proceedings;

(c) pledges or deposits in connection with workers' compensation, unemployment insurance and other social security legislation and deposits securing liability to insurance carriers under insurance or self-insurance arrangements;

(d) deposits to secure the performance of letters of credit, bids, trade contracts (other than for borrowed money), leases, statutory obligations, surety and appeal bonds, performance bonds and other obligations of a like nature incurred in the ordinary course of business; (e) easements, rights-of-way, restrictions and other similar encumbrances that, in the aggregate, are not substantial in amount and which do not in any case materially detract from the value of the Property subject thereto or materially interfere with the ordinary conduct of the business of the Primary Borrower or any of its Subsidiaries;

(f) Liens in existence on the date hereof listed on Schedule 7.3(f), securing Indebtedness permitted by Section 7.2(e), PROVIDED that no such Lien is spread to cover any additional Property after the Closing Date and that the amount of Indebtedness secured thereby is not increased;

(g) Liens securing Indebtedness of the Primary Borrower or any other Subsidiary incurred pursuant to Section 7.2(c) to finance the acquisition of fixed or capital assets, PROVIDED that (i) such Liens shall be created substantially simultaneously with the acquisition of such fixed or capital assets, (ii) such Liens do not at any time encumber any Property other than the Property financed by such Indebtedness, or (iii) the amount of Indebtedness secured thereby is not increased;

(h) Liens created pursuant to the Security Documents;

(i) any interest or title of a lessor under any lease entered into by the Primary Borrower or any other Subsidiary in the ordinary course of its business and covering only the assets so leased;

(j) Liens on certain real property securing Indebtedness permitted by Section 7.2(d);

(k) Liens on the property of assets of a corporation that becomes a Subsidiary after the date hereof securing Indebtedness otherwise permitted, PROVIDED that (i) such Liens existed at the time such corporation became a Subsidiary and were not created in anticipation thereof, (ii) any such Lien is not expanded to cover any property or assets of such corporation after the time such corporation becomes a Subsidiary (other than after-acquired title in or on such property and proceeds of the existing collateral in accordance with the instrument creating such Lien), and (iii) the amount of Indebtedness secured thereby is not increased;

 (1) licenses, leases or subleases permitted hereunder granted to other Persons not interfering in any material respect in the business of the Primary Borrower or any of its Subsidiaries;

(m) attachment or judgment Liens in respect of judgments or decrees that have not been vacated, discharged, stayed or bonded pending appeal within 60 days from the entry thereof and, in addition, attachment or judgment Liens in an aggregate amount outstanding at any one time not in excess of \$5,000,000 (not paid or fully covered by insurance as to which the relevant insurance company has acknowledged in writing coverage above applicable deductibles); (n) Liens arising from precautionary U.C.C. financing statement filings with respect to operating leases or consignment arrangements entered into by the Primary Borrower or any of its Subsidiaries in the ordinary course of business;

(o) Liens in favor of a banking institution arising by operation of law encumbering deposits (including the right of set-off) held by such banking institutions incurred in the ordinary course of business and that are within the general parameters customary in the banking industry;

(p) Liens on assets of an Insurance Subsidiary securing Indebtedness of such Insurance Subsidiary permitted by Section 7.2(n); and

(q) Liens not otherwise permitted by this Section 7.3 so long as neither (i) the aggregate outstanding principal amount of the obligations secured thereby nor (ii) the aggregate fair market value (determined, in the case of each such Lien, as of the date such Lien is incurred) of the assets subject thereto exceeds (as to the Primary Borrower and all Subsidiaries) \$10,000,000 at any one time.

7.4 LIMITATION ON FUNDAMENTAL CHANGES. Enter into any merger, consolidation or amalgamation, or liquidate, wind up or dissolve itself (or suffer any liquidation or dissolution), or Dispose of all or substantially all of its Property or business, except that:

> (a) any Non-Insurance Subsidiary of the Primary Borrower may be merged, consolidated with or liquidated into the Primary Borrower (PROVIDED that the Primary Borrower shall be the continuing or surviving corporation) or with or into any Subsidiary Guarantor (PROVIDED that (i) the Subsidiary Guarantor shall be the continuing or surviving corporation or (ii) simultaneously with such transaction, the continuing or surviving corporation shall become a Subsidiary Guarantor and the Primary Borrower shall comply with Section 6.10 in connection therewith);

(b) any Insurance Subsidiary may be merged, consolidated with or liquidated into any other Insurance Subsidiary, and any Subsidiary of any Insurance Subsidiary may be merged, consolidated with or liquidated into any other Subsidiary of such Insurance Subsidiary;

(c) any Subsidiary of the Primary Borrower may Dispose of any or all of its assets (upon voluntary liquidation or otherwise) to the Primary Borrower or any Subsidiary Guarantor; and

(d) any Insurance Subsidiary may Dispose of any or all of its assets (upon voluntary liquidation or otherwise) to any other Insurance Subsidiary.

7.5 LIMITATION ON DISPOSITION OF PROPERTY. Dispose of any of its Property (including, without limitation, receivables and leasehold interests), whether now owned or hereafter acquired, or, in the case of any Subsidiary, issue or sell any shares of such Subsidiary's Capital Stock to any Person, except: (a) the Disposition of obsolete, condemned, surplus or worn out property in the ordinary course of business;

(b) the sale of inventory in the ordinary course of business; the Disposition of Cash or Cash Equivalents for fair value; the license of Intellectual Property in the ordinary course of business; leases or subleases not materially interfering with the ordinary conduct of business;

(c) Dispositions permitted by Section 7.4(c) and (d) and Section 7.10;

(d) the sale or issuance or contribution of any Subsidiary's Capital Stock to the Parent, the Primary Borrower or any Subsidiary Guarantor;

(e) the sale or issuance or contribution of any Insurance Subsidiary's Capital Stock to any other Insurance Subsidiary;

(f) the Disposition by any Insurance Subsidiary in the ordinary course of business of assets held in its investment portfolio, and factoring of receivables;

(g) the Disposition of any Investment made and permitted pursuant to Section 7.7, except the Investment permitted by Section 7.7(f).

(h) the Disposition in any fiscal year of the Primary Borrower of any non-insurance businesses or assets having a fair market value not to exceed \$25,000,000 in the aggregate;

(i) the Disposition of any insurance businesses or assets, or insurance related businesses or assets; PROVIDED, that in any fiscal year the Primary Borrower and its Subsidiaries shall not Dispose of, in the aggregate, either (x) insurance businesses or assets, or insurance-related businesses or assets, representing 25% of combined Capital and Surplus of the Insurance Subsidiaries as of the end of the prior fiscal year (it being understood that any Disposition by an Insurance Subsidiary of Capital and Surplus shall be included in the calculation of the aggregate amount of Capital and Surplus Disposed of notwithstanding that such Insurance Subsidiary retains the proceeds of such Disposition as Capital and Surplus) or (y) insurance businesses or assets, or insurance-related businesses or assets, contributing 25% of consolidated net written premium as of the end of the prior fiscal year (excluding the impact of the Reinsurance Agreement referenced in Section 5.1(m) in the year executed);

(j) any Recovery Event, PROVIDED, that the requirements of Section 2.12(b) are complied with in connection therewith.

7.6 LIMITATION ON RESTRICTED PAYMENTS. Declare or pay any dividend on, or make any payment on account of, or set apart assets for a sinking or other analogous fund for, the purchase, redemption, defeasance, retirement or other acquisition of, any Capital Stock of the Parent, the Primary Borrower or any Subsidiary, whether now or hereafter outstanding, or make any other distribution in respect thereof, either directly or indirectly, whether in cash or property or in obligations of the Parent, the Primary Borrower or any Subsidiary, or enter into any derivatives or other transaction with any financial institution, commodities or stock exchange or clearinghouse (a "DERIVATIVES COUNTERPARTY") obligating the Parent, the Primary Borrower or any Subsidiary to make payments to such Derivatives Counterparty as a result of any change in market value of any such Capital Stock (collectively, "RESTRICTED PAYMENTS"), except that:

> (a) any Subsidiary may make Restricted Payments to the Primary Borrower or any Subsidiary Guarantor and any Insurance Subsidiary may make Restricted Payments to any other Insurance Subsidiary;

(b) the Parent may make Restricted Payments in the form of common stock of the Parent;

(c) the Primary Borrower may redeem the Berkshire Preferred Stock as permitted pursuant to Section 7.8;

(d) the Primary Borrower may pay a dividend to the Parent in an amount equal to the proceeds of any issuance of Indebtedness used for the purpose of refinancing the Subordinated Seller Note as permitted pursuant to Section 6.11; PROVIDED that the amount of such dividend received by the Parent is used to repay the Subordinated Seller Note; and

(e) so long as no Default or Event of Default shall have occurred and be continuing, (i) the Primary Borrower may pay dividends at the stated rate on the Berkshire Preferred Stock (in any event in an aggregate amount not to exceed \$40,000,000 in any fiscal year), (ii) the Primary Borrower may make Restricted Payments to the Parent, and the Parent may make Restricted Payments to its shareholders, in each case in an aggregate amount not to exceed \$20,000,000 in any fiscal year, (iii) if on the Closing Date there is Excess Cash, the Primary Borrower may, at any time prior to the date which is eighteen months after the Closing Date, pay a dividend to the Parent in an amount equal to the lesser of (x) the amount of such Excess Cash (less the amount, if any, of Excess Cash used on the Closing Date to replace any shortfall in other amounts pursuant to Section 5.1(b)) and (y) \$45,000,000, PROVIDED, that the amount of such dividend received by the Parent is used to prepay the Subordinated Seller Note and/or defease or redeem the Third Party Preferred Stock and (iv) the Primary Borrower may pay a dividend to the Parent in an amount of up to \$20,000,000 with the proceeds of Indebtedness issued pursuant to Section 7.2(u), such dividend to be used by the Parent solely to defease or redeem the Third Party Preferred Stock.

7.7 LIMITATION ON INVESTMENTS. Make any advance, loan, extension of credit (by way of guaranty or otherwise) or capital contribution to, or purchase any Capital Stock, bonds, notes, debentures or other debt securities of, or any assets constituting an ongoing business from, or make any other investment in, any other Person (all of the foregoing, "Investments"), except:

(a) extensions of trade credit in the ordinary course of business;

(b) investments in Cash Equivalents and other obligations, instruments or securities guaranteed or backed by the United States Government which are comparable in quality to Cash Equivalents and have maturities of not longer than five years; (c) investments in short-term money market investments denominated in currencies other than Dollars and comparable in quality and maturity to investments constituting Cash Equivalents, and investments in other obligations or instruments which, although not guaranteed or backed by the United States Government, are comparable in quality to Cash Equivalents and have maturities of not longer than five years;

(d) investments arising in connection with the incurrence of Indebtedness permitted by Section 7.2(b), (e), (g), (j), (m) and (q);

(e) (i) loans and advances to officers, directors and employees of the Parent, White Mountains and its Subsidiaries, the Primary Borrower or any Subsidiaries of the Primary Borrower in the ordinary course of business (including, without limitation, for travel, entertainment and relocation expenses) in an aggregate amount for the Parent, White Mountains and its Subsidiaries, the Primary Borrower and Subsidiaries of the Primary Borrower not to exceed \$15,000,000 at any one time outstanding and (ii) investment of assets (which support deferred compensation liabilities of Subsidiaries of White Mountains and CGU) in White Mountains, not to exceed \$47,500,000;

(f) the Acquisition;

(g) investments in assets useful in the Primary Borrower's business made by the Primary Borrower or any of its Subsidiaries with the proceeds of any Reinvestment Deferred Amount;

(h) investments (other than those relating to the incurrence of Indebtedness permitted by Section 7.2) by the Parent, the Primary Borrower or any of its Subsidiaries in the Primary Borrower or any Person that, prior to such Investment, is a Subsidiary Guarantor;

(i) loans or guarantees of loans to insurance agents in the ordinary course of business and loans to insurance agents outside the ordinary course of business so long as the aggregate principal amount of non-ordinary course loans does not exceed \$25,000,000 in the aggregate at any time;

(j) investments in, and additional capital contributions to, Insurance Subsidiaries of the Primary Borrower by the Primary Borrower and the Subsidiaries of the Primary Borrower;

(k) investments by any Insurance Subsidiary in the ordinary course of business in investment portfolio investments;

(1) investment of the proceeds of any Disposition permitted by Section 7.5(i) in insurance-related businesses;

(m) investments by Folksamerica in an aggregate amount not to exceed \$100,000,000 in the aggregate for any fiscal year in insurance-related businesses; (n) acquisitions by the Primary Borrower or any of its
 Subsidiaries of insurance-related businesses in an aggregate amount
 (valued at cost at any time invested) not to exceed \$200,000,000 during
 the term of this Agreement;

(o) in addition to Investments otherwise expressly permitted by this Section, Investments by the Primary Borrower or any of its Subsidiaries of any type in an aggregate amount (valued at cost at any time invested) not to exceed \$25,000,000 during the term of this Agreement.

7.8 LIMITATION ON OPTIONAL PAYMENTS, REDEMPTIONS AND MODIFICATIONS OF $\ensuremath{\mathsf{DEBT}}$ INSTRUMENTS AND PREFERRED STOCK, ETC. (a) Make or offer to make any optional or voluntary payment, prepayment, repurchase or redemption of, or otherwise voluntarily or optionally defease, (i) the Berkshire Preferred Stock (except with the proceeds received from the exercise of warrants pursuant to the Warrant Agreement), (ii) the Third Party Preferred Stock (other than any such redemption or defeasance with the proceeds of dividends received from the Primary Borrower as permitted by Sections 7.6(e)(iii) and 7.6(e)(iv) and/or the proceeds of common equity issued by the Parent to White Mountains) or (iii) the Seller Subordinated Note (other than any such prepayment of the Seller Subordinated Note with proceeds of dividends from the Primary Borrower permitted by Section 7.6(e)(iii), or with the proceeds of common equity contributed to the Parent by White Mountains, and other than any refinancing of the Seller Subordinated Note permitted by Sections 7.2(h) and 7.2(i), or segregate funds for any such payment, prepayment, repurchase, redemption or defeasance, (b) amend, modify or otherwise change, or consent or agree to any amendment, modification, waiver or other change to, any of the terms of the Berkshire Preferred Stock, the Third Party Preferred Stock or the Seller Subordinated Note (other than any such amendment, modification, waiver or other change which (i) would extend the maturity or reduce the amount of any payment of principal thereof, reduce the rate or extend the date for payment of interest or dividend thereon or relax any covenant or other restriction applicable to the Parent, the Primary Borrower or any of its Subsidiaries and (ii) does not involve the payment of a consent fee), (c) make any amendment of the Tax Sharing Agreement that would make the terms thereof materially less favorable to the Primary Borrower or (d) amend its certificate of incorporation in any manner determined by the Administrative Agent to be materially adverse to the Lenders.

7.9 LIMITATION ON TRANSACTIONS WITH AFFILIATES. Enter into any transaction, including, without limitation, any purchase, sale, lease or exchange of Property, the rendering of any service or the payment of any management, advisory or similar fees, with any Affiliate (other than the Parent, the Primary Borrower or any Subsidiary Guarantor) unless such transaction is (a) otherwise permitted under this Agreement, (b) in the ordinary course of business of the Parent, the Primary Borrower or such Subsidiary, as the case may be, and (c) upon fair and reasonable terms no less favorable to the Parent, the Primary Borrower or as the case may be, than it would obtain in a comparable arm's length transaction with a Person that is not an Affiliate.

7.10 LIMITATION ON SALES AND LEASEBACKS. Enter into any arrangement with any Person providing for the leasing by the Parent, the Primary Borrower or any Non-Insurance Subsidiary of real or personal property which has been or is to be sold or transferred by the Parent, the Primary Borrower or such Subsidiary to such Person or to any other Person to whom

80

funds have been or are to be advanced by such Person on the security of such property or rental obligations of the Parent, the Primary Borrower or such Subsidiary; PROVIDED, that sale and leaseback transactions yielding proceeds in an aggregate amount up to \$10,000,000 may be consummated while this agreement is in effect, and the proceeds received from such transactions shall not be required to be applied to the mandatory prepayment of the Loans and permanent reduction of the Revolving Credit Facility.

7.11 LIMITATION ON CHANGES IN FISCAL PERIODS. Permit the fiscal year of the Primary Borrower to end on a day other than December 31 or change the Primary Borrower's method of determining fiscal quarters.

7.12 LIMITATION ON NEGATIVE PLEDGE CLAUSES. Enter into or suffer to exist or become effective any agreement that prohibits or limits the ability of the Parent, the Primary Borrower or any of its Subsidiaries to create, incur, assume or suffer to exist any Lien upon any of its Property or revenues, whether now owned or hereafter acquired, to secure the Obligations or, in the case of any guarantor, its obligations under the Guarantee and Collateral Agreement, other than (a) this Agreement and the other Loan Documents, (b) any agreements governing Investments in joint ventures, (c) any agreement containing customary provisions prohibiting Liens upon certain intangible assets including licenses and Intellectual Property and (d) any agreements governing any purchase money Liens or Capital Lease Obligations otherwise permitted hereby (in which case, any prohibition or limitation shall only be effective against the assets financed thereby).

7.13 LIMITATION ON LINES OF BUSINESS. Enter into any business, either directly or through any Subsidiary, except for those businesses in which the Primary Borrower and its Subsidiaries are engaged on the date of this Agreement (after giving effect to the Acquisition) or that are reasonably related thereto.

SECTION 8 EVENTS OF DEFAULT

If any of the following events shall occur and be continuing:

(a) Any Borrower shall fail to pay any principal of any Loan made to such Borrower or Reimbursement Obligation owing by such Borrower when due in accordance with the terms hereof; or any Borrower shall fail to pay any interest on any Loan made to such Borrower or Reimbursement Obligation owing to such Borrower, or any other amount payable by such Borrower hereunder or under any other Loan Document, within five days after any such interest or other amount becomes due in accordance with the terms hereof; or

(b) Any representation or warranty made or deemed made by any Loan Party herein or in any other Loan Document or that is contained in any certificate, document or financial or other statement furnished by it at any time under or in connection with this Agreement or any such other Loan Document shall prove to have been inaccurate in any material respect on or as of the date made or deemed made or furnished; or (c) Any Loan Party shall default in the observance or performance of any agreement contained in Section 6.4(a)(i) (with respect to the Parent and the Primary Borrower only), Section 6.4(a)(ii), Section 6.7(a) or Section 7, or in Section 5 of the Guarantee and Collateral Agreement; or

(d) Any Loan Party shall default in the observance or performance of any other agreement contained in this Agreement or any other Loan Document (other than as provided in paragraphs (a) through (c) of this Section), and such default shall continue unremedied for a period of 30 days; or

(e) The Parent, any Borrower or any of its Subsidiaries shall (i) default in making any payment of any principal of any Indebtedness (including, without limitation, any Guarantee Obligation, but excluding the Loans and Reimbursement Obligations) on the scheduled or original due date with respect thereto (after giving effect to any applicable grace periods); or (ii) default in making any payment of any interest on any such Indebtedness beyond the period of grace, if any, provided in the instrument or agreement under which such Indebtedness was created; or (iii) default in the observance or performance of any other agreement or condition relating to any such Indebtedness or contained in any instrument or agreement evidencing, securing or relating thereto, or any other event shall occur or condition exist, the effect of which default or other event or condition is to cause, or to permit the holder or beneficiary of such Indebtedness (or a trustee or agent on behalf of such holder or beneficiary) to cause, with the giving of notice if required, such Indebtedness to become due prior to its stated maturity or to become subject to or mandatory offer to purchase by the obligor thereunder or (in the case of any such Indebtedness constituting a Guarantee Obligation) to become payable; $\ensuremath{\mathsf{PROVIDED}}$, that a default, event or condition described in clause (i), (ii) or (iii) of this paragraph (e) shall not at any time constitute an Event of Default unless, at such time, one or more defaults, events or conditions of the type described in clauses (i), (ii) and (iii) of this paragraph (e) shall have occurred and be continuing with respect to Indebtedness the outstanding principal amount of which exceeds in the aggregate \$25,000,000; or

(f) (i) The Parent, any Borrower or any of its Subsidiaries shall commence any case, proceeding or other action (A) under any existing or future law of any jurisdiction, domestic or foreign, relating to bankruptcy, insolvency, reorganization or relief of debtors, seeking to have an order for relief entered with respect to it, or seeking to adjudicate it a bankrupt or insolvent, or seeking reorganization, arrangement, adjustment, winding-up, liquidation, dissolution, composition or other relief with respect to it or its debts, or (B) seeking appointment of a receiver, trustee, custodian, conservator or other similar official for it or for all or any substantial part of its assets, or the Parent, any Borrower or any of its Subsidiaries shall make a general assignment for the benefit of its creditors; or (ii) there shall be commenced against the Parent, any Borrower or any of its Subsidiaries any case, proceeding or other action of a nature referred to in clause (i) above that (A) results in the entry of an order for relief or any such adjudication or appointment or (B) remains undismissed, undischarged or unbonded for a period of 60 days; or (iii) there shall be commenced against the Parent, any Borrower or any of its Subsidiaries any case, proceeding or other action seeking issuance of a warrant of attachment, execution, distraint or similar process against all or any substantial part of its assets that results in the entry of an order for any such relief that shall not have been vacated, discharged, or stayed or bonded pending appeal within 60 days from the entry thereof; or (iv) the Parent, any Borrower or any of its Subsidiaries shall take any action in furtherance of, or indicating its consent to, approval of, or acquiescence in, any of the acts set forth in clause (i), (ii), or (iii) above; or (v) the Parent, any Borrower or any of its Subsidiaries shall generally not, or shall be unable to, or shall admit in writing its inability to, pay its debts as they become due; or

(g) (i) Any Person shall engage in any "prohibited transaction" (as defined in Section 406 of ERISA or Section 4975 of the Code) involving any Plan, (ii) any "accumulated funding deficiency" (as defined in Section 302 of ERISA), whether or not waived, shall exist with respect to any Plan or any Lien in favor of the PBGC or a Plan shall arise on the assets of any Borrower or any Commonly Controlled Entity of any Borrower, (iii) a Reportable Event shall occur with respect to, or proceedings shall commence to have a trustee appointed, or a trustee shall be appointed, to administer or to terminate, any Single Employer Plan, which Reportable Event or commencement of proceedings or appointment of a trustee is, in the reasonable opinion of the Required Lenders, likely to result in the termination of such Plan for purposes of Title IV of ERISA, (iv) any Single Employer Plan shall terminate for purposes of Title IV of ERISA, (v) any Borrower or any Commonly Controlled Entity of any Borrower shall, or in the reasonable opinion of the Required Lenders is likely to, incur any liability in connection with a withdrawal from, or the Insolvency or Reorganization of, a Multiemployer Plan or (vi) any other event or condition shall occur or exist with respect to a Plan; and in each case in clauses (i) through (vi) above, such event or condition, together with all other such events or conditions, if any, could, in the sole judgment of the Required Lenders, reasonably be expected to have a Material Adverse Effect; or

(h) One or more judgments or decrees shall be entered against the Parent, any Borrower or any of its Subsidiaries involving for the Parent, any Borrower and its Subsidiaries taken as a whole a liability (to the extent not paid or fully covered by insurance above applicable deductions) of \$25,000,000 or more, and all such judgments or decrees shall not have been vacated, discharged, stayed or bonded pending appeal within 30 days from the entry thereof; or

(i) Any of the security interests created pursuant to the Security Documents shall cease, for any reason (other than by reason of the express release thereof pursuant to Section 10.15), to be a valid, perfected security interest, or any Loan Party or any Affiliate of any Loan Party shall so assert; or

(j) The guarantee contained in Section 2 of the Guarantee and Collateral Agreement shall cease, for any reason (other than by reason of the express release thereof pursuant to Section 10.15), to be in full force and effect or any Loan Party or any Affiliate of any Loan Party shall so assert; or

(k) Prior to any initial public offering of the Parent's common stock, White Mountains shall cease to own, directly or indirectly, at least 50% of all of the outstanding Capital Stock of the Parent, and (b) after any initial public offering of the Parent's Capital Stock, (i) White Mountains shall cease to own, directly or indirectly, at least 25% of all of the outstanding Capital Stock of the Parent, or (ii) any Person or group of Persons other than White Mountains shall own a greater percentage than White Mountains of all of the outstanding Capital Stock of the Parent or (iii) a majority of the directors of the Parent ceases to be persons appointed or approved by White Mountains; or

(1) (i) Any Material Insurance Subsidiary shall fail to comply with any material term or provision of any Insurance Regulations pertaining to fiscal soundness, solvency or financial condition; or (ii) an Insurance Regulator shall assert in writing that it is taking administrative action against any Material Insurance Subsidiary to revoke or modify any contract of insurance, license, permit, certification, authorization, accreditation or charter or to enforce the fiscal soundness, solvency or financial provisions or requirements of any Insurance Regulations against any Material Insurance Subsidiary and such Insurance Subsidiary shall have been unable to cause such Insurance Regulator to withdraw such written notice within five Business Days following receipt of such written notice by such Insurance Subsidiary, in each of clauses (i) and (ii), if such event could reasonably be expected to have a Material Adverse Effect.

(m) The Subordinated Seller Note thereof shall cease, for any reason, to be validly subordinated to the obligations of the Loan Parties hereunder and under the other Loan Documents, or any Loan Party, any Affiliate of any Loan Party or the holder of the Subordinated Seller Note shall so assert;

then, and in any such event, (A) if such event is an Event of Default specified in clause (i) or (ii) of paragraph (f) above with respect to any Borrower, automatically the Commitments shall immediately terminate and the Loans hereunder (with accrued interest thereon) and all other amounts owing under this Agreement and the other Loan Documents (including, without limitation, all amounts of L/C Obligations, whether or not the beneficiaries of the then outstanding Letters of Credit shall have presented the documents required thereunder) shall immediately become due and payable, and (B) if such event is any other Event of Default, either or both of the following actions may be taken: (i) with the consent of the Required Lenders, the Administrative Agent may, or upon the request of the Required Lenders, the Administrative Agent shall, by notice to each Borrower declare the Commitments to be terminated forthwith, whereupon the Commitments shall immediately terminate; and (ii) with the consent of the Required Lenders, the Administrative Agent may, or upon the request of the Required Lenders, the Administrative Agent shall, by notice to each Borrower, declare the Loans hereunder (with accrued interest thereon) and all other amounts owing under this Agreement and the other Loan Documents (including, without limitation, all amounts of L/C Obligations, whether or not the beneficiaries of the then outstanding Letters of Credit shall have presented the documents required thereunder) to be due and payable forthwith, whereupon the same shall immediately become due and payable. In the case of all Letters of Credit with respect to which presentment for honor shall not have occurred at the time of an acceleration pursuant to this paragraph, the relevant Borrower shall at such time deposit in a cash collateral account opened by the Administrative Agent an amount equal to the aggregate then undrawn and unexpired amount of such Letters of Credit. Amounts held in such cash collateral account shall be applied by the Administrative Agent to the payment of drafts drawn under such Letters of Credit, and the unused portion thereof after all such Letters of Credit shall have expired or been fully drawn

upon, if any, shall be applied to repay other obligations of the relevant Borrower hereunder and under the other Loan Documents. After all such Letters of Credit shall have expired or been fully drawn upon, all Reimbursement Obligations shall have been satisfied and all other obligations of the relevant Borrower hereunder and under the other Loan Documents shall have been paid in full, the balance, if any, in such cash collateral account shall be returned to the relevant Borrower (or such other Person as may be lawfully entitled thereto).

SECTION 9 THE ADMINISTRATIVE AGENT

9.1 APPOINTMENT. Each Lender hereby irrevocably designates and appoints the Administrative Agent as the agent of such Lender under this Agreement and the other Loan Documents, and each Lender irrevocably authorizes the Administrative Agent, in such capacity, to take such action on its behalf under the provisions of this Agreement and the other Loan Documents and to exercise such powers and perform such duties as are expressly delegated to the Administrative Agent by the terms of this Agreement and the other Loan Documents, together with such other powers as are reasonably incidental thereto. Notwithstanding any provision to the contrary elsewhere in this Agreement, the Administrative Agent shall not have any duties or responsibilities, except those expressly set forth herein, or any fiduciary relationship with any Lender, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Loan Document or otherwise exist against the Administrative Agent.

9.2 DELEGATION OF DUTIES. The Administrative Agent may execute any of its duties under this Agreement and the other Loan Documents by or through agents or attorneys-in-fact and shall be entitled to advice of counsel concerning all matters pertaining to such duties. The Administrative Agent shall not be responsible for the negligence or misconduct of any agents or attorneys-in-fact selected by it with reasonable care.

9.3 EXCULPATORY PROVISIONS. Neither the Administrative Agent nor any of its officers, directors, employees, agents, attorneys-in-fact or affiliates shall be (i) liable for any action lawfully taken or omitted to be taken by it or such Person under or in connection with this Agreement or any other Loan Document (except to the extent that any of the foregoing are found by a final and nonappealable decision of a court of competent jurisdiction to have resulted from its or such Person's own gross negligence or willful misconduct) or (ii) responsible in any manner to any of the Lenders for any recitals, statements, representations or warranties made by any Loan Party or any officer thereof contained in this Agreement or any other Loan Document or in any certificate, report, statement or other document referred to or provided for in, or received by the Administrative Agent under or in connection with, this Agreement or any other Loan Document or for the value, validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement or any other Loan Document or for any failure of any Loan Party a party thereto to perform its obligations hereunder or thereunder. The Administrative Agent shall not be under any obligation to any Lender to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Loan Document, or to inspect the properties, books or records of any Loan Party.

9.4 RELIANCE BY ADMINISTRATIVE AGENT. The Administrative Agent shall be entitled to rely, and shall be fully protected in relying, upon any instrument, writing, resolution,

notice, consent, certificate, affidavit, letter, telecopy, telex or teletype message, statement, order or other document or conversation believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons and upon advice and statements of legal counsel (including, without limitation, counsel to the Loan Parties), independent accountants and other experts selected by the Administrative Agent. The Administrative Agent may deem and treat the payee of any Note as the owner thereof for all purposes unless such Note shall have been transferred in accordance with Section 10.6 and all actions required by such Section in connection with such transfer shall have been taken. The Administrative Agent shall be fully justified in failing or refusing to take any action under this Agreement or any other Loan Document unless it shall first receive such advice or concurrence of the Required Lenders (or, if so specified by this Agreement, all Lenders) as it deems appropriate or it shall first be indemnified to its satisfaction by the Lenders against any and all liability and expense that may be incurred by it by reason of taking or continuing to take any such action. The Administrative Agent shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement and the other Loan Documents in accordance with a request of the Required Lenders (or, if so specified by this Agreement, all Lenders or any other instructing group of Lenders specified by this Agreement), and such request and any action taken or failure to act pursuant thereto shall be binding upon all the Lenders and all future holders of the Loans.

9.5 NOTICE OF DEFAULT. The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default hereunder unless the Administrative Agent shall have received notice from a Lender, the Parent or any Borrower referring to this Agreement, describing such Default or Event of Default and stating that such notice is a "notice of default". In the event that the Administrative Agent shall receive such a notice, the Administrative Agent shall give notice thereof to the Lenders. The Administrative Agent shall take such action with respect to such Default or Event of Default as shall be reasonably directed by the Required Lenders (or, if so specified by this Agreement, all Lenders or any other instructing group of Lenders specified by this Agreement); provided that unless and until the Administrative Agent shall have received such directions, the Administrative Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as it shall deem advisable in the best interests of the Lenders.

9.6 NON-RELIANCE ON ADMINISTRATIVE AGENT AND OTHER LENDERS. Each Lender expressly acknowledges that neither the Administrative Agent nor any of its respective officers, directors, employees, agents, attorneys-in-fact or affiliates have made any representations or warranties to it and that no act by the Administrative Agent hereafter taken, including any review of the affairs of a Loan Party or any affiliate of a Loan Party, shall be deemed to constitute any representation or warranty by the Administrative Agent to any Lender. Each Lender represents to the Administrative Agent that it has, independently and without reliance upon the Administrative Agent or any other Lender, and based on such documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, operations, property, financial and other condition and creditworthiness of the Loan Parties and their affiliates and made its own decision to make its Loans hereunder and enter into this Agreement. Each Lender also represents that it will, independently and without reliance upon the Administrative Agent or any other Lender, and based on such documents and information as it shall deem appropriate at the time. continue to make its own credit analysis, appraisals and

decisions in taking or not taking action under this Agreement and the other Loan Documents, and to make such investigation as it deems necessary to inform itself as to the business, operations, property, financial and other condition and creditworthiness of the Loan Parties and their affiliates. Except for notices, reports and other documents expressly required to be furnished to the Lenders by the Administrative Agent hereunder, the Administrative Agent shall have no duty or responsibility to provide any Lender with any credit or other information concerning the business, operations, property, condition (financial or otherwise), prospects or creditworthiness of any Loan Party or any affiliate of a Loan Party that may come into the possession of the Administrative Agent or any of its officers, directors, employees, agents, attorneys-in-fact or affiliates.

9.7 INDEMNIFICATION. The Lenders agree to indemnify the Administrative Agent in its capacity as such (to the extent not reimbursed by the Parent or the Primary Borrower and without limiting the obligation of the Parent or the Primary Borrower to do so), ratably according to their respective Aggregate Exposure Percentages in effect on the date on which indemnification is sought under this Section (or, if indemnification is sought after the date upon which the Commitments shall have terminated and the Loans shall have been paid in full, ratably in accordance with such Aggregate Exposure Percentages immediately prior to such date), for, and to save the Administrative Agent harmless from and against, any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind whatsoever that may at any time (including, without limitation, at any time following the payment of the Loans) be imposed on, incurred by or asserted against the Administrative Agent in any way relating to or arising out of, the Commitments, this Agreement, any of the other Loan Documents or any documents contemplated by or referred to herein or therein or the transactions contemplated hereby or thereby or any action taken or omitted by the Administrative Agent under or in connection with any of the foregoing; provided that no Lender shall be liable for the payment of any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements that are found by a final and nonappealable decision of a court of competent jurisdiction to have resulted from the Administrative Agent's gross negligence or willful misconduct. The agreements in this Section shall survive the payment of the Loans and all other amounts pavable hereunder.

9.8 ADMINISTRATIVE AGENT IN ITS INDIVIDUAL CAPACITY. The Administrative Agent and its Affiliates may make loans to, accept deposits from, hold equity interests in and generally engage in any kind of business with any Loan Party or its Affiliates as though the Administrative Agent were not an Administrative Agent. With respect to its Loans made or renewed by it and with respect to any Letter of Credit issued or participated in by it, the Administrative Agent shall have the same rights and powers under this Agreement and the other Loan Documents as any Lender and may exercise the same as though it were not an Administrative Agent, and the terms "Lender" and "Lenders" shall include the Administrative Agent in its individual capacity.

9.9 SUCCESSOR ADMINISTRATIVE AGENT. The Administrative Agent may resign as Administrative Agent upon 15 days' notice to the Lenders and the Borrowers. If the Administrative Agent shall resign as Administrative Agent under this Agreement and the other Loan Documents, then the Required Lenders shall appoint from among the Lenders a successor agent for the Lenders, which successor agent shall (unless an Event of Default under Section 8(a) or Section 8(f) with respect to the Primary Borrower shall have occurred and be continuing) be

subject to approval by the Primary Borrower (which approval shall not be unreasonably withheld or delayed), whereupon such successor agent shall succeed to the rights, powers and duties of the Administrative Agent, and the term "Administrative Agent" shall mean such successor agent effective upon such appointment and approval, and the former Administrative Agent's rights, powers and duties as Administrative Agent shall be terminated, without any other or further act or deed on the part of such former Administrative Agent or any of the parties to this Agreement or any holders of the Loans. If no successor agent has accepted appointment as Administrative Agent by the date that is 15 days following a retiring Administrative Agent's notice of resignation, the retiring Administrative Agent's resignation shall nevertheless thereupon become effective, and the Lenders shall assume and perform all of the duties of the Administrative Agent hereunder until such time, if any, as the Required Lenders Administrative Agent's resignation as Administrative Agent, the provisions of this Section 9 shall inure to its benefit as to any actions taken or omitted to be taken by it while it was Administrative Agent under this Agreement and the other Loan Documents.

9.10 AUTHORIZATION TO RELEASE LIENS AND GUARANTEES. The Administrative Agent is hereby irrevocably authorized by each of the Lenders to effect any release of Liens or guarantee obligations contemplated by Section 10.15.

9.11 THE ARRANGER, THE SYNDICATION AGENT AND THE DOCUMENTATION AGENTS. The Arranger, the Syndication Agent and the Documentation Agents, in their respective capacities as such, shall have no duties or responsibilities, and shall incur no liability, under this Agreement and the other Loan Documents.

SECTION 10 MISCELLANEOUS

10.1 AMENDMENTS AND WAIVERS. Neither this Agreement or any other Loan Document, nor any terms hereof or thereof may be amended, supplemented or modified except in accordance with the provisions of this Section 10.1. The Required Lenders and each Loan Party party to the relevant Loan Document may, or (with the written consent of the Required Lenders) the Administrative Agent and each Loan Party party to the relevant Loan Document may, from time to time, (a) enter into written amendments, supplements or modifications hereto and to the other Loan Documents (including amendments and restatements hereof or thereof) for the purpose of adding any provisions to this Agreement or the other Loan Documents or changing in any manner the rights of the Lenders or of the Loan Parties hereunder or thereunder or (b) waive, on such terms and conditions as may be specified in the instrument of waiver, any of the requirements of this Agreement or the other Loan Documents or any Default or Event of Default and its consequences; PROVIDED, HOWEVER, that no such waiver and no such amendment, supplement or modification shall:

> (i) forgive or reduce the principal amount or extend the final scheduled date of maturity of any Loan or Reimbursement Obligation, extend the scheduled date of any amortization payment in respect of any Term Loan, reduce the stated rate of any interest or fee payable hereunder or extend the scheduled date of any payment thereof, or increase the amount or extend the expiration date of any Commitment of any Lender, in each case without the consent of each Lender directly affected thereby;

(ii) amend, modify or waive any provision of this Section, reduce any percentage specified in the definition of Required Lenders or Required Prepayment Lenders, amend, modify or waive the last sentence of the definition of Required Lenders, or otherwise amend or modify the definition of Required Lenders if the effect of such amendment or modification would be to eliminate reference in such definition to Loans and/or Commitments under any particular Facility, consent to the assignment or transfer by any Borrower of any of its rights and obligations under this Agreement and the other Loan Documents, release all or substantially all of the Collateral or release the Parent or all or substantially all of the Subsidiary Guarantors from their guarantee obligations under the Guarantee and Collateral Agreement, in each case without the consent of all Lenders;

(iii) reduce the percentage specified in the definition of Majority Facility Lenders with respect to any Facility without the written consent of all Lenders under such Facility;

(iv) amend, modify or waive any provision of Section 9 without the consent of the Administrative Agent;

(v) amend, modify or waive any provision of Section 2.6 or 2.7 without the written consent of the Swing Line Lender;

(vi) amend, modify or waive any provision of Section 2.18 without the consent of each Lender directly affected thereby;

(vii) amend, modify or waive any provision of Section 3
without the consent of the Issuing Lender;

(viii) amend, modify or waive (x) any provision of Sections 2.12 or 7.2(l) requiring by its terms the agreement of the Required Prepayment Lenders or (y) any other provision hereof if the effect is to subordinate one of the Facilities in right of payment to any other of the Facilities, in either case without the consent of the Required Prepayment Lenders; or

(ix) amend, modify or waive the provisions of the definition of Interest Period regarding nine or twelve month Interest Periods for Eurodollar Loans without the consent of each relevant Lender.

Any such waiver and any such amendment, supplement or modification shall apply equally to each of the Lenders and shall be binding upon the Loan Parties, the Lenders, the Administrative Agent and all future holders of the Loans. In the case of any waiver, the Loan Parties, the Lenders and the Administrative Agent shall be restored to their former position and rights hereunder and under the other Loan Documents, and any Default or Event of Default waived shall be deemed to be cured and not continuing; but no such waiver shall extend to any subsequent or other Default or Event of Default, or impair any right consequent thereon. Any such waiver, amendment, supplement or modification shall be effected by a written instrument signed by the parties required to sign pursuant to the foregoing provisions of this Section; PROVIDED, that delivery of an executed signature page of any such instrument by facsimile transmission shall be effective as delivery of a manually executed counterpart thereof.

For the avoidance of doubt, this Agreement may be amended (or amended and restated) with the written consent of the Required Lenders, the Administrative Agent and each Loan Party to each relevant Loan Document (x) to add one or more additional credit facilities to this Agreement and to permit the extensions of credit from time to time outstanding thereunder and the accrued interest and fees in respect thereof to share ratably in the benefits of this Agreement and the other Loan Documents with the Term Loans and Revolving Extensions of Credit and the accrued interest and fees in respect thereof and (y) to include appropriately the Lenders holding such credit facilities in any determination of the Required Lenders, Required Prepayment Lenders and Majority Revolving Facility Lenders.

10.2 NOTICES. All notices, requests and demands to or upon the respective parties hereto to be effective shall be in writing (including by telecopy), and, unless otherwise expressly provided herein, shall be deemed to have been duly given or made when delivered, or three Business Days after being deposited in the mail, postage prepaid, or, in the case of telecopy notice, when received, addressed (a) in the case of the Parent, the Primary Borrower, the Additional Borrower and the Administrative Agent, as follows and (b) in the case of the Lenders, as set forth in an administrative questionnaire delivered to the Administrative Agent or on Schedule I to the Lender Addendum to which such Lender is a party or, in the case of a Lender which becomes a party to this Agreement pursuant to an Assignment and Acceptance, in such Assignment and Acceptance or (c) in the case of any party, to such other address as such party may hereafter notify to the other parties hereto:

The Parent:	TACK Holding Corp. 80 South Main Street Hanover, NH 03755-2053
The Primary Borrower:	TACK Acquisition Corp. 80 South Main Street Hanover, NH 03755-2053
The Additional Borrower:	White Mountains Insurance Group, Ltd. 80 South Main Street Hanover, NH 03755-2053
The Administrative Agent (in respect of credit matters):	Lehman Commercial Paper Inc. 3 World Financial Center New York, New York 10285 Attention: Michele Swanson Telecopy: 212-526-0242 Telephone: 212-526-0330

The Administrative Agent (in respect of administrative agency matters):	Lehman Commercial Paper Inc. 3 World Financial Center New York, New York 10285 Attention: Diane Albanese Telecopy: 212-528-8725 Telephone: 212-526-4979
Issuing Lender:	As notified by the Issuing Lender to the Administrative Agent and the Primary Borrower

PROVIDED that any notice, request or demand to or upon the Administrative Agent, the Issuing Lender or any Lender shall not be effective until received.

10.3 NO WAIVER; CUMULATIVE REMEDIES. No failure to exercise and no delay in exercising, on the part of the Administrative Agent or any Lender, any right, remedy, power or privilege hereunder or under the other Loan Documents shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privileges. The rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.

10.4 SURVIVAL OF REPRESENTATIONS AND WARRANTIES. All representations and warranties made hereunder, in the other Loan Documents and in any document, certificate or statement delivered pursuant hereto or in connection herewith shall survive the execution and delivery of this Agreement and the making of the Loans and other extensions of credit hereunder.

10.5 PAYMENT OF EXPENSES. The Primary Borrower agrees (a) to pay or reimburse the Administrative Agent for all its reasonable out-of-pocket costs and expenses incurred in connection with the syndication and administration of the Facilities (other than fees payable to syndicate members) and the development, preparation and execution of, and any amendment, supplement or modification to, this Agreement and the other Loan Documents and any other documents prepared in connection herewith or therewith, and the consummation and administration of the transactions contemplated hereby and thereby, including, without limitation, the reasonable fees and disbursements and other charges of counsel to the Administrative Agent and the charges of Intralinks, (b) to pay or reimburse each Lender and the Administrative Agent for all their costs and expenses incurred in connection with the enforcement or preservation of any rights under this Agreement, the other Loan Documents and any such other documents, including, without limitation, the fees and disbursements of counsel to each Lender and of counsel to the Administrative Agent, (c) to pay, indemnify, or reimburse each Lender and the Administrative Agent for, and hold each Lender and the Administrative Agent harmless from, any and all recording and filing fees and any and all liabilities with respect to, or resulting from any delay in paying, stamp, excise and other taxes, if any, which may be payable or determined to be payable in connection with the execution and delivery of, or consummation or administration of any of the transactions contemplated by, or any amendment, supplement or modification of, or any waiver or consent under or in respect of, this Agreement, the other Loan Documents and any such other documents, and (d) to pay, indemnify or

reimburse each Lender, the Administrative Agent, its respective affiliates, and their respective officers, directors, trustees, employees, advisors, agents and controlling persons (each, an "Indemnitee") for, and hold each Indemnitee harmless from and against any and all other liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever with respect to the execution, delivery, enforcement, performance and administration of this Agreement, the other Loan Documents and any such other documents, including, without limitation, any of the foregoing relating to the use of proceeds of the Loans or the violation of, noncompliance with or liability under, any Environmental Law applicable to the operations of the Parent, any Borrower or any of its Subsidiaries or any of the Properties and the fees and disbursements and other charges of legal counsel in connection with claims, actions or proceedings by any Indemnitee against any Borrower hereunder (all the foregoing in this clause (d), collectively, the "Indemnified Liabilities"), provided, that the Primary Borrower shall have no obligation hereunder to any Indemnitee with respect to Indemnified Liabilities to the extent such Indemnified Liabilities are found by a final and nonappealable decision of a court of competent jurisdiction to have resulted from the gross negligence or willful misconduct of such Indemnitee. No Indemnified Person shall be liable for use by unauthorized Persons of information sent through electronic media. It is understood and agreed that, to the extent not precluded by a conflict of interest (as reasonably determined by the relevant Indemnitee), each Indemnitee shall endeavor to work cooperatively with the Borrowers with a view towards minimizing the legal and other expenses associated with any defense and any potential settlement or judgment. To the extent reasonably practicable and not disadvantageous to any Indemnitee (as reasonably determined by the relevant Indemnitee), it is anticipated that a single counsel selected by the affected Lenders may be used. Without limiting the foregoing, and to the extent permitted by applicable law, each Borrower agrees not to assert and to cause its Subsidiaries not to assert, and hereby waives and agrees to cause its Subsidiaries to so waive, all rights for contribution or any other rights of recovery with respect to all claims, demands, penalties, fines, liabilities, settlements, damages, costs and expenses of whatever kind or nature with respect to the execution, delivery, enforcement, performance and administration of this Agreement, the other Loan Documents and any such other documents, under or related to Environmental Laws, that any of them might have by statute or otherwise against any Indemnitee, except to the extent such rights result from the gross negligence or willful misconduct of such Indemnitee. All amounts due under clauses (a), (b) and (c) of the first sentence of this Section shall be payable not later than 30 days after written demand therefor. Statements payable by the Primary Borrower pursuant to this Section shall be submitted to Reid T. Campbell (Telephone No. (603) 640-2203) (Fax No. (603) 643-4562), at the address of the Primary Borrower set forth in Section 10.2, or to such other Person or address as may be hereafter designated by the Primary Borrower in a written notice to the Administrative Agent. The agreements in this Section shall survive repayment of the Loans and all other amounts payable hereunder.

10.6 SUCCESSORS AND ASSIGNS; PARTICIPATIONS AND ASSIGNMENTS. (a) This Agreement shall be binding upon and inure to the benefit of the Parent, each Borrower, the Lenders, the Administrative Agent, all future holders of the Loans and their respective successors and assigns, except that neither Borrower may not assign or transfer any of its rights or obligations under this Agreement without the prior written consent of the Administrative Agent and each Lender.

(b) Any Lender may, without the consent of any Borrower or the Administrative Agent (but with notice to the Administrative Agent), in accordance with applicable law, at any time sell to one or more banks, financial institutions or other entities (each, a "PARTICIPANT") participating interests in any Loan owing to such Lender, any Commitment of such Lender or any other interest of such Lender hereunder and under the other Loan Documents. In the event of any such sale by a Lender of a participating interest to a Participant, such Lender's obligations under this Agreement to the other parties to this Agreement shall remain unchanged, such Lender shall remain solely responsible for the performance thereof, such Lender shall remain the holder of any such Loan for all purposes under this Agreement and the other Loan Documents, and the Borrowers and the Administrative Agent shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement and the other Loan Documents. In no event shall any Participant under any such participation have any right to approve any amendment or waiver of any provision of any Loan Document, or any consent to any departure by any Loan Party therefrom, except to the extent that such amendment, waiver or consent would, pursuant to Section 10.1, require the consent of all Lenders under any Facility in which such Participant has a participating interest. Each Borrower agrees that if amounts owing by such Borrower under this Agreement and the Loans are due or unpaid, or shall have been declared or shall have become due and payable upon the occurrence of an Event of Default, each Participant shall, to the maximum extent permitted by applicable law, be deemed to have the right of setoff in respect of its participating interest in amounts owing under this Agreement to the same extent as if the amount of its participating interest were owing directly to it as a Lender under this Agreement, PROVIDED that, in purchasing such participating interest, such Participant shall be deemed to have agreed to share with the Lenders the proceeds thereof as provided in Section 10.7(a) as fully as if such Participant were a Lender hereunder. Each Borrower also agrees that each Participant shall be entitled to the benefits of Sections 2.19, 2.20 and 2.21 with respect to its participation in the Commitments and the Loans outstanding from time to time as if such Participant were a Lender; PROVIDED that, in the case of Section 2.20, such Participant shall have complied with the requirements of said Section, and PROVIDED, FURTHER, that no Participant shall be entitled to receive any greater amount pursuant to any such Section than the transferor Lender would have been entitled to receive in respect of the amount of the participation transferred by such transferor Lender to such Participant had no such transfer occurred.

(c) Any Lender (an "ASSIGNOR") may, in accordance with applicable law and upon written notice to the Administrative Agent, at any time and from time to time assign to any Lender or any affiliate or Control Investment Affiliate thereof or, with the consent of the Primary Borrower and the Administrative Agent and, in the case of any assignment of Revolving Credit Commitments, the written consent of the Issuing Lender and the Swing Line Lender (which, in each case, shall not be unreasonably withheld or delayed) (PROVIDED that the consent of the Primary Borrower need not be obtained with respect to (x) any assignment by the Administrative Agent or by any Lehman Entity and (y) any assignment of "ASSIGNEE") all or any part of its rights and obligations under this Agreement pursuant to an Assignment and Acceptance, substantially in the form of Exhibit E, executed by such Assignee and such Assignor (and, where the consent of the Primary Borrower, the Administrative Agent or the Issuing Lender or the Swing Line Lender is required pursuant to the foregoing provisions, by the Primary Borrower and such other Persons) and delivered to the Administrative Agent for its acceptance and recording in the Register; PROVIDED

that (i) no such assignment of (x) Tranche A Term Loans or (y) Revolving Credit Commitments and/or Revolving Credit Loans to an Assignee (other than any Lender or any Control Investment Affiliate thereof) shall be in an aggregate principal amount of less than \$5,000,000 (other than in the case of an assignment of all of a Lender's interests under this Agreement), (ii) no such assignment of (x)Tranche A Term Loans or (y) Revolving Credit Commitments and/or Revolving Credit Loans (other than in the case of an assignment of all of a Lender's interests under this Agreement) shall result in such Lender's Commitments, together with the Commitments of such Lender's Control Investment Affiliates, being reduced to less than \$5,000,000, (iii) no such assignment of Tranche B Term Loans to an Assignee (other than any Lender or any Control Investment Affiliate thereof) shall be in an aggregate principal amount of less than \$1,000,000 (other than in the case of an assignment of all of a Lender's interests under this Agreement), and (iv) no such assignment of Tranche B Term Loans (other than in the case of an assignment of all of a Lender's interests under this Agreement) shall result in such Lender's Tranche B Term Loan Commitment (or, after the Closing Date, Tranche B Term Loans), together with the Tranche B Term Loan Commitment (or, after the Closing Date, Tranche B Term Loans) of such Lender's Control Investment Affiliates, being reduced to less than \$1,000,000 (but more than \$0), in each case unless otherwise agreed by the Primary Borrower and the Administrative Agent. Any such assignment need not be ratable as among the Facilities. Upon such execution, delivery, acceptance and recording, from and after the effective date determined pursuant to such Assignment and Acceptance, (x) the Assignee thereunder shall be a party hereto and, to the extent provided in such Assignment and Acceptance, have the rights and obligations of a Lender hereunder with Commitments and/or Loans as set forth therein, and (y) the Assignor thereunder shall, to the extent provided in such Assignment and Acceptance, be released from its obligations under this Agreement (and, in the case of an Assignment and Acceptance covering all of an Assignor's rights and obligations under this Agreement, such Assignor shall cease to be a party hereto, except as to Section 2.19, 2.20, 10.5 and 10.14 in respect of the period prior to such effective date). Notwithstanding any provision of this Section, the consent of the Primary Borrower shall not be required for any assignment that occurs at any time when any Event of Default shall have occurred and be continuing.

(d) The Administrative Agent shall, on behalf of the Borrowers, maintain at its address referred to in Section 10.2 a copy of each Assignment and Acceptance delivered to it and a register (the "REGISTER") for the recordation of the names and addresses of the Lenders and the Commitment of, and principal amount of the Loans owing to, each Lender from time to time. The entries in the Register shall be conclusive, in the absence of manifest error, and each Borrower, the Administrative Agent and the Lenders shall treat each Person whose name is recorded in the Register as the owner of the Loans and any Notes evidencing such Loans recorded therein for all purposes of this Agreement. Any assignment of any Loan, whether or not evidenced by a Note, shall be effective only upon appropriate entries with respect thereto being made in the Register (and each Note shall expressly so provide). Any assignment or transfer of all or part of a Loan evidenced by a Note shall be registered on the Register only upon surrender for registration of assignment or transfer of the Note evidencing such Loan, accompanied by a duly executed Assignment and Acceptance; thereupon one or more new Notes in the same aggregate principal amount shall be issued to the designated Assignee, and the old Notes shall be returned by the Administrative Agent to the relevant Borrower marked "canceled". The Register shall be available for inspection by any Borrower or any Lender (with

respect to any entry relating to such Lender's Loans) at any reasonable time and from time to time upon reasonable prior notice.

(e) Upon its receipt of an Assignment and Acceptance executed by an Assignor and an Assignee (and, in any case where the consent of any other Person is required by Section 10.6(c), by each such other Person) together with payment to the Administrative Agent of a registration and processing fee of \$3,500 (except that (x) no such registration and processing fee shall be payable in the case of an Assignee which is an affiliate of the Assignor or a Person under common management with the Assignor and (y) only one registration and processing fee shall be payable in cases of contemporaneous multiple assignments by an Assignor to Assignees which are Control Investment Affiliates of each other), the Administrative Agent shall (i) promptly accept such Assignment and Acceptance and (ii) on the effective date determined pursuant thereto record the information contained therein in the Register and give notice of such acceptance and recordation to the Administrative Agent and the Borrowers. On or prior to such effective date, the relevant Borrower, at its own expense, upon request, shall execute and deliver to the Administrative Agent (in exchange for the Revolving Credit Note and/or applicable Term Notes, as the case may be, of the assigning Lender) a new Revolving Credit Note and/or applicable Term Notes, as the case may be, to the order of such Assignee in an amount equal to the Revolving Credit Commitment and/or applicable Term Loans, as the case may be, assumed or acquired by it pursuant to such Assignment and Acceptance and, if the Assignor has retained a Revolving Credit Commitment and/or Term Loans, as the case may be, upon request, a new Revolving Credit Note and/or Term Notes, as the case may be, to the order of the Assignor in an amount equal to the Revolving Credit Commitment and/or applicable Term Loans, as the case may be, retained by it hereunder. Such new Note or Notes shall be dated the Closing Date and shall otherwise be in the form of the Note or Notes replaced thereby.

(f) For avoidance of doubt, the parties to this Agreement acknowledge that the provisions of this Section concerning assignments of Loans and Notes relate only to absolute assignments and that such provisions do not prohibit assignments creating security interests, including, without limitation, any pledge or assignment by a Lender of any Loan or Note to any Federal Reserve Bank in accordance with applicable law.

10.7 ADJUSTMENTS; SET-OFF. (a) Except to the extent that this Agreement provides for payments to be allocated to a particular Lender or to the Lenders under a particular Facility, if any Lender (a "Benefitted Lender") shall at any time receive any payment of all or part of the Obligations owing to it, or receive any collateral in respect thereof (whether voluntarily or involuntarily, by set-off, pursuant to events or proceedings of the nature referred to in Section 8(f), or otherwise), in a greater proportion than any such payment to or collateral received by any other Lender, if any, in respect of such other Lender's Obligations, such Benefitted Lender shall purchase for cash from the other Lenders a participating interest in such portion of each such other Lender's Obligations, or shall provide such other Lenders with the benefits of any such collateral, as shall be necessary to cause such Benefitted Lender to share the excess payment or benefits of such collateral ratably with each of the Lenders; provided, however, that if all or any portion of such excess payment or benefits is thereafter recovered from such Benefitted Lender, such purchase shall be rescinded, and the purchase price and benefits returned, to the extent of such recovery, but without interest. (b) In addition to any rights and remedies of the Lenders provided by law, each Lender shall have the right, without prior notice to the Parent or any Borrower, any such notice being expressly waived by the Parent and each Borrower to the extent permitted by applicable law, upon any amount becoming due and payable by the Parent or any Borrower hereunder (whether at the stated maturity, by acceleration or otherwise), to set off and appropriate and apply against such amount any and all deposits (general or special, time or demand, provisional or final), in any currency, and any other credits, indebtedness or claims, in any currency, in each case whether direct or indirect, absolute or contingent, matured or unmatured, at any time held or owing by such Lender or any branch or agency thereof to or for the credit or the account of the Parent or such Borrower, as the case may be. Each Lender agrees promptly to notify the Borrower and the Administrative Agent after any such setoff and application made by such Lender, PROVIDED that the failure to give such notice shall not affect the validity of such setoff and application.

10.8 COUNTERPARTS. This Agreement may be executed by one or more of the parties to this Agreement on any number of separate counterparts, and all of said counterparts taken together shall be deemed to constitute one and the same instrument. Delivery of an executed signature page of this Agreement or of a Lender Addendum by facsimile transmission shall be effective as delivery of a manually executed counterpart hereof. A set of the copies of this Agreement signed by all the parties shall be lodged with each Borrower and the Administrative Agent.

10.9 SEVERABILITY. Any provision of this Agreement that is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

10.10 INTEGRATION. This Agreement and the other Loan Documents represent the entire agreement of the Parent, each Borrower, the Administrative Agent, the Arranger and the Lenders with respect to the subject matter hereof and thereof, and there are no promises, undertakings, representations or warranties by the Arranger, the Administrative Agent or any Lender relative to subject matter hereof not expressly set forth or referred to herein or in the other Loan Documents.

10.11 GOVERNING LAW. THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES UNDER THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK.

10.12 SUBMISSION TO JURISDICTION; WAIVERS. Each of the Parent and each Borrower hereby irrevocably and unconditionally:

(a) submits for itself and its Property in any legal action or proceeding relating to this Agreement and the other Loan Documents to which it is a party, or for recognition and enforcement of any judgment in respect thereof, to the non-exclusive general

jurisdiction of the courts of the State of New York, the courts of the United States of America for the Southern District of New York, and appellate courts from any thereof;

(b) consents that any such action or proceeding may be brought in such courts and waives any objection that it may now or hereafter have to the venue of any such action or proceeding in any such court or that such action or proceeding was brought in an inconvenient court and agrees not to plead or claim the same;

(c) agrees that service of process in any such action or proceeding may be effected by mailing a copy thereof by registered or certified mail (or any substantially similar form of mail), postage prepaid, to the Parent or any Borrower, as the case may be, at its address set forth in Section 10.2 or at such other address of which the Administrative Agent shall have been notified pursuant thereto;

(d) agrees that nothing herein shall affect the right to effect service of process in any other manner permitted by law or shall limit the right to sue in any other jurisdiction; and

(e) waives, to the maximum extent not prohibited by law, any right it may have to claim or recover in any legal action or proceeding referred to in this Section any special, exemplary, punitive or consequential damages.

10.13 ACKNOWLEDGMENTS. Each of the Parent and each Borrower hereby acknowledges that:

(a) it has been advised by counsel in the negotiation, execution and delivery of this Agreement and the other Loan Documents;

(b) neither the Arranger, the Administrative Agent nor any Lender has any fiduciary relationship with or duty to the Parent or any Borrower arising out of or in connection with this Agreement or any of the other Loan Documents, and the relationship between the Arranger, the Administrative Agent and the Lenders, on one hand, and the Parent and each Borrower, on the other hand, in connection herewith or therewith is solely that of debtor and creditor; and

(c) no joint venture is created hereby or by the other Loan Documents or otherwise exists by virtue of the transactions contemplated hereby among the Arranger, the Administrative Agent and the Lenders or among the Parent any Borrower and the Lenders.

10.14 CONFIDENTIALITY. Each of the Administrative Agent and the Lenders agrees to keep confidential all non-public information (including oral information) provided to it by any Loan Party pursuant to this Agreement that is designated by such Loan Party as confidential; PROVIDED that nothing herein shall prevent the Administrative Agent or any Lender from disclosing any such information (a) to the Arranger, the Administrative Agent, any other Lender or any Affiliate of any thereof (PROVIDED, that such Affiliates shall expressly agree to be bound by the provisions of this Section, and the disclosing Lender shall be responsible for insuring compliance by such Affiliate with this Section), (b) to any Participant or Assignee (each, a

"TRANSFEREE") or prospective Transferee that expressly agrees to comply with confidentiality provisions similar to the provisions of this Section, (c) to any of its employees, directors, agents, attorneys, accountants and other professional advisors, (d) to any financial institution that is a direct or indirect contractual counterparty in swap agreements or such contractual counterparty's professional advisor (so long as such contractual counterparty or professional advisor to such contractual counterparty expressly agrees to be bound by the provisions of this Section or confidentiality provisions similar to the provisions of this Section), (e) upon the request or demand of any Governmental Authority having jurisdiction over it, (f) in response to any order of any court or other Governmental Authority or as may otherwise be required pursuant to any Requirement of Law, (g) in connection with any litigation or similar proceeding, (h) that has been publicly disclosed other than in breach of this Section, (i) to the NAIC or any similar organization or any nationally recognized rating agency that requires access to information about a Lender's investment portfolio in connection with ratings issued with respect to such Lender or (j) in connection with the exercise of any remedy hereunder or under any other Loan Document.

10.15 RELEASE OF COLLATERAL AND GUARANTEE OBLIGATIONS.

(a) Notwithstanding anything to the contrary contained herein or in any other Loan Document, upon request of the Primary Borrower in connection with any Disposition of Property permitted by the Loan Documents, the Administrative Agent shall (without notice to, or vote or consent of, any Lender, or any affiliate of any Lender that is a party to any Specified Hedge Agreement) take such actions as shall be required to release its security interest in any Collateral being Disposed of in such Disposition, and to release any guarantee obligations under any Loan Document of any Person being Disposed of in such Disposition, to the extent necessary to permit consummation of such Disposition in accordance with the Loan Documents.

(b) Notwithstanding anything to the contrary contained herein or any other Loan Document, when all obligations of the Borrowers hereunder and under the other Loan Documents (other than obligations in respect of any Specified Hedge Agreement) have been paid in full, all Commitments have terminated or expired and no Letter of Credit shall be outstanding, upon request of the Primary Borrower, the Administrative Agent shall (without notice to, or vote or consent of, any Lender, or any affiliate of any Lender that is a party to any Specified Hedge Agreement) take such actions as shall be required to release its security interest in all Collateral, and to release all guarantee obligations under any Loan Document, whether or not on the date of such release there may be outstanding obligations in respect of Specified Hedge Agreements. Any such release of guarantee obligations shall be deemed subject to the provision that such guarantee obligations shall be reinstated if after such release any portion of any payment in respect of the obligations guaranteed thereby shall be rescinded or must otherwise be restored or returned upon the insolvency, bankruptcy, dissolution, liquidation or reorganization of any Borrower or any Guarantor, or upon or as a result of the appointment of a receiver, intervener or conservator of, or trustee or similar officer for, any Borrower or any Guarantor or any substantial part of its property, or otherwise, all as though such payment had not been made.

10.16 ACCOUNTING CHANGES. In the event that any "Accounting Change" (as defined below) shall occur and such change results in a change in the method of calculation of financial covenants, standards or terms in this Agreement, then the Primary Borrower and the Administrative Agent agree to enter into negotiations in order to amend such provisions of this Agreement so as to equitably reflect such Accounting Changes with the desired result that the criteria for evaluating the Primary Borrower's financial condition shall be the same after such Accounting Changes as if such Accounting Changes had not been made. Until such time as such an amendment shall have been executed and delivered by each Borrower, the Administrative Agent and the Required Lenders, all financial covenants, standards and terms in this Agreement shall continue to be calculated or construed as if such Accounting Changes had not occurred. "Accounting Changes" refers to changes in accounting principles required by the promulgation of any rule, regulation, pronouncement or opinion by the Financial Accounting Standards Board of the American Institute of Certified Public Accountants, applicable Insurance Regulators, the NAIC or, if applicable, the SEC.

10.17 DELIVERY OF LENDER ADDENDA. Each initial Lender shall become a party to this Agreement by delivering to the Administrative Agent a Lender Addendum duly executed by such Lender, the Primary Borrower and the Administrative Agent.

10.18 WAIVERS OF JURY TRIAL. THE PARENT, EACH BORROWER, THE ADMINISTRATIVE AGENT AND THE LENDERS HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVE TRIAL BY JURY IN ANY LEGAL ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT AND FOR ANY COUNTERCLAIM THEREIN. IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered by their proper and duly authorized officers as of the day and year first above written.

TACK HOLDING CORP.

By:

Name: Title:

TACK ACQUISITION CORP.

By: _____ Name:

Title:

WHITE MOUNTAINS INSURANCE GROUP, LTD.

By:

Name: Title:

LEHMAN BROTHERS INC. as Arranger

By:

Name: Title:

LEHMAN COMMERCIAL PAPER, INC. as Administrative Agent

By:

Name: Title: FLEET NATIONAL BANK as Syndication Agent

By:

Name: Title:

BANK OF AMERICA, N.A., as Documentation Agent

By:

Name: Title: CONSENTS, AUTHORIZATIONS, FILINGS AND NOTICES

SCHEDULE 4.15

SUBSIDIARIES

UCC FILING JURISDICTIONS

Loan Party	Filing Office

[Primary Borrower to list name of each Loan Party which is a party to any Security Document and each filing office in which a UCC financing statement must be filed in respect of such Loan Party and its collateral] UCC FINANCING STATEMENTS TO REMAIN ON FILE

UCC FINANCING STATEMENTS TO BE TERMINATED

SCHEDULE 7.2(e)

EXISTING INDEBTEDNESS

SCHEDULE 7.3(f)

EXISTING LIENS

SUBSIDIARIES OF THE REGISTRANT AS OF DECEMBER 31, 2000

FULL NAME OF SUBSIDIARY PLACE OF INCORPORATION -----. BARBADOS, WEST INDIES WHITE MOUNTAINS HOLDINGS (BARBADOS) SRL WM ASSET MANAGEMENT (BARBADOS) SRL BARBADOS, WEST INDIES BRITISH INSURANCE COMPANY OF CAYMAN CAYMAN ISLANDS WHITE MOUNTAINS SERVICES LLC DELAWARE, USA WATERFORD INSURANCE COMPANY KANSAS, USA PENINSULA INSURANCE COMPANY MARYLAND, USA PENINSULA INDEMNITY COMPANY MARYLAND, USA AMERICAN CENTENNIAL HOLDINGS, INC. DELAWARE, USA AMERICAN CENTENNIAL INSURANCE COMPANY DELAWARE, USA FOLKSAMERICA HOLDING COMPANY, INC. NEW YORK, USA FOLKSAMERICA REINSURANCE COMPANY NEW YORK, USA

CONSENT OF INDEPENDENT ACCOUNTANTS

We hereby consent to the incorporation by reference in the Registration Statements, as amended, pertaining to the Long Term Incentive Plan (Form S-8, No. 33-5297), Medium-Term Notes Series A (Form S-3, No. 33-54006), Common Stock Warrants (Form S-3, No. 33-54749), Folksamerica Holding Company 401(K) Savings and Investment Plan (Form S-8, No. 333- 82563), the Executive Bonus Plan (Form S-8, No. 333-89381), the Directors' Retirement Benefit Plan (Form S-8, No. 333-89385) and the Voluntary Deferred Compensation Plan (Form S-8, No. 333-89387) of White Mountains Insurance Group, Ltd. of our report dated February 9, 2001, except for Note 17, which is as of March 16, 2001, relating to the 2000 and 1999 financial statements and financial statement schedules, which appears in this Form 10-K.

/s/ PricewaterhouseCoopers

Hamilton, Bermuda March 26, 2001

CONSENT OF INDEPENDENT ACCOUNTANTS

The Board of Directors White Mountains Insurance Group, Ltd.:

We consent to the incorporation by reference in the registration statements, as amended, No. 33-5297 on Form S-8 pertaining to the Long-Term Incentive Plan, No. 33-54006 on Form S-3 pertaining to the Medium-Term Notes Series A, No. 33-54749 on Form S-3 pertaining to Common Stock Warrants, No. 333-82563 on Form S-8 pertaining to Folksamerica Holding Company 401(k) Savings and Investment Plan, No. 333-89381 on Form S-8 pertaining to the Executive Bonus Plan, No. 333-89385 on Form S-8 pertaining to the Directors Retirement Benefit Plan and No. 333-89387 on Form S-8 pertaining to the Voluntary Deferred Compensation Plan of our report dated February 12, 1999, with respect to the consolidated statements of income and comprehensive income, shareholders' equity, and cash flows of White Mountains Insurance Group, Ltd. and subsidiaries (formerly "Fund American Enterprises Holdings, Inc.") and all related schedules for the year ended December 31, 1998, which appears in this Form 10-K.

/s/ KPMG LLP

Providence, Rhode Island March 26, 2001

CONSENT OF INDEPENDENT ACCOUNTANTS

We consent to the incorporation by reference in the Registration Statements, as amended, pertaining to the Long-Term Incentive Plan (Form S-8, No. 33-5297), Medium-Term Notes Series A (Form S-3, No. 33-5406), Common Stock Warrants (Form S-3, No. 33-54749), Folksamerica Holding Company 401(K) Savings and Investment Plan (Form S-8, No. 333-82563), the Executive Bonus Plan (Form S-8, No. 333-89381), the Directors' Retirement Benefit Plan (Form S-8, No. 333-89385) and the Voluntary Deferred Compensation Plan (Form S-8, No. 333-89387) of White Mountains Insurance Group, Ltd. of our report dated January 25, 2000, except for Note 22, as to which the date is March 14, 2000, with respect to the consolidated financial statements of Financial Security Assurance Holdings, Ltd. and Subsidiaries as of December 31, 1999 and 1998 and for each of the three years in the period ended December 31, 1999 and our report dated February 2, 1999, except for Note 17 as to which the date is February 24, 1999 with respect to the consolidated financial statements of Folksamerica Holding Company, Inc. and its subsidiaries for the year ended December 31, 1998.

/s/ PricewaterhouseCoopers LLP

New York, New York March 26, 2001

KNOW ALL MEN by these presents, that Patrick M. Byrne does hereby make, constitute and appoint K. Thomas Kemp the true and lawful attorney-in-fact of the undersigned, with full power of substitution and revocation, for and in the name, place and stead of the undersigned, to execute and deliver the Annual Report on Form 10-K, and any and all amendments thereto; such Form 10-K and each such amendment to be in such form and to contain such terms and provisions as said attorney or substitute shall deem necessary or desirable; giving and granting unto said attorney, or to such person or persons as in any case may be appointed pursuant to the power of substitution herein given, full power and authority to do and perform any and every act and thing whatsoever requisite, necessary or, in the opinion of said attorney or substitute, able to be done in and about the premises as fully and to all intents and purposes as the undersigned might or could do if personally present, hereby ratifying and confirming all that said attorney or such substitute shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has duly executed these presents this 1st day of March 2001.

/s/ Patrick M. Byrne

KNOW ALL MEN by these presents, that Howard L. Clark Jr. does hereby make, constitute and appoint K. Thomas Kemp the true and lawful attorney-in-fact of the undersigned, with full power of substitution and revocation, for and in the name, place and stead of the undersigned, to execute and deliver the Annual Report on Form 10-K, and any and all amendments thereto; such Form 10-K and each such amendment to be in such form and to contain such terms and provisions as said attorney or substitute shall deem necessary or desirable; giving and granting unto said attorney, or to such person or persons as in any case may be appointed pursuant to the power of substitution herein given, full power and authority to do and perform any and every act and thing whatsoever requisite, necessary or, in the opinion of said attorney or substitute, able to be done in and about the premises as fully and to all intents and purposes as the undersigned might or could do if personally present, hereby ratifying and confirming all that said attorney or substitute shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has duly executed these presents this 1st day of March 2001.

/s/ Howard L. Clark Jr.

KNOW ALL MEN by these presents, that Robert P. Cochran does hereby make, constitute and appoint K. Thomas Kemp the true and lawful attorney-in-fact of the undersigned, with full power of substitution and revocation, for and in the name, place and stead of the undersigned, to execute and deliver the Annual Report on Form 10-K, and any and all amendments thereto; such Form 10-K and each such amendment to be in such form and to contain such terms and provisions as said attorney or substitute shall deem necessary or desirable; giving and granting unto said attorney, or to such person or persons as in any case may be appointed pursuant to the power of substitution herein given, full power and authority to do and perform any and every act and thing whatsoever requisite, necessary or, in the opinion of said attorney or substitute, able to be done in and about the premises as fully and to all intents and purposes as the undersigned might or could do if personally present, hereby ratifying and confirming all that said attorney or substitute shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has duly executed these presents this 1st day of March 2001.

/s/ Robert P. Cochran

KNOW ALL MEN by these presents, that Steven E. Fass does hereby make, constitute and appoint K. Thomas Kemp the true and lawful attorney-in-fact of the undersigned, with full power of substitution and revocation, for and in the name, place and stead of the undersigned, to execute and deliver the Annual Report on Form 10-K, and any and all amendments thereto; such Form 10-K and each such amendment to be in such form and to contain such terms and provisions as said attorney or substitute shall deem necessary or desirable; giving and granting unto said attorney, or to such person or persons as in any case may be appointed pursuant to the power of substitution herein given, full power and authority to do and perform any and every act and thing whatsoever requisite, necessary or, in the opinion of said attorney or substitute, able to be done in and about the premises as fully and to all intents and purposes as the undersigned might or could do if personally present, hereby ratifying and confirming all that said attorney or substitute shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has duly executed these presents this 1st day of March 2001.

/s/ Steven E. Fass

KNOW ALL MEN by these presents, that George J. Gillespie, III does hereby make, constitute and appoint K. Thomas Kemp the true and lawful attorney-in-fact of the undersigned, with full power of substitution and revocation, for and in the name, place and stead of the undersigned, to execute and deliver the Annual Report on Form 10-K, and any and all amendments thereto; such Form 10-K and each such amendment to be in such form and to contain such terms and provisions as said attorney or substitute shall deem necessary or desirable; giving and granting unto said attorney, or to such person or persons as in any case may be appointed pursuant to the power of substitution herein given, full power and authority to do and perform any and every act and thing whatsoever requisite, necessary or, in the opinion of said attorney or substitute, able to be done in and about the premises as fully and to all intents and purposes as the undersigned might or could do if personally present, hereby ratifying and confirming all that said attorney or substitute shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has duly executed these presents this 1st day of March 2001.

/s/ George J. Gillespie, III

KNOW ALL MEN by these presents, that John D. Gillespie does hereby make, constitute and appoint K. Thomas Kemp the true and lawful attorney-in-fact of the undersigned, with full power of substitution and revocation, for and in the name, place and stead of the undersigned, to execute and deliver the Annual Report on Form 10-K, and any and all amendments thereto; such Form 10-K and each such amendment to be in such form and to contain such terms and provisions as said attorney or substitute shall deem necessary or desirable; giving and granting unto said attorney, or to such person or persons as in any case may be appointed pursuant to the power of substitution herein given, full power and authority to do and perform any and every act and thing whatsoever requisite, necessary or, in the opinion of said attorney or substitute, able to be done in and about the premises as fully and to all intents and purposes as the undersigned might or could do if personally present, hereby ratifying and confirming all that said attorney or substitute shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has duly executed these presents this 1st day of March 2001.

/s/ John D. Gillespie

KNOW ALL MEN by these presents, that Gordon S. Macklin does hereby make, constitute and appoint K. Thomas Kemp the true and lawful attorney-in-fact of the undersigned, with full power of substitution and revocation, for and in the name, place and stead of the undersigned, to execute and deliver the Annual Report on Form 10-K, and any and all amendments thereto; such Form 10-K and each such amendment to be in such form and to contain such terms and provisions as said attorney or substitute shall deem necessary or desirable; giving and granting unto said attorney, or to such person or persons as in any case may be appointed pursuant to the power of substitution herein given, full power and authority to do and perform any and every act and thing whatsoever requisite, necessary or, in the opinion of said attorney or substitute, able to be done in and about the premises as fully and to all intents and purposes as the undersigned might or could do if personally present, hereby ratifying and confirming all that said attorney or substitute shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has duly executed these presents this 1st day of March 2001.

/s/ Gordon S. Macklin

KNOW ALL MEN by these presents, that Frank A. Olson does hereby make, constitute and appoint K. Thomas Kemp the true and lawful attorney-in-fact of the undersigned, with full power of substitution and revocation, for and in the name, place and stead of the undersigned, to execute and deliver the Annual Report on Form 10-K, and any and all amendments thereto; such Form 10-K and each such amendment to be in such form and to contain such terms and provisions as said attorney or substitute shall deem necessary or desirable; giving and granting unto said attorney, or to such person or persons as in any case may be appointed pursuant to the power of substitution herein given, full power and authority to do and perform any and every act and thing whatsoever requisite, necessary or, in the opinion of said attorney or substitute, able to be done in and about the premises as fully and to all intents and purposes as the undersigned might or could do if personally present, hereby ratifying and confirming all that said attorney or substitute shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has duly executed these presents this 1st day of March 2001.

/s/ Frank A. Olson

KNOW ALL MEN by these presents, that Arthur Zankel does hereby make, constitute and appoint K. Thomas Kemp the true and lawful attorney-in-fact of the undersigned, with full power of substitution and revocation, for and in the name, place and stead of the undersigned, to execute and deliver the Annual Report on Form 10-K, and any and all amendments thereto; such Form 10-K and each such amendment to be in such form and to contain such terms and provisions as said attorney or substitute shall deem necessary or desirable; giving and granting unto said attorney, or to such person or persons as in any case may be appointed pursuant to the power of substitution herein given, full power and authority to do and perform any and every act and thing whatsoever requisite, necessary or, in the opinion of said attorney or substitute, able to be done in and about the premises as fully and to all intents and purposes as the undersigned might or could do if personally present, hereby ratifying and confirming all that said attorney or substitute shall lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has duly executed these presents this 1st day of March 2001.

/s/ Arthur Zankel